# **Rachel Wojdak**

Studies in Natural Language and Linguistic Theory

73

# The Linearization of Affixes: Evidence from Nuu-chah-nulth



# THE LINEARIZATION OF AFFIXES: EVIDENCE FROM NUU-CHAH-NULTH

# VOLUME 73

## Managing Editors

Marcel den Dikken, *City University of New York* Liliane Haegeman, *University of Lille* Joan Maling, *Brandeis University* 

#### Editorial Board

Guglielmo Cinque, University of Venice Carol Georgopoulos, University of Utah Jane Grimshaw, Rutgers University Michael Kenstowicz, Massachusetts Institute of Technology Hilda Koopman, University of California, Los Angeles Howard Lasnik, University of Maryland Alec Marantz, Massachusetts Institute of Technology John J. McCarthy, University of Massachusetts, Amherst Ian Roberts, University of Cambridge

# THE LINEARIZATION OF AFFIXES: EVIDENCE FROM NUU-CHAH-NULTH

by

RACHEL WOJDAK

University of Ottawa, Canada



A C.I.P. Catalogue record for this book is available from the Library of Congress.

ISBN 978-1-4020-6549-1 (HB) ISBN 978-1-4020-6548-4 (e-book)

> Published by Springer, P.O. Box 17, 3300 AA Dordrecht, The Netherlands.

> > www.springer.com

Printed on acid-free paper

All Rights Reserved © 2008 Springer

No part of this work may be reproduced, stored in a retrieval system, or transmitted in any form or by any means, electronic, mechanical, photocopying, microfilming, recording or otherwise, without written permission from the Publisher, with the exception of any material supplied specifically for the purpose of being entered and executed on a computer system, for exclusive use by the purchaser of the work.

## Preface

This book is a revision of my doctoral dissertation, completed at the University of British Columbia in 2005. The members of my doctoral committee – Henry Davis, Rose-Marie Déchaine, Martina Wiltschko – were instrumental in shaping the content and presentation of this work, and I owe them my gratitude for their considerable input. As my supervisor, Henry gave me great latitude in this project, and his encouragement and enthusiasm made fieldwork all the more exciting. This final product would not have been nearly as strong without the benefit of Rose-Marie's sharp intuition and critiques, and I appreciate the memories I have of leaving meetings with her feeling absolutely invigorated. Martina always knew all the right questions to ask, and I came to count on her reliable feedback. It was an honour to be a part of the UBC linguistics department, and I thank all its members for their part in creating the lively environment I enjoyed for my years there.

My greatest thanks are due to the speakers of Nuu-chah-nulth who shared with me their time, their humour, and their love for their language: Mary Jane Dick, Katherine Fraser, Carrie Little, Archie Thompson, Barbara Touchie, Barney Williams Jr., Barney Williams Sr., Christine Nicolaye, Sarah Webster. Their patience and dedication was humbling, and I am indebted to them for the chance they gave me to learn so much. It was a true privilege to be welcomed into their lives and their homes, and I enjoyed the time I spent with them. Thanks in particular to Mary Jane for being a tireless consultant, a devoted researcher, and my determined teacher from the very start back in January 2000. I am also especially grateful to Katherine for her leadership over the years. The aim of this book is to do justice to the elegance that nuučaanut speakers have shown me characterizes their language. It can be a challenge to bridge the gap between theoretical and practical linguistics, but it is possible to transfer the examples of "*Au*-verbs" in this book to a practical use for those wanting to learn the language. This book is written in a technical style, but really it is just about how to form sentences using those special *Au*-verbs.

A number of people provided thoughtful comments on various aspects of this project over the years, including Peter Ackema, David Adger, Wallace Chafe, Matthew Davidson, Carrie Gillon, Eun-Sook Kim, Felicia Lee, Éric Mathieu, Marianne Mithun, Gunnar Hansson, Doug Pulleyblank, Christine Ravinski, Eric Reuland, Maria-Luisa Rivero, Naomi Sawai, Lisa Matthewson, Ryan Waldie, Adam Werle, Florence Woo, and Susi Wurmbrand. I would also like to thank two anonymous reviewers for the extensive feedback I received. The questions and comments I received from participants of the NIK (Ottawa, 2006), CLA (Halifax, 2003; Winnipeg, 2004), GLOW (Lund, 2003), and ICSNL (Lillooet, 2003) conferences were particularly encouraging to me. On the fieldwork side, I would like to thank the folks at the University of Victoria for allowing access to department facilities for elicitation sessions.

Finally, I would like to thank all of my family members for their support, curiousity, and encouragement. I owe a great deal to my husband Antal Járai for all he has added to my life over the past several years; I am tremendously lucky to have

#### PREFACE

had such a thoughtful non-linguist audience for this book's "puzzles". My son Zoltán came into being as this book took shape, and I cannot imagine it any other way.

I gratefully acknowledge funding for my fieldwork from the Jacobs Research Fund (Whatcom Museum Society, Bellingham WA), Phillips Fund (American Philosophical Society), UBC Hampton Fund Research Grant in the Humanities and Social Sciences (to Henry Davis), SSHRC 410-1998-1597 (to Henry Davis), SSHRC 410-2002-1715 (to Lisa Matthewson), and SSHRC 410-2002-1078 (to Martina Wiltschko). My graduate studies at UBC were financially supported by a SSHRC doctoral fellowship, an Izaak Walton Killam fellowship, a Patrick David Campbell fellowship, and a University Graduate Fellowship.

# **Table of Contents**

Pre	eface		V
1.	Int	roduction	1
	1.0	Introduction	1
	1.1	Theoretical Assumptions	3
		1.1.1 Interface Requirements	4
		1.1.2 "Bottom-to-top" Syntactic Derivation	5
		1.1.3 Syntax "All the Way Down"	6
	1.2	Linearization	7
		1.2.1 Proposal: Local Spell-out	7
		1.2.2 Alternative Models for the Timing of Spell-out	10
	1.3	Affixal Predicates in Nuu-chah-nulth	13
	1.4	Research Context	18
		1.4.1 Genetic Affiliations	19
		1.4.2 Previous Literature	20
		1.4.3 Overview of Word Structure in Nuu-chah-nulth	21
		1.4.4 Methodology	23
	1.5	Outline of the Book	24
2.	PF	Incorporation	27
	2.0	Introduction	27
	$\frac{2.0}{2.1}$	Mornho-nhonological Dependency	32
	2.1 2.2	L inearization is L ocal	
	2.2	2.2.1 Iterativity	40
		2.2.1 Inclutivity	40 43
		2.2.2.1 Adjective Incorporation	
		2.2.2.2 Adverbial Incorporation	
		2.2.3 Coordinated Objects	
	2.3	Insensitivity to Syntactic Category	
	2.4	Absence of LF Effects	
		2.4.1 Discourse Transparency	59
		2.4.2 Absence of Scopal Effects	60
	2.5	Opacity Effects	61
		2.5.1 Opaque DPs and the Bare Nominal Requirement	63
		2.5.2 Opaque CPs and "Restructuring" Effects	64
		2.5.3 Impermeability of Saturated Domains	65
	26	Conclusion	67

3.	Cla	usal Architecture of Nuu-chah-nulth	69
	3.0	Introduction	69
	3.1	Linearization of Syntactic Terminals	70
		3.1.1 A Universal Linearization Scheme?	71
		3.1.2 Directionality is Determined at Spell-out	73
		3.1.2.1 Directionality Conventions	73
		3.1.2.2 Comparison to Headedness "Parameters"	75
		3.1.2.3 Affixation Patterns	76
		3.1.2.4 Directionality Conventions of Nuu-chah-nulth	76
	3.2	Configurationality	78
		3.2.1 Partial Head Marking	79
		3.2.2 An Asymmetry in Possessor Raising	81
		3.2.3 An Asymmetry in Incorporation	83
		3.2.4 Weak Crossover Effects	84
	3.3	Predicate-initial Word Order	84
		3.3.1 Proposal: "Basic" Predicate-initial Order (VOS)	87
		3.3.2 Evidence for Right-branching Specifiers	90
		3.3.3 Evidence Against Verb-raising	91
		3.3.4 Evidence Against Raising of Verbal Phrase	94
		3.3.5 Implications for the Linearization of Affixal Predicates	96
		3.3.5.1 The "Complement" Effect	96
		3.3.5.2 Evidence Against Directionality of Affixation	98
	3.4	Cliticization Domains	100
		3.4.1 DP Domain	100
		3.4.2 CP Domain	103
	3.5	Conclusion	105
4.	Noi	minal Complements of Affixal Predicates	107
	4.0	Introduction	107
	4.1	Predication Configurations	109
		4.1.1 Basic Structures	109
		4.1.2 Transitivization	111
		4.1.3 Flexibility in Theta-role Mapping	112
		4.1.4 Predicate Inventory	114
		4.1.5 Unergatives are Necessarily Non-affixal	115
	4.2	Unaccusatives	116
		4.2.1 Incorporation	117
		4.2.2 Intransitivity in Nuu-chah-nulth	118
		4.2.3 Absence of Unergative Affixal Predicates	119
		4.2.4 "Long" Possessor Raising as a Diagnostic	
		for Unaccusativity	121
	4.3	Transitives	129
		4.3.1 Incorporation	130

		4.3.2 Subject Agreement	132
		4.3.3 Possessor Raising	134
	4.4	Extended Unaccusatives	136
		4.4.1 Locatum Predicates	141
		4.4.1.1 Incorporation	142
		4.4.1.2 Subject Agreement	144
		4.4.1.3 Possessor Raising	145
		4.4.2 Location Predicates	146
		4.4.2.1 Incorporation	147
		4.4.2.2 Subject Agreement	148
		4.4.2.3 Possessor Raising	149
	4.5	Ditransitives	150
		4.5.1 Locatum-type Ditransitives	150
		4.5.2 Location-type Ditransitive	152
	4.6	Serial Verb Affixation	154
	4.7	Conclusion	158
5.	Ver	bal Complements of Affixal Predicates	159
		-	
	5.0	Introduction	159
	5.1	Two Classes of "Verb Incorporating" Affixal Predicates	160
		5.1.1 Affixal Main Predicates	160
		5.1.2 Affixal Auxiliary Predicates	161
		5.1.3 Affixal Auxiliaries are Non-thematic	163
		5.1.3.1 Rigid vs. Alternating Complementation	164
		5.1.3.2 Ability to License a "Mismatched" Subject	164
		5.1.3.3 Citation Forms	169
		5.1.4 Summary	169
	5.2	PF Incorporation Occurs Only in Infinitival Environments	169
	5.3	Absence of Clausal Morphology	172
		5.3.1 No Tense	173
		5.3.2 No Person/Mood Agreement	174
		5.3.3 No Complementizer	175
	5.4	Lack of Clause-boundedness Effects	176
		5.4.1 "Long" wh-movement	177
		5.4.2 "Long" Possessor Raising	179
		5.4.2.1 "Long" Possessor Raising with Affixal	
		Auxiliaries	180
		5.4.2.2 No "Long" Possessor Raising	
		with Affixal Main Predicates	182
	5.5	Conclusion	185
	_		
6.	Imp	plications	187
	6.0	Introduction	187
	6.1	Summary	187

#### TABLE OF CONTENTS

6.2	Theor	etical Im	plications	189
	6.2.1	Spell-or	it is Strictly Interpretive	189
		6.2.1.1	Comparison to "Phasal" Spell-out	190
		6.2.1.2	Variation in Permeability of CP Domains	193
		6.2.1.3	The Status of the vP Domain	196
	6.2.2	The Hea	ad Movement Question	199
		6.2.2.1	Linearization is Purely Phonological	201
		6.2.2.2	Syntactic Head Movement is More Complex	204
	6.2.3	Local Sp	pell-out Creates "Outside-in" Dependencies	206
		6.2.3.1	"Inside-out" Dependencies are not Possible	
			via Local Spell-out	207
		6.2.3.2	Evidence from Kwakw'ala (Northern Wakashan)	207
		6.2.3.3	Affixation as Primitive or Derived	208
6.3	Typol	ogical Im	plications	210
	6.3.1	The Typ	pology of Noun Incorporation	211
	6.3.2	The Gov	verning/Restrictive Hypothesis in Wakashan	213
	6.3.3	Lexical	Suffixation as an Areal Feature	
		of the P	acific Northwest	214
6.4	Concl	usion		215
Referen	ices	•••••		217
Append	lix	•••••		227
Abbrev	iation	5		229
Index	•••••	•••••		231

## 1. Introduction

What we shall call the beginning is often the end And to make an end is to make a beginning. The end is where we start from. ~T.S. Eliot Four Quartets no. 4

#### **1.0 Introduction**

This book proposes a new understanding of the means by which affixes in natural language come to be linearized. Affixal elements, whether prefixal or suffixal, share the property of requiring a "host" with which they may form a word. An affix is not permitted to stand on its own. For example, the English morpheme *un*- must occur as an affix (specifically, a prefix), and not as an independent word.

(1)	a.		I am <u>un</u> hurried.	cf.	I am not rushed.
	b.	*	I am <u>un</u> .	cf.	I am <u>not</u> .

It is often said that *morphology* governs affixation, as it imposes restrictions on the internal composition of words. It is this aspect of the grammar which distinguishes between "bound" morphemes (such as *un*-), which form subparts of words, and "free" morphemes (such as *not*), which are permitted as independent words. The system of *syntax*, on the other hand, can be understood to be the means by which words are grouped together to form larger phrases.

This book presents an argument that syntax plays a determining role in the combinatory properties not only of words, but also of affixes (cf. Baker 1988; Lieber 1992, among others). According to the analysis, affixes have a syntax which is indistinct from that of non-affixes. I propose that the local syntactic environment of a morpheme conditions the available patterns of affixation. Under this view, affixes are subject to the same sorts of structural relationships in the syntax as are non-affixes. However, morphemes with affixal status introduce a tension to the grammar in a way in which non-affixal elements do not. If one or the other of  $\alpha$  or  $\beta$  in the syntactic construct in (2) is an affix, then a host for the affix must be found in order for the expression to be pronounced.

What this study undertakes is an examination of the way that this phonological "neediness" of affixes is resolved by the grammar. I will argue that affixation derives a restricted set of linearizations for syntactic constructs such as (2). If, for example,  $\alpha$  is a prefix, then a linearization of  $\alpha$ - $\beta$  results. If, however,  $\alpha$  is a suffix, then an ordering of  $\beta$ - $\alpha$  arises. The consequence of this analysis is that phonological

considerations induce linearization on inherently *unordered* syntactic entities (cf. Chomsky 1995; Fox and Pesetsky 2005).

The language of investigation for this study is Nuu-chah-nulth (Nootka), a member of the Wakashan family spoken in British Columbia, Canada. Nuu-chah-nulth has a class of *affixal predicates* which participate in a linearization strategy of suffixation. In the following example, the affixal predicate  $-\dot{c}aas$  "bet" suffixes to the nominal  $ki \hat{\pi}uuk$  "dishes". The affixal predicate and its gloss are indicated by highlighting.

(3)	ki <del>∕u</del> ukćasitniš?aa†	huu?ak?uyi
	ki⊁uuk- <u><b>ćaas</b></u> -mit-niiš-?aa†	huu?ak-?uyi
	dishes- <u>bet</u> -PST-1PL.IND-HAB	early-ago
	We always used to bet dishes lo	ng ago.

Affixal predicates in Nuu-chah-nulth are obligatorily bound, and may never occur as non-suffixes. This is illustrated by the example in (4), which shows that it is impossible for the affixal predicate  $-\dot{c}aas$  "bet" to appear without a host which it may suffix to. In this ungrammatical example, the nominal  $ki \mathcal{R}uuk$  "dishes" occurs separately from the affixal predicate.

(4)	*	ċaasitniš?aa₁	ki⊁uuk	huu?ak?uyi	
		<u><b>čaas</b></u> -mit-niiš-?aa†	ki⊁uuk	huu?ak-?uyi	
		<b>bet</b> -PST-1PL.IND-HAB	dishes	early-ago	
		We always used to bet dishes long ago.			

This book develops a Minimalist approach to the linearization of affixal predicates in Nuu-chah-nulth and argues that suffixation provides a means of satisfying the requirement that linguistic outputs be linearized. According to the proposal, Nuuchah-nulth is "proof-of-concept" for the hypothesis that linearization is induced in stages corresponding to strictly minimal syntactic units (cf. Epstein et al. 1998). This analysis employs only the smallest necessary constructs for the syntax to phonology mapping.

The linearization of affixal predicates in Nuu-chah-nulth may be observed to induce a "displacement" effect. Displacement effects are ubiquitous in natural language, and can be found at various levels in the grammar. When displacement applies at a syntactic level, words or phrases are "dislocated", such that they are pronounced in one position in a sentence although they are interpreted in another. For example, in the English sentence *Who did Kyle see?* the word *who* is felt to be the object of the verb *see*, even though it occurs in initial position within the clause and not in canonical object position following the verb (parallel to *Sarah* in the sentence *Did Kyle see Sarah?*). In such contexts, transformational linguists analyse the fronted word as having undergone an abstract form of movement. Considerable syntactic research lies in the examination of the displacement properties of words and larger constituents (e.g., Ross 1967; Chomsky 1977).

Dislocation may also be examined at a subword level. In the case of Nuuchah-nulth suffixation, the "piece" of word which acts as a host for an affixal

predicate is removed from the post-verbal position it would otherwise occupy. In (5a), the nominal *?aapinis* "apple" serves as a host for the affixal predicate – *iic* "consume". Non-nominal hosts are also available to affixal predicates. In (5b), the modifier *ha?um* "tasty" acts as a host for the affixal predicate – *iic* "consume", while the nominal *?aapinis* follows the initial predicate complex.

- (5) a. ?aapiniyic?iš?a<sup>+</sup> ?aapinis-<u>'iic</u>-?iiš-?a<sup>+</sup> apple-<u>consume</u>-3.IND-PL They are eating apples.
  - b. ha?um?ic?iš?a<sup>+</sup> ?aapinis ha?um-<u>'iic</u>-?iiš-?a<sup>+</sup> ?aapinis tasty-<u>consume</u>-3.IND-PL apple They are eating delicious apples.

I will argue that the post-verbal position of the nominal *?aapinis* "apple" in (5b) is the standard orientation for objects in Nuu-chah-nulth. Thus, the ordering of *?aapinis* "apple" as a pre-verbal host for the affixal predicate in (5a) represents a departure from the general linearization pattern.

What drives the displacement properties of affixation? This question is to be addressed here from the perspective of the linearization of affixal predicates in Nuu-chah-nulth. In a sense to be made more explicit over the course of this chapter, suffixation in Nuu-chah-nulth is triggered by the need to pronounce an affix. Affixes, unlike free morphemes, are bound elements which cannot be pronounced in isolation. When an affixal predicate attaches to a host, its boundedness requirement is met, and a well-formed word results (cf. Lasnik 1981). It is a consequence of the affixal predicate's status as a suffix that the internal components (host and affix) of the resulting word are fixed in a particular (host-initial) order.

The next sections of this introductory chapter are organized into four parts. The following section, §1.1, presents an overview of the Minimalist framework which is employed for the analysis of affixal predicates. The theoretical backdrop is further developed in §1.2, in which the linearization of syntactic constructs is discussed. This leads to the central claim of this book, introduced in §1.3, that the affixation pattern of Nuu-chah-nulth is a reflex of the linearization of linguistic outputs. In §1.4, I situate the discussion of Nuu-chah-nulth affixal predicates within the broader context of Wakashan linguistics. Finally, §1.5 outlines the form which the remaining chapters of this book takes.

#### **1.1 Theoretical Assumptions**

This section lays out the Minimalist theoretical framework which this book adopts. Following Chomsky (1995, 2001, 2005), I pursue a strongly derivational approach to syntactic structure.

#### **1.1.1 Interface Requirements**

The grammar is charged with the task of delivering linguistic expressions which are serviceable to two external systems: the system of thought, and the sensorimotor system (Chomsky 1995, 2001, 2005). The grammar thereby allows spoken languages to map an abstract form to meaning and to sound. According to the Minimalist approach, a linguistic expression exiting the generative system is viable only if it meets the interface requirements imposed by these external levels. In Chomsky's (1995) terminology, linguistic expressions must be "legible" to each interface level, Logical Form (LF) and Phonetic Form (PF).

(6) *Model of the grammar* 



In this system, the lexicon acts as the source of the elements which enter the computation. The lexicon codes the semantic, syntactic, and phonological properties which are specific to each lexical item. Lexical items enter the computation from the lexical array known as the numeration. Syntactic structures are composed using the lexical building blocks provided by the numeration, exhausting this lexical array. The syntactic constructs are interpreted at the LF and PF interfaces at the point of *spell-out*.

#### (7) Spell-out: map syntax to the interfaces

Spell-out to the PF portion feeds the outputs of syntactic structure-building (Merge and Move) to the phonological system, where they may receive a pronounceable form.

With the exception of the interfaces at LF and PF, no other levels exist in the Minimalist grammar. Representational levels such as "deep structure" and "surface structure", which existed in earlier principles and parameters models, are eschewed in favour of a more barebones model which contains only the conceptually necessary meaning/sound interfaces. This move away from representational levels corresponds to the Minimalist ideal of paring down the grammar to those elements which are necessary design features of natural language. All constraints are abandoned save those which hold at the interfaces and "are motivated by the properties of the interface" (Chomsky 1995: 171). These interface requirements are known as *bare output* conditions: external to the syntax, these interpretative conditions are mandated by the requirements of the conceptual and sensorimotor systems, and ensure that the requirements of lexical items are met over the course of the derivation.

#### 1.1.2 "Bottom-to-top" Syntactic Derivation

This book adopts from Chomsky (1995, 2001, 2005) the notion that syntactic derivations are built up from "bottom-to-top", through successive applications of two concatenative operations: Merge and Move. Merge operates on elements selected from the numeration, and conjoins pairs of items in a binary fashion:

(8) Merge: concatenate  $\alpha$  with  $\beta$ , forming  $\gamma$ 

If X and Y are merged, the category label of one of these conjoined elements is projected. For example, in (9), the category of X is projected as X(P).



Merge applies iteratively, building a syntactic structure by pairing the output of a prior instance of Merge with a lexical item freshly introduced from the numeration. In the following representation, Z is added to the structure of (9) via an additional application of Merge.

(10) Merge(Z, XP)



All binary merger creates two sisters – a pairing which Epstein et al. (1998) label *derivational sisterhood*. In the trees above, [X, Y] are derivational sisters, as are [Z, XP].

The operation of Move (or "remerge") parallels Merge in that it also pairs two syntactic objects and projects a single category label (Kitahara 1994, 1995; Epstein et al. 1998). Move differs from pure Merge, however, in that it reinserts a syntactic object already introduced in the derivation, rather than selecting a new item from the numeration. Thus, while Merge applies to lexical items external to the existing syntactic construct, Move looks internally to the derivation to "recycle" a previously introduced lexical item (Chomsky 2005). Like Merge, Move is an instance of binary concatenation. Move can be captured by a restatement of the simple Merge operation, as in (11). (11) (re)Merge: concatenate  $\alpha$  (where  $\alpha$  is an existing terminal) with  $\beta$ , forming  $\gamma$ 

By reducing all operations of the syntax to operations of binary concatenation, nonbranching nodes are eliminated from the syntax. That is, there will be no instances in which an element does not have a derivational sister (Epstein et al. 1998).

#### 1.1.3 Syntax "All the Way Down"

Following Halle and Marantz (1993) and other work in the Distributed Morphology framework, I assume that word formation parallels sentence formation in that both occur outside of the lexicon. There is no independent module for word-forming operations: "morphology" is a cover term for syntactic or post-syntactic processes. That is, concatenation of morphemes may apply through syntactic processes of head movement, or it may be conditioned by the post-syntactic interface with PF (Embick and Noyer 2001).

Under this view, a complex string such as the Nuu-chah-nulth sentence in (12a) has a syntactic structure similar to a sentence in which the individual morphemes are expressed as separate words, as in the English sentence in (12b).

(12)	a.	?aapiniýicmaḥsa?iš?a‡
		?aapinis- <u><b>'iic</b>-<b>mahsa</b>-?iiš-?a+</u>
		apple- <u>consume</u> - <u>want.to</u> -3.IND-PL
		They want to eat apples.

b. They want to eat apples.

Syntactic structures for the Nuu-chah-nulth and English sentences are illustrated in (13).<sup>1</sup>



In each case, the morphemes occupy syntactic terminals. The claim of this book is that affixation requirements of bound morphemes are responsible for the distinct linearization patterns of the sentences in (12). In Nuu-chah-nulth, affixes are linearized attached to a host with which they may form a word. English, which lacks

<sup>&</sup>lt;sup>1</sup> The trees in (13) abstract away from several syntactically relevant properties. For one, I assume that Nuu-chah-nulth makes use of a covert pronominal (*pro*) in cases such as (13a) in which the 3rd person argument is phonologically null. The third person plural ending -2is(2a+) is not to be taken to be equivalent to English "they". I refer the reader to Chapter 3 for discussion of Nuu-chah-nulth clausal structure.

affixal predicates parallel to -iic "consume" or -mahsa "want to", linearizes these morphemes as independent words. According to the maxim of "syntax all the way down", the affixal or non-affixal status of the morphemes has no consequence for syntactic representation.

#### 1.2 Linearization

In the theoretical framework I have adopted, there are two sets of requirements which must be met over the course of the derivation by elements entering the syntactic computation. In the bifurcated model of the Minimalist grammar, requirements may be necessitated by the interface to LF, or the interface to PF.

With respect to the PF branch, how must elements be arranged so that the sensorimotor systems can make use of them? Crucially, a (spoken) linguistic expression must be *sequentially ordered* so that it may be represented as a speech stream (Kayne 1997). Linearization is a bare output condition on PF (Chomsky 1995). A linearization scheme is not provided by the inherent mechanics of the syntax. As described in §1.1.2, syntactic structure-building reduces to two concatenative operations, Merge and Move, which are not inherently oriented for directionality. When Merge unites two elements,  $\alpha$  and  $\beta$ , there is no restriction whether  $\alpha$  must precede  $\beta$ , or whether  $\alpha$  follows  $\beta$ . All that binary concatenation requires is that  $\alpha$  combine with  $\beta$ , joining an unordered set of { $\alpha$ ,  $\beta$ }. Given the unordered nature of binary concatenation, the representations in (14) are therefore to be interpreted as syntactically equivalent.



1.2.1 Proposal: Local Spell-out

If one or the other of the nodes { $\alpha$ ,  $\beta$ } are understood to be an affix, then a specific type of linearization requirement is imposed on the orientation of these elements. I hypothesize that this requirement applies at spell-out, the point at which syntactic structures such as (14) take on a phono-temporal form.<sup>2</sup> Affixation induces a particular linearization of terminal elements: if an element  $\alpha$  is an affix, it must realized as a suffix ( $-\alpha$ ) or as a prefix ( $\alpha$ –). Suffixation occurs when the affix is pronounced to the right of a host with which it forms a word; prefixation takes place when the affix forms a word with a host by attaching to its left. I do not consider "infix" to be a distinct boundedness requirement, as I assume that infixation is reducible to either prefixation or suffixation. In Nuu-chah-nulth, for example, the plural "infix" *-t*–(e.g., *Si-t-niii*t "dogs") can be analysed as a suffix which is positioned prosodically following the first syllable of its host (Stonham 1999; Wojdak 2002).

<sup>&</sup>lt;sup>2</sup> Alternative affixation mechanisms which are employed in other languages are discussed in Chapter 6. As will be shown, affixation in Nuu-chah-nulth is a strictly lexical requirement of bound morphemes; it does not arise from prosodic deficiency.

I take the choice of prefixation or suffixation for a given bound element to be a spell-out convention, relatable to language- or morpheme-specific considerations.<sup>3</sup> The logical possibilities for linearization of the syntactic terminals in (14) are listed in Table 15:

	α	-α	α–
β	αβ	β–α	α-β
	βα		
-β	α-β	-α-β	α–β
		$-\beta-\alpha$	
β–	β–α	β–α	α-β-
			β-α-

#### (15) Linearization of bound and free elements

In three cases (shaded in the above table), there is more than one option available for linearization. When neither  $\alpha$  nor  $\beta$  is an affix, affixation cannot serve as an ordering mechanism. When both  $\alpha$  and  $\beta$  are suffixes, their relative orientation is indeterminate; the same applies when both  $\alpha$  and  $\beta$  are prefixes. In these indeterminate cases, one or the other of the affixes does not receive its required type of host. In the case of  $(-\alpha-\beta)$ , for example, the element  $-\alpha$  is not bound as a suffix. Indeterminacy in this sense therefore entails that the affixation requirement of the elements is not met: no appropriate host has been provided for the affixes. As such, there is no interpretable output for the linearization of these affixes.

The remaining, fully interpretable, orientations of affixes have just two surface realizations:  $\alpha - \beta$  or  $\beta - \alpha$ . For each of these linearization patterns, there are three distinct types of underlying bound/free dependencies. These structural ambiguities with respect to the surface forms of  $\alpha - \beta$  and  $\beta - \alpha$  are listed in (16).

#### (16) Structural ambiguities with respect to outputs

output: $\alpha - \beta$	output: $\beta - \alpha$
a. $\alpha$ , $-\beta$ ( $\alpha$ is free; $\beta$ is a suffix)	d. $\alpha$ , $\beta$ - ( $\alpha$ is free; $\beta$ is a prefix)
b. $\alpha$ -, $\beta$ ( $\alpha$ is a prefix; $\beta$ is free)	e. $-\alpha$ , $\beta$ ( $\alpha$ is a suffix; $\beta$ is free)
c. $\alpha$ -, - $\beta$ ( $\alpha$ is a prefix; $\beta$ is a	f. $-\alpha$ , $\beta$ - ( $\alpha$ is a suffix; $\beta$ is a
suffix)	prefix)

Thus, although the syntactic device of binary concatenation provides no instructions for linearization, a restricted set of linearizations arises when the merged element is an affix. This linearization is, by its very nature, non-syntactic: the syntax itself can be assumed to be unordered. I refer to the means by which the relative ordering of

<sup>&</sup>lt;sup>3</sup> As described in Chapter 3, the status of an affix as a prefix or a suffix is determined for a language learner during the process of acquisition, from salient evidence in the input.

affixes in Nuu-chah-nulth is fixed as *local spell-out*. This mechanism provides the input to interpretation at PF (and LF) based on strictly minimal syntactic domains. This linearization mechanism is defined in statement 17:

#### (17) Local spell-out: for Merge $(\alpha, \beta)$ , interpret $(\alpha, \beta)$

This mechanism is "local" in that it is hypothesized to apply at each minimal step of the syntactic derivation, to derivational sisters conjoined by Merge (see also Epstein et al. 1998; Matushansky 2006). It is inherently a pairwise function, because each step of the derivation is an operation of binary concatenation.

In the Minimalist framework, the necessity of orienting an affix with respect to a host is a consequence of spell-out to PF. Affixes require linearization so that the arrangement may be phono-temporally ordered. An earlier formulation of this affixation requirement is the Stranded Affix Filter of Lasnik (1981, 2000). Although this filter does not make reference to phono-temporal sequencing, it does capture the notion that a derivation is not viable if an affix does not find a host. A mechanical apparatus for affixation is supplied by the Morphological Merger operation of Marantz (1988, 1989; see also Bobaljik 1994), and its more recent incarnations, Lowering and Local dislocation (Embick and Noyer 2001). In each of these variations, an affixation rule forces two elements to "switch places", with a single word resulting.

#### (18) *Morphological merger*: $X \dots Y \rightarrow [Y + X]$

In the Minimalist programme, bare output conditions are adopted instead of derivational filters such as the Stranded Affix Filter or rules such as Morphological Merger. The local spell-out hypothesis provides a Minimalist alternative to these filter- or rulebased mechanisms for affixation.<sup>4</sup>

In the PF branch, spell-out necessarily induces linearization: linguistic outputs must be linearized. Thus, the two elements treated by local spell-out inevitably undergo linearization with respect to each other. According to the local spell-out hypothesis, the PF branch receives directions to resolve affixation requirements each time the syntactic tree is expanded, as indicated in (19). We can take the diagram in (19) to be the output of three successive applications of Merge: the first uniting  $\alpha$  and  $\beta$  (Merge  $\alpha$ ,  $\beta$ ); the second uniting  $\delta$  and  $\gamma$  (Merge  $\delta$ ,  $\gamma$ ); the third uniting  $\theta$  and  $\phi$  (Merge  $\theta$ ,  $\phi$ ).

<sup>&</sup>lt;sup>4</sup> An additional type of output "filter" on the positioning of affixes are Optimality Theory constraints on alignment. In Optimality Theory, constraint rankings determine whether an affix is right-aligned to a host as a suffix by AlignR[Affix] or left-aligned to a host as a prefix by AlignL[Affix]. A minimalist solution for the positioning of affixes with respect to hosts is developed in Chapter 3.

(19) *Iterative application of local spell-out* 



For each application of Merge, local spell-out applies to the derivational sisters. The interface requirements of elements introduced in the derivation are thus subject to aggressive interpretation, in which PF and LF needs are assessed at each step of the derivation. This entails that syntactic outputs are richly "phonologized" over the course of the derivation, via addition of directions for pairwise interpretations at PF (Epstein et al. 1998).

According to the proposal, spell-out enriches the derivation, due to iterative interpretations at the interfaces. This echoes the argument of Fox and Pesetsky (2005), who hypothesize that the sole function of spell-out is to add information. By their *Order Preservation hypothesis*, information established in one cycle of spell-out is never deleted over the course of the derivation. This determines that linearization is established cumulatively:

Each time a new Spell-out domain D' is constructed, Spell-out linearizes the new material in D' and adds information about its linearization to the information cumulatively produced by previous applications of Spell-out.

(Fox and Pesetsky 2005: 6)

Fox and Pesetsky argue for the Order Preservation hypothesis in the context of successive-cycle syntactic movement, assuming a model in which spell-out applies at certain designated syntactic nodes, rather than at each instance of Merge. This timing of spell-out contrasts with the present proposal, in which local spell-out applies incrementally, at each step of the syntactic derivation. The next section compares the local spell-out hypothesis to alternative models of the timing of spell-out.

#### **1.2.2** Alternative Models for the Timing of Spell-out

There are three logical possibilities for the point at which spell-out occurs. Mapping of the syntax to the interfaces may apply once, multiple times, or at every stage possible in the derivation. The third option corresponds to the local spell-out hypothesis. (20) Timing of spell-out

a.	SINGLE SPELL-OUT HYPOTHESIS:	root branch node
b.	MULTIPLE SPELL-OUT HYPOTHESIS:	multiple branch nodes
c.	LOCAL SPELL-OUT HYPOTHESIS:	every branch node

This section briefly compares the characteristics of each model, and discusses a unique prediction of the local spell-out hypothesis.

If spell-out is restricted to applying to the root branch node, then spell-out applies once per utterance. This represents the traditional view, in which mapping between syntax and phonology takes place at a single point, after the completion of the syntactic derivation (Chomsky 1995). In the following diagram, spell-out applies at  $\pi$ , the root node.

(21) Single spell-out hypothesis



The multiple spell-out hypothesis was introduced as an alternative to the single spellout hypothesis (Uriagereka 1999; Chomsky 2001). In a multiple spell-out system, it is proposed that spell-out occurs once the derivation reaches certain designated syntactic nodes, such as CP,  $\nu$ P, and possibly DP (Chomsky 2001, 2005). Spell-out thus applies cyclically over the course of a derivation. This is represented abstractly in the following diagram, in which spell-out applies at the node  $\pi$ , as well as at the mid-derivational point  $\gamma$ . Spell-out does not apply at every branching node: for example, in (22), spell-out does not occur at the point  $\phi$ . Instead  $\delta$ , the terminal node of  $\phi$ , is not spelled-out until the higher spell-out node of  $\pi$  is reached.

(22) Multiple spell-out hypothesis



In the terminology of Chomsky (2001), spell-out is induced at distinct *phases* of the derivation. A node such as  $\gamma$  belongs to a phase separate from the mid-derivational point  $\phi$ .

Multiple spell-out entails that phonological operations have access to midderivational units formed by syntactic structure-building. By this view, PF and LF operations are limited by the same derivational units which constrain the syntax, because mid-derivational constructs created by the syntax are translated simultaneously to the PF and LF components (Chomsky 2001; Svenonius 2001, 2004). The local spell-out hypothesis takes this isomorphism one step further by imposing phonological-semantic integrity at each step of the derivation (Epstein et al. 1998).

Where the local spell-out and multiple spell-out hypotheses diverge is the stages at which spell-out to the interfaces obtains. In a multiple spell-out system, only certain syntactic nodes – such as the sisters to CP, vP, and possibly DP (Chomsky 2001) – are spell-out domains. With the local spell-out hypothesis, there is no such stipulation. Instead, PF and LF requirements of lexical elements are assessed aggressively over the course of the derivation: interpretation at the interfaces applies after each step of the syntactic derivation, rather than in larger derivational chunks.

A consequence of the local spell-out hypothesis is that derivational sisterhood is predicted to be a uniquely privileged relationship for relationships established at spell-out. Specifically, lexical requirements of elements must be met at the strictly minimal stage of the derivation in which only the element and its derivational sister are present. Let us take the linearization of affixes as an illustration of this prediction. By hypothesis, the linearization of affixes in Nuu-chah-nulth applies at spell-out, due to the PF requirement that linguistic outputs be temporally ordered. According to the local spell-out hypothesis, if an element  $\alpha$  takes  $\beta$  as its derivational sister, then  $\alpha$  is necessarily linearized at spell-out with respect to  $\beta$ . If  $\alpha$  is lexically specified as an affix, then for the tree in (23), spell-out of  $\gamma$  induces a linearization in which the affix  $\alpha$  is linearized with  $\beta$  as its host.

#### (23) Local spell-out of derivational sisters



According to the local spell-out hypothesis, elements which are not the derivational sister of  $\alpha$  are ineligible to act as the host of  $\alpha$ .<sup>5</sup> Thus, it is predicted that elements

 $<sup>^{5}</sup>$  In fact, the choice of host faces an additional restriction: it is only an element at the *initial edge* of the derivational sister which is eligible as a host. This restriction is established in Chapter 2.

such as  $\delta$  or  $\theta$  should not serve as the host for  $\alpha$ : these elements are not derivational sisters of  $\alpha$ . Only derivational sisters of affixes are present at the point of local spellout of the affix.

In contrast, for the single and multiple spell-out models, derivational sisterhood is not predicted to be a uniquely privileged relationship. Both the single and multiple spell-out models allow for the possibility that certain nodes are not spellout points. In (24), for example, spell-out applies at  $\phi$ , but it does not apply at  $\gamma$ .

#### (24) Delayed spell-out: single and multiple spell-out



If  $\alpha$  in (24) is an affix, then it is predicted that there are multiple elements which are available to serve as the host for the affix. All else being equal, if spell-out does not apply until  $\phi$ , then single and multiple spell-out models predict that  $\delta$  or  $\beta$  should be equally eligible as hosts for  $\alpha$ . Thus, for these models, the derivational sisterhood which  $\alpha$  shares with  $\beta$  does not guarantee that  $\alpha$  will be uniquely linearized relative to  $\beta$  at spell-out.

In the next section, I introduce the empirical ground on which the local spell-out hypothesis will be tested. Over the course of this book, it will be demonstrated that affixation in Nuu-chah-nulth is sensitive to derivational sisterhood. Affixal predicates in Nuu-chah-nulth find a host at spell-out which is chosen from the phrase with which they were syntactically merged. Moreover, there is critical evidence that this sensitivity to derivational sisterhood is not reducible to an independent effect such as directionality: in Nuu-chah-nulth,  $\beta$  is not chosen as the host for  $\alpha$  simply because it is right-adjacent to  $\alpha$ . Instead, I will argue that the Nuu-chah-nulth facts fall out elegantly from a model in which linearization consistently occurs between derivational sisters.

#### 1.3 Affixal Predicates in Nuu-chah-nulth

Affixal predicates in Nuu-chah-nulth participate in two types of bound/free dependencies. An affixal predicate  $(-\alpha)$  suffixes to a free  $(\beta)$  or bound  $(\beta-)$  host. In either case, the surface realization is  $\beta-\alpha$ . The place of these Nuu-chah-nulth linearization patterns within the typology of bound/free dependencies is indicated in (25). The Nuu-chah-nulth behaviour corresponds to (25e, f).

output: $\alpha - \beta$	output: $\beta - \alpha$
a. $\alpha$ , $-\beta$ ( $\alpha$ is free; $\beta$ is a suffix)	d. $\alpha$ , $\beta$ - ( $\alpha$ is free; $\beta$ is a prefix)
b. $\alpha$ -, $\beta$ ( $\alpha$ is a prefix; $\beta$ is free)	e. $-\alpha$ , $\beta$ ( $\alpha$ is a suffix; $\beta$ is free)
c. $\alpha$ -, - $\beta$ ( $\alpha$ is a prefix; $\beta$ is a	f. $-\alpha$ , $\beta$ - ( $\alpha$ is a suffix; $\beta$ is a
suffix)	prefix)

(25)	Typology	of bound/free	dependencies
< - /	21 02		

An illustration of the dependency of (25e) is supplied by the following examples. In (26a), the affixal predicate -siik "make" suffixes to the host 4učin "dress", a free noun. In (26b), the affixal predicate -mahsa "want to" suffixes to the host wat-sik "go home (PERF)", a verbal complex which is likewise a non-affixal host.

(26)	a.	nupititsa nupit-mit-sa once-PST-1SG.DEP I made a dress once.	⁺uč?insiik ⁺uč?in- <u>siik</u> dress- <u>make</u>

 b. wałśiλmaḥsak wał-šiλ-<u>maḥsa</u>-k go.home-PERF-<u>want.to</u>-2SG.Q Do you want to go home?

Descriptively, these suffixation patterns may be labelled as "noun incorporation", and "verb incorporation", respectively. In the first case, the affixal predicate suffixes to (or "incorporates") a noun, while in the second case, it suffixes to (or "incorporates") a verb. However, despite the difference in these descriptive labels of noun and verb incorporation, both types of incorporation share an identical phonological dependency: the affixal predicate suffixes to a free host.

In contrast, an example of the dependency of (25f) is given in (27). Here, the affixal predicate –'*iic* "consume" suffixes to a bound nominal host, *suuh*-"spring salmon". This pattern of suffixation to a noun may also be considered to be a type of noun incorporation.

 (27) suuwiicsiš suuh-'<u>iic</u>-siiš salmon-<u>consume</u>-1SG.IND I'm eating salmon.

Superficially, this pattern resembles (26a), since in each case, the affixal predicate has suffixed to a noun. However, (26a) and (27) differ in that in (26a), the nominal host *tuč2in* "dress" is free, while in (27), the nominal host *suu*<sub>4</sub>- "spring salmon" is bound.

Bound hosts in Nuu-chah-nulth, such as *suuḥ*- "spring salmon" in (26), belong to a closed-class set of nominals which have been referred to in the Nuu-chah-nulth literature as "combining forms" (Rose 1981: 287; Davidson 2002). These

bound nominals are often truncated versions of free-standing nominals in the language (Rose 1981). For example, the bound nominal *čapx*- "man" is a truncated allomorph of the free-standing nominal *čakup* "man". Other "combining forms" are equivalent to the free form minus its aspectual or "absolutive" affix (Rose 1981: 286-287). The bound nominal suuh-"spring salmon", for example, is related to the free form *suuh-aa* "spring salmon", which is inflected for continuative aspect -(y)aa(CONT). Another common pattern for bound nominals, in particular vowel- or nasalfinal forms, is the stem plus a final -q- (Rose 1981). An example of this type of alternation is *taana/taanaq*- "money", in which the bound form ends with a -q-. For the bound/free alternants of Nuu-chah-nulth, the bound form occurs if and only if it is suffixed to an affixal predicate. However, not all free nominals have a bound alternant. In fact, for the youngest generation of Nuu-chah-nulth speakers, free nominal forms are often preferred over bound variants in the case of alternations with truncated allomorphs.<sup>6</sup> For the remainder of this book, I set aside the issue of alternation of bound and free nominals, and focus instead on the properties of affixal predicates.

Affixal predicates in Nuu-chah-nulth do not show an allomorphic alternation with free predicates. Instead, free predicates constitute a distinct class, unrelated in form to affixal predicates.

#### (28) Free and bound classes of predicates in Nuu-chah-nulth



I label this class of free elements *independent predicates*. While affixal predicates require suffixation to a host, independent predicates never occur as suffixes. For example, although an affixal predicate such as -siik "make" incorporates a host, this option is unavailable to a non-affixal predicate such as ii-iik "sew (PERF)". In (29a), the affixal predicate -siik "make" suffixes to tucilin "dress". As shown in (29b), the independent predicate ii-cik "sew (PERF)" cannot suffix to this nominal.

(29) a. łuč?insiikitsiš łuč?in-<u>siik</u>-mit-siiš dress-<u>make</u>-PST-1SG.IND I made a dress.

<sup>&</sup>lt;sup>6</sup> As Nakayama (2000: 39) notes, this novel use of free nominal allomorphs is testament to the productive nature of Nuu-chah-nulth incorporation. If affixal predicates and their hosts constituted lexicalized chunks, then such novel forms would be unanticipated.

b. \* <sup>1</sup>uč?inničiλitsiš
<sup>1</sup>uč?in-ni-čiλ-mit-siiš
dress-sew-PERF-PST-1SG.IND
I sewed a dress.

Affixal predicates require a host which they may suffix to. The ungrammaticality of (30a) arises because the affixal predicate *-siik* "make" fails to find an appropriate host. Independent predicates, in contrast, do not occur as suffixes. In (30b), the independent predicate  $\dot{ni}$ - $\dot{ciik}$  "sew (PERF)" appears separately from the nominal  $tu\dot{c}iin2akqs$  "my dress".

(30)	a.	*	siikitsiš <u>siik</u> -mit-siiš <u>make</u> -PST-1SG.IND I made my dress.	↓uč?in?akqs ↓uč?in-?ak-qs dress-₽OSS-1SG.PS
	b.		ničiλitsiš ni-čiλ-mit-siiš sew-PERF-PST-1SG.IND I sewed my dress.	⁺uč?in?akqs ⁺uč?in-?ak-qs dress-POSS-1SG.PS

Thus, affixal and independent predicates can be seen to have distinct behaviours.

A similar distinction can be drawn between the affixal predicate -mahsa"want to" and the independent predicate *Sapaak* "willing to". As noted earlier, the affixal predicate -mahsa "want to" takes a verbal host, in a suffixation pattern I descriptively labelled "verb incorporation". This is shown in (31a), in which the affixal predicate suffixes to wat-sit" "go home (PERF)". This example may be contrasted with the one in (31b), which shows the independent predicate *Sapaak* "willing to". As an independent predicate, *Sapaak* "willing to" is incompatible with being linearized as a suffix. As shown in (31b), it is ungrammatical for the independent predicate to suffix to the verbal host wat-sit" "go home (PERF)".

- (31) a. watši紀maḥsak wat-ši紀-<u>maḥsa</u>-k go.home-PERF-<u>want.to</u>-2SG.Q Do you want to go home?
  - b. \* wałśiλſapaakk wał-šiλ-ſapaak-k go.home-PERF-willing-2SG.Q Are you willing to go home?

The inverse pattern is shown in (32), in which the verb wat-sit "go home (PERF)" follows the predicate, as a separate word. An affixal predicate is impossible in an environment such as (32), since this would entail that it would not be linearized as a suffix. The ungrammaticality of (32a) confirms that the affixal predicate -mahsa "want to" must appear as a suffix. In contrast, it is grammatical for the independent

predicate *fapaak* "willing to" to occur in this environment. As (32b) shows, *fapaak* "willing to" is not a suffix.

(32)	a.	*	maḥsak <u>maḥsa</u> -k <u>want.to</u> -2sg.q Do you want to	wałši <del>k</del> wał-šik go.home-PERF go home?
	b.		Sapaakk Sapaak-k willing-2SG.Q Are you willing	wałši⊁ wał-ši⊁ go.home-PERF to go home?

I hypothesize that the classes of affixal and independent predicates are lexically differentiated. In particular, I propose that affixal predicates in Nuu-chah-nulth are marked in the lexicon with an affixation requirement, [affix].

(33) [affix]: 
$$\alpha \neq \omega$$

This lexical specification states that the morpheme ( $\alpha$ ) is not equivalent to a phonological word ( $\omega$ ). The consequence of this requirement is that the affix requires a host with which it may form a phonological word. This lexical requirement must be met over the course of the derivation. Specifically, since this lexical requirement involves morpho-phonological instructions, this requirement must be met in the PF branch. Sample lexical entries for affixal and independent predicates in Nuu-chahnulth are given in (34–35), which indicate the properties of the predicates which are idiosyncratic to the lexical items.<sup>7</sup>

#### (34) Lexical entries for affixal predicates



<sup>&</sup>lt;sup>7</sup> This abstracts away from the families of features (formal, semantic, phonological) which lexical properties fall into.

b. *Sapaak* "willing to":



The status of affixal predicates in Nuu-chah-nulth as bound morphemes thus arises from lexical specification, rather than from phonological deficiency. As will be described in Chapter 6, affixal predicates often have enough phonological content to satisfy the minimal foot requirement of Nuu-chah-nulth words. Thus, affixal status in Nuu-chah-nulth is a lexically marked primitive, rather than a derived property which arises from consideration of prosody (cf. Anderson's 1992 "special" or "simple" clitics).

To foreshadow the content of the following chapters, we will see how the affixation requirement of Nuu-chah-nulth is met at spell-out, resulting in a type of suffixation I term *PF Incorporation*. This incorporation process is sensitive to linear adjacency. That is, an affixal predicate incorporates any element which abuts it, showing an insensitivity to syntactic constituency (e.g., the Coordinate Structure Constraint), as well as syntactic category. Yet, this condition on string adjacency will be shown to exist hand-in-hand with a "complement" effect with incorporation: an affixal predicate only incorporates an element from its syntactic complement, and not from projections which c-command the predicate. I will argue that the local spell-out hypothesis allows an elegant means of reconciling these dual sensitivities to string adjacency and syntactic configuration.

This book explicitly argues against a "hybrid" treatment of PF as a module operating both on hierarchically and linearly arranged constructs (contra Embick and Noyer 2001). Rather, I illustrate that the seemingly syntactic complement restriction on incorporation is a by-product of the derivation. By adopting the notion that syntax-to-PF mapping occurs at each instance of syntactic merge (Epstein et al. 1998), local spell-out is predicted to apply only to derivational sisters. This derives the complement effect as an epiphenomenal consequence of the mechanism of spell-out.

We now turn to a discussion of how this study of the linearization properties of affixal predicates in Nuu-chah-nulth fits in with existing research on the Wakashan language family.

#### 1.4 Research Context

Nuu-chah-nulth (*nuučaanuf*) is a member of the Southern branch of the Wakashan language family. This dialectally diverse language is the cultural heritage of a people whose occupation of the coastal territory of British Columbia's western Vancouver Island can be traced back archeologically at least 4,000 years (Dewhirst 1978). The language was previously referred to by the name *Nootka*, a misnomer which speakers of the language reject (see Davis and Wojdak 2007b).

The next subsections present an overview of genetic affiliations of Nuuchah-nulth, followed by a brief description of existing research on the language, and an overview of Nuu-chah-nulth word structure. The section concludes with discussion of the methodology for the present study.

#### 1.4.1 Genetic Affiliations

This section gives an overview of the relationships of Nuu-chah-nulth within the Wakashan language family. The family is spoken in western British Columbia, Canada and extends to the north-western tip of Washington State, USA. The family splits sharply into two divisions, Southern and Northern Wakashan, which have also been labelled the Nootkan and Kwakiutlan branches, respectively (Boas 1891; Rath 1974; Jacobsen 1979).

(36)	Wakashan family classification (Howe 2000)
	a. Northern (Kwakiutlan):
	Haisla, Heiltsuk, Ooweky'ala, Kwakw'ala (Kwakiutl)
	b. Southern (Nootkan):
	Nuu-chah-nulth, Ditidaht (Nitinat), Makah

The time depth separating the two branches of Wakashan is great, and has been estimated at approximately 29 centuries (Swadesh 1953: 26, 41; Jacobsen 1979c: 769). The Northern and Southern branches share relatively few cognate roots, but Sapir (1911) notes that there are at least about 90 lexically contentful suffixes in Nuu-chah-nulth and Kwakw'ala which have a common etymological source. The origins of Nuu-chah-nulth affixal predicates may therefore be plausibly traced to Proto-Wakashan.

The Southern branch of the Wakashan language family is comprised of the closely related languages Nuu-chah-nulth, Ditidaht and Makah. Nuu-chah-nulth shares strong grammatical and lexical ties to Ditidaht and Makah (Davidson 2002), although pervasive phonological changes separate these sister languages (see among others Jacobsen 1969a, b, 1979, 2007; Haas 1969). The languages of Southern Wakashan - like all other First Nations languages of the Pacific Northwest - face the threat of extinction. Contact with Europeans, dating to the late 18th century, had a profound impact on the indigenous societies of the Pacific Northwest, as introduced diseases decimated the local populations (Arima 1983; Boyd 1999), while colonization disrupted the traditional socio-economic order. Language loss in Nuuchah-nulth communities was hastened in the 20th century by the imposition of a compulsory residential schooling system which had the effect of removing young people from their families (Golla 1987; Behrend and Kammler 2003). Today, there are estimated to be fewer than 200 remaining speakers of Nuu-chah-nulth, most of whom are elderly. Ditidaht has fewer than ten fluent speakers remaining, while Makah is no longer spoken as a first language (Adam Werle, p.c. 2005). Language revitalization initiatives are underway in the communities, but face considerable challenges (see Behrend and Kammler 2003). Recent language maintenance successes for Nuu-chah-nulth include a pocket dictionary developed by the Barkley Sound Dialect Working Group (2004), and texts with accompanying audio recordings of the late Ahousaht elders Caroline Little (Nakayama 2003a) and George Louie (Nakayama 2003b).

Prior to contact with Europeans there was no self-designation for the Nuuchah-nulth community at large. Component groups had largely autonomous identities (Drucker 1951; Arima 1983), which are reflected today in the 15 or so distinct dialects of the language. The dialects of Nuu-chah-nulth may be broadly categorized into northern, central, and southern varieties, following Stonham (2004).

- (37) Subclassification of Nuu-chah-nulth
  - a. Northern Nuu-chah-nulth:
    - Chicklisaht, Kyuquot, Ehattesaht, Nuchatlaht, Mowachaht, Muchalaht
  - b. Central Nuu-chah-nulth:
    - Hesquiaht, Ahousaht, Tla-o-qui-aht
  - c. Southern Nuu-chah-nulth:
    - Ucluelet, Toquaht, Tseshaht, Huu-ay-aht, Uchucklesit, Hupachasath

These dialects are spoken along western Vancouver Island from Cape Cook in the north to Barkley Sound in the south. The mountainous terrain of western Vancouver Island, with its punctuated coastline of inlets and small islands, played a key role in the differentiation of the socio-political and linguistic identities of the inhabitants of the area (Drucker 1951). The data presented in this study come from original field-work on Ahousaht, a central Nuu-chah-nulth dialect spoken on Flores Island, off the west coast of Vancouver Island.

#### 1.4.2 Previous Literature

The linguistic study of Nuu-chah-nulth has been spurred by three major waves of fieldwork activity, conducted over the course of the last century. The roots of Nuuchah-nulth linguistics are in the work of Edward Sapir, who engaged in intensive fieldwork on the language in 1910 and 1913-1914. Working together with Nuuchah-nulth speaker Alex Thomas and Sapir's one-time student Morris Swadesh, Sapir oversaw the collection of extensive textual materials, which were published in part as Nootka Texts (Sapir and Swadesh 1939). This project yielded a concise dictionary of the southern Nuu-chah-nulth dialect Tseshaht and a brief but foundational grammatical description of this dialect, published together as an appendix to *Nootka* Texts. A second volume of texts was published as Sapir and Swadesh (1955), while additional linguistic notes appeared in Sapir (1911b, 1924, 1929), and Swadesh (1939, 1948). The material collected during the Sapir-Thomas collaboration forms the empirical basis for much subsequent analytic work on the language, including Stonham (1999, 2004), and the Nuu-chah-nulth component of Davidson (2002). A dictionary of the Tseshaht dialect has also been published (Stonham 2005), based on Sapir's fieldnotes. Textual materials which were residual to the original two published volumes of the Sapir-Thomas texts (Sapir and Swadesh 1939, 1955) were recently brought to light with the appearance of Sapir et al. (2000, 2004). The final two installments of the Sapir-Thomas series are due to be published in the near future (Eugene Arima and Terry Klokeid, p.c.).

After a lull in linguistic fieldwork on Southern Wakashan stretching from the 1920s into the 1960s, a second wave of activity ensued with research expanding to Ditidaht (e.g., Haas 1969, 1972; Klokeid 1976, 1978a, b) and Makah (e.g., Jacobsen 1969, 1979; Renker 1987), as well as a subset of Nuu-chah-nulth dialects

(Paik 1968; Rose and Carlson 1984; Kess and Kess 1986, among others). This period saw the completion of the first grammar of Nuu-chah-nulth, based on the northern dialect Kyuquot (Rose 1981).

Recent years have seen a renewed period of concerted fieldwork on Nuuchah-nulth, largely in response to the advancing age of the last fluent speakers of the Southern Wakashan languages. This ongoing wave of activity has produced the following recent doctoral dissertations on Southern Wakashan: a treatment of grammatical properties of the Ahousaht dialect of Nuu-chah-nulth (Nakayama 1997), a study of this dialect's morpho-phonology (Kim 2003), a comparison of Makah grammar to that of the Tseshaht dialect of Nuu-chah-nulth (Davidson 2002), and an earlier version of the present study (Wojdak 2005). Other publications include Nakayama (1998, 2001), Davis and Sawai (2001) and Wojdak (2001). This renewal of research activity is marked with the publication of a special edition of the *Canadian Journal of Linguistics* devoted to Southern Wakashan (Davis and Wojdak 2007a).

The topic of affixal predicates – or "lexical suffixes", as they have frequently been labelled – is a recurrent theme in research on Nuu-chah-nulth. The role of these morphemes in Nuu-chah-nulth was initially highlighted in Sapir's (1921) discussion of Nuu-chah-nulth polysynthesis, as well as in Swadesh's (1939) article entitled Nootka Internal Syntax. Inventories of these predicates are found in Sapir and Swadesh (1939), Rose (1981), and Davidson (2002), while detailed descriptive treatments of their suffixation patterns are discussed in Rose (1981) and Nakayama (1997, 1998, 2001). To my knowledge, the first reference to the word formation properties of Southern Wakashan affixal predicates as a type of incorporation is found in Klokeid (1976). A noun incorporation analysis in terms of syntactic head movement is developed in Woo (2000), Davis and Sawai (2001), Yiu (2001), and Stonham (2004). An alternative to the noun incorporation analysis is presented by Waldie (2004), who analyses the suffixation as a type of denominal verb formation. While suffixation to nominals ("noun incorporation") has been the primary focus of this previous literature, suffixation to verbs ("verb incorporation") has been relatively underinvestigated. Cursory treatments are found in Rose (1981) and Nakayama (1997, 1998, 2001). The topic of Nuu-chah-nulth verb incorporation plays a central role in the present study.

### 1.4.3 Overview of Word Structure in Nuu-chah-nulth

Affixal predicates fall within an inner layer of suffixes in the Nuu-chah-nulth word (Rose 1981; Nakayama 2001; Davidson 2002). The following is a simplified description of suffixal bound morphemes, adapted from Davidson (2002: 321):

(38) *"Inner" and "outer" layers of suffixes in Nuu-chah-nulth* 

[=PREDICATE=ASP] =TR=TEMP=PAS=POSS=TENSE=AGR/MOOD=PL=AGAIN=HAB

"core" suffixes

"peripheral" suffixes

In addition to affixal predicates, the core inner layer also includes aspectual suffixes, which are marked only for those predicates with no "inherent" aspectual value (Davidson 2002: 97–99). The examples in (39) illustrate two aspectual suffixes, the perfective  $-\dot{sir}$  (PERF) and the continuative -aa (CONT), shown suffixed to the independent predicate  $\dot{mir}$  "rain". Following the aspectual suffixes are markers for tense and pronominal agreement.

- (39) a. miiえšiえit?iš miえ-šiえ-mit-?iiš rain-PERF-PST-3.IND It rained.
  - b. miiλaamit?iš miiλ-aa-mit-?iiš rain-CONT-PST-3.IND It was raining.

As will be described in Chapter 5, only the inner layer of suffixes is included together with a verb when the verb is "incorporated" into an affixal predicate. In (40a), the aspectually marked verb  $mi\lambda$ -si $\lambda$  "rain (PERF)" is suffixed by the affixal predicate -*ii*-ci $\lambda$  "hear (PERF)". As is demonstrated in (40b), it is not possible for an outer suffix such as past tense -*mit* (PST) to be marked on the "incorporated" verb.

- (40) a. miiえšiえ?iičiえitsiš miえ-šiえ-<u>?ii</u>-čiえ-mit-siiš rain-PERF-<u>hear</u>-PERF-PST-1SG.IND I heard it rained.
  - b. \* miikšikit?iičikitsiš miik-šik-mit-?ii-čik-mit-siiš rain-PERF-PAST-<u>hear</u>-PERF-PST-1SG.IND I heard it rained.

Outside of affixal predicates and aspectual markers lies a strictly ordered clitic sequence which includes argument structure affecting enclitics such as the passive and possessive, as well as markers for tense (Rose 1981; Nakayama 2001; Davidson 2002). Pronominal agreement enclitics fall towards the outer limits of the clitic sequence and simultaneously encode clause-typing "mood" distinctions. In (41), the affixal predicate -mahsa "want" is followed by two valency-affecting morphemes, the transitivizer - 'ap (TR) and the passive – 'at (PAS). The past tense morpheme -mit (PST) is next in the string, followed by the fusional mood/agreement marker -?iiš (3.IND), which marks a third person subject and "indicative" mood.

 (41) kukċuqšmaḥsapanit?iš kukċuqš-maḥsa-'ap-'at-mit-?iiš spoon.feed-want-TR-PAS-PST-3.IND S/he wants to be fed with a spoon.

The placement of inflectional clitics is addressed in Chapter 3. According to the analysis, inflectional clitics owe their linearization to the same local spell-out mechanism which induces the linearization of affixal predicates.

In addition to suffixation, Nuu-chah-nulth word-formation strategies include word-initial reduplication (see §2.1), and restricted cases of infixation (see Stonham 2004: 139–142).

#### 1.4.4 Methodology

This study is based on fieldwork conducted during 2000–2005 with fluent speakers of Nuu-chah-nulth. Sentences and short stretches of discourse were elicited in a series of person-to-person interviews. This elicitation method is crucial to syntactic research as it allows for the targeted study of phenomena which may be rare in texts. Moreover, syntactic elicitations permit access to native speaker intuitions about grammaticality, which are otherwise inaccessible in the format of textual analysis. The data obtained in elicitation sessions were transcribed and subsequently inputted to a computer database. Some recordings were also made to supplement the transcriptions. Transcriptions were proofread by Mary Jane Dick, a fluent speaker of the Ahousaht dialect of Nuu-chah-nulth who has formal training in linguistics.

The majority of the data in this book has its genesis in a project I began in 2002 with Mary Jane Dick to document Ahousaht usages of the "lexical suffixes" catalogued in Sapir and Swadesh (1939), Rose (1981), and Davidson (2002). Mary Jane Dick worked independently and with her mother, Sarah Webster, to create illustrative example sentences for more than 200 suffixes. These example sentences were an invaluable resource, and provided a large data-set which formed the basis for follow-up elicitation sessions during 2002–2005.

The two primary language consultants for this project are Mary Jane Dick and her mother, Sarah Webster, who are each native speakers of the Ahousaht dialect. Both were born in Ahousat, British Columbia – Sarah in 1924, and Mary Jane in 1945. Nuu-chah-nulth was the dominant household language while both were young, and they each began to learn English as a second language when they started schooling. Currently, the two usually see each other daily and use their language as much as possible with each other.

This study benefited from elicitations with several secondary language consultants, who are speakers of Ahousaht or other Nuu-chah-nulth dialects. Katherine Fraser is a speaker of the Ahousaht dialect and has formal training in linguistic analysis. The late Caroline Little was also a speaker of the Ahousaht dialect. Barbara Touchie is a speaker of the Ucluelet dialect, while her brother, Archie Thompson self-identifies as a speaker of the Toquaht dialect. Josephine Thompson is a speaker of the Ahousaht dialect. Barney Williams, Sr., identifies with both the Tla-o-qui-aht and Chickliset dialects, while his son, Barney Williams, Jr., speaks the Tla-o-qui-aht dialect. Each of these language consultants speaks Nuu-chah-nulth as their first language, and English as their second. Sessions with secondary language consultants were used to confirm various patterns indicated by primary language consultants.

#### 1.5 Outline of the Book

There are five chapters which follow this introduction. This section gives an overview of their contents.

Chapter 2 argues that local spell-out gives rise in Nuu-chah-nulth to a specific type of dependency termed *PF incorporation*. I argue that mid-derivational units constructed in the syntax form minimal spell-out domains for the post-syntactic linearization mechanism for affixal predicates. An affixal predicate is linearized as a suffix to the leftmost element in the string which is linearly adjacent to it at spell-out, a condition I label the *string adjacency effect*. The linearization shows an insensitivity to syntactic category and an absence of LF effects. However, opacity effects come into play in that an affixal predicate cannot incorporate a host taken from a DP or CP, two units which I analyse as *saturated domains*. Instead, when an affixal predicate has a DP or CP as its complement, an expletive host surfaces to rescue the potentially stranded affix.

Chapter 3 presents in detail the Nuu-chah-nulth clausal architecture which underlies PF incorporation. I argue that Nuu-chah-nulth has a syntactically underived VOS word order, with arguments introduced within the verbal projections. Evidence for a structural asymmetry between subject and object comes several sources, including possessor raising and Weak Crossover effects. With respect to the linear ordering, I present the argument that linearization of syntactic terminals is achieved exclusively at PF, as syntactic constructs are not inherently ordered for directionality. Furthermore, I argue that syntactic c-command does not unambiguously determine linear precedence (contra Kayne 1994). Instead, directionality is imposed at the point of spell-out via choice of one of two logically possible linearizations. Regularities in directionality (i.e., left or right syntactic "headedness") are proposed to be spell-out conventions which arise through the process of language acquisition.

Chapter 4 gives an inventory of the argument structures of affixal predicates which take nominal complements. The structure-building operation of Merge generates the nominal arguments of affixal predicates in a range of orientations, from unaccusative to extended unaccusative, transitive to ditransitive. Unergatives, however, are absent from the inventory of affixal predicates in Nuu-chah-nulth – a systematic gap which I attribute to the need for an affixal predicate to have a phonologically distinct complement. For each type of affixal predicate, the predicate incorporates a host from its complement, often giving rise to noun incorporation. I present a range of independent diagnostics for syntactic structure, including possessor raising and subject agreement.

Chapter 5 surveys types of affixal predicates which take verbal complements. I propose that these predicates fall into two general classes: affixal main predicates and affixal auxiliary predicates, both of which permit verb incorporation. The PF incorporation pattern of these predicates may result in incorporation in contexts in which the complement is a reduced verbal projection (vP), rather than an opaque, fully inflected clause (CP). Given that these PF incorporation contexts are monoclausal, a range of "restructuring" effects are predicted, including "long" wh-movement and "long" possessor raising.

Chapter 6 contains concluding remarks. A brief summary of the book is given, followed by discussion of theoretical and typological implications of the analysis. On the theoretical side, I review evidence for the strictly interpretive function of spell-out, contrasting the local spell-out model to "phasal" spell-out (Chomsky 2001, 2005). Under the view that derivational opacity associated with DP and CP domains is tied to saturation, Nuu-chah-nulth presents evidence for variation in the permeability of these domains - specifically, against the universality of "escape hatches" for long-range processes. Turning to cross-linguistic variation in affixation, this chapter discusses affixation mechanisms other than local spell-out. First, affixation arising from local spell-out is contrasted with the more complex process of syntactic head movement. Second, evidence is presented for "inside-out" affixation dependencies - the antithesis of Nuu-chah-nulth affixation - which arise at late stages of the derivation, via prosody. Here, the Nuu-chah-nulth affixation pattern is contrasted with that of the Northern Wakashan language Kwakw'ala. Additional typological factors are also considered. I state that there is a range of variation attested in noun incorporation dependencies cross-linguistically. Furthermore, the "lexical suffixes" of the Pacific Northwest Sprachbund do not, as has been previously assumed, reduce to a single type of "bound nominal" pattern (Gerdts 1998).

We now turn to the chapter on PF Incorporation.
# 2. PF Incorporation

So I find words I never thought to speak In streets I never thought I should revisit ~T.S. Eliot Four Quartets no. 4

# 2.0 Introduction

In Chapter 1, the observation was made that a syntactic construct, formed derivationally by binary concatenation, is not inherently ordered. In the syntax, the elements joined by Merge are an unordered set  $\{\alpha, \beta\}$ .

(1)



Before such a derivational output can be pronounced, however, the elements occupying the terminal positions  $\alpha$  and  $\beta$  must be linearized. I hypothesize that it is strictly non-syntactic principles in the grammar which are responsible for ordering them. In Chapter 1, I proposed that affixation requirements constitute one means by which the linguistic elements  $\alpha$  and  $\beta$  may be ordered. If one or the other of  $\alpha$  and  $\beta$  is an affix, a restricted set of linearization patterns arises: either  $\alpha$ - $\beta$  or  $\beta$ - $\alpha$ . These orderings emerge in the syntax-to-PF mapping, via local spell-out. Repeated from Chapter 1, this claim may be defined as the following:

# (2) Local spell-out: for Merge $(\alpha, \beta)$ , interpret $(\alpha, \beta)$

The need for an affix to be linearized with respect to its host is a bare output condition on PF – the portion of the grammar sensitive to temporal sequencing.

The Wakashan language Nuu-chah-nulth was introduced as the empirical ground on which the local spell-out hypothesis will be assessed. This chapter develops the notion that spell-out induces in Nuu-chah-nulth a particular arrangement which I refer to as *PF incorporation*. In this linearization, an affixal predicate  $-\alpha$  suffixes to a host  $\beta(-)$ , yielding an ordered pair of  $\beta-\alpha$ . The affixal predicate "incorporates" its host in order to achieve a pronounceable form, that of a linearized affix. In (3), this pattern is exemplified by the string k'waaq-caaqa "busy with spawned herring eggs", which is comprised of the affixal predicate –*caaqa* "busy with" and its nominal host k'waq "spawned herring eggs".

 (3) kwaaqcaaqa?iš Saahuus?ath kwaq-caaqa[+L]-?iiš Saahuus-'ath s.h.eggs-busy.with-3.IND place.name-from The Ahousahts are busy with spawned herring eggs. According to my analysis, the string  $k^{iwaaq}$ -caaqa emerges as a reflex of the need to linearize the affixal predicate –caaqa "busy with". The host chosen for the affix is its derivational sister, the nominal  $k^{iwaq}$  "spawned herring eggs".

(4)

-caaqa  $k^{waq}$ busy with s.h.eggs

The resulting dependency is a case of PF incorporation.

The term "incorporation" has a lengthy history in Amerindian linguistics. In the context of what has been referred to as "noun incorporation", this label applies to instances in which a noun and verb are combined into a single word. Over the last century, a series of high-profile debates have occurred regarding this phenomenon. Kroeber (1909, 1911) and Sapir (1911a) capitalize on the free-bound contrast to make a distinction between noun incorporation languages in which the verb is a free stem, and those "verbalizing suffix" languages in which the verb is bound. This notion resurfaces in an exchange between Mithun (1984, 1986) and Sadock (1980, 1986). My contribution to this discussion is to specify a four-way range of affixation relationships which exist in the typology of noun–verb dependencies. If the noun and verb are each either free or bound, four patterns of dependencies are available: a bound noun may affix to a bound verb or to a free verb; a free noun, in turn, may affix to a bound verb or to a free verb. Thus, the affixation dependencies of "noun incorporation" are not a uniform phenomenon (see Chapter 6 for further discussion).<sup>1</sup>

	Bound noun	Free noun
Bound verb	(a) N–V	(b) N–V
	(Nuu-chah-nulth)	(Nuu-chah-nulth)
Free verb	(c) N–V	(d) N–V

## (5) Affixation dependencies of noun incorporation

Nuu-chah-nulth instantiates two of these four options. A sequence of N–V in Nuu-chah-nulth arises only via the presence of a bound verb, but there is variation in whether the noun is bound (5a) or free (5b). (Chapter 6 addresses the remaining typological options.) Thus, in a Nuu-chah-nulth complex denoted by  $\beta-\alpha$ , the bound verb  $-\alpha$  takes a bound or free nominal  $\beta(-)$  as its host. Because the affixal predicate is obligatorily bound in Nuu-chah-nulth, it can never go without a host. The example

(6)	a.	maḥťa?amit?iš	čakup
		maḥťa- <b>'aap</b> -mit-?iiš	čakup
		house- <u>buy</u> -PST-3.IND	man
		A man bought a house.	

<sup>&</sup>lt;sup>1</sup> Baker (1988: 143) reaches a similar conclusion, although he proposes a three-way morphological distinction: root noun and root verb, root noun and affixal verb, and affixal noun and root verb. Baker does not discuss the possibility of an affixal noun attaching to an affixal verb.

b.	*	?aamit?iš	mahtii	čakup
		<u>'aap-</u> mit-?iiš	maḥťii	čakup
		<u>buy</u> -pst-3.ind	house	man
		A man bought a	house.	

in (6a) indicates a grammatical instance of the affixal predicate *-?aap* "buy" suffixing to the bound nominal *mahta-* "house". In (6b), even though the free form of "house", *mahtii*, is used, it is not possible for the noun and verb to be separated.

The example in (6b) is ruled out because the affixal predicate *-?aap* "buy" must be linearized as a suffix.

A purely syntactic account of incorporation does not capture the significance of the bound or free status of the verb in affixation contexts.<sup>2</sup> In languages in which the "incorporating verb" is obligatorily bound, two logical possibilities emerge for the host for the verb, according to the local spell-out hypothesis. As previously noted, if  $\alpha$  in (7) is a bound verb, then  $\beta$  may take on the role of host for  $\alpha$  at spell-out.



There is a logically possible alternative, however; an expletive host may be inserted at the point of spell-out to act as "dummy" placeholder for the bound verb. This predicted pattern is attested in Nuu-chah-nulth, which utilizes an expletive host, 2u-, in contexts in which the affixal predicate has not incorporated a host.<sup>3</sup> An example is given below, in which 2u- acts as a host for the affixal predicate -2aap "buy".

(8)	a.	?u?aamit?iš ?u- <u>'<b>aap-</b></u> mit-?iiš Ø- <u>buy</u> -PST-3.IND A man bought a house.	maḥťii maḥťii house	čakup čakup man
	b.	maḥťaʔamitʔiš maḥťa- <b>'aap</b> -mit-ʔiiš house- <u>buy</u> -PST-3.IND A man bought a house.	čakup čakup man	

In (8a), " $\lambda$ -support" occurs as an alternative to the noun incorporation of (8b). This expletive is also employed in cases in which the affixal predicate takes a sentential complement. In (9a), for example, the expletive  $\lambda$ - appears as a host for the affixal predicate -cuk "necessary". In (9a), the predicate takes the conditional complement

 $<sup>^{2}</sup>$  Baker (1988: 72) argues that it is a morphological component of the grammar which determines whether (syntactic) incorporation is obligatory in some cases, or forbidden in others. This is ascribed to a filtering effect of the morphology.

<sup>&</sup>lt;sup>3</sup> In the framework of Distributed Morphology,  $\partial t$ - in Nuu-chah-nulth qualifies as a "dissociated" morpheme – that is, one that is inserted at the point of spell-out (Embick 1997; Noyer and Embick 2001).

 $\dot{c}uk^{w}i\partial atquu \ qaawic\partial i$  "that the potatoes be washed". For the predicate -cuk "necessary", the presence of  $\partial u$ - alternates with the incorporation strategy. In (9b), the affixal predicate -cuk "necessary" suffixes to the verbal host,  $\dot{c}u$ - $k^{w}i\partial$  "wash (PERF)", rather than to the expletive  $\partial u$ .

(9)	a.	?ucuk?iš	ċuk <sup>w</sup> i?atquu	qaawic?i
		?u- <u>cuk</u> -?iiš	ċu-k <sup>w</sup> i <del>λ</del> -'at-quu	qaawic-?ii
		Ø- <u>need</u> -3.ind	wash-PERF-PAS-3.COND	potato-DET
		It is best to wash potatoes be wash	h the potatoes. ( <i>lit</i> : "it is ne hed")	cessary that the
	b.	ċuk <sup>w</sup> i <del>λ</del> cuk?iš	qaawic	?i
		ću-k <sup>w</sup> i≁- <b>cuk</b> -?iiš	qaawic	-?ii
		wash-PERF- <u>need</u>	<u>l</u> -3.IND potatoe	es-DET
		The potatoes ne	ed washing.	

The expletive  $\lambda u$ - in Nuu-chah-nulth is regularly used in the citation forms of affixal predicates, as in  $\lambda u$ - $\lambda u$  "buy" and  $\lambda u$ -cuk "need". From this point onwards, I adopt the convention of writing affixal predicates in their citation forms when I mention them.

Greenlandic (Eskimo) provides another example of this expletive pattern. In the Greenlandic language, "incorporating verbs" are suffixes, just as in Nuu-chahnulth (Waldie 2004). There is a "placeholder" morpheme, *pi*-, in Greenlandic which surfaces in contexts in which no incorporation occurs (Sadock 1980). The following examples are adapted from Sadock (1980: 306, ex. 18a and 307, ex. 24).

(10) *Greenlandic* 

a.	Qimme-qar-poq dog-have-3SG.IND He has a dog.		
b.	Qimmi-mik dog-INST He has a dog.	pe-qar-poq Ø-have-3SG.IND	

In (10a), incorporation unites the nominal *qimme* "dog" with the suffixal verb -qar "have". In (10b), in contrast, no incorporation of the nominal occurs, and instead, the bound verb is attached to the empty form pi- (surfacing as pe-).

Unlike languages with bound verbs, languages with free verbs make no use of an expletive host for a verb in contexts in which incorporation fails to apply. An example of a noun-incorporating language with free verbs is Mohawk (Iroquoian). The incorporation pattern of Mohawk is indicated in (11a), in which the nominal *–nuhs-* "house" is incorporated into the inflected verb *ye-nuhwe'-s* "like 3FS/3N". Note that in Mohawk the verb can surface detached from the noun, as in (11b).

#### PF INCORPORATION

a.	Yao-wir-a'a	ye-nuhs-nuhwe'-s
	PRE-baby-SUF	3FS/3N-house-like-ASP
	The baby house	-likes.

b. Yao-wir-a'a ye-nuhwe'-s ne ka-nuhs-a' PRE-baby-SUF 3FS/3N-like-ASP DET PRE-house-SUF The baby likes the house.

When incorporation of the noun into the verb does not occur, there is no process similar to 2u-support for the Mohawk verb. The inflected verb ye-nuhwe'-s "like" in (11b) does not receive an expletive host. This difference symbolizes a key contrast between the Nuu-chah-nulth and Mohawk patterns of incorporation. The contrasting patterns fall out from an analysis which takes into account the underlying bound/free status of the verb in affixation contexts.

To summarize, we have seen two ways in which the affixation requirement of an affixal predicate may be met in Nuu-chah-nulth. On the one hand, local spellout may attach an affixal predicate to an incorporated host, yielding PF incorporation. On the other hand, the expletive element 2u-may be introduced at spell-out as a host. Under this view, 2u-insertion receives an analysis similar to that which Lasnik (1981, 2000) proposes for *do*-support in English: the "dummy" *do* is inserted to meet the requirements of a potentially stranded affix. Over the following chapters, we will return to the discussion of *u*-support as a spell-out solution which applies in cases in which an "edge" separates the affixal predicate from a potential incorporable host. In Nuu-chah-nulth, the use of an expletive host for an affixal predicate is obligatory when the predicate takes a DP or CP complement: incorporation of a host cannot occur across these opaque domains. For example, the use of the expletive is necessary when the object of an affixal predicate is marked with a determiner. The example in (12a) shows a grammatical instance of incorporation when the nominal host 4uč2in "dress" is bare. Incorporation is impossible when the determiner -2iiappears, either in (12b) as the host for the affixal predicate, or in (12c) when suffixed to the nominal. The example in (12d) shows the mandatory  $\lambda$ -support which occurs when the nominal is marked with the determiner.

- (12) a. łuč?insiikitsiš łuč?in-<u>siik</u>-mit-siiš dress-<u>make</u>-PST-1SG.IND I made a dress.
  - b. \* łuč?in?isiikitsiš łuč?in-?ii-<u>siik</u>-mit-siiš dress-DET-<u>make</u>-PST-1SG.IND I made the dress.

c.	*	?iisiikmitsiš	łuč?in
		?ii- <u>siik</u> -mit-siiš	łuč?in
		DET- <u>make</u> -PST-1SG.IND	dress
		I made the dress.	
d.		?usiikitsiš	łuč?in?i
		?u- <u>siik</u> -mit-siš	łuč?in-?ii
		Ø- <u>make</u> -PST-1SG.IND	dress-DET
		I made the dress.	

We return to the topic of opacity effects in PF incorporation in §2.5.

For either expletive insertion or incorporation of a host, the satisfaction of the affixation requirement in Nuu-chah-nulth constitutes a bare output operation on PF. An inserted or incorporated host allows the bound predicate to meet its linearization requirement. Bound status and linearization are not relevant to the syntax proper; instead, these are conditions on phonological representation.

We now turn to discussion of the trademark properties of PF incorporation in Nuu-chah-nulth which serve to corroborate the claim that this phenomenon is derived post-syntactically, at the point of spell-out. In §2.1, independent evidence is presented in support of the claim that an affixal predicate and its host, whether incorporated or expletive, share an intimate phonological relationship. This is followed in §2.2 with discussion of the observation that incorporation in Nuu-chah-nulth is sensitive to string adjacency, rather than to syntactic c-command. In turn, §2.3 argues that incorporation in Nuu-chah-nulth displays an insensitivity to syntactic category, in the sense that the hosts of incorporation come from a range of lexical and functional categories. Evidence is presented in §2.4 that incorporation in Nuu-chah-nulth shows an absence of LF effects. In §2.5, I illustrate that incorporation in Nuu-chah-nulth, although otherwise insensitive to syntactic and semantic properties, cannot cross a DP or CP. I attribute this opacity effect to the status of DPs and CPs as saturated domains (cf. the notion of "phase" in Chomsky 2001). The chapter concludes with §2.6, which sums up evidence against a strictly syntactic view of Nuu-chah-nulth affixation.

## 2.1 Morpho-phonological Dependency

The analysis which I am proposing states than an affixal predicate is united with a host at spell-out so that it may be linearized. According to the local spell-out hypothesis, an affixal predicate  $-\alpha$  is spelled-out with its host  $\beta(-)$ , inducing a  $\beta-\alpha$  linearization.

# (13) Local spell-out: for Merge $(\alpha, \beta)$ , interpret $(\alpha, \beta)$

This section presents independent evidence for a phonological dependency between  $\alpha$  and  $\beta$ . This evidence comes from the morpho-phonological "subcategorization" of affixal predicates.

In Nuu-chah-nulth, a striking property of bound morphemes is their ability to prosodically condition their hosts (Sapir and Swadesh 1939; Rose 1981; Davidson 2002; Kim and Wojdak 2002; Kim 2003). For example, the repetitive iterative

#### PF INCORPORATION

suffix -(y)a [+R+L] (ITER) causes vowel lengthening and reduplication of the first syllable of a monosyllabic root, as well as vowel lengthening of both the base and the reduplicant (Sapir and Swadesh 1939; Wojdak 2002; Kim and Wojdak 2002).

(14)	a.	čuusčuusa	b.	kii⊁kii⊁a
		ċus-a[+R+L]		ki <del>≈</del> -a[+R+L]
		dig-ITER		break-ITER
		'digging continuously'		'breaking continuously'

Affixal predicates share this ability to prosodically condition their hosts, while independent predicates never induce prosodic conditioning. Each affixal predicate is associated with a characteristic pattern, although many affixal predicates are "neutral" in that they do not impose changes on their host. A given affixal predicate may induce reduplication, a long vowel, a short vowel, or some combination of the three. (I refer the reader to Kim and Wojdak 2002 and Kim 2003 for a thorough description of which combinations are available.)

Prosodic conditioning		
a.	[+R]	reduplication
b.	[+L]	long vowel
c.	[+S]	short vowel
	Prose a. b. c.	Prosodic conditioning a. [+R] b. [+L] c. [+S]

Both the expletive morpheme  $\partial \mu$  and incorporated hosts are affected by the prosodic requirements of affixal predicates.

The examples below illustrate the behaviour of the affixal predicate  $\lambda uu-h\dot{w}a+$  "use", which triggers vowel lengthening of the first syllable of its host. In (16a), the vowel of the expletive morpheme  $\lambda u$ - is lengthened to  $\lambda uu$ -, while in (16b) the first vowel of *yaxyak* "broom" is lengthened to *yaaxyak*.

(16) Long initial vowel (+L)

a.	?uuḥwa†?i	yaxyak
	?u- <b>ḥẃa1</b> [+L]-'ii	yaxyak
	Ø- <u>use</u> -2sg.imp>30bj	broom
	Use a broom!	

 b. yaaxýakhwał?i yaxýak-hwał[+L]-'ii
 broom-use-2SG.IMP>30BJ Use a broom!

The following example shows how reduplication is triggered by the affixal predicate 2u/2u-q "travel with (in a vessel)". In (17a), the expletive morpheme 2u-surfaces as 2u/2u-, while in (17b), the morpheme 2uus "someone" appears as 2uu/2uus. In this pattern, the first consonant and vowel of the host are reduplicated. The vowel length of the reduplicant is determined by the underlying vowel length of the morpheme which serves as the base.

#### THE LINEARIZATION OF AFFIXES

#### (17)Reduplication with neutral vowel length (+R)

a.	?u?uq?iš	Louis	huupuuk <sup>w</sup> asuk?i	Robin
	?u- <b>g</b> [+R]-?iiš	Louis	huupuuk <sup>w</sup> as-uk-?i	Robin
	Ø- <u>travel.with</u> -3.IND	Louis	car-POSS-3.PS	Robin
	Louis is travelling in	Robin's	s car.	
	Louis is travelling in	Robin's	s car.	

Louis b. ?uu?uuša?iš ?uuš-a[+R]-?iiš Louis someone-travel.with-3.IND Louis Louis is travelling with someone (e.g., in his canoe).

Some affixal predicates impose restrictions on vowel length as well as inducing reduplication. For example, the affixal predicate *2u2uu-sapi* "depend on" requires reduplication, as well as a short vowel in the reduplicant and a long vowel in the initial syllable of the base. In the examples below, the expletive surfaces as 2u2uu-(18a), while *2uuš* "someone" surfaces as *2u2uuš* (18b).

- (18)Reduplication with short initial vowel and long second vowel (+R+S+L)
- taana waa+ak Vancouver. Louis ?in a. ?u?uusapi?iš ?ayanak ?u-sapi[+R+S+L]-?iiš Louis ?in ?aya-naak taana waa+ak Vancouver. Ø-depend.on-3.IND Louis COMP many-have money go.to Vancouver Louis is depending on having a lot of money in going to Vancouver.
- b. ?u?uušsapi?iš ?uuš-sapi[+R+S+L]-?iiš someone-depend.on-3.IND S/he is depending on someone.

In contrast, the predicate *?u?u-sum* "want" triggers reduplication with a short vowel in both the reduplicant and the initial syllable of the base. In (19a), the expletive appears as 2u2u-, while in (19b) the morpheme taanag- "money" surfaces as tatanaq-.

*Reduplication with short initial vowel and short second vowel* (+R+S+S)(19)

a.	?u?usum?iš	Louis	taanaak?i
	?u- <u>sum[</u> +R+S+S]-?iiš	Louis	taana-?ak-?ii
	Ø- <u>want</u> -3.ind	Louis	money-POSS-DET
	Louis wants his money.		
b.	tatanaqsum?iš	Louis	
	taanaq- <b>sum</b> [+R+S+S]-?ii	š Louis	

Louis

money-want-3.IND

Louis wants money.

34

It is also possible for an affixal predicate to require reduplication of the first syllable of the base, as well as a long vowel in the initial syllable of base. In this pattern, the reduplicated syllable retains the underlying vowel length of the base. This is demonstrated in (20) with the predicate  $2u^2uu-yuk$  "cry for". In (20a), the host is reduplicated to  $2u^2uu-$ , while in (20b), the host is reduplicated to *siisii*- "you (PL)".<sup>4</sup>

# (20) Reduplication with neutral initial vowel and long second vowel (+R+L)

a.	?u?uuyuk?anitwa?iš?a <sup>↓</sup> ?u- <u>yuk[</u> +R+L]-'at-mit-wa?iš-?a <sup>↓</sup> Ø- <u>crv.for</u> -PAS-PST-3.QUOT-PL Kyle was crying for his uncles/aunts.	nana?iiqsak na?iiqsu-[+R]-?ak aunt/uncle-PL-POSS ( <i>lit</i> : "his uncles were	?uḥ?at ?uḥ?at by cried f	Kyle Kyle Kyle for by Kyle")
b.	siisiiyuk?anitwa?icuuš	Kyle		
	sii- <b>yuk</b> [+R+L]-'at-mit-wa?icuuš	Kyle		
	you- <u>cry.for</u> PAS-PST-2PL.QUOT	Kyle		
	Kyle was crying for you (pl). (lit: "you	(PL) were cried for	(by) Ky	yle")

Only in the "neutral" pattern is the host prosodically unaffected by the affixal predicate. As the examples in (21) show, the phonemically contrastive vowel lengths of the host are unaltered by the affixal predicate -u2aat "find", and no reduplication is triggered. Accordingly, in (21a), 2u- surfaces without reduplication or a change in vowel length, and in (21b), the same applies for *taanaq*- "money".

# (21) Neutral pattern

a.	?uyu?aa†siš	taana
	?u- <u>u?aa†</u> -siiš	taana
	Ø- <u>find</u> -1sg.ind	money
	I found money.	

 taanaqu?aa+siš taanaq-<u>u?aa+</u>-siiš money-<u>find</u>-1SG.IND I found money.

Many affixal predicates in the language display this neutral pattern, and impose no prosodic conditioning (see Sapir and Swadesh 1939).

Kim (2003) provides an analysis of the reduplicative patterns triggered by affixal predicates within the framework of Optimality Theory. I present here a brief

<sup>&</sup>lt;sup>4</sup> Pronouns such as sii- (2PL) are possible as hosts for affixal predicates in Nuu-chah-nulth, despite the evidence that DP is a closed domain for affixation (see §2.5). Investigation is required into the status of Nuu-chah-nulth pronominals in the cross-linguistic typology of pronominal phrase types, as in the DP/ $\phi$ P/nP contrast drawn by Déchaine and Wiltschko (2002).

sketch of the form which an analysis of prosodic conditioning may take if it is articulated in accordance with Minimalist assumptions. I assume that the prosodic "subcategorization" of affixal predicates is specified at the level of the lexical entry. A predicate such as *?uu-hwa+* "use" has the following lexical entry:



Together with lexical semantics, phonemic representation, and information on the affixal status of this morpheme, this lexical entry contains a specification for a [+L] feature. This vowel lengthening requirement forms a condition on convergence at the PF interface. When the affixal predicate  $\partial uu$ -hwa+"use" is spelled-out to PF, the lengthening requirement is obligatorily realized on its host. If *yaxyak* "broom" is the derivational sister of the affixal predicate, then as a host, it must undergo vowel lengthening.

(23)

-hŵat yaxýak

This induces a form of yaaxyak-hwat "broom-use" at PF.

I present this prosodic conditioning as independent evidence for the idea that local spell-out links an affixal predicate with its host at spell-out to PF. The prosodic conditioning of a host by an affixal predicate indicates that the two must be interpreted together at PF. In effect, prosodic conditioning leaves a detectable "footprint" of local spell-out. The local spell-out hypothesis offers a maximally restrictive prediction regarding which morphemes may be treated by prosodic conditioning, and which may not. According to this hypothesis, the only element which may be prosodically influenced is the derivational sister of the element which is specified for a prosodic requirement. For instance, for (24), the local spell-out hypothesis determines that the morpheme *yaxyak* "broom" is eligible for prosodic conditioning by the affixal predicate  $\lambda uu-hwat$  "use", but the past tense marker -(m)it (PST) is ineligible. In (24a), the lengthening is realized on *yaxyak* "broom" as *yaaxyak*; however, as indicated by (24b), -(m)it (PST) may never be lengthened to -(m)iit.

- (24) a. yaaxýakhwałit?iš yaxýak-hwał[+L]-mit-?iiš broom-<u>use</u>-PST-3.IND S/he used a broom.
  - b. \* yaxýakhwałit?iš yaxýak-hwał[+L]-mit-?iiš broom-use-PST-3.IND S/he used a broom.

#### PF INCORPORATION

According to the local spell-out hypothesis, a derivational sister to the affixal predicate is present at the point of local spell-out of the affix, but the past tense marker is not. In the diagram in (25), the past tense marker is shown to occupy a higher position in the tree than the affixal predicate  $\partial uu - h w a + u u = h w a + u =$ 

(25) -mit local spell-out PST -hŵat yaxyak use broom

In contrast, the single spell-out and multiple spell-out hypotheses open the door for the possibility that morphemes from the derivation *other than derivational sisters* may be prosodically influenced by the affixal predicate. With these delayed spell-out models, an additional stipulation would be required to rule out why the past tense marker is not equally eligible for prosodic conditioning. One possible stipulation could be directionality – that is, prosodic conditioning should only affect a morpheme to the left of the affixal predicate. The local spell-out hypothesis does not need to resort to directionality in determining the site of prosodic conditioning. Chapter 3 discusses problematic aspects of a directionality-sensitive mechanism for affixation (see §3.3.5.2). Moreover, evidence is shown in Chapter 4 that affixal predicates with unergative semantics are absent in Nuu-chah-nulth: a systematic gap attributable to the need for an affixal predicate to find a host in its complement – not simply a rightward host. Thus, the relationship between an affixal predicate and its host is sensitive to derivational sisterhood, not directionality.

## 2.2 Linearization is Local

When an element reaches spell-out, it must be linearized with respect to its neighbour. This is the essence of the local spell-out proposal. In the discussion up until this point, the locality constraint on this linearization process has been trivial in that only two syntactic terminals,  $\alpha$  and  $\beta$ , were represented as the input to the spell-out rule:



The syntactic configuration in (26) can be considered to be the basic step of the syntactic derivation, equivalent to a single application of Merge ( $\alpha$ ,  $\beta$ ). An example of this simple arrangement is when an affixal predicate selects a bare noun complement, as in *taanaq-u?aa+* "find money" (from 21b).



The linearization forced at spell-out for *taanaq-u?aa1* is a case of PF incorporation. (We will return to the topic of the nominal complements of affixal predicates in Chapter 4.) In this section, we take a first step towards defining the linearization of more complex syntactic constructs.

As we will see in this section, the linearization of affixal predicates in Nuuchah-nulth is strictly local. For an explanation of this locality constraint, consider the following syntactic construct:



This configuration is derived via two separate applications of binary concatenation. In the first,  $\delta$  and  $\pi$  are selected from the numeration [ $\alpha$ ,  $\delta$ ,  $\pi$ ] and are joined through Merge ( $\delta$ ,  $\pi$ ). The output of Merge ( $\delta$ ,  $\pi$ ) is  $\beta$ , the abstract node label designating the contents of the pairing. For the second concatenation,  $\alpha$  is introduced from the numeration. This concatenation unites  $\alpha$  with  $\beta$ , through Merge ( $\alpha$ ,  $\beta$ ). The syntactic output of this sequence of operations is  $\gamma$ , the root node label.

According to the local spell-out hypothesis, after the first round of Merge,  $\delta$  and  $\pi$  are spelled-out. Let us assume that a consequence of spell-out is that the two elements,  $\delta$  and  $\pi$ , are ordered with respect to each other. (The means by which this ordering takes place is addressed in Chapter 3, but for now we can adopt this assumption.) Take this ordering to be specified first to last as  $\langle \delta, \pi \rangle$ . When the next element,  $\alpha$ , enters the computation and attains spell-out,  $\alpha$  must be linearized with respect to  $\beta$ , just as it was in the simpler case of (26).

# (29) Local spell-out: for Merge $(\alpha, \beta)$ , interpret $(\alpha, \beta)$

With the derivation in (28), however,  $\beta$  is not a simplex construct; in (28),  $\beta$  is equal to the linearized object  $\langle \delta, \pi \rangle$ .

Assume that  $\alpha$  is an affixal predicate ( $-\alpha$ ). At spell-out to PF, an ordering of  $\alpha$  with respect to  $\beta$  requires that the affixal predicate ( $-\alpha$ ) must be linearized relative to the ordered object  $<\delta$ ,  $\pi$ >. The claim that this section makes is that spellout of this arrangement consistently yields in Nuu-chah-nulth a linearization of  $<(\delta-\alpha)$ ,  $\pi>$ : an affixal predicate in Nuu-chah-nulth only ever suffixes to the element at the initial edge of its derivational sister. For the linearized object  $<\delta$ ,  $\pi>$ , the host for an affixal predicate is identified as  $\delta$ . The alternative of  $<\delta$ ,  $(\pi-\alpha)>$  never arises in Nuu-chah-nulth.

#### PF INCORPORATION

Furthermore, when even larger derivational samples  $\langle \theta, \delta, \pi \rangle$  are considered, we will also see that linearization never "skips" a potential host. Take  $\langle \theta, \delta, \pi \rangle$  to be the linearized object specified after two initial applications of Merge. These two concatenations (and corresponding local spell-outs) are following by a third application of Merge, introducing the affixal predicate  $\alpha$ .

(30)



At the spell-out point of  $\gamma$ , the affixal predicate  $-\alpha$  must be linearized with respect to the ordered object  $\langle \theta, \delta, \pi \rangle$ . Affixation is based on linear adjacency in that the resulting linearization is  $\langle (\theta - \alpha), \delta, \pi \rangle$  and not  $\langle \theta, (\delta - \alpha), \pi \rangle$ . It is the single leftmost element in the string which can serve as the host for the affixal predicate. I label this constraint in Nuu-chah-nulth the *string adjacency effect*.

## (31) *String adjacency effect:*

An affixal predicate must be linearized as a suffix to the leftmost element in the string which is linearly adjacent to it.

Which property of the grammar is responsible for this adjacency requirement? I consider this effect to be a reflex of the spell-out of the affix. It arises from local spell-out, in which the affixal predicate is evaluated relative to its derivational sister. The simplest view of the interpretative capacity of spell-out is that it operates without recourse to counting. Formally, linearization is therefore insensitive to ordinal position within a complex string  $<\theta$ ,  $\delta$ ,  $\pi$ >. Instead, the linearization mechanism attends to the *boundaries* of the string (cf. Klavans 1985). The initial element of the derivational sister is chosen as host for an affixal predicate in Nuu-chah-nulth due to its peripheral position in the string. This derives the string adjacency effect.

The string adjacency effect follows from the need to linearize an affix with respect to the boundary of its derivational sister. In Nuu-chah-nulth, affixes select as host the element at the left periphery of their derivational sister. There also exists a logically possible alternative: orientation to the right boundary of a derivational sister. Although this option does not arise in Nuu-chah-nulth, it may be observed in languages with "phrasal" affixes which are positioned relative to the final element in a syntactic constituent (Klavans 1985). The two options for orientation of an affix at the boundaries of its derivational sister are illustrated in (32). In (32), the affix  $\alpha$  has as its derivational sister the linearized object  $\langle \theta, \delta, \pi \rangle$ .

## (32) Peripheral positions within the derivational sister of an affix



Orientation to the initial boundary will yield a string adjacency effect in which the element  $\theta$  will be selected as host for  $\alpha$ , as in Nuu-chah-nulth. Positioning relative to the final boundary will determine that  $\pi$  is chosen as host for  $\alpha$ . Thus, the local spellout model predicts two alternative orientations for choice of hosts. These alternative realizations appear to be attested cross-linguistically (Klavans 1985). This prediction distinguishes the local spell-out model from Marantz's (1988, 1989) Morphological Merger, which allows only string adjacent elements to be chosen as host for an affix.

In the following subsections, we will see empirical evidence for the string adjacency effect in Nuu-chah-nulth affixation. I start in §2.2.1 by showing that affixation in Nuu-chah-nulth does not skip potential hosts, but instead feeds a potentially iterative affixation process. In §2.2.2, I present evidence that PF incorporation targets non-heads of a syntactic constituent, so long as these elements are positioned at spell-out contiguous to the affixal predicate. In §2.2.3, it is shown that this operation breaks up coordinated objects – disregarding the Coordinate Structure Constraint (CSC) – by targeting the conjoined element which abuts the affixal predicate at spell-out.

#### 2.2.1 Iterativity

This section discusses the make-up of complex strings of dependencies. Consider (33), in which a sequence of affixes (including two affixal predicates, -*iiḥ* "try to" and *-maḥsa* "want to") are suffixed to the verb *huḥtak* "know".

(33)	huuḥtakšiiḥmaḥsaʔiš	Lucy	quuquu?aca
	huḥtak-šiૠ- <b>ʾiiḥ</b> [+L]- <b>maḥsa</b> -?iiš	Lucy	quu?ac-[+R]-(y)a
	know-PERF- <u>try.to</u> -want.to-3.IND	Lucy	person-speak-CONT
	Lucy wants to learn how to speak Nuu-	-chah-nulth.	

I have described the process of affixation in Nuu-chah-nulth as one applying to pairs of items: an affix and a host. If the linearization specified by local spell-out is a pairwise function, then how can complex sequences such as (33) be formed? It seems on the surface that there are many affixes, and only a single host (the verb *huhtak* "know"). If the host for an affixal predicate must be linearly adjacent to the affixal

PF INCORPORATION

predicate, then why is that -mahsa "want to" is attaching to another bound element (-'*iih* "trying to"), and not attaching directly to the free form *huhtak* "know"? Strictly speaking, the affixal predicates -mahsa and -'iih cannot be serving as hosts for each other. Each of these affixal predicates are suffixes, so if they must find a host between them, then one will necessarily be left without. Recall from Chapter 1 that the combination of two suffixes,  $-\alpha$  and  $-\beta$ , results in an indeterminate ordering of  $(-\alpha-\beta)$  or  $(-\beta-\alpha)$ . If the ordering of  $(-\alpha-\beta)$  is selected, the result fails in that the element  $-\alpha$  is not bound as a suffix. Conversely, if the ordering of  $(-\beta-\alpha)$  is picked, then  $-\beta$  fails to appear as a suffix. Thus, the relationship between the elements is inherently incompatible, and as such, no dependency obtains.

The solution to this problem is iterative application of local spell-out, an idea first introduced in Chapter 1. In the framework I am assuming, spell-out applies for each occurrence of Merge. By the Order Preservation hypothesis (Fox and Pesetsky 2005), spell-out of later cycles adds information to previous cycles. A consequence of this proposal is that successive applications of local spell-out enforce a build-up of hosts, induced when one affixal predicate finds a host, and then this affix-host complex in turn serves as the host for another affix. We can take the data in (34) as an illustration of this process.

(34) čamaspał-ċuqši?in
 čamas-jał-ċuq-šiλ-'in
 sweets-taste-in.mouth-PERF-1PL.IMP
 Let us put something sweet in our mouths.

In (34), there are two affixal predicates: 2u-pa? "taste of" and 2u-cuq "in mouth". The affixal predicates are followed by the perfective suffix -si? (PERF), and the imperative marker -in (1PL.IMP).

Recall that syntactic derivations are built from bottom to top. We assume the first step of the syntactic derivation to be one in which the predicate -pat "taste of" joins with *čamas* "sweet" via Merge (pat, čamas). (In Chapter 4, I show in detail how arguments of affixal predicates such as 2u-pat "taste of" are syntactically introduced.)

(35)

-*pa*+ čamas taste sweet

Because  $-\dot{p}a t$  is a suffix, this arrangement must be linearized at spell-out as  $<\dot{c}amas-\dot{p}at$ ? "sweet-tasting". Successive steps of Merge result in a longer string of morphemes. In the next stage,  $-\dot{c}uq$  "in mouth" is merged into the derivation, via Merge ( $\dot{c}uq$ ,  $\dot{c}amas-\dot{p}at$ ).

(36)



Because  $-\dot{c}uq$  is a suffix, this string is linearized as  $<\dot{c}amas\dot{p}a\dot{\tau}-\dot{c}uq>$  "something sweet tasting in the mouth".

Suffixes introduced later in the derivation *follow* a previously positioned suffix. "Tucking in" of suffixes is not permitted, ruling out a form such as  $* < \dot{c}amas-\dot{c}uq-\dot{p}a+$ , in which a later suffix (- $\dot{c}uq$  "in mouth") would adjoin directly to the original host of the derivation ( $\dot{c}amas$  "sweet"), rather than falling outside the last suffix (- $\dot{p}a+$  "taste of"). "Tucking in" is countercyclic, and its impermissibility follows directly from the role that local spell-out plays in fixing phonological content. If suffixes were to adjoin iteratively to the original host, rather than the last suffix, then this would disrupt the placements of earlier cycles. Under the local spell-out model, linearization is fixed at each cycle.

Returning to the derivation of (34), the non-predicative suffixes,  $-\dot{s}i\dot{\kappa}$  (PERF) and  $-\dot{i}n$  (1PL.IMP), are also eligible for positioning through local spell-out. When the perfective suffix  $-\dot{s}i\dot{\kappa}$  (PERF) is merged into the derivation, it is spelled-out to PF with its derivational sister  $\dot{c}amas\dot{p}a\dot{+}\dot{c}uq$ , inducing a linearization of  $<\dot{c}amas\dot{p}a\dot{+}\dot{c}uq-\dot{s}i\dot{\kappa}>$  "put something sweet tasting in the mouth".

(37)



Finally, the imperative suffix -in (1PL.IMP) is then positioned at spell-out after it has been merged.

(38)



As a suffix, the imperative marker -in (1PL.IMP) is spelled-out following the previously linearized components. The resulting arrangement is < camas ja + cuq si - 2in >"let us put something sweet in our mouths". The principles of iterative local spellout therefore correspond to the effects of Baker's (1988) Mirror Principle: the left to right arrangement of suffixes reflects the first to last steps of the syntactic derivation. Suffixes introduced later in the derivation will be linearized towards the end of the word. In effect, each step of the syntax induces a "phonologization" of the elements of the syntactic tree.

According to my proposal, the "phonologizing" effects of local spell-out are reflected in a sensitivity to string adjacency at PF. Sensitivity to linear ordering is a property of the phonological system, not the syntax (Chomsky 1995; Fox and Pesetsky 2005). As I have described, the syntax does not operate on the basis of linear arrangements: it is simply a device of binary concatenation. The next sections add weight to the argument that PF incorporation is a non-syntactic phenomenon. The data which I will present demonstrate that PF incorporation operates in Nuuchah-nulth on linearly adjacent items, irrespective of their internal syntactic structure.

# 2.2.2 Modifier Incorporation

The analysis predicts that any element which is string adjacent to an affixal predicate should be chosen to act as its host; elements which are not linearly adjacent to the affix should not be eligible as hosts. Here, we examine the phenomenon of modifier incorporation as evidence for sensitivity to string adjacency in Nuu-chah-nulth incorporation. The prediction is that a modifier will serve as host for an affixal predicate if it occurs at the left-periphery of its derivational sister. Two types of modifier incorporation will be considered in detail: the first, targeting adjectives; the second, targeting adverbials.

# 2.2.2.1 Adjective Incorporation

Within nominal phrases in Nuu-chah-nulth, there is a strict ordering relationship between constituent elements, such that a modifier necessarily precedes the nominal. (In Chapter 3, I will attribute this regularity to a spell-out "convention"). A modifier such as *hafum* "tasty" must obligatorily precede a nominal such as *faapinis* "apples".

(39)	a.		?u?iic?iš?a†	ha?um	<b>?aapinis</b>	
			?u- <u><b>'iic</b></u> -?iiš-?a <del>1</del>	ha?um	?aapinis	
			Ø- <u>consume</u> -3.ind-pl	tasty	apples	
			They are eating delicious	s apples.		
	b.	*	?u?iic?iš?a <del>1</del>	?aapinis		ha?um
			?u- <u><b>'iic</b></u> -?iiš-?a <del>1</del>	?aapinis		ha?um
			Ø- <u>consume</u> -3.ind-pl	apples		tasty
			They are eating delicious	s apples.		

Let us assume for the present discussion that in (39), the affixal predicate *?u-?iic* "consume" takes the nominal phrase *ha?um ?aapinis* "tasty apples" as its complement. (Chapter 4 provides an analysis of the syntactic configuration of arguments of affixal predicates.) In this section, we will see that, as predicted, the choice of host for an affixal predicate is determined by the linear ordering of elements with the nominal phrase that it takes as a complement. An affixal predicate incorporates whatever host is "leftmost in the order Q[uantifier] > Q[uantity] > A[djective] > N[oun]" (Rose 1981: 294). In other words, an affixal predicate obligatorily attaches to the element in the complement which is *string adjacent* to the affixal predicate.

It is this sensitivity to linear ordering which determines that PF incorporation is not "noun incorporation" in a strict sense. Although affixal predicates may select a noun as host in a simplex nominal complement, this preference switches once a prenominal constituent enters the picture. For example, although the nominal *?aapinis* "apples" is the host for *?u-?iic* "consume" in (40a), it cannot act as a host when the prenominal modifier *ha?um* "tasty" appears, as in (40b).

- (40) a. ?aapiniyic?iš?a<sup>+</sup> ?aapinis-<u>'iic</u>-?iiš-?a<sup>+</sup> apples-<u>consume</u>-3.IND-PL They are eating apples.
  - b. \* ?aapiniýic?iš?a‡ ha?um ?aapinis-<u>'iic</u>-?iiš-?a‡ ha?um apples-<u>consume</u>-3.IND-PL tasty They are eating delicious apples.

In this context, the adjective is necessarily chosen to be the host, rather than the modified noun. In (41), the affixal predicate  $\lambda u$ - $\lambda i c$  "consume" suffixes to the modifier *halum* "tasty".

(41) ha?um?ic?iš?a<sup>+</sup> ?aapinis ha?um-<u>'iic</u>-?iiš-?a<sup>+</sup> ?aapinis tasty-<u>consume</u>-3.IND-PL apples They are eating delicious apples.

Quantifiers, which like adjectives necessarily precede a nominal, show a parallel pattern of being selected as host for the affixal predicate. The example in (42a) shows this quantifier-initial order. An affixal predicate incorporates a quantifier (42b), rather than the quantified noun (42c).

(42)	a.	?u?is?iš	?aya	muks?i
		?u- <b>'is</b> -?iiš	?aya	muks?i
		Ø- <b>on.beach</b> -3.IND	many	rocks
		There's lots of rocks of	n the beach.	
	b.	?ayiis?iš	muks?i	
		?aya- <b>'is</b> -?iiš	muks?i	
		many-on.beach-3.IND	rocks	
		There's lots of rocks of	n the beach.	
	c.	* muks?i?is?iš	?ava	
		muks?i- <b>'is</b> -?iiš	?aya	
		rock-on.beach-3.IND	many	
		There's lots of rocks or	n the beach.	

This sensitivity to linear adjacency extends to object *wh*-questions formed by incorporation into an affixal predicate. In "which"-questions, the *wh*-word hosts the predicate, while the restriction is stranded (Davis and Sawai 2001):

(43)	waayaSamith	Louis	č'upč'upšum†
	waayaq- <u>'aap</u> -mit-ḥ	Louis	čupčupšum1
	which-buy-PST-3.Q	Louis	sweater
	Which sweater did Lou	is buy?	

The restriction which Nuu-chah-nulth incorporation has on targeting the "leftmost" element (Rose 1981: 295) is not in the vocabulary of the syntax. This is because incorporation in Nuu-chah-nulth is not a syntactic process. In the Minimalist model I have adopted, syntactic processes operate on the basis of hierarchical relationships – created by binary concatenation – while PF processes operate on the basis of linearly defined relationships. In the terminology of Lasnik (2000), the two types of analyses may be teased apart in contexts in which *linear* adjacency does not correspond to *hierarchical* adjacency. Let us consider how these types of adjacency differ.

Syntactic head movement is possible only in accordance with the Head Movement Constraint, a restriction which operates on the basis of hierarchical dominance relations within a syntactic tree:

(44) Head Movement Constraint (Travis 1984):A head Y may only move to X if Y is the sister of X

Matushansky (2006) terms this relation a "head-of-the-complement" locality. Following Matushansky, I will assume that it is fundamentally a syntactic restriction, as the same notion of locality is at play in c-selection, the means by which heads select the syntactic category of their complement.<sup>5</sup> Indeed, as Matushansky argues, head movement is possible only between elements when one is c-selected by the other.

For the tree in (45), Y is the sister of X, but Z is not. In other words, Y is the head of the complement of X.



Thus, according to the Head Movement Constraint, only Y should be eligible to move to X.<sup>6</sup> Movement of Z–X should be impossible. The "head-of-the-complement"

<sup>&</sup>lt;sup>5</sup> Matushansky's (2006) proposal counters the speculation of Chomksy (2001) that head movement is non-syntactic. See also Donati (2003) and Surányi (2003), among others for related arguments that head movement is a syntactic phenomenon. I return to this topic in Chapter 6.

<sup>&</sup>lt;sup>6</sup> I wait until Chapter 6 to discuss a particular technical implementation of syntactic head movement. The traditional model of head adjunction (Travis 1984, Baker 1988) is at odds with the Minimalist requirement that movement (remerge) only be effected at the root node (the Extension Condition). Surányi (2003) and Matushansky (2006) each offer analyses of syntactic head movement compatible with the Extension Condition. Either of these approaches would be compatible with my assumptions here. See Chapter 6 for further discussion of syntactic head movement.

locality of head movement is a hierarchically defined restriction. This locality is sensitive to which element of a syntactic pairing is the one to project. In (45), Y projects after Merge (Z, Y). Therefore, Y is the head-of-the-complement. Y may be said to be "hierarchically" adjacent to X, because it is Y(P) which is united with X in the next round of structure-building, Merge (Y, X). This notion of hierarchical adjacency is not equivalent to linear adjacency. If the syntactic object of (45) is realized as the linearized object  $\langle X, Z, Y \rangle$ , then X shares a linear adjacency to Z *despite the fact that Z is not hierarchically adjacent to X*.

As Baker (2003: 152) describes, it is a consequence of this hierarchical adjacency restriction that adjective incorporation is explicitly forbidden syntactically: "the head movement constraint implies that one can never incorporate the attributive modifier of a noun to form a  $A_k + V [_{NP} t_k N]$  structure" Thus, even if an adjective in complement position of a verb is linearly adjacent to the verb ("leftmost" of the complement), it is not "hierarchically adjacent" to the verb in the syntax. In the tree in (46), the N is the head-of-the-complement of V; A is not the head-of-the-complement.



By the Head Movement Constraint, it is therefore predicted to be impossible for A to move to V.

In contrast, operations which are processed on the basis of linear adjacency are not restricted by head-of-the-complement locality. Let us consider the case of adjectival incorporation in Nuu-chah-nulth as evidence that incorporation operates on the basis of linear adjacency within derivational sisters, and not on the basis of hierarchical adjacency. Recall from the preceding discussion that an affixal predicate incorporates an adjective contained in its complement. I repeat here the example of adjective incorporation shown earlier in (41).

(47)	ha?um?ic?iš?a+	?aapinis
	ha?um- <u><b>'iic</b></u> -?iiš-?a†	?aapinis
	tasty- <u>consume</u> -3.IND-PL	apples
	They are eating delicious	apples.

It is impossible for an affixal predicate to incorporate the noun if an adjective is present. Crucially for our discussion, it was also noted that an adjective necessarily precedes a modified noun in Nuu-chah-nulth. Thus, where  $\delta$  is a modifer and  $\pi$  a noun, the affixal predicate  $-\alpha$  is attached to the leftmost element in its complement  $<\delta$ ,  $\pi$ >, yielding a linearization of  $<(\delta-\alpha)$ ,  $\pi>$ .

Adjective incorporation is an ideal test case for determining whether linearization of affixal predicates is a syntactic or PF phenomenon because for an adjective contained with the complement of a verb, *linear* adjacency between the PF INCORPORATION

adjective and verb does not correspond to *hierarchical* adjacency. That is, while an adjective may directly follow a verb, it is the noun which heads the complement that is hierarchically adjacent to the verb, not the adjective contained within the complement. The diagram in (48) shows the proposed syntactic structure for a modified noun phrase *harum ?aapinis* "tasty apples" which occurs as the complement of the affixal predicate *?u-?iic* "consume".



If incorporation in Nuu-chah-nulth were a case of syntactic head movement, then the movement should be in accordance with the Head Movement Constraint. This constraint determines that for (48), the noun should incorporate, but the adjective should not. This is not what occurs in Nuu-chah-nulth incorporation. In Nuu-chah-nulth, it is ungrammatical if the noun incorporates into the affixal predicate when an adjective is present. This is shown in (49a), in which the affixal predicate — *iic* "consume" incorrectly suffixes to the noun *?aapinis* "apples". As indicated in (49b), the affixal predicate must instead suffix to the adjective *ha?um* "tasty".

- (49) a. \* ?aapiniýic?iš?ał ha?um ?aapinis-<u>?iic</u>-?iiš-?ał ha?um apples-<u>consume</u>-3.IND-PL tasty They are eating delicious apples.
  - b. ha?um?ic?iš?a‡ ?aapinis ha?um-**?iic**-?iiš-?a‡ ?aapinis tasty-<u>consume</u>-3.IND-PL apples They are eating delicious apples.

This behaviour follows from the generalization that an affixal predicate suffixes to whatever element is leftmost in the noun phrase. The fact that incorporation in Nuuchah-nulth targets whatever element is contiguous to the affixal predicate, irrespective of hierarchical adjacency, is predicted by a local spell-out analysis in which incorporation operates on the basis of string adjacency. However, such behaviour is not predicted by a syntactic head movement analysis of incorporation.

In order to maintain a syntactic head-movement analysis of Nuu-chah-nulth incorporation, it would become necessary to abandon a noun-headed analysis of the complement phrase, in favour of a representation in which modifiers head the phrase containing the nominal (Stonham 2004). Such an analysis is represented in (50).



However, this putative case of syntactic incorporation is at odds with independent evidence for the nominal headedness of adjective-noun combinations. This independent evidence comes from categorial restrictions on modification in Nuu-chah-nulth.

Modification in Nuu-chah-nulth is sensitive to a constraint on syntactic category (Wojdak 2000, 2001). While adjectives may modify a noun (such as  $haak^waa \lambda$  "girl"), an adjective (such as *?aaphii* "friendly") is barred from serving as the modifier of another adjective.

(51)	a.	?u?uk <sup>w</sup> inkitsiš	[q <sup>w</sup> aća <del>1</del> aq?i	ḥaakʷaaૠ]
		?u-k <sup>w</sup> ink-[+R]-mit-siiš	[qwaca1-aq[+S]-?ii	ḥaakʷaa⊁]
		Ø-with-speak-PST-1SG.I	ND beautiful-AUG-DET	girl
		I talked with the beautiful	ıl girl.	

b.	*	?u?uk <sup>w</sup> inkitsiš	[q <sup>w</sup> aćataq?i	?aapḥii]
		?u-k <sup>w</sup> ink-[+R]-mit-siiš	[qwacat-aq[+S]-?ii	?aapḥii]
		Ø-with-speak-PST-1SG.IND	beautiful-AUG-DET	friendly
		I talked with the beautiful fi	riendly (one).	

The distinction between nominals and adjectives in such modification constructions is clearly not reducible to a non-categorial difference such as a contrast between stage-level and individual-level predication (Kratzer 1995) or between transitives and intransitives. Lexical items like  $haak aa \pi$  "girl" and  $qaa aa \pi$  "beautiful" are each one-place predicates which denote individual-level (non-transient) properties. Thus, I follow the argument which Demirdache and Matthewson (1995) make for Salish in proposing that the factor which distinguishes the class of these lexical items is inherent lexical category.

The argument for the headedness of adjective–noun combinations comes from consideration of iteratively modified forms such as (52).

(52)	?u?uk <sup>w</sup> inkitsiš	[q <sup>w</sup> aćataq?i	?aapḥii	ḥaakʷaa⊁]
	?u-k <sup>w</sup> ink-[+R]-mit-siiš	[qwacat-aq[+S]-?ii	?aaphii	ḥaakʷaa⊁]
	Ø-with-speak-PST-1SG.IND	beautiful-AUG-DET	friendly	girl
	I talked with the beautiful fr	riendly girl.		

Given the category-sensitive restriction in (53a, b), we can deduce for an iteratively modified form that adjectival headedness of the phrase is ruled out (54b), in favour of nominal headedness (54a). In other words, since we know that adjective-adjective modification is impossible, it must be the case that the internal constituent denoted

by bracketing in (54) must be behaving syntactically as a noun phrase, and not an adjective phrase. If the internal constituent in (54) were behaving as an adjective phrase, then it would be incorrectly predicted that combination with another adjective, as in (54b), should be impossible.

(53) a.  $AP + NP \implies (54)$  a.  $AP + _{NP}[AP + NP]$ b. \*  $AP + AP \implies b.$  \*  $AP + _{AP}[AP + NP]$ 

Thus, in an adjectivally modified noun, it must be the noun that is the head of the phrase.

This, in turn, implies that incorporation in Nuu-chah-nulth applies to morphemes (e.g., adjectives) which are not the syntactic heads of the complement of an affixal predicate. This constitutes strong evidence against a syntactic account of Nuu-chah-nulth incorporation in which the head movement is subject to the Head Movement Constraint. Adjectives are not hierarchically adjacent to the affixal predicate; however, they do maintain a linear adjacency to the affixal predicate. A linearization mechanism sensitive to linear adjacency, but not hierarchical adjacency, predicts this incorporation pattern in Nuu-chah-nulth.

## 2.2.2.2 Adverbial Incorporation

This sensitivity to linear ordering may also be observed with affixal predicates which take verbal, rather than nominal, complements. Affixal predicates in this class include -qaath "claim" and 2u-Gi $\lambda$  "come upon". These predicates allow incorporation of a verb from their logical complement.

(55)	a.	wałši⊁qatḥitsiš wał-ši⊁- <b>gaatḥ</b> -mit-siiš go.home-PERF- <u>claim</u> -PST-1SG.IND I claimed I went home.	
	1.	0: *C:0:4-:*	V

b.	warichikitsis	Ken
	wa?ič- <u><b>Siૠ</b>-mit-siiš</u>	Ken
	sleep- <u>come.upon</u> -PST-1SG.IND	Ken
	I came upon Ken sleeping.	

This section considers the pattern of adverbials which respect to these "verb-incorporating" affixal predicates.

Outside of incorporation contexts, adverbials in Nuu-chah-nulth split into two classes, depending on whether they are subject to flexible or rigid positioning relative to the main predicate. "High" adverbials (Cinque 1999), such as subjectoriented or temporal adverbs, have a flexible order, and may either precede or follow a main predicate. This class includes subject-oriented *qwaluuh* "purposely" and temporals *čaani* "first" and *nalik* "immediately". The two patterns for these flexibly positioned adverbs are shown in the examples below. In each of the

#### THE LINEARIZATION OF AFFIXES

(a) cases, the adverb appears before the main verb. In the (b) examples, the adverb follows the main verb. (Inflectional morphemes, as "second position" enclitics, suffix to whatever word is first in the clause – it will be argued in Chapter 3 that these inflectional morphemes are linearized at local spell-out, just as with the linearization of affixal predicates.)

(56)	a.	q <sup>w</sup> a?uuḥ?iš	Florence	nunuuk	
		q <sup>w</sup> a?uuḥ-?iiš	Florence	nunuuk	
		purposely-3.IND	Florence	sing	
		Florence is purpos	sely singing.		
	b.	nunuuk?iš	Florence	q <sup>w</sup> a?uuḥ	
		nunuuk-?iiš	Florence	q <sup>w</sup> a?uuḥ	
		sing-3.IND	Florence	purposely	
		Florence is purpos	sely singing.		
		(context: Florence	e's neighbour ke <sub>l</sub>	pt her awake las	st night and now
		she want.	s to get even by b	peing loud)	
(57)	a.	čaani?aq <del>x</del> ?iš	wa?ič		
		čaani-?aqx-?iiš	wa?ič		
		first-FUT-3.IND	sleep		
		He will sleep first	(i.e., before doin	g something els	se).
	b.	wa?ič?aa <del>λ</del> ?iš	čaani		
		wa?ič-?ag <del>\u00cb</del> -?iiš	čaani		
		sleep-FUT-3.IND	first		
		He will sleep first	(i.e., before doin	g something els	se).
(58)	a.	na?iik?aa⊁siš	watsiz	hawii	?azauu
(00)		na?iik-?aox-siiš	wat-šit	hawii	-'ax-quu
		immediately-FUT-	1SG.IND go.hom	e-PERF finish	-TEMP-3.COND
		I will immediately	go home when i	it's finished.	
	b.	watsi?aaksiš	na?ii	k hawii	?a <del>x</del> auu
	2.	wat-šiz-?agz-siiš	na?ii	k hawii	-'az-quu
		go.home-PERF-FU	T-1SG.IND imme	ediately finish	-TEMP-3.COND
		I will immediately	go home when	it's finished	

Manner adverbials, in contrast, belong to a second class which must rigidly precede the main predicate. These manner adverbials include *wityax* "slowly", *čamaq* $\hat{x}$ "properly" and *hacuk* "(sleep) deeply". As shown in the following (a) examples, it is grammatical for the manner adverbial to precede the predicate it modifies. In the (b) examples, in contrast, ungrammaticality arises when the manner adverbial follows the main predicate.

50

(59)	a.		witýaxits witýax-mit-s slow-PST-1SG.ABS I was going home	S e slowly.	waa†ši⊁ wa†-[+L] go.home	-šîλ -CONT-PERF
	b.	*	waałšikits wał-[+L]-šik-mit- go.home-CONT-PI I was going home	-s ERF-PST-1 e slowly.	SG.ABS	witỷax witỷax slow
(60)	a.		čamaq <del>%</del> ?iš čamaq <del>%</del> -?iiš properly-3.IND Florence is dryinş	titiqs titiqs dry g dishes p	Florence Florence Florence properly.	
	b.	*	titiqs?iš titiqs-?iiš dry-3.IND Florence is drying	čamaq⊁ čamaq⊁ properly g dishes p	properly.	Florence Florence Florence
(61)	a.		ḥacuk <sup>w</sup> itʔiš ḥacuk-mit-ʔiiš deeply-PST-3.IND Ken was sleeping	wa?ič wa?ič sleep deeply.	Ken Ken Ken	
	b.	*	wa?ičit?iš wa?ič-mit-?iiš sleep-PST-3.IND Ken was sleeping	ḥacuk ḥacuk deeply deeply.	Ken Ken Ken	

What is the suffixation pattern of affixal predicates which take adverbially modified complements? Rose (1981: 296) makes the following general statement about sentential complements: "[p]arallel to NP incorporation, it is the leftmost and highest constituent of the clause governed by the suffix which serves as base to the suffix". Rose's generalization makes the correct predictions about the incorporation pattern of adverbials. In the case of flexibly positioned adverbials, an affixal predicate has the option of attaching to either the adverbial or the verb. In the following (a) examples, the affixal predicate suffixes to the adverbial; in the (b) examples, suffixation is to the verb.

(62)	a.	q <sup>w</sup> a?uuḥqatḥ?iš	ta?i <del>1</del>	Ken
		q <sup>w</sup> a?uuḥ- <b>gaatḥ</b> -?iiš	ta?i <del>1</del>	Ken
		purposely- <u>claim</u> -3.IND	sick	Ken
		Ken is pretending to be s	sick on pu	irpose.
	b.	ta?i <del>1</del> qatḥ?iš	Ken	q <sup>w</sup> a?uuḥ
		ta?i1- <b>qaath</b> -?iiš	Ken	q <sup>w</sup> a?uuh
		sick- <u>claim</u> -3.IND	Ken	purposely
		Ken is pretending to be s	sick on pu	irpose.

## THE LINEARIZATION OF AFFIXES

(63)		a.	čaaniwitassiš	wa?ič		
			čaani- <b>witas</b> -siiš	wa?ič		
			first- <u>gonna</u> -1SG.IND	sleep		
			I'm gonna sleep first (i	.e., before doin	ng somethin	ng else)
		b.	wa?ičwiťassiš	čaani		
			wa?ič- <b>wiťas</b> -siiš	čaani		
			sleep-gonna-1SG.IND	first		
			I'm gonna sleep first (i	.e., before doin	ng somethi	ng else)
(64)	a.	na?iikqa	ıthitsiš	wa⁺ši⊁	?atquu	wiktumsa
. ,		na?iik- <b>q</b>	aath-mit-siiš	wa†-ši⊁	?atquu	wik-tum-sa
		immedia	ately-claim-PST-1SG.INE	go.home-PER	F although	NEG-PST-1SG.DEP
		I claime	ed I went home immedia	tely, but I didr	n't.	
				•		

b. wałśiżqathitsiś na?iik ?atquu wiktumsa wał-šiż-**qaath**-mit-siiš na?iik ?atquu wik-tum-sa go.home-PERF-<u>claim</u>-PST-1SG.IND immediately although NEG-PST-1SG.DEP I claimed I went home immediately, but I didn't.

For adverbials which rigidly precede a predicate, however, the choice of host for the affixal predicate is inflexible: the affix must attach to the adverbial, rather than to the verb. As the following (a) examples illustrate, it is grammatical for the affixal predicate to attach to the manner adverbial. However, the (b) examples show that it is ungrammatical for the affixal predicate to suffix to the verb, rather than the manner adverbial.

(65)	a.		witýaxmaḥsasiš witýax- <u>maḥsa</u> -siiš	waa†šiî wa†-[+I	⊦ _]-šiλ	
			slow- <u>want.to</u> -1SG.IND	go.hom	e-CONT-P	ERF
			I want to go home slowly			
	b.	*	waałši∂maḥsasiš wał-ši∂{[+L]- <u>maḥsa</u> -siiš go.home-CONT-PERF- <b>wan</b>	<b>it.to-</b> 1sG.	IND	witỷax witỷax slow
			I want to go home slowly	•		
(66)	a.		čamaqî~qatḥ?iš čamaqî≁- <b>qaatḥ</b> -?iiš properly- <b>claim</b> -3.IND Florence is pretending to	titiqs titiqs dry dry dishe	Florenc Florenc Florenc s properl	e e y.
	b.	*	titiqsqath?iš titiqs- <b>qaath</b> -?iiš dry- <u>claim</u> -3.IND	čamaq <del>a</del> čamaq <del>a</del> properly	e e y	Florence Florence Florence
			Florence is pretending to dry dishes properly.			

52

(67)	a.		ḥacuƙiiૠitsiš ḥacuk- <u><b>ſiૠ</b></u> -mit-siiš deeply- <u>come.upon</u> -PST-1SG.IND I came upon Ken in a deep sleep.	wa?ič wa?ič sleep	Ken Ken Ken
	b.	*	wa?ičʕiૠitsiš wa?ič- <u>ʕiૠ</u> -mit-siiš sleep- <u>come.upon</u> -PST-1SG.IND I came upon Ken in a deep sleep.	ḥacuk ḥacuk deeply	Ken Ken Ken

This difference in the incorporation pattern of the two sets of adverbials follows from an analysis in which PF incorporation is sensitive to the linear ordering of elements. If an adverbial permits a post-verbal syntactic positioning, then it is possible for the verb to serve as a host for the affixal predicate at local spell-out. However, if an adverbial is only ever linearized preceding the verb, then the verb will not be in the "leftmost" position which allows it to be string adjacent to the affixal predicate at spell-out. The distinct linearization schemes of pre-verbal and post-verbal adverbials are indicated in (68), in which the verbal phrase is the derivational sister to an affixal predicate  $-\alpha$ . Variably positioned adverbs allow either ordering, while manner adverbials require (68a).<sup>7</sup>



At spell-out, the affixal predicate  $-\alpha$  requires a host. In the case of pre-verbal adverbials as in (68a), it is the adverbial which is contiguous to the affixal predicate; as such, the adverbial acts as the host for the affixal predicate. For post-verbal adverbials such as (68b), however, it is the verb which is "leftmost" in the phrase; consequently, the verb is selected as the host for the affixal predicate. Thus, adverbial incorporation in Nuu-chah-nulth shows a sensitivity to independently attested restrictions on linear ordering.

## 2.2.3 Coordinated Objects

According to the local spell-out hypothesis, an affix is linearized with respect to an immediately neighbouring element. The host for an affix is necessarily selected from the derivational sister of the affix. Since this linearization process operates on the basis of string adjacency, the analysis predicts that the internal syntactic structure of the derivational sister to the affix should be irrelevant to the linearization process. This section concerns the specific prediction of this model for coordinated

<sup>&</sup>lt;sup>7</sup> This raises the question of what determines this ordering. See Chapter 3 for discussion.

structures. The string adjacency effect determines that an affix should be linearized relative to a linearly adjacent element in a coordinated object. In the following diagram, a conjunction (CONJ) joins two conjuncts,  $\beta$  and  $\chi$ . This coordinated phrase is the derivational sister of the affix  $-\alpha$ .

(69)



The prediction of the analysis is that in Nuu-chah-nulth  $\beta$  should act as the host of the affix  $-\alpha$ . This section illustrates that this prediction holds.

The behaviour of coordinated objects provides evidence that PF incorporation operates on the basis of linear adjacency in Nuu-chah-nulth. The conjunction  $\lambda u h \partial i i \delta$  (CONJ) is used exclusively to conjoin nominals in Nuu-chah-nulth. Its use is shown in (70), in which it appears between the two conjuncts,  $haak waa \lambda$  "girl" and malikqac "boy" in (70a) and Bill and Mary in (70b).

(70)	a.	naačpiihamitsiš naačpiiha-mit-siiš see-glimpse-PST-1SG.IND I caught a glimpse of a gir	ḥaak <sup>w</sup> aa ḥaak <sup>w</sup> aa girl l and a b	dλ dλ oy.	?uḥ?iiš ?uḥ?iiš CONJ	maʔiઋqac maʔiઋqac boy
	b.	huuł-huuł-amit?iš huuł-a[+R]-mit-?iiš dance-ITER-PST-3.IND Bill and Mary were dancii	Bill Bill Bill ng.	?uḥ?iiš ?uḥ?iiš CONJ	Mary Mary Mary	

First, we must note that the CSC is operative in syntactic movement in Nuu-chahnulth, as with the *wh*-questions shown below. The examples in (71) are object *wh*-questions, while the examples in (72) are subject *wh*-questions. In the (a) examples, we have a grammatical case of *wh*-movement which does not make use of conjunction. In the (b) and (c) examples, however, it is shown that it is ungrammatical for *wh*-movement to target a single conjunct of the argument. These CSCviolating examples are ruled out in Nuu-chah-nulth, parallel to the English cases which are given as their literal translation.

(71)	a.		?aačači†itk ?ačaq-či†[+L]-mit-k who-AUX-PST-2SG.Q Who did you catch a glir	ňaačpiiha ňaač-piiha see-glimpse npse of?		
	b.	*	?aačači†itk ?ačaq-či†[+L]-mit-k who-AUX-PST-2SG.Q ( <i>lit</i> : "Who did you catch	ňaačpiiḥa ňaač-piiḥa see-glimpse a glimpse of and a	?uḥ?iiš ?uḥ?iiš CONJ boy?")	maʔi⊁qac maʔi⊁qac boy

	c.	*	?aačačiłitk ?ačaq-čił[+L]-mit who-AUX-PST-2S ( <i>lit</i> : "Who did yo	t-k G.Q u cat	naačpiiha naač-piiha see-glimpse tch a glimpse	ḥaakʷaa ḥaakʷaa girl of a girl a	ት ት and?")	?uḥ?iiš ?uḥ?iiš CONJ
(72)	a.		?ačaqith ?ačaq-mit-h who-PST-3.Q Who was dancing	huu huu dan g?	tthuuta tt-a[+R] ice-ITER			
	b.	*	?ačaqitḥ ?ačaq-mit-ḥ who-PST-3.Q ( <i>lit</i> : "Who and M	huu huu dan lary	14huu4a 14-a[+R] 1ce-ITER was dancing?	?uḥ?iiš ?uḥ?iiš CONJ ")	Mary Mary Mary	
	c.	*	?ačaqitḥ ?ačaq-mit-ḥ who-PST-3.Q ( <i>lit</i> : "Bill and wh	huu huu dan o wa	rthuuta rt-a[+R] ice-ITER as dancing?"	Bill Bill Bill	?uḥ?iiš ?uḥ?iiš CONJ	

In the case of PF incorporation, however, a different pattern emerges. In the Ahousaht dialect of Nuu-chah-nulth, the first word of a coordinated object is chosen to host an affixal predicate, in striking contrast to the pattern of CSC-obeying syntactic movement. Examples of this characteristic of PF incorporation are shown below. For example, in (73b), the affixal predicate  $2u-k^wistap$  "take away" incorporates the nominal hamuut "bones", leaving stranded the remainder of the conjunction 2uh2iis kuuna "and gold". Additional cases of incorporation targeting the first word of the conjunct are shown in (74b) and (75b).

(73)	a.	?uk <sup>w</sup> ist <sup>a</sup> mit?iš ?u- <u>k<sup>w</sup>ist</u> -'ap-mit-?iiš Ø- <u>move.away</u> -TR-PST-3.IND White people took away the b	mama†'ni mama†'ni white.peo ones and g	hamu hamu ple bone old.	iut ?uḥi iut ?uḥi s COM	?iiš kuuna ?iiš kuuna N gold
	b.	hamuutk <sup>w</sup> ist <sup>2</sup> amit?iš hamuut- <u>k<sup>w</sup>ist</u> -'ap-mit-?iiš bones- <u>move.away</u> -TR-PST-3.1 White people took away the b	mama mama ND white ones and g	rtni rtni .people old.	?uḥ?iiš ?uḥ?iiš CONJ	kuuna kuuna gold
(74)	a.	?u?aamitsiš ?u- <b>'aap</b> -mit-siiš ∅- <u>buy</u> -PST-1SG.IND I bought a canoe and a n	č'apac č'apac canoe et.	?uḥ?iiš ?uḥ?iiš CONJ	čima čima net	
	b.	čapac?amitsiš čapac- <u>'aap</u> -mit-siiš canoe- <u>buy</u> -PST-1SG.IND I bought a canoe and a n	et.	?uḥ?iiš ?uḥ?iiš CONJ	čima čima net	

(75)	a.	?uḥaayʾasči ?u- <b>ḥaa-ʾas</b> -čii Ø- <u>buy-go</u> -2sG.GO Go buy flour and sugar!	⊀i⊀ickuk ⊀i⊀ickuk flour		?uḥ?iiš ?uḥ?iiš CONJ	šuuk <sup>w</sup> aa šuuk <sup>w</sup> aa sugar
	b.	⊀i⊀ickukḥaaỳasči ⊀i⊀ickuk- <u>ḥaa</u> - <u>'as</u> -čii flour- <u>buy</u> -go-2sG.GO Go buy flour and sugar!	?uḥ?iiš šuu ?uḥ?iiš šuu CONJ sug	k <sup>w</sup> aa k <sup>w</sup> aa ;ar		

Under a purely syntactic analysis of incorporation, examples such as (73b–75b) should be banned by the CSC. However, since sensitivity to linear order is a property of phonology (Chomsky 1995), this behaviour is directly predicted under an account which derives these linearizations from spell-out to PF.

In this section, we saw that the linearization of affixal predicates in Nuuchah-nulth shows an insensitivity to syntactic constituency. The next section discusses the observation that this process is similarly blind to syntactic category.

# 2.3 Insensitivity to Syntactic Category

The local spell-out hypothesis states that an affix is linearized with whatever element is adjacent to the affix at the point of spell-out. This linearization is predicted to be insensitive to syntactic category, as the process is attuned to string adjacency. As this section demonstrates, PF incorporation in Nuu-chah-nulth is unselective for syntactic category.

Affixal predicates which select nominal complements may incorporate a noun, adjective, quantifier, *wh*-pronoun, or relative pronoun. This range of possible hosts is illustrated in (76) for the affixal predicate *?u-?aap* "buy".

(76)	a.	č'upč'upšumł?amitsiš č'upč'upšumł- <b>'<u>aap</u>-mit-sii</b> š sweater- <u>buy</u> -PST-1SG.IND I bought a sweater.	ŝ		(noun)
	b.	ởiḥ?aamitsiš ởiḥ- <u>'aap</u> -mit-siiš red- <u>buy</u> - PST-1SG.IND <i>I</i> bought a red sweater.	čupčupšum† čupčupšum† sweater	siỷa siỷa 1SG	(adjective)
	c.	hiỷaapatuk?iš hiš- <u>'aap</u> -'at-uk-?iiš all- <u>buy</u> -PASS-POSS-3.IND All his/her necklaces were	nuutinum(minh) nuutinum(-minh) necklace(-PL) bought.		(quantifier)

d.	?aqi?amitḥ ?aqi- <u><b>'aap</b>-</u> mit-ḥ what- <u>buv</u> -PST-3. What did Louis t	Q ouy?	Louis Louis Louis	(wh-pronoun)
e.	k <sup>w</sup> in?a†it?iš	John	⁺uč?in	yaSaamit?itk
	k <sup>w</sup> in?a†-mit-?iiš	John	⁺uč?in	yaq- <u>'<b>aap</b>-mit-?iitk</u>
	like-PST-3.IND	John	dress	REL- <u>buy</u> -PST-2SG.RL
	John liked the dr	ess you	bought.	<i>(relative pronoun)</i>

In a strict sense, PF incorporation is therefore not equivalent to "noun incorporation". Although an affixal predicate (which takes a nominal complement) is capable of incorporating a noun, elements with a range of other syntactic categories can serve the role of host.

The same is true for the phenomenon described as "verb incorporation" in Nuu-chah-nulth. As previously discussed, adverbial modifiers show the ability to incorporate, along with verbs.

(77)	a.	kamatqukmaḥsaʔiš	Flore	nce
		kamatq-uk- <u><b>maḥsa</b></u> -?iiš	Flore	nce
		run-DUR- <u>want.to</u> -3.IND	Flore	nce
		Florence wants to run.		
	b.	≁a?ixmaḥsa?iš	kamatquk	Florence
		≁a?ix- <u>maḥsa</u> -?iiš	kamatq-uk	Florence
		fast- <u>want.to</u> -3.IND	run-DUR	Florence
		Florence wants to run fas	st.	

The negative particle *wik* can also incorporate into an affixal predicate. However, this pattern is restricted by constraints on string adjacency, and is thus indirectly conditioned by syntactic structure. In Nuu-chah-nulth, suffixation to the negative particle *wik* applies only with "low" scope negation, and not with "high" scope negation. In the "low" negation example of (78a), the affixal predicate -qaath "claim" suffixes to *wik* NEG; in the "high" negation example of (78b), the affixal predicate instead suffixes to the verb  $\lambda u$ -uc "own".

(78)	a.	wikqaath?iš	Ken	?uuc	šuwis?i
		wik- <b>qaath</b> -?iiš	Ken	?u-ic	šuwis-?ii
		neg- <u>claim</u> -3.ind	Ken	Ø-own	shoes-DET
		Ken claims he do	besn't own the sho	es.	('low' negation)
	b.	wik?iiš	?uucqatḥ	Ken	šuwis?i
		wik-?iiš	?u-ic-qaath	Ken	šuwis-?ii
		neg-3.ind	Ø-own- <u>claim</u>	Ken	shoes-DET
		Ken doesn't claim	m to own the shoe	s.	('high' negation)

As with adverbial incorporation, the availability of *wik* NEG to incorporate relates to linear ordering restrictions on the incorporation process. As Rose (1981: 296) describes, "it is the left-most and highest constituent of the clause governed by the suffix which serves as base to the suffix". With "low" negation, *wik* NEG hosts the affixal predicate, while in "high" negation, a verb takes over as host. This can be related to the spell-out position of the respective hosts. In "low" negation, the negative particle *wik* appears in the derivational sister to the affixal predicate –*qaath* "claim", and is spelled-out adjacent to the affixal predicate. (Chapter 5 provides argumentation for the syntactic representation that I assume here.) In "high" negation, however, the verb  $\partial u$ -uc "own" is spelled-out adjacent to the affixal predicate. Only in "low" negation does the negation particle *wik* (NEG) fall into a position where it is the leftmost element in the derivational sister of –*qaath* "claim", as shown in (79a). In (79b), in contrast, it is the verb  $\partial u$ -uc "own" which is the leftmost element of the derivational sister of –*qaath* "claim".



Thus, due to the string adjacency effect on incorporation, for (79a), a linearization of  $\langle wik-qaath ?uuc šuwis?i Ken \rangle$  is anticipated. In (79b), the ordering of  $\langle ?uucqaath suwis?i Ken \rangle$  is predicted. This linearization mechanism is not sensitive to a difference in syntactic category between the negation host wik (NEG) and the verbal host ?u-uc "own".

# 2.4 Absence of LF Effects

Under a model in which LF effects are restricted to the narrow syntax (Chomsky 1995), spell-out to PF is predicted to have no semantic effects. This section considers two domains in which an observable LF effect is absent from Nuu-chahnulth PF incorporation: first, the referentiality of the incorporee (§2.4.1), and second, scopal effects (§2.4.2).

#### 2.4.1 Discourse Transparency

In many languages in which verbs affix to nouns, the incorporated noun shows distinct referential properties from independent nouns in the language (Mithun 1984). As Mithun reports, in Huahtla Nahuatl, when a noun is first introduced in the discourse, it must not be incorporated. The following examples are from Mithun (1984: 860–861, ex. 58).

(80)	Nahuai	tl			
	A:	askem	nan	ti-'-kwa	nakatl.
		never		you-it-eat	meat
		You n	ever eat me	eat.	
	B:	na'	ipanima	ima ni-naka-kwa.	
		Ι	always	I-me	eat-eat
		I eat it	t (meat) all	the time.	

In the first portion of this discourse, the noun *nakatl* "meat" occurs independently of the verb *kwa* "eat". In the following sentence, however, *naka* "meat" is able to be incorporated into the verb. The incorporability of the noun corresponds to its discourse role in Nahuatl. The restriction that incorporated nouns may not appear as the initial token of the argument suggests that incorporated nouns in Nahuatl cannot establish a discourse referent.

If incorporation in Nuu-chah-nulth occurs at spell-out to PF, then it is predicted that incorporated nouns should be discourse transparent. That is, the spell-out properties of the noun should have no consequence for discourse effects. Indeed, as this section demonstrates, incorporation of an element into an affixal predicate in Nuu-chah-nulth does not affect this element's ability to occur as a discourse antecedent (Rose 1981). The referentiality of a Nuu-chah-nulth incorpore is indicated in examples (81a–c). In (81a), the incorporated bound nominal *čapx-* "man" serves as the discourse referent for the pronoun (*pro*) "he" in *?aphiqaqit?is* "He was very friendly". Similarly, the free nominal *siičpax* "cougar" which is incorporated into the affixal predicate in (81b) acts as the antecedent for the object of the following sentence, *qaaccanakuuhitwa?is John* "John saw three (of them)". A final example of the discourse transparency of incorporated elements is given in (81c), in which *?aħtaqumf* "two kinds" is incorporated into the affixal predicate, but still serves as the discourse referent for the following question *waayaqhwatmahsak* "Which of them do you want to use?".

(81)	a.	čapxčaasitsiš	hintši⊁	?apḥiqaqit?iš
		čap <sub>.</sub> x- <u>čaas</u> -mit-siiš	hint-ši⊁	?apḥiiq-aq[+S]-mit-?iiš
		man- <u>beside</u> -PST-1SG.IND	come-PERF	friendly-AUG-PST-3.IND
		I was (sitting) beside a ma	in coming this	way. He was very friendly.

b.	sičpaxpa†?iš	qaaccanakuuhitwa?iš	John
	siičpax- <b>pa½</b> [+S]-?iiš	qacca- <b>nakuuh</b> [+L]-mit-wa?iš	John
	cougar-around-3.IND	three-observe-PST-3.QUOT	John
	There's cougars aroun	d. John saw three (of them).	

c. ?aλtaqumłnaksiš waayaqhŵałmahsak
 ?aλ-taqumł-<u>naak</u>-siiš waayaq-<u>hŵał[</u>+L]-<u>mahsa</u>-k
 two-kinds-<u>have</u>-1SG.IND which-<u>use-want</u>-2SG.Q
 I have two sets (e.g., of lahal bones). Which (of them) do you want to use?

The referentiality of the incorpore follows if PF incorporation has no LF consequences.

# 2.4.2 Absence of Scopal Effects

Additional evidence for the lack of an LF effect comes from quantifier scope. As predicted by the local spell-out hypothesis, the choice of host for an affixal predicate should have no effect on scope at LF. As predicted, the surface order derived by PF incorporation appears to have no consequence for quantifier scope. Quantified subjects are ambiguous between wide and narrow scope over objects if the object is incorporated into the affixal predicate or if it is not.<sup>8</sup> In (82), the object *muunaa* "engine" of the affixal predicate *?uu-taq* "fix, work on" hosts the predicate, and linearly precedes the quantifier *hišuk* "all". Both a wide scope and a narrow scope interpretation of the quantifier are permitted.

- (82) a. muunaataqit?iš hišuk čaakupiih. čaawacnasaat?at muunaa-taq[+L]-mit-?iiš hišuk čakup-iih[+L] čawa-nasaat[+L]-Ø-?at motor-fix-PST-3.IND all-DUR man-PL one-handle-3.ABS-PL All the men were working on an engine. They were all working on the same one.
  - b. muunaataqit?iš *hišuk* čaakupiiḥ ćaćawanak muunaa-<u>taq[+L]-mit-?iiš</u> hiš-uk čakup-iiḥ[+L] ćawa-naak-∅-[+R] motor-<u>fix</u>-PST-3.IND all-DUR man-PL one-have-3.ABS-PL *All* the men were working on an engine. They each had one.

In the following example, *muunaa* "engine" does not host the affix, and it instead appears below the quantifier. Here, the expletive  $\lambda \mu$  hosts the affixal predicate  $\lambda uu$ -taq "fix, work on". Still, both scope interpretations are possible.

(83) a. ?uutaqit?iš hišuk čakupiih muunaa ćaawacna?aa+?a+?u-taq[+L]-mit-?iiš hiš-uk čakup-iih[+L] muunaa ćawa-na?aa+?+L]-Ø-?a+Ø-fix-PST-3.IND all-DUR man-PL motor one-handle-3.ABS-PL All the men were working on an engine. They were all working on the same one.

<sup>&</sup>lt;sup>8</sup> The source of this ambiguity remains to be determined.

b.	?uutaqit?iš	hišuk	čakupiiḥ	muunaa	ćaćawanak
	?u- <u>taq</u> [+L]-mit-?iiš	hiš-uk	čakup-iiḥ[+L]	muunaa	ċawa-naak-∅-[+R]
	Ø- <u>fix</u> -pst-3.ind	all-DUR	man-PL	motor	one-have-3.ABS-PL
	All the men were w	orking on	an engine. The	ey each had	l one.

The equivalence of (82a–83a) and (82b–83b), respectively, follows under an analysis in which spell-out to PF has no LF effect.

## 2.5 Opacity Effects

As I have described, an affixal predicate in Nuu-chah-nulth incorporates a host chosen from its derivational sister, the complement of the verb. This linearization process has been stated to be insensitive to syntactic category. All else being equal, we should therefore expect that incorporation should be possible when any of NP, DP, vP, or CP are the complements of the affixal predicate.



This section presents evidence that not all else is equal. I hypothesize that DP and CP differ from other projections in that they are *saturated domains* of the derivation (cf. the notion of "phase" in Chomsky 2001, 2005). A variety of syntactic evidence has been presented for an inherent symmetry between D and C, to the exclusion of other categories (Abney 1987; Szabolcsi 1994). According to these analyses, DPs and CPs possess parallel functional structures. Endowed by their make-up is the capacity for propositional independence – a symmetry reflected in (85), in which the DP and CP qualify equally as the propositional complement of *recall*.

(85) a. I recalled [the city's destruction]<sub>DP</sub>.
b. I recalled [that the city was destroyed]<sub>CP</sub>.

According to the saturated domain hypothesis, DPs and CPs constitute completed units of the derivation, in that the needs of the members of DP and CP are met within these respective domains.

(86) Saturated domain hypothesis: The contents of DP and CP are fully interpreted.

As will be described in this section, the consequence of saturation is derivational inertness. The opacity effects associated with these domains arise as a section of the derivation is completed, and consequently abandoned. As a D(P) or C(P) is introduced, the contents of these saturated phrases thus become impervious to further derivational manipulations (cf. Chomsky 2001; Svenonius 2004; Fox and Pesetsky 2005). In effect, the presence of a D(P) or C(P) imposes a "border" between portions of the derivation, as the details of the make-up of DPs and CPs are no longer open to the workspace of the derivation. An attractive conceptual consequence of this

postulation is that it reduces the amount of information held in active memory (Chomsky 2001; Matushansky 2005), as only one portion of the derivation is considered at a time. In this way, saturation is tied to derivational independence: once a domain becomes saturated, the workspace is closed as it turns to unsaturated material. This characteristic reflects economy, as the derivation works to satisfy "needy" elements, and no more. As this section demonstrates, the opacity effects of Nuu-chah-nulth affixation arise due to the impermeability of saturated domains in this language: Nuu-chah-nulth lacks any "escape hatch" at the border of CP and DP which would allow a host to cross these opaque derivational domains.

In Nuu-chah-nulth, there is salient prosodic evidence that DPs and CPs constitute phonologically independent constituents. This evidence comes from the two distinct cliticization domains found in Nuu-chah-nulth. (See Chapter 3 for further discussion.) Clitic strings may be built up within a DP, or at a clausal level which excludes the DP(s). In (87), there are two separate cliticization domains, indicated by bracketing. In the CP domain, the clitics -mit (PST) and -siis (1SG.IND) are found; in the DP domain are placed the clitics -uk (POSS) and -2iitk (2SG.PS).

(87)	[hiixtaqči <i>mitsiš</i> ] <sub>CP domain</sub>	[huupuuk <sup>w</sup> as <b>uk?itk</b> ] <sub>DP domain</sub>
	hiixtaq-čip- <i>mit-siiš</i>	huupuuk <sup>w</sup> as- <b>uk-?iitk</b>
	have.accident-BEN-PST-1SG.IND	car-POSS-2SG.PS
	I had an accident with your car.	

As presented in more detail in Chapter 3, these DP-level and CP-level clitics may each be described as "second position" morphemes: they occur (potentially in a string) at the left edge of the phrase, attached in "second position" as a suffix to a host. The saturated domain hypothesis presents a straightforward means of explaining how two sets of items (DP-level and CP-level clitics) may each equally be described as "second position" morphemes, when in fact they occur in distinct positions in the clause. According to the saturated domain hypothesis, there is no single "second position" in a clause, because the derivation is considered in separate chunks: CPs and DPs are independent domains, and each has their own "second position".

In §2.5.1 and §2.5.2, I show that PF incorporation can never cross a DP or CP in Nuu-chah-nulth. According to my proposal, members of DPs and CPs are inaccessible as hosts for a sister affixal predicate because these derivational units are saturated domains. This opacity effect is schematized in (88). As illustrated, a "border" marks the edge between the affixal predicate and the saturated domains taken as complement. When an affix  $-\alpha$  taking a DP or CP complement is spelled out in Nuu-chah-nulth, the expletive  $\lambda - \mu$  must surface in order to provide a host for the affix and rescue the derivation.



In §2.5.3, I discuss the source of this opacity effect under the local spell-out model. This explanation relies on the distinction between the interpretive capacity of
spell-out, and the inaccessibility of saturated domains in the derivation. The opacity effects of Nuu-chah-nulth arise as domains become saturated.

# 2.5.1 Opaque DPs and the Bare Nominal Requirement

In Nuu-chah-nulth, an affixal predicate may not suffix to an element of its complement when the phrase contains the determiner -2*ii*. The consequence of this restriction is a bare nominal requirement on incorporation. In (89a), incorporation targets the bare nominal  $\frac{1}{2}u\ddot{c}2in$  "dress". The example in (89b) indicates that a nominal marked with the determiner -2*ii* cannot be incorporated. Furthermore, as shown in (89c), the determiner cannot itself act as a host for the affixal predicate When the nominal complement of an affixal predicate is marked with a determiner,  $\frac{1}{2}u$ -support must occur, as shown in (89d).

- (89) a. <sup>1</sup>uč?insiikitsiš <sup>1</sup>uč?in-<u>siik</u>-mit-siiš dress-<u>make</u>-PST-1SG.IND I made a dress.
  - b. \* łuč?in?isiikitsiš łuč?in-?ii-<u>siik</u>-mit-siiš dress-DET-<u>make</u>-PST-1SG.IND I made the dress.
  - c. \* ?iisiikmitsiš +uč?in ?ii-<u>siik</u>-mit-siiš +uč?in DET-<u>make</u>-PST-1SG.IND dress I made the dress.
     d. ?usiikitsiš +uč?in?i
  - d. rushklisis fuchtarrow ruchtarrow ruchta

This restriction may be attributed to the identity of DPs as saturated units. The presence of a determiner, as in (89b, c), closes a "border" between the affixal predicate and its complement. In such contexts, an expletive host must be inserted to satisfy the affixal predicate's affixation requirement, as in (89d).

This restriction on incorporation in contexts with the determiner -2ii can also be illustrated in the following stretch of discourse. The sentences in (90) establish the context for the test sentences in (91).

(90)	wa†aakitsiš	naquwas	?atḥiimit?i
	wa†aak-mit-siiš	naqu-was	?atḥii-mit-?ii
	go.to-PST-1SG.IND	drink-building	night-PST-DET

?uk <sup>w</sup> ink	?uušḥỷ	umsukqs	Lucy		qacciii+it?iš
?u-k <sup>w</sup> ink	?uušḥỷı	ums-uk-qs	Lucy		qacca-ii+-mit-?iiš
∅-with	friend-F	vOSS-1SG.PS	Lucy		three-AUX-PST-3.IND
čaakupiiḥ	[+L]	mama†ỉi	huwayii?atḥ	?uḥ?iiš	quu?as
čakup-iiḥ[		mama†ỉi	huwayii?atḥ	?uḥ?iiš	quu?as
man-PL		white	black	and	First.Nations

I went to the bar last night with my friend Lucy. There were three guys there: a white guy, a black guy, and a First Nations guy.

This context establishes the nominal *huwayii?ath* "black person" as discourse-familiar. In this context, the preferred means of stating "Lucy liked the black man" is with an expletive host for the affixal predicate *?uu-?a+sumhi* "to like", as in (91a). It is marked for the definite nominal *huwayii?ath* "black person" to be incorporated, as shown in (91b). Furthermore, it is not possible to incorporate the nominal if it is marked by the determiner *-?ii*, as in (91c).

(91)	a.		?uu?a†sumḥimit?iš	Lucy	huwayii?atḥ?i
			?u- <u><b>'a1sumḥi</b>[</u> +L]-mit-?iiš	Lucy	huwayii?ath-?ii
			Ø- <u>like</u> -pst-3.ind	Lucy	black.person-DET
			Lucy liked the black man.		
	b.	#	huwayii?ath?a+sumhimit?iš	Lucy	
			huwayii?atḥ- <u><b>'a1sumḥi</b></u> -mit-?iiš	Lucy	
			black.person-like-PST-3.IND	Lucy	
			Lucy liked the black man.	-	
	c.	*	huwayii?ath?i?a+sumhimit?iš	Lucy	
			huwayii?atḥ-?ii- <u>'alsumḥi</u> -mit-?iiš	Lucy	
			black.person-DET-like-PST-3.IND	Lucy	
			Lucy liked the black man.		

The restriction that a discourse-familiar nominal cannot serve as a host for the affixal predicate follows from the bare nominal requirement on Nuu-chah-nulth incorporation. I attribute this effect to the status of DP as a saturated constituent. The content of this saturated domain is hypothesized to be inaccessible as a host for a higher affixal predicate.

## 2.5.2 Opaque CPs and "Restructuring" Effects

Parallel examples may be supplied to illustrate a ban on incorporation across CPs. Incorporation is possible only out of uninflected complements, not full CPs (see Chapter 5). In (92a), an example is given showing a full CP complement for the affixal predicate 2uu-*nakuuh* "observe". This full complement contains the complementizer 2in (COMP), the past tense marker -mit (PST), and the dependent mood inflection -suuk (2SG.DEP). In contexts of verb incorporation, it is ungrammatical for such

clausal demarcations to appear, as indicated in (92b). The grammatical instance of verb incorporation in (92c) shows no complementizer, no tense marking, and no dependent mood inflection. In effect, the clausal content of (92c) is "restructured" to exclude these clausal demarcations.

(92)	a.		?uunakuuhitsiš?in?u-nakuuh[+L]-mit-siiš?inØ-observe-PST-1SG.INDCOMPI observed that you were jumping.		tuuxtuux <sup>w</sup> amitsuk tuux <sup>w</sup> -a[+R]-mit-suuk jump-IT-PST-2SG.DEP	
	<ul> <li>tuuxtuux<sup>w</sup>amitňakuuhitsiš</li> <li>tuux<sup>w</sup>-a[+R]-mit-<u>ňakuuh</u>[+L]-mit-siiš</li> <li>jump-IT-PST-<u>observe</u>-PST-1SG.IND</li> <li>I observed you jumping.</li> </ul>		siiš	(?in) (?in) (COMP)	suwa suwa you	
	c.		tuuxtuux <sup>w</sup> anakuuhitsiš tuux <sup>w</sup> -a[+R]- <b>nakuuh</b> [+L]-mit-siiš jump-IT- <u>observe</u> -PST-1SG.IND I observed you jumping.	suẁa suẁa you		

In Chapter 5, I will analyse the verb incorporation case in (92c) as having a vP complement, rather than the sort of CP complement in (92a). The inability of incorporation to occur across a CP complement may be attributed to the status of CP as a saturated domain, whose "border" has closed. When no CP structure is present, there is no demarcation imposed between the affixal predicate and its complement.

# 2.5.3 Impermeability of Saturated Domains

This section presents additional evidence for the derivational independence of the saturated domains CP and DP, based on proof of a ban on cross-domain syntactic movement in Nuu-chah-nulth. With respect to the CP domain, Nuu-chah-nulth has been shown to lack cross-clausal *wh*-movement (Davis and Sawai 2001). This restriction is illustrated below with an example in which the matrix predicate taquk "believe" takes an embedded clause as complement. A grammatical declarative sentence is shown in (93a). In (93b,c), it is shown to be ungrammatical to *wh*-question the subject of the embedded clause.

- (93) a. ťaaquk-čiλ-'aλ-?iiš John ?in kuuŵił-itḥuk Mary č'apac ťaaquk-čiλ-'aλ-?iiš John ?in kuuŵił-mit-ḥuuk Mary č'apac believe-PERF-TEMP-3.IND John COMP steal-PST-3.DEP Mary canoe John believes that Mary stole the canoe. (cf. Davis and Sawai 2001: 133)
  - b. \* ʔačaqh ťaaquk<sup>w</sup>iʔaλ John ʔin kuuŵiłithuk čapac ʔačaq-h ťaaquk-čiλ-ʾaλ John ʔin kuuŵił-mit-huuk čapac who-3.Q believe-PERF-TEMP John COMP steal- PST-3.DEP canoe Who does John believe stole the canoe? (cf. Davis and Sawai 2001: 133)

c. \* ?ačaqḥ ťaaquk<sup>w</sup>i?aૠ John (?in) kuuŵiłitḥ č'apac ?ačaq-ḥ ťaaquk-čiૠ-'aૠ John (?in) kuuŵił-mit-ḥ č'apac who-3.Q believe-PERF-TEMP John (COMP) steal-PST-3.Q canoe Who does John believe stole the canoe?

*Wh*-movement which crosses a CP is not possible in Nuu-chah-nulth. Instead, as I describe in Chapter 5, "long" *wh*-movement in Nuu-chah-nulth only occurs in monoclausal "restructuring" environments. The sentence in (94) provides an example of such a monoclausal *wh*-question.

(94)	?ačaqqatḥ?apḥ	John	kuuwi1	čapac			
	?ačaq- <b>qaatḥ</b> -'ap-ḥ	John	kuuwi1	čapac			
	who-claim-TR-3.Q	John	steal	canoe			
	Who does John claim stole the canoe?						

As argued in Chapter 5, the embedded clause in (94) is "reduced" in that it fails to project up to a CP. As such, movement is possible out of this unsaturated domain.

Likewise, there is also indication that syntactic movement out of a DP is restricted in Nuu-chah-nulth. Possessor raising in Nuu-chah-nulth is possible only out of reduced nominal phrases, which lack the outer housing of a DP. Ravinski (2005) illustrates that possessor raising is illicit out of a nominal phrase marked with the determiner -2ii (DET). The sentences in (95) exemplify this pattern. In these examples, the predicate is suffixed by the possessive morpheme -uk (POSS), as is characteristic of possessor raising in the language. The contrast in grammaticality between (95a) and (95b) indicates that possessor-raising may not occur when the determiner appears in the possessive nominal.

(95)	a.		λu†uk?iš λu†-uk-?iiš	huupuk <sup>w</sup> as huupuk <sup>w</sup> as	Lucy Lucy
			good-POSS-3.IND	car	Lucy
			Lucy's car is nice	e. (Ravir	ski 2005: 65, ex. 177a)
	b.	*	λu≁uk?iš λu≁-uk-?iiš	huupuk <sup>w</sup> as?i huupuk <sup>w</sup> as-?ii	Lucy Lucy
			good-POSS-3.IND	car-DET	Lucy
			Lucy's car is nice	e. (Ravin	ski 2005: 65, ex. 177b)

The determiner -2ii overtly marks a DP. The necessary absence of this determiner in contexts of possessor raising is consistent with an analysis in which only reduced nominals permit extraction. Such a restriction is in line with the status of DPs as impermeable domains in Nuu-chah-nulth.

Thus, restrictions on *wh*-movement and possessor-raising corroborate the claim that CPs and DPs in Nuu-chah-nulth are independent domains of the derivation whose borders are closed to derivational manipulation. Taken together with the evidence for two separate prosodic domains of cliticization, we see that phonological and syntactic criteria in Nuu-chah-nulth converge on a representation of CPs and

66

DPs as saturated domains. In Nuu-chah-nulth, these saturated domains are impermeable in that elements from one domain may not "escape" to another. In Chapter 6, we revisit the permeability of saturated domains, and examine cross-linguistically attested "escape hatches" for movement. It is suggested that long-range movement arises from incomplete interpretation.

# 2.6 Conclusion

In this chapter, evidence was presented for the analysis that the positioning of affixal predicates in Nuu-chah-nulth is a reflex of the need to linearize bound morphemes. According to the proposal, the affixal predicate must find a host from within its derivational sister at the point of spell-out. This derivational sister is a linearized object, leading to the restriction that suffixation operates on the basis of linear adjacency. I have argued that the notion of local spell-out elegantly captures the dual sensitivities which affixal predicates show to derivational sisterhood and linear adjacency. The syntax indirectly conditions the input to the linearization process through its composition of local spell-out domains. Two conditioning effects of the syntax were emphasized: the first is the locality requirement induced by the binary concatenation of the syntax, which yields a complement restriction in Nuu-chahnulth; the second, the creation of DP and CP constituents which form distinct domains for affixation processes.

A purely syntactic analysis of incorporation fails to predict the sensitivity which Nuu-chah-nulth affixal predicates have to linear adjacency. As described in §2.2.2, an affixal predicate incorporates a modifier, rather than the element which syntactically heads the complement. Such insensitivity to hierarchical adjacency is at odds with the syntactic incorporation process described by Baker (1988), as it violates the syntactic Head Movement Constraint (Travis 1984). Moreover, in §2.2.3, it was shown that incorporation in Nuu-chah-nulth displays an insensitivity to a restriction on extraction from a conjoined object, the CSC. This constraint was shown to hold in Nuu-chah-nulth in contexts of true syntactic movement.

Just as a purely syntactic analysis of Nuu-chah-nulth fails, so too does a strictly phonological one. Note that the host for an affixal predicate cannot be defined in strictly phonological terms, such as a syllable or a prosodic foot (cf. Halpern 1992). Instead, the host is equivalent to a unit of the syntactic derivation (i.e., one of the two elements treated by syntactic Merge). As the following examples illustrate, hosts for an affixal prediate in Nuu-chah-nulth can be mono- or poly-syllabic. The host in (96a), *?uuš* "someone", is monosyllabic. The host in (96b), *quu?ac-*"(First Nations) person", is bisyllabic. Finally, the host in (96c), *mama†niq-*"white person" is trisyllabic.

(96) a. ?uušił?aλ?iš
 ?uuš-'<u>ił</u>-'aλ-?iiš
 someone-<u>inside</u>-TEMP-3.IND
 There's someone inside (the house) now.

- b. quu?acił?iš quu?ac-'<u>ił</u>-?iiš person-<u>inside</u>-3.IND There's a person inside (the house).
- c. mamałńiqił?iš mamałńiq-'<u>ił</u>-?iiš white.person-<u>inside</u>-3.IND There's white people inside (the house).

Feet in Nuu-chah-nulth are maximally bisyllabic (Werle 2002; Stonham 2004). Thus, hosts for an affixal predicate may be equal to, smaller than, or larger than a prosodic foot. These mismatches with prosodic definitions lend support to a syntactically constrained definition of the host (i.e., a derivational sister).

In the following chapters, it will be shown that the combinatory properties of affixal predicates are conditioned by their argument structure. The syntactic positions in which arguments are introduced indirectly constrain the post-syntactic linearization operation of affixation. Argument structure determines derivational sisterhood, which in turn defines the domains which local spell-out applies to. The next chapter serves as an introduction to the clausal architecture of Nuu-chah-nulth.

# 3. Clausal Architecture of Nuu-chah-nulth

Every phrase and every sentence is an end and a beginning... ~T.S. Eliot Four Quartets no. 4

#### **3.0 Introduction**

Up until this point, there has been an implicit assumption that affixal predicates take an argument as their derivational sister. The following chapters will provide evidence for such a conclusion. In this chapter, I take a first step towards this analysis by arguing that nominal arguments of Nuu-chah-nulth predicates are introduced within verbal projections of the affixal predicate. This analysis has consequences for the local spell-out hypothesis, since it determines that an argument (or part of an argument), as the derivational sister of an affixal predicate (V), will be selected as the host for an affixal predicate.



This chapter adopts the analysis that the "basic" word order for Nuu-chah-nulth is VOS. I propose that this word order is not derived via movement; instead this ordering results from a syntactically underived predicate-initial system in which subjects appear in right-linearized specifier positions. The consequence of this predicate-initial system is that when an affixal predicate (V<sub>1</sub>) takes a propositional complement, the embedded verb (V<sub>2</sub>) precedes the embedded arguments as the leftmost element in the complement. This is represented in (2), in which the affixal predicate's complement (circled in the diagram) is assumed to be equivalent to vP.



The *v* head is represented here as phonologically null  $(\emptyset)$ .<sup>1</sup> As such, the embedded verb  $(V_2)$  is string adjacent to the affixal predicate,  $(V_1)$ . By the string adjacency

<sup>&</sup>lt;sup>1</sup> The v may be overt or covert in Nuu-chah-nulth. If overt, the v is realized as an aspectual suffix on the embedded verb.

effect, it therefore follows naturally that the verb  $(V_1)$  should select the embedded verb  $(V_2)$  as its host at spell-out.

In the next section, we turn to a discussion of how the PF component is responsible for the linearization of syntactic terminals. According to the analysis, syntactic outputs are linearized at the point of spell-out.

#### 3.1 Linearization of Syntactic Terminals

In the framework assumed for this book, the syntax is equivalent to the structurebuilding operations of Merge or Move ("remerge"). The syntactic constructs formed by this binary concatenation are unspecified for linear order.

# (3) Merge: concatenate $\alpha$ with $\beta$ , forming $\gamma$

Although the binary concatenation of Merge or Move fails to impose a linearization of the joined elements, it is nonetheless an unmistakable characteristic of natural language that linear orderings do exist. In the following English sentence, for example, *not* must precede *rushed*, and not the reverse.

(4) a. I am [not rushed]. b. \* I am [rushed not].

If linearization is not reducible to the structure-building properties of the syntax, then what is responsible for its effects? In the Minimalist model, the linear ordering of linguistic constructs constitutes a bare output condition at PF (Chomsky 1995). The temporal nature of speech determines that linguistic representations must be mapped to a linearized speech stream. Thus, the "phonologization" of syntactic constructs entails that relative orderings must be imposed on the syntactic terminals  $\{\alpha, \beta\}$ . In Chapters 1 and 2, I introduced the idea that affixation is one means by which ordering may be imposed on these elements. However, this linearization mechanism is only applicable if one or the other of  $\alpha$  and  $\beta$  is an affix. Nothing has yet been said about linearization in non-affixal contexts.

Linearization is itself necessary, but what particular linearization scheme of syntactic terminals is required? Two hypotheses may be distinguished, one of which attributes linearization to an invariant syntax-phonology mapping, and the other which opens the door to variations in linearization schemes for syntactic terminals.

(5) *PF Linearization mechanisms* (i) Universal linearization scheme
 (ii) Language-specific linearization

In the following subsections, I will consider each of these possibilities in turn, and will make arguments for the latter.

# 3.1.1 A Universal Linearization Scheme?

According to the Universal Base Hypothesis (Lakoff 1970; Bach 1968), all languages share a common architecture. With respect to linearization schemes, Kayne (1994) presents the hypothesis that languages have an invariant mapping algorithm from syntax to linear ordering. According to Kayne's Linear Correspondence Axiom (LCA), the structural relationships of the syntax unambiguously determine ordering in that asymmetric c-command entails precedence. Originally formalized as a syntactic filter, this principle has been recast in various Minimalist approaches as a post-syntactic principle which applies at spell-out (e.g., Chomsky 1995; Dobashi 2003). This linearization scheme may be defined by statement (6):

# (6) *Linear correspondence axiom (LCA)*

If X asymmetrically c-commands Y, then the terminals in X precede the terminals of Y.

This axiom entails that the sole linearization option available for a head projecting a specifier (Spec) and taking a complement (Comp) is that shown in (7a).



#### (7) *Linearization schemes*

C-command relationships of the above configurations can be evaluated according the following definition in (8), supplied by Reinhart (1979).<sup>2</sup>

 $<sup>^2</sup>$  Epstein (1999) proposes a derivational notion of c-command which follows from the properties of binary concatenation. According to derivational c-command, an element c-commands those elements with which it was paired in the course of a derivation. The choice of a representational or derivational view of c-command does not affect the argument here.

(8) X c-commands Y iff

(i) The first branching node dominating X dominates Y, *and*(ii) X does not dominate Y, *and*(iii) X is not equal to Y

Asymmetric c-command, in turn, may be determined through reference to the following additional statement:

(9) X asymmetrically c-commands Y iff
 (i) X c-commands Y, and
 (ii) Y does not c-command X

Let us now consider how the LCA applies to each of the linearization schemes in (7). The "Spec – Head – Comp" configuration of (7a) satisfies the LCA, since the specifier ( $\gamma$ ) both asymmetrically c-commands and precedes the head (terminal  $\alpha$ ); the head (terminal  $\alpha$ ), in turn, asymmetrically c-commands and precedes the complement ( $\beta$ ). Assuming transitivity of precedence relations, asymmetric c-command therefore exhaustively orders the terminals in (7a). The structure of (7b), in contrast, violates the precedence requirement with respect to sequencing the head ( $\alpha$ ) *following* the complement ( $\beta$ ). According to the LCA, since  $\alpha$  asymmetrically c-commands  $\beta$  in (7b),  $\alpha$  should precede  $\beta$ , not follow it. The configuration in (7c) shows a similar precedence violation involving the specifier ( $\gamma$ ). By the LCA, it is impermissable for  $\gamma$  to follow the head and complement, since this node asymmetrically c-commands them. Finally, the option in (7d) is ruled out by the LCA, since its linear ordering is the reverse of that required by the LCA. In (7d), nodes follow, not precede, the terminals they asymmetrically c-command.

If the LCA is adopted, directionality parameters are inapplicable as a means of accounting for different surface word orders. By the LCA, only a single linearization scheme is universally available, whether it applies to the base-generated syntax (as in the pre-Minimalist formulation originally proposed by Kayne 1994), or in the syntax to PF mapping (as in Chomsky 1995). The consequence of this proposal is that word order variations between languages (Greenberg 1966) must derive from different combinations of movement. Let us now consider a key conceptual problem facing the LCA hypothesis.

According to the LCA, asymmetric c-command induces a linear ordering between syntactic terminals. Kayne (1994: 36) argues that this hierarchical relationship consistently results in a *precedence* relation. However, this particular linearization requirement is a stipulation. There is an alternative ordering which would equally allow a consistent mapping between asymmetric c-command and linear sequencing; if c-commanding terminals follow rather than precede other elements of the tree, then syntactic relations will also strictly determine linear correspondence. The two possible options for invariant mappings between c-command and ordering are given below.

# (10) LCA (reformulated)

option (i):

If X asymmetrically c-commands Y, then the terminals in X *precede* the terminals of Y.

option (ii):

If X asymmetrically c-commands Y, then the terminals in X *follow* the terminals of Y.

If the first option is selected, then a "Spec – Head – Comp" order is expected (7a). However, if the second option is chosen, then a "Comp – Head – Spec" order arises (7d). Note that the two configurations are mirror images of each other and whatever asymmetric c-command relations hold for one will necessarily hold for the other (see Uriagereka 2002 for discussion).

In a crucial sense, asymmetric c-command therefore *underspecifies* linear ordering. Kayne observes this complication, and rules out the "Comp – Head – Spec" order through an independent stipulation. According to Kayne (1994: 36–38), this alternative order is inconsistent with the asymmetry of time. However, I contend that this is simply a restatement of the phono-temporal mapping problem, rather than a solution. A linearization of any of (7a–d) would be compatible with forward-moving time, in that they each specify a temporal sequence for the terminals. Either of (7a) or (7d) allows an invariant branching-to-linearization mechanism. I therefore conclude that the conceptual motivation is lacking for a single, optimally "harmonic" linearization scheme induced by the c-command relation of the syntax (cf. Uriagereka 2002).

In the next section, an alternative linearization mechanism is investigated. According to this hypothesis, language-specific variation in linearization is permitted.

#### 3.1.2 Directionality is Determined at Spell-out

As argued in the previous sections, syntactic structures are indeterminate for linear ordering. Yet, linearization is a bare output constraint at the PF interface. An independent mechanism is therefore necessary for fixing the sequencing of non-affixal syntactic terminals. I present here the hypothesis that linearization schemes arise at the point of spell-out to PF (cf. Chomsky 1995). For Merge ( $\alpha$ ,  $\beta$ ), the syntactic object may be spelled-out as the phonological objects  $<\alpha$ ,  $\beta >$  or  $<\beta$ ,  $\alpha >$ .

Regularities in linearization may be attributed to language-specific *directionality conventions*, as will be discussed in §3.1.2.1. In §3.1.2.2, these conventions are compared to the notion of "parameters". I propose that these conventions are not specific to the linearization of non-affixal syntactic terminals, and argue in §3.1.2.3 that directionality conventions are also at play in the choice of a prefixation or suffixation pattern for affixal elements. This discussion concludes in §3.1.2.4 with a summary of the directionality conventions assumed for Nuu-chah-nulth.

## 3.1.2.1 Directionality Conventions

In many languages, syntactic objects and phonological objects stand in an implicational relationship. That is, a syntactic object may be consistently mapped to a particular linearization. The examples in (11) may be understood to be distinct linearization schemes for a syntactic object in which  $\alpha$  "heads" the phrase (owing to the fact that it is the category which projects its label). In a language in which syntactic form has implications for ordering, the syntactic object may be realized regularly as an  $\alpha$ -initial linearization; alternatively, it may be consistently ordered as an  $\alpha$ -final pattern.



These regularities in linearization may be attributed to language-specific ordering "conventions" imposed at spell-out. The *directionality convention* hypothesis may be stated as follows:

## (12) Directionality convention: Order $(\alpha, \beta)$

Ordering is stipulated phrase by phrase, and language by language. I assume that these ordering conventions are established through language acquisition. The decision for an  $\alpha$ -initial or  $\alpha$ -final ordering is conditioned by the language which the learner is exposed to. I assume that an  $\alpha$ -initial or  $\alpha$ -final ordering may hold uniformly across the language or it may be specific for certain syntactic categories, dependent on the regularities of the language. In either case, the convention establishes one out of the two logically possible orderings as a regularity at spell-out. Languages with free word order, which show no regularity in ordering, are assumed to lack directionality conventions for spell-out: a linearization is necessarily induced at spell-out, but one linearization scheme is not preferable over the other.

Let us consider the contrast between English prepositions and Hungarian postpositions as an illustration of the proposed spell-out directionality conventions. Each of these language show regular patterns. In English, prepositions (e.g., *above*, *beside*, *under*) reliably precede a nominal. In Hungarian, however, postpositions (e.g., *fölött* "above", *mellett* "beside", *alatt* "under") consistently follow a nominal. This difference is illustrated in (13) with *above/fölött*. In (13a), *above* precedes *the table*; in (13b), *fölött* "above" follows *az asztal* "the table".

(13)	a.	The pi	The picture is above the table.				(English)	
	b.	A Det	kép picture	az DET	asztal table	fölött above	van. BE-3SG	
		The picture is above the table.						(Hungarian)

It is ungrammatical for English prepositions to follow the nominal, as shown in (14a, b). Conversely, it is ungrammatical in Hungarian for postpositions to precede the nominal, as shown in (15a, b).

# (14) English a. There are pictures everywhere: under, over and beside the table. b. \* There are pictures everywhere: the table under, over and beside.

#### (15) Hungarian

- a. Képek vannak mindenütt: az asztal alatt, fölött, mellett. picture-PL BE-3PL every-where DET table under above beside There are pictures everywhere: under, over, and beside the table.
- b. \* Képek vannak mindenütt: alatt, fölött, mellett az asztal. picture-PL BE-3PL every-where under above beside DET table There are pictures everywhere: under, over, and beside the table.

How are the distinct English and Hungarian patterns derived? Note that if a preposition (P) is concatenated with a nominal (DP) via Merge (P, DP), then there are two logically possible linearizations for this unordered syntactic object: the preposition may precede (16a) or follow (16b) the nominal.



According to the directionality convention hypothesis, the difference between English prepositions and Hungarian postpositions reduces to a spell-out regularity in linearization. English learners adopt (16a) as a linearization convention, while Hungarian learners opt for (16b). In the case of English, the pattern for prepositions is consistent with the general head-initial nature of the language. In Hungarian, however, head-final pattern for prepositions may be viewed as a convention specific to this syntactic category: determiners, for example, are subject to a head-initial pattern, as in *az asztal* "the table".

## 3.1.2.2 Comparison to Headedness "Parameters"

How does the notion of directionality "conventions" compare to the principles-andparameters options for headedness "parameters"? In the principles-and-parameters framework, learning a language entails selecting one of the possible settings made available by the parameters of Universal Grammar. By this view, language learners are "hard-wired" (or innately endowed) with a variety of possible parameters. For example, Universal Grammar is postulated to allow either an  $\alpha$ -initial or  $\alpha$ -final setting for the headedness parameter(s).

(17) *Headedness parameter(s)* 

- (i) SETTING A ( $\alpha$ -initial):  $\alpha$  precedes  $\beta$
- (ii) Setting B (a-final):  $\alpha$  follows  $\beta$

If a learner is exposed to English prepositions, then SETTING A is switched on. If a learner is exposed to Hungarian postpositions, then SETTING B is switched on.

What the directionality convention hypothesis shares with this notion of parameter-setting is the aspect of acquisitional choices. However, the directionality convention hypothesis diverges from the parameter hypothesis in the grammatical source of these choices. According to the directionality convention hypothesis, a language learner simply selects one of only two *logically possible* linearizations for a syntactic object at local spell-out: the choices are not innately endowed in the language faculty. This follows from the reasoning that the choices of  $\alpha$ -initial and  $\alpha$ -final orderings need not be designated biologically, because they are inherently learnable (see Newmeyer 2005).

## 3.1.2.3 Affixation Patterns

The previous discussion was concerned with the linearization schemes of nonaffixal elements. According to the directionality convention hypothesis, a language learner selects one of the two logically possible linearizations when it comes time to order  $\alpha$  with respect to  $\beta$  at spell-out. This choice determines whether a language utilizes, for example, a head-initial or specifier-final pattern.

How does this specification of directionality for syntactic terminals compare with word-internal directionality? Within a word, a bound element may be designated as either prefixal or suffixal. A prefixal ordering linearizes the bound element to the left of its host; a suffixal ordering linearizes the bound element to the right of its host. I propose that the choice between a prefixation and a suffixation pattern is also a spell-out convention. In Nuu-chah-nulth, for example, affixal predicates are consistently linearized as suffixes. A language learner adopts this pattern over the logically possible alternative of ordering these affixal predicates as prefixes. Note that the learner is exposed to abundant evidence for the language-specific choice of suffixation or prefixation in Nuu-chah-nulth. Thus, prefixation–suffixation patterns may be seen as a subtype of directionality convention. If  $\alpha$  or  $\beta$  is an affix, then Order ( $\alpha$ ,  $\beta$ ) establishes a preference for a regular prefixation or suffixation linearization.

#### 3.1.2.4 Directionality Conventions of Nuu-chah-nulth

In this book, five linearization regularities for spell-out are proposed for Nuu-chahnulth. These order a specifier relative to a head (18a), a head relative to a complement (18b), an adjective relative to a noun (18c), and a manner adverb relative to a modified verb (18d). A final convention establishes the suffixation pattern of affixal elements in Nuu-chah-nulth (18e).<sup>3</sup>

(18) Directionality conventions of Nuu-chah-nulth

- a. SPECIFIER-FINAL: a head precedes a specifier
- b. HEAD-INITIAL: a head precedes a complement

<sup>&</sup>lt;sup>3</sup> Clearly, this is a heterogenous set. It remains to be shown whether these regularities reduce to a more abstract characterization.

- c. ADJECTIVE-INITIAL: an adjective precedes a modified noun
- d. MANNER ADVERB-INITIAL: a manner adverb precedes a modified verb
- e. SUFFIXATION: a host precedes an affixal predicate

As determined by (18a), specifiers are realized in a "right-branching" orientation in which they follow the head of the phrase. I will propose in §3.3 that this convention applies to subjects of clauses, inducing a subject-final ordering of VOS.

According to the convention of (18b), Nuu-chah-nulth has a head-initial pattern in which complements follow heads (Davis and Sawai 2001; Stonham 2004; Ravinski 2005, among others). In Nuu-chah-nulth, heads of phrases are often affixes, which may obscure this head-initial linearization pattern. However, non-affixal heads such as *wik* (NEG) show a consistent head-initial pattern with respect to their complements.<sup>4</sup>



In sentence (20), *wik* must precede, not follow, the phrase which it takes scope over,  $wat-si\lambda$  "go.home (PERF)".

(20)	a.	wikitsiš	wa†ši⊁
		wik-mit-siiš	wa†-ši <del>≀</del>
		NEG-PST-1SG.IND	go.home-PERF
		I didn't go home.	

b.	*	wa†ši⊁itsiš	wik
		wa⁺-ši⊁-mit-siiš	wik
		go.home-PERF-PST-1SG.IND	NEG
		I didn't go home.	

As was discussed in Chapter 2, a consistent ordering relation holds between an adjective and a modified noun in Nuu-chah-nulth (§2.2.2.1). This is captured by the convention of (18c). As shown in (21), the adjective *halum* "tasty" must obligatorily precede the nominal *lapinis* "apples".

(21)	a.	?u?iic?iš?a†	ha?um	?aapinis
		?u- <u><b>'iic</b></u> -?iiš-?a <del>1</del>	ha?um	?aapinis
		Ø- <u>consume</u> -3.ind-pl	tasty	apples
		They are eating deliciou		

<sup>&</sup>lt;sup>4</sup> For the sake of simplicity, I exclude tense (*-mit*) and subject agreement (*-siiš*) from this diagram. I assume these morphemes are introduced in higher projections, and surface suffixed to *wik* in (20a) through local spell-out of these affixes.

b.	*	?u?iic?iš?a†	?aapinis	ha?um			
		?u- <u><b>'iic</b></u> -?iiš-?a <del>1</del>	?aapinis	ha?um			
		Ø- <u>consume</u> -3.ind-pl	apples	tasty			
		They are eating delicious apples.					

This initial ordering of the modifier also applies to manner adverbials, as determined by the convention of (18d). As discussed in Chapter 2, a manner adverbial such as *wityax* "slow" precedes a verb in Nuu-chah-nulth (§2.2.2.2).

(22)	a.	witýaxits v witýax-mit-s v slow-PST-1SG.ABS g I was going home slowly.	vaałši⊁ vał-[+L]-ši⊁ go.home-CONT-PERF
	b. *	* waałšiλits wał-[+L]-šiλ-mit-s go.home-CONT-PERF-PST-1S I was going home slowly.	witỷax witỷax G.ABS slow

The suffixation convention of (18e) determines that affixal predicates are linearized as suffixes to their hosts, and not as prefixes.<sup>5</sup> In (23), the affixal predicate -aap "buy" attaches to the right of its host, *mahta-* "house", and not to its left.

(23)	a.		maḥtʾa?amit?iš maḥtʾa- <b>ʾaap</b> -mit-ʔiiš house- <b>buy</b> -PST-3 IND	čakup čakup man
			A man bought a house.	man
	b.	*	?aamaḥťamit?iš <u>'<b>aap</b>-maḥťa-mit-?iiš <b>buy</b>-house-PST-3.IND A man bought a house.</u>	čakup čakup man

In the next section, we return to the topic of the linearization schemes for syntactic terminals, with a focus on the syntactic structure of Nuu-chah-nulth.

# 3.2 Configurationality

Before discussing evidence for a VOS linearization scheme in Nuu-chah-nulth in §3.3, let us first cover background topics on the position of arguments in a Nuu-chah-nulth clause. In this section, I argue that Nuu-chah-nulth is a "configurational" language (cf. Hale 1983; Baker 1996) in which DP arguments are introduced with the verbal projections, as in (25).

<sup>&</sup>lt;sup>5</sup> This suffixation appears to hold for all bound forms in Nuu-chah-nulth, with the exception of bound nominal allomorphs which host an affixal predicate (§1.3).



This section addresses the basic structural properties of the Nuu-chah-nulth language, and argues for a syntactic asymmetry between subjects and objects.

#### 3.2.1 Partial Head Marking

In "partial head marking" languages, agreement morphology associated with a predicate obligatorily registers some, but not all, arguments of the predicate. Nuuchah-nulth is a partial head marking language (Davis, Waldie, and Wojdak 2007): portmanteau mood/agreement enclitics agree with *subjects* (Rose 1981; Davidson 2002). Objects are not registered via agreement, with the exception of a limited form of object marking which appears in imperative environments (see Davidson 2002). The subject agreement paradigm for the Ahousaht dialect is indicated in the following table. This table is closely based on Nakayama (1997, 2001), although I distinguish between "absolutive" and "dependent" moods, as well as propose a "confirmation" mood. For third persons, plurality is optionally indicated by the plural enclitic -2at (Nakayama 1997: 30).

Mood/Person	lsg	2sg	3	1pl	2pl
INDICATIVE	-siiš	-?iick	-?iiš	-niiš	-?iicuuš
INTERROGATIVE	-ḥs	-k	-ḥ	-ḥin	-ḥsuu
CONFIRMATION	-ḥaas	_	-ḥaa(č)	_	_
QUOTATIVE	-waa?ič'as	-waa?ick	-waa?iš	-waa?ič'in	waa?icuuš
ABSOLUTIVE	-S	?	Ø	-na	-suu
SUBORDINATE	-qs	-k	-q	-qin	-qsuu
DEPENDENT	-sa	-suuk	-ḥuuk	-na	-suu
SIMPLE RELATIVE	-qs	-?iitk	-?iitq	-qin	-?iitqsuu
"INDEFINITE"	-(y)iis	-(y)iik	-(y)ii	-(y)in	-(y)iisuu
RELATIVE					
CONDITIONAL	-quus	-quuk	-quu	-q <sup>w</sup> in	-quusuu

(25) Subject agreement in Nuu-chah-nulth (Ahousaht dialect)

The contrast between subjects and objects with respect to registering agreement correlates with the ability to license *pro*-drop. Subject DPs are freely omissible in Nuu-chah-nulth, given appropriate discourse contexts. The sentences in (26) may equally be uttered, for example, in the following specified scenerio:

(26) *Context:* You have been out with Kyle on the beach and he saw a sea anemone for the first time. You want to report what happened.

a.	naatsiičiλit?iš naatsii-šiλ-mit-?iiš see-PERF-PST-3.IND Kyle saw a sea anemone.	kin†umc kin†umc sea.anemone	Kyle Kyle Kyle
b.	naatsiičiλit?iš naatsii-šiλ-mit-?iiš see-PERF-PST-3.IND He saw a sea anemone.	kinłumc kinłumc sea.anemone	

For objects, however, *pro*-drop is not freely available, even with appropriate context. An overt object (27a) is permitted in the context below, but a covert object (27b) is not.

(27)	Context:	You know that Ken is planning to buy a sweater for his mother.
		You see a new sweater on the couch, and want to know if Ken
		bought it.

- a. maakuk<sup>w</sup>ith č<sup>'</sup>upč<sup>'</sup>upšumł?i maakuk-mit-h č<sup>'</sup>upč<sup>'</sup>upšumł-?ii buy-PST-3.Q sweater-DET Did he buy the sweater?
- b. \* maakuk<sup>w</sup>ith maakuk-mit-h buy-PST-3.Q Did he buy it?

Following Rizzi (1986), I adopt the analysis that "rich" subject inflection formally licenses a null pronominal (*pro*) as subject in cases where overt subject DPs are absent (Davis et al. 2007). In effect, the properties of the subject can be recovered by the inflection. Objects, which lack rich inflectional agreement, do not formally license a null pronominal argument.<sup>6</sup>

The asymmetry between subjects and objects with respect to agreement and the licensing of *pro* has key implications for the structural representations of DPs in the language. Specifically, Nuu-chah-nulth does not meet the definition of a "poly-synthetic" language in the technical sense of Baker (1996).<sup>7</sup> Baker (1996) draws a link between the rich inflection exhibited by certain polysynthetic languages, and the lack of structural asymmetries between lexical DPs. Under Baker's analysis, lexical DPs in these non-configurational "polysynthetic" languages occupy adjunct positions. Pronominal inflection, which agree with null *pro* arguments, are licensed

<sup>&</sup>lt;sup>6</sup> Nuu-chah-nulth does allow limited *pro*-drop of objects under certain discourse conditions; this is discourse-licensed and not agreement-licensed.

<sup>&</sup>lt;sup>7</sup> Note that Sapir's (1921) discussion of "polysynthesis" was partly in response to observations about Nuu-chah-nulth. In the original sense of the term, a "polysynthetic" language is one which exhibits a high morpheme to word ratio.

for theta role assignment by the verb via the Morphological Visibility Condition. This condition states that theta roles of the verb are to be found within the same word as the verbal head, either as (i) inflection or (ii) an incorporated noun.

#### (28) *Morphological visibility condition* (Baker 1996: 17)

A phrase X is visible for  $\theta$ -role assignment from a head Y only if it is coindexed with a morpheme in the word containing Y via:

(i) An agreement relationship

(ii) A movement relationship

In Nuu-chah-nulth, however, objects violate Baker's Morphological Visibility Condition on the licensing of pronominal arguments because they are not registered morphologically. In a similar vein, it is also the case that Nuu-chah-nulth does not match the definition of a "pronominal argument" language (Jelinek and Demers 1994) in which pronominal inflection occur directly as arguments: objects in the language may be directly represented by non-pronominal DPs. Thus, the absence of object inflection in Nuu-chah-nulth implies that at least object DPs must occupy argument positions in Nuu-chah-nulth.

In the next section, I turn to additional evidence for a syntactic asymmetry between the structural representation of subjects and objects in Nuu-chah-nulth.

### 3.2.2 An Asymmetry in Possessor Raising

Evidence for an asymmetry between subjects and objects in Nuu-chah-nulth comes from restrictions on a construction known as "possessor raising". In this construction, the possessive marker -uk/-(?)ak (POSS) is suffixed to a predicate, instead of (or in addition to) its base position suffixed to the possessum (Davidson 2002; Ravinski 2005).<sup>8</sup> The basic pattern of possessor raising is illustrated below with the intransitive predicate wiwisiaq $\lambda$  "lazy". Note that either possessor raising or possessor doubling behaviours are generally possible. In (29a), the nominal *taña* "child" is suffixed by the possessive marker and is interpreted as the possessum. In (29b), the predicate wiwisiaq $\lambda$  "lazy" is suffixed by -uk (POSS), while the nominal *taña* "child" receives no such marking but is still interpreted as the possessum. The example in (29c) shows both the predicate and the nominal suffixed by possessive markers.

(29)	a.	wiwišʕaq≁?iš	ťanaak?i	?uuštaqyu?i
		wiwišʕaqૠ-?iiš	ťaňa-?ak-?ii	?uuštaqyu-?ii
		lazy-3.IND	child-POSS-3.PS	healer-DET
		The doctor's chi	ild is lazy.	(unraised)

<sup>&</sup>lt;sup>8</sup> The choice between -uk (POSS) and -(i)ak (POSS) is phonologically constrained. As Ravinski (2005: 25) notes, -uk follows consonants, while -(i)ak follows vowels. Both -uk (POSS) and -(i)ak (POSS) indicate alienable possession. A different suffix, -iat (POSS) occurs with inalienable

possession (see Rose 1981; Davidson 2002; Ravinski 2005).

#### THE LINEARIZATION OF AFFIXES

b.	wiwišʕaq≁uk?iš wiwišʕaq≁-uk-?iiš lazy-POSS-3.IND	ťaňa ťaňa child	?uuštaqy ?uuštaqy healer-D	ru?i ru-?ii ET
	The doctor's child is lazy.			(possessor raising)
с.	wiwišSaq≁uk?iš wiwišSaq≁-uk-?iiš lazy-POSS-3.IND The doctor's child is lazy.	ťanaak?i ťana-?ak child-PO	i x-7ii DSS-3.PS	?uuštaqyu?i ?uuštaqyu-?ii healer-DET (possessor doubling)

The subject agreement in possessor raising (and possessor doubling) constructions necessarily agrees with the possessor, unlike in unraised examples. This is indicated in the examples below. In the unraised (30a), the subject agreement is third person *-7iiš* (3.IND), corresponding to the nominal *tanakqs* "my child". In the raised example in (30b), the subject agreement is first person *-siiš* (1SG.IND), matching the person of the possessor.

(30)	a.	wiwišʕaq≁ʔiš wiwišʕaq≁-ʔiiš lazy-3.IND My child is lazy.	ťaňaakqs ťaňa-?ak-qs child-POSS-1SG.PS	(unraised)
	b.	wiwišʕaq≁uksiš wiwišʕaq≁-uk-siiš lazy-POSS-1SG.IND My child is lazy.	ťaňa ťaňa child	(possessor raising)

In Chapter 4, I discuss the analysis in which the possessive marker -uk (POSS) on the predicate licenses a position for a raised DP possessor (Ravinski 2005). For our present purposes, we can set aside the mechanics of this raising operation and focus instead on the use of possessor raising as a diagnostic for distinguishing between subjects and objects.

The behaviour of transitives with respect to possessor raising gives evidence for a distinction between subjects and objects in Nuu-chah-nulth. For transitives, there is a restriction that a possessive marker on the predicate is only ever associated with a possessum subject, and not a possessum object (Davidson 2002; Ravinski 2005). As indicated by (31a, b), a subject ("cat") and an object ("bird") are equally compatible with possessive marking in "unraised" contexts.<sup>9</sup> Crucially, however, it is shown by the interpretation of the possessor-raised (31c) that objects face a restriction which subjects do not.

(31)	a.	hink <sup>w</sup> a?iiḥit?iš hin-k <sup>w</sup> a?iiḥ-mit-?iiš	piišpiš piišpiš	maamaatakqs maamaati-?ak-qs	
		LOC-go.aner-PS1-5.IND	cat	DIFG-POSS-15G.PS	
		A cat was after my bird.			(unraised)

<sup>&</sup>lt;sup>9</sup> Post-predicative word order in these examples, and in the language generally, is largely variable (see §3.3 for discussion).

b.	hinkwa?iihit?iš hin-kwa?iih-mit-?iiš LOC-go.after-PST-3.IND	piišpišukqs piišpiš-uk-qs cat-POSS-1SG.PS	maamaati maamaati bird	
	My cat was after a bird.		(unraised)	
с.	hink <sup>w</sup> a?iiḥuksiš hin-k <sup>w</sup> a?iiḥ-uk-siiš LOC-go.after-PST-POSS-1S = (i) My cat was after a bi ≠ (ii) A cat was after my b	piišpiš piišpiš G.IND cat rd. pird.	maamaati maamaati bird (possessor raising)	

In (31c), the raised possessor controls clausal subject agreement, as is standard for Nuu-chah-nulth possessor raising. The sole available interpretation for (31c) is one in which the nominal "cat" is the possessum; an interpretation in which the nominal "bird" is the possessum is unavailable. This difference in the behaviour of the two nominals reflects a systematic difference between subjects and objects: possessor raising is only possible out of a subject in Nuu-chah-nulth (Ravinski 2005).

# 3.2.3 An Asymmetry in Incorporation

Further evidence for the configurationality of arguments in Nuu-chah-nulth is the restriction that only *complements* of affixal predicates are able to act as hosts (Woo 2000; Davis and Sawai 2001; Yiu 2001; Wojdak 2003a, b, 2004). In the realm of "noun incorporation", arguments which occur as objects of an affixal predicate can be suffixed by the affixal predicate, but, as is indicated by (32b), subjects do not undergo a similar type of suffixation.

(32)	a.		haa?um?iixas?iis ha?um- <u>?iix[+L]</u> -'as-?iis food- <u>take</u> -go-3.IND Father went to get food.	nuwiiq nuwiiq father
	b.	*	nuwiic?iૠas?iš nuwiic- <u>?iૠ</u> -'as-?iiš father- <u>take</u> -go-3.IND Father went to get food.	ha?um ha?um food

In (32a), the affixal predicate  $2u-2i\lambda$  "take" suffixes to the object nominal *ha2um* "food". The example in (32b) shows that it is illicit for the subject *nuwiic*- "father" to be suffixed by the affixal predicate. Such asymmetries are found across all affixal predicates in Nuu-chah-nulth, a topic that we will return to in Chapter 4.

Possessor raising and incorporation thus pick out complementary sets of arguments: incorporation applies to objects, while possessor-raising applies to

subjects.<sup>10</sup> In the next section, we turn to a final argument for a structural asymmetry between subjects and objects.

# 3.2.4 Weak Crossover Effects

Davis et al. (2007) introduce Weak Crossover effects as proof of a distinction between subjects and objects in Nuu-chah-nulth. Examples of this asymmetry are presented in (33), which indicate that a subject *wh*-phrase in Nuu-chah-nulth may bind a possessive pronominal inside an object DP, but an object *wh*-phrase may not bind a possessive pronominal inside a subject DP. This directly parallels the available interpretations of the English sentences given as translations.

(33)	a.	?ačaqit <u>ḥ</u>	k™ik™ixasi⊁	?um?iiqsak?i				
		?ača-q-mit-h	k̇̀™ix-asi⊁[+R]	?um?iiqsu-ak-?i				
		who-AUX-PST-3.Q	kiss-on.cheek	mother-POSS-3.PS				
		= (i) Who <sub>i</sub> kissed their <sub>i</sub> mother on the cheek?						
b.		= (ii) Who <sub>i</sub> kissed th	eir <sub>j</sub> mother on the c	heek?				
	b.	?aačači†itḥ	?um?iiqsak?i	k™ik™ixasi⊁				
		?ača-či1[+L]-mit-h	?um?iiqsu-ak-?i	k̇́ <sup>w</sup> ix−asi⊁[+R]				
		who-AUX-PST-3.Q	mother-POSS-3.PS	kiss-on.cheek				
		$\neq$ (i) Who <sub>i</sub> did their <sub>i</sub>	mother kiss on the o	cheek?				
		= (ii) Who <sub>i</sub> did their	mother kiss on the	cheek?				

This can be taken as strong evidence that Nuu-chah-nulth must distinguish hierarchically between subjects and objects, and therefore that clausal structure cannot be "flat" (Davis et al. 2007).

In the following section, I introduce a particular analysis for the asymmetrical representation of subjects and objects in Nuu-chah-nulth. My analysis of Nuuchah-nulth clausal structure proposes that canonical predicate-initial word order arises from a "basic" VOS linearization.

# 3.3 Predicate-initial Word Order

Word order in Nuu-chah-nulth is predicate-initial. When both subject and object are overtly expressed, either VSO or VOS word orders are generally available (Rose 1981; Whistler 1985; Jacobsen 1993).

(34)	a.	kuuwitit?iš kuuwitt_mit_?iiš	čakup	huupuk <sup>w</sup> as
		Kuuwi1-IIIIt-IIIS	сакир	nuupuuk as
		steal-PST-3.IND	man	car
		A man stole a car. (VSO)		

<sup>&</sup>lt;sup>10</sup> Note that the two constructions do not pick out completely complementary sets (Davis et al. 2007). Incorporation operates on *underlying* objects, including the subjects of unaccusative verbs; in contrast, possessor raising is linked to *surface subjects*, also including the subjects of unaccusative verbs. As Davis et al. (2007) note, however, this should make no difference to the argument against non-configurationality.

b.	kuuwi1it?iš	huupuk <sup>w</sup> as	čakup
	kuuwi1-mit-?iiš	huupuukwas	čakup
	steal-PST-3.IND	car	man
	A man stole a car. (VOS)		

There is evidence that post-predicative word order is not strictly free, and that it is linked to constraints on animacy and discourse (Rose 1981; Woo 2004). As Rose (1981: 179) maintains, alternate word orders in Nuu-chah-nulth are "a function of the communicative salience of the constituents, e.g. newness of information, definiteness, particularity, contrastiveness, and role in the discourse." Woo (2004) generalizes that VOS in Nuu-chah-nulth most readily obtains when the object is "unmarked" – inanimate and/or indefinite. Consonant with Woo's generalization, the primary Ahousaht consultants for this book show a strong preference for the subject to precede the object when the object is animate. These speakers generally disprefer VOS with an animate object, as indicated in the sentence below, in which word order serves to restrict the possible interpretations.<sup>11</sup>

(35)	?u?uuyuk?iš	Ken	Kay			
	?u- <u>yuk</u> [+R]-?iiš	Ken	Kay			
	Ø- <u>cry.for</u> -3.ind	Ken	Kay			
	= (i) Ken is crying for Kay. (VSO)					
	$\neq$ (ii) Kay is crying for Ken. (VOS)					

In (35), the nominal *Ken* must precede *Kay* in order for *Ken* to interpret as a subject. This may be contrasted with the case in (34), in which the inanimate object *huupuuk<sup>w</sup>as* "car" has the option of preceding or following the subject.

Rose (1981: 112) proposes that an object may precede a subject when the object is new and salient, as she illustrates with the following examples from the Kyuquot dialect of Nuu-chah-nulth.

(36)	a.	?uk <sup>w</sup> ii†iš	qa?uuc	Mary		
		?u- <b>čii†</b> -?iiš	qa?uuc	Mary		
		Ø- <u>make</u> -3.ind	basket	Mary		
		Mary is making	a BASKET	. (VOS)		
		(Kyuquot; Rose	1981: 11	3, ex. 373)		
	b.	hišimyuup		maacqwin	łuucma	
		hiš-qimt-'uup-Q	5	maacq <sup>w</sup> in	łuucma	
		all-around-CAUS	-3.ABS	fly	woman	
		The lady collected HOUSEFLIES. (VOS)				
		(Kyuquot; Rose	1981: 113	3, ex. 375)		

<sup>&</sup>lt;sup>11</sup> A topic worthy of future research is inter-speaker (or inter-dialectal) variation in word order. As a whole, factors influencing word order in this language have not been adequately investigated to date.

However, a flavour of optionality must be recognized, as she notes that an order of VSO is also possible in this discourse context.

The presence of the determiner -2ii is correlated with word order permutations in the Ahousaht dialect of Nuu-chah-nulth. If an object is marked with this determiner, then the object preferentially appears after the subject. This is illustrated in (37–38). In (37), the determiner-marked nominal *maamaati-2i* "the bird" must follow *piišpiš* "cat" in order for *maamaati-2i* "the bird" to be interpretable as the object. Similarly, in (38), the determiner-marked nominal *Simii* $\lambda$ -2i "the dog" must follow *čakup* "man" if it is to be interpreted as an object.

(37)	a.		?u?iicit?iš ?u- <u>'iic</u> -mit-?iiš Ø- <u>consume</u> -PST-: A cat is eating the	3.IND bird. (V	piišpiš piišpiš cat SO)		maamaa maamaa bird-DET	ti?i ti-?ii
	b.	#	?u?iicit?iš ?u- <u>'iic</u> -mit-?iiš $\emptyset$ - <u>consume</u> -PST- = (i) The bird is e $\neq$ (ii) A cat is eati	3.IND ating a canng the bi	maamaa maamaa bird-DET at. (VSO) rd. (VOS	ti?i ti-?ii ) )	piišpiš piišpiš cat	
(38)	a.		hisqin $\lambda$ ?iš his-qin $\lambda$ -?iiš hit-on.head-3.IND A man hit the dog	čakup čakup man g on the h	Sinii⊁?i Sinii⊁-?ii dog-DET nead using	?uuḥŵ ?u-ḥŵ ⊘-use g a hatche	ra∔ a⁺[+L] et. (VSO-	hisýak hisýak hatchet - OBL)
	b.	!	hisqin $\Re$ ?iš his-qin $\Re$ -?iiš hit-on.head-3.IND = (i) The dog hit $\neq$ (ii) A man hit th	Sinii⊁?i Sinii⊁-?ii dog-DET a man on ne dog or	čakup čakup man the heac the heac	?uuḥẁa+ ?u-ḥẁa+ Ø-use I using a I using a	, [+L] hatchet. ( hatchet. (	hisýak hisýak hatchet (VSO- OBL) (VOS- OBL)

A variety of analyses has been proposed to account for the derivation of the world's predicate-initial systems. Although these analyses are constrained by theory-internal restrictions on admissable clause structure (see, e.g., Anderson 1984; Lee 2000), there is mounting cross-linguistic evidence that there are multiple "routes" available to achieving predicate-initial word order (Chung 1998; Davis 2005). That is, predicate-initial status amounts to a superficial characteristic which masks potentially distinct underlying syntactic systems. On the one hand, predicate-initial systems may be directly linearized as VOS, under a right-linearized specifier analysis (Chung 1991). This is represented schematically below, abstracting away from node labelling.



0

With this line of analysis, VSO is also possible as a derived word order, surfacing after movement of the object, as has been argued for mixed VOS/VSO systems including Mayan (England 1991), Austronesian (Chung 1991, 1998, 2004), and Salish (Davis 2005).

On the other hand, predicate-initial systems may be derived from an underlying SVO configuration. Within this option, two general possibilities are to be distinguished: V-raising targeting the predicate head, as in McCloskey's (1991) treatment of VSO in tensed clauses in Irish; or VP-raising which targets the predicate at the level of the phrase, as has been argued for Malagasy (Pearson 1998), Niuean (Massam 2000), and Zapotec (Lee 2000). In the latter case, VOS straightforwardly obtains. However, VSO may be derived in conditions of VP remnant-raising (Lee 2000), in which the object evacuates the VP before the VP "remnant" is raised. The diagrams in (40) illustrate head-raising and phrase-raising, respectively.

### (40) "Derived" predicate-initial word orders



Conclusive empirical evidence is lacking for the superiority of one of these analyses over the others for Nuu-chah-nulth. Indeed, any of these analyses are compatible with the central claims of this book. What is crucial to the analysis at hand is an asymmetrical representation of subjects and objects, within a predicate-initial system for both "small" and "full" clauses: these criteria may be satisfied by either "basic" or "derived" predicate-initial systems. In the interest of explicitness, I adopt in this book an analysis of Nuu-chah-nulth predicate-initial word order as a "basic" VOS system. In what follows, I will describe this proposal in more detail, followed by a discussion of existing evidence against alternative raising approaches. In §3.3.1, a right-branching specifier analysis to possessive phrases. Evidence against a verbraising analysis of Nuu-chah-nulth predicate-initial word is presented in §3.3.3, and discussion of problems for a verb phrase raising treatment is given in §3.3.4. The analysis of predicate-initial word order concludes in §3.3.5 with a summary of the implications of variable word order for the linearization of affixal predicates.

# 3.3.1 Proposal: "Basic" Predicate-initial Order (VOS)

According to the analysis proposed here, predicate-initial word order in Nuu-chahnulth is not derived by syntactic movement. That is, no syntactic movement operation is necessary in order for the verb to precede the arguments of the clause. This is achieved via implementation of a right-branching specifier system, as in (41a), rather than a left-branching one, as in (41b). These systems represent two distinct linearizations schemes for identical syntactic objects.



Right-branching specifiers are unavailable under the model of syntax proposed by Kayne (1994). However, according to the directionality convention hypothesis, rightand left-branching systems are equally available across languages as distinct postsyntactic linearization schemes for universally *unlinearized* syntactic objects. By hypothesis, a directionality convention at spell-out in Nuu-chah-nulth determines that specifiers follow, rather than precede, the inner layers of the projection (the head and complement).

Based on this right-branching specifier analysis, I propose the following representation of Nuu-chah-nulth predicates relative to their arguments. In (42), I adopt the analysis that a verbal head v introduces a subject, and takes a VP as complement. According to this arrangement, arguments appear on the right-periphery of morphemes which head the verbal projection(s) of the clause.



This yields an underived VOS word order. This linearization is consistent with the following generalization of Sapir (1924: 83): "verb, object, subject- this is the most common Nootka order".

As indicated at the outset, however, VSO word orders compete with VOS in Nuu-chah-nulth, generating controversy as to which, if either, of these word orders can be construed as more "basic" (Woo 2004). Indeed, Jacobsen (1993) counters Sapir's assertion by showing that VSO is in fact statistically more common than VOS in the text examined by Sapir (1924), and furthermore, that sentences with two overt arguments are rare in text contexts. I follow England (1991) in maintaining that frequency of natural occurrence is not a determining criterion in the analysis of underlying word order (see also Woo 2004), as syntactic and non-syntactic factors conspire on a language by language basis to determine the specific word order permutations which arise. Definiteness effects, animacy restrictions, and topic/focus are

all observed to affect the available word orders of the daughter languages of Proto-Mayan (England 1991). Strictly "stylistic" factors such as prosodic heaviness have also been argued to play a role in the argument order in predicate-initial Salish systems (Davis 2005). There is inadequate evidence at present to allow for a conclusive characterization of which mix of factors are at play in restricting the word order permutations of Nuu-chah-nulth. However, existing research is consistent with the hypothesis that argument order is influenced by information structure.

Adopting England's (1991) analysis of the mixed VOS/VSO systems of Mayan, I therefore propose that VSO in Nuu-chah-nulth may plausibly be derived from a "basic" VOS order when a "reordering" rule moves a marked object to the right-periphery.

$$(43)$$
 [V S] 'reordered' O

(England 1991: 480)

This falls in line with claims that "marked" animate and/or definite objects in Nuuchah-nulth often follow a subject (Rose 1981; Woo 2004). The "reordering" of the object can be represented by movement of the object to a specifier position above the subject, as in (44).



The XP projection can be understood to be a privileged position for animate and/or definite objects. For example, in the sentence in (45a), the determiner-marked object *maamaati-îi* "bird-DET" necessarily raises above the subject *piišpiš* "cat". A VOS ordering is unavailable, as indicated by (45b).

(45)	a.		?u?iicit?iš ?u- <u><b>'iic</b></u> -mit-?iiš Ø- <u>consume</u> -pst-3.ind A cat is eating the bird. (	piišpiš piišpiš cat VSO)	maamaati?i maamaati-?ii bird-DET
	b.	#	?u?iicit?iš ?u- <u>'iic</u> -mit-?iiš $\varnothing$ - <u>consume</u> -pst-3.ind = (i) The bird is eating a $\neq$ (ii) A cat is eating the	maamaati?i maamaati-?ii bird- DET cat. (VSO) bird. (VOS)	piišpiš piišpiš cat

### 3.3.2 Evidence for Right-branching Specifiers

A right-branching specifier system elegantly captures the canonical word order of Nuu-chah-nulth possessed phrases. As Ravinski (2005) describes, the unmarked word order for possessed noun phrases when the possessum is adjectivally modified is adjective-possessor.

 (46) naatsiičiλitwa?iš Christine [ċušukuk?i maḥťii Rachel] naatsii-šiλ-mit-wa?iš Christine [ċuš-uk-uk-?i maḥťii Rachel] see-PERF-PST-3.QUOT Christine new-DUR-POSS-3.PS house Rachel Christine saw [Rachel's new house]. (Ravinski 2005: 41, ex 114c)

The availability of this word order is straightforwardly predicted by an analysis in which the possessor (*Rachel*) occupies a specifier position to the right of the possessed nominal. This may be represented by a structure in which a possessor occurs as the specifier of the PossP phrase, as in (47).<sup>12</sup>

## (47) Right-branching possessor



Following Ravinski's (2005) analysis, the possessive morpheme -uk (POSS) is shown to head the Possessive Phrase (PossP).<sup>13</sup> An Agreement Phrase (AgrP) is postulated, which is headed by an agreement marker registering the possessor. In (47), the possessor agreement is third person -2i (3.PS), to match the third person possessor *Rachel*.

If a left-branching analysis of possessors is assumed, then the possessor is predicted to *precede*, not follow, the noun possessum (see Braithwaite 2003 for discussion). This is illustrated in the tree below.

<sup>&</sup>lt;sup>12</sup> This departs from Ravinski (2005), who represents the possessor as occuring as a rightbranching internal argument of the NP.

<sup>&</sup>lt;sup>13</sup> The diagram in (47) abstracts away from syntactic locus of the aspectual marking -uk (DUR) on the adjective *cus-uk* "new (DUR)". The aspectual properties of Nuu-chah-nulth adjectives require additional research.





As Ravinski (2005) notes, a right-branching specifier analysis straightforwardly predicts the attested possessor-final word order, as well as the placement of the inflectional clitics on the (first word of) the possessum.

#### 3.3.3 Evidence Against Verb-raising

Previous treatments of Nuu-chah-nulth word order have analysed predicate-initial word order as arising from head-movement of the verb (Davis and Sawai 2001; Stonham 2004, among others). These analyses assume that head movement applies to an underlying SVO structure, raising the head of the predicate to a position past the subject, where it syntactically adjoins to tense and/or agreement occupying higher functional projection (e.g., TP, Mood). This may be represented by the following:

(49)



In this section, I illustrate empirical problems with this analysis as applied to Nuuchah-nulth.

The first hurdle confronting a head-raising analysis is the question of evidence for verb movement to a functional projection above the vP. Since inflectional morphemes such as tense and subject agreement are "second position" enclitics in Southern Wakashan (Klokeid 1978; Davidson 2002), their appearance suffixed to the verb is phonologically conditioned, and as such, does not necessarily entail a syntactic mode of placement (see §3.4.2 for a local spell-out analysis of this cliticization pattern). The examples below illustrate the "second position" effect in which tense and subject agreement encliticize to the first word of the sentence, whether it is the predicate itself (50a), a preceding adverbial (50b), or a negation particle (50c).

(50)	a.	waa†ši≿ <i>its</i> wa†-[+L]-ši≿- <i>mit-s</i> go.home-CONT-PERF-PST-1SG.ABS I was in the process of going home.				
	b.	witỷax <i>its</i> witỷax- <i>mit-s</i> slowly-PST-1SG.ABS I was slow in going hor	waa†ši <del>k</del> wa†-[+L]-šik go.home-CONT-PERF ne.			
	c.	wik <i>its</i> wik- <i>mit-s</i> NEG-PST-1SG.ABS I wasn't slow in going l	witýax waałšiλ witýax wał-[+L]-šiλ slowly go.home-CONT-PER 10me.	F		

The fact that a verb in Nuu-chah-nulth may bear tense and/or agreement morphology does not therefore constitute evidence for syntactic raising of the verb to these functional projections. If a verb is the first word in the predicate phrase, it will bear tense and/or agreement morphology; if it is not first, then it will not.

Moreover, there is the broader issue of a trigger for the putative verbraising. Based on the observation that VSO word order obtains in Modern Irish in tensed clauses, while SVO occurs in infinitival ones, McCloskey (1991) argues that Irish [+finite] verbs are attracted to an inflectional projection. In Nuu-chah-nulth, however, no parallel argument can be constructed.<sup>14</sup> In Nuu-chah-nulth, predicateinitial word order is possible in small clause environments. On the assumption that small clauses lack functional projections above the vP which could house a raised verb, predicate-initial word order in this environment should be impossible. This verb-initial pattern is illustrated for the non-finite complements of the perception verb  $\lambda uu-\dot{n}a\dot{k}uub$ "observe". (In Chapter 5, I provide evidence that the complement here is equivalent to a vP.)

<sup>&</sup>lt;sup>14</sup> Davis and Sawai (2001: 125) argue based on the behaviour of the perception verb *haatsiičik* "to see" that SVO word order occurs in non-finite complements. However, given that *haatsiičik* "see" is compatible with nominal complements (e.g., *haatsiičikitiš Mary John* "Mary saw John"), combined with the availability of null pronouns and null "absolutive" third person subject agreement in Nuu-chah-nulth, it is unclear whether their test sentence involves nominal complementation (as indicated by the bracketing below) or clausal complementation. Note that the third person inflection (-*huk*/ $\varnothing$ ) in the second clause is optionally overt.

 <sup>(</sup>i) [naatsiičiitiiš Mary John] k<sup>\*</sup>wik<sup>\*</sup>wisasit(huk) pro Wanda
 [naatsii-šit-mit-?iiš Mary John] k<sup>\*</sup>wik<sup>\*</sup>wisas-šit(-huk) pro Wanda
 see-PERF-PST-3.IND Mary John kiss.on.cheek-PERF-(3.DEP) pro Wanda
 Mary saw John, (he was) kissing Wanda on the cheek. (Davis and Sawai 2001: 125, ex. 5)

(51)	?uunakuuḥitsiš	[wa?ič'as	ḥaa	čakup?i]		
	?u-nakuuh़[+L]-mit-siiš	[wa?ič-'as	ḥaa	čakup-?ii]		
	Ø-observe-PST-1SG.IND	sleep-on.ground	DEIC	man-DET		
	I observed that man sleeping on the ground.					

The same generalization may be made of non-finite complements of negation (52a), as well as in contexts with auxiliaries (52b) and non-affixal modals (52c). Each of these environments allows the predicate to precede the subject, in the absence of an overt syntactic trigger (such as tense or finiteness).<sup>15</sup> In the negation context in (52a), the predicate *wa?ič* "sleep" precedes the subject, *Ken*. The example in (52b) shows a relative clause which is formed when the relative pronoun *yaq* (REL) is suffixed by the auxiliary *?uu-k<sup>w</sup>i?* (AUX): what follows this auxiliary is the verb *maakuk* "buy", crucially *preceding* the subject of the relative clause, *čakup-?i* "the man". In (52c), the verb *wa?i-ši?* "go home (PERF)" is sandwiched between the modal *Sapaak* "willing" and the subject *Kay*.

(52)	a.	wikit?iš wik-mit-?iiš NEG-PST-3.IND		ḥacuk ḥacuk deeply	wa?ič wa?ič sleep	Ken Ken Ken	<i>.</i>
		Ken wasn't in a	a deep sleep	p.			(negation)
	b.	ởiiḥumł?iš ởiiḥ-umł-?iiš red-RD-3SG.IND The shoes the n	šuwis [ya šuwis [ya shoes RE nan bought	aaq <sup>w</sup> i†itii aq-či†[+L L-AUX-Pa are red.	i .]-mit-ii st-3sg.ir ( <i>c</i>	maakuk maakuk L buy uuxiliary ei	čakup?i] čakup-?ii] man-DET nvironment)
	c.	Sapaak?iš Sapaak-?iiš willing-3.IND	wałši⊁ wał-ši⊁ go.home	e-PERF	Kay Kay Kay		

(modal environment)

What these environments share is a word order in which the verb precedes the subject. However, in the negation context in (52a), the predicate phrase is irrealis, and so the trigger for the pututive raising cannot therefore be finiteness. Similar arguments can be constructed for (52b, c): why would the presence of the pre-verbal auxiliaries not preclude the need for the verb itself to raise?

Kay is willing to go home.

Complex nominal predicates also constitute a challenge to a head-raising analysis of predicate-initial word order.<sup>16</sup> Since head movement can apply to only a single head in the predicate phrase, this operation is predicted to "break up" a

<sup>&</sup>lt;sup>15</sup> An alternative word order is also possible where the verb phrase follows the subject. The availability of this word order requires further research, as do other cases of word order variability in the language (see Rose 1981; Davidson 2002). What is crucial for the present argument against verb-raising is that the predicate-initial word should not be possible in this environment.

<sup>&</sup>lt;sup>16</sup> Nuu-chah-nulth allows any of the lexical categories (A, V, N) to occur directly as predicates in clause-initial position; there is no copula in the language (Wojdak 2000, 2001).

complex predicate composed of a predicate and its modifier, by forcing the subject to intervene between the raised element and unraised residue left in the VP. In fact, contrary to expectations, the standard pattern is for the subject to *follow* a complex nominal in Nuu-chah-nulth (Davidson 2002: 128).<sup>17</sup> In such contexts, the "fronted" element resembles a maximal projection, not a head. In (53), the nominal predicate  $\lambda u + tuucma$  "good woman" precedes the subject *Kay*.

# (53) *Complex nominal predicate (subject-peripheral)*

[zuł?iiš	⁺uucma]	Kay
[zut-?iiš	⁺uucma]	Kay
good-3.IND	woman	Kay
Kay is a good	woman.	

In contrast, for an underived predicate-initial system, this subject-peripheral word order follows naturally, as the nominal occupies initial position in the clause, together with its modifer. This word order is represented in the following structure, adopted from Ravinski's (2005) analysis of nominal predicates.<sup>18</sup> In (54), the nominal head *n* introduces the subject (*Kay*) of the nominal predicate.

# (54) Complex nominal predicates in right-branching specifier system



In conclusion, I suggest that a plausible solution to the problems posed by a headraising account – lack of evidence for raising, lack of a syntactic trigger for raising, unpredicted word orders – is to assume that Nuu-chah-nulth clause structure is VOS, with rightward movement of the object deriving the VSO variant.

## 3.3.4 Evidence Against Raising of Verbal Phrase

The issues noted above with respect to head-raising also create complications for a predicate-raising analysis of Nuu-chah-nulth. On the topic of word order variability,

 $<sup>^{17}</sup>$  Davidson (2002: 128) notes that "the words in a multi-word nominal predicate. are usually strictly ordered: quantifier/number > property > noun." Subject-intervening patterns, though marked, are attested in my own fieldwork, however. Additional research is required into this pattern, as with other instances of word order variability in Nuu-chah-nulth.

<sup>&</sup>lt;sup>18</sup> I assume that encliticization of the subject agreement marker -2iis to the adjective is achieved through local spell-out (§3.4.1).

any analysis of Nuu-chah-nulth clausal structure will need to employ special machinery to account for the VOS/VSO alternations and other word order permutations in the language. A potential advantage of a VP-raising analysis over a head-raising approach, however, is that it has been demonstrated independently that it can successfully deal with VOS/VSO alternations (Massam 2000, 2001). In the predicateinitial system of Niuean, post-predicative word order is not strictly flexible, as it is tied to a definiteness effect on the object. In Massam's VP-fronting analysis of Niuean, VSO word order is obtained when a definite object vacates the VP before the remnant of the phrase raises. Indefinite objects which are "pseudo-incorporated" into the V remain as part of the VP complex, generating VOS order when the VP raises. The distinct derivations of VOS and VSO orders are represented below.

#### (55) a. *VP-raising* (= VOS) b. *VP-remnant raising* (= VSO)



The VP-remnant raising derivation is essentially a "two-step" process in which the object raises on its own first, before the rest of the VP undergoes movement.

It is apparent, however, that the analysis which Massam applies to Niuean cannot be directly translated to Nuu-chah-nulth. As Woo (2004) observes, this analysis makes the crucial prediction that existential clauses must have VOS order, since the indefinite object has no motivated escape route out of the VP. Under this view, the indefinite object of an existential clause should be "pseudo-incorporated" into the V, and should never raise outside of the VP. This runs counter to the observation that VSO word orders are the preferred pattern for locative existentials in Nuu-chah-nulth (Wojdak and Woo 2004). In (56a), the object *haium* "food" follows the subject *niisyak-ii* "the pot"; in (56b), the indefinite object *ciixsac* "frying pan" comes after the subject *čamaqkýak-ii* "the oven".<sup>19</sup>

(56)	a.	?ućuu?iš	niisyak?i	ha?um
		?u-ċuu-?iiš	niisyak-?ii	ha?um
		Ø-in.container-3.IND	pot-DET	food
		There's food in the pot.	(VSO)	

<sup>&</sup>lt;sup>19</sup> The indefinite locatum argument (*halum* "food", *ciixsac* "frying pan") in these locative existentials can be clearly shown to be objects of the predicate (§4.4.1).

b.	?uuq≁?iiš	č'amํaq≁y'ak?i	ciixsac
	?u-'aq <del>λ</del> -?iiš	č'aṁaq≁y'ak-?ii	ciixsac
	Ø-inside-3.IND	oven-DET	frying.pan
	There's a frying	pan in the oven. (	VSO)

An indefinite object is incorrectly predicted to follow the verb for these existentials.<sup>20</sup>

In sum, given the challenges faced by derivational mechanisms for generating predicate-initial word order in Nuu-chah-nulth, I present a "basic" VOS configuration as a plausible alternative for this language. The factors which govern post-predicative word order variability in the language await clarification by future research.

In the next section, I examine the implications of post-predicative word order variability for the linearization of affixal predicates.

#### 3.3.5 Implications for the Linearization of Affixal Predicates

According to the local spell-out hypothesis, an affixal predicate finds a host chosen from its derivational sister. In §3.3.5.1, I illustrate how this has the implication of creating a "complement" effect in incorporation, whereby only elements from the complement of an affixal predicate are eligible as hosts. I then argue that the local spell-out analysis is superior to alternative models of linearization which select a host for the affixal predicate via constraints on directionality. In §3.3.5.2, it is shown that direction-sensitive mechanisms for affixations have difficulty coping with the variable post-predicative word order of Nuu-chah-nulth.

#### 3.3.5.1 The "Complement" Effect

According to the analysis I have proposed, affixal predicates are spelled-out in a minimal domain containing only the affix and its derivational sister. Under the assumption that objects, but not subjects, occur as complements to a predicate, the local spell-out hypothesis predicts that only objects of an affixal predicate are eligible as hosts. This is illustrated in (57), in which an object acts as the derivational sister to the verb, while the subject is introduced in a higher projection. The verb is spelled-out with the object, its complement.

(57) The "complement" effect in Nuu-chah-nulth affixation



 $<sup>^{20}</sup>$  A topic for future research is the surface position of the definite subjects in (56). As proposed in §3.3.1, determiner-marked *objects* are moved rightward in Nuu-chah-nulth. It is not known whether determiner-marked *subjects* ever show a similar preference in Nuu-chah-nulth.

#### CLAUSAL ARCHITECTURE OF NUU-CHAH-NULTH

For verbs which take nominal complements, there is robust evidence for an affixation asymmetry between the subjects and objects of affixal predicates (Woo 2000; Davis and Sawai 2001; Stonham 2004, among others). (This is argued in detail in Chapter 4.) This effect is illustrated in (58) for the affixal predicate *2u-2ap* "buy". The object of this verb, *maḥta-* "house" may serve as host for the affix, but the subject *čapx-* "man" cannot.

(58)	a.		maḥťa?amit?iš	čakup
			maḥťa- <b>'aap</b> -mit-?iiš	čakup
			house- <u>buy</u> -PST-3.IND	man
			A man bought a house.	
	b.	*	čapx?aamit?iš	maḥťii
			čapx- <b>'aap</b> -mit-?iiš	mahťii
			man- <b>buy</b> -PST-3.IND	house
			A man bought a house.	

This complement effect is also observed with affixal predicates which take verbal complements, such as *Auu-ňakuuh* "observe". The syntactic structure for an affixal predicate which takes a verbal (sentential) complement may be represented abstractly as in (59). (A detailed analysis of the structure of such predicates is presented in Chapter 5.)

(59) *local spell-out* 



The affixal predicate occupies matrix position as  $V_1$ , and takes the sentential complement (VOS) as its complement (circled). According to the local spell-out hypothesis,  $V_1$  is linearized with respect to this complement, since it is its derivational sister. The subject of  $V_1$  (i.e.,  $S_1$ ) falls outside of this local spell-out domain, predicting an asymmetry between  $S_1$  and the complement. Evidence for such an asymmetry is given in (60). In (60a), the embedded verb *tuuxtuux*<sup>w</sup>a "jump (ITER)" hosts the affixal predicate *luu-nakuuh* "observe". As shown in (60b), the subject of the affixal predicate, *čapx*-"man", is ineligible as host.

(60)	a.	tuuxtuux <sup>w</sup> anakuuhit?iš	čakup	ťaatňa?	is
		tux <sup>w</sup> -a[+R]- <u><b>'nakuuḥ</b></u> -mit-?iiš	čakup	ťaatňa?	is
		jump-ITER- <u>observe</u> -PST-3.IND	man	childre	n
		A man observed the children jumping.			
	b.	* čapxňakuuhit?iš	tuuxtuu	xwa	ťaatňa?is
		čapx- <u>ňakuuh</u> -mit-?iiš	tux <sup>w</sup> -a[-	-a[+R] ťaat	
		man- <u>observe</u> -PST-3.IND	jump-ITER		children
		A man observed the children jumping.			

This asymmetry follows if *čapx*- "man" is not a derivational sister of the affix. By the "complement" effect in affixation, only derivational sisters of affixal predicates are able to act as hosts.

The following section presents an argument against an alternative analysis of the observed affixation asymmetry.

#### 3.3.5.2 Evidence Against Directionality of Affixation

This section argues that sensitivity to derivational sisterhood is not reducible to an independent effect of directionality. Let us label this alternative analysis the *RIGHT-directionality* hypothesis. This hypothesis is defined by statement (61):

## (61) *RIGHT-directionality* hypothesis: An affixal predicate attaches to whatever host is found to its right

Given a syntactic structure as in (62), this rule would determine that an affix  $-\alpha$  choses  $\beta$  as its host because  $\beta$  is right-adjacent to  $-\alpha$ . The element  $\delta$ , in contrast, would be ineligible as a host for  $-\alpha$  because it occurs to the left of  $-\alpha$ .



Under such an analysis, it is irrelevant that  $\beta$  is the derivational sister of  $-\alpha$ . All that matters according to the RIGHT-directionality hypothesis is that  $\beta$  follows  $-\alpha$ .

Evidence against the RIGHT-directionality hypothesis comes from two independent properties: first, the post-predicative word order possibilities of Nuuchah-nulth; second, the systematic absence of unergative affixal predicates in the language. We will consider each of these characteristics in turn, beginning with word order variability. As noted in the earlier discussion, in Nuu-chah-nulth, there is often variation in the word order of an object with respect to an overt subject. This variability is illustrated in (63) for the affixal predicate *?u-?aap* "buy". In (63a), the object *mahtii* "house" precedes the subject *čakup* "man". In (63b), the ordering of the arguments is reversed.

(63)	a.	?u?aamit?iš	maḥťii	čakup
		?u- <u><b>'aap</b></u> -mit-?iiš	maḥťii	čakup
		Ø- <u>buy</u> -pst-3.ind	house	man
		A man bought a house.		
b.	?u?aamit?iš	čakup	maḥťii	
----	----------------------------------	-------	--------	
	?u- <u><b>'aap</b></u> -mit-?iiš	čakup	maḥťii	
	Ø- <u>buy</u> -pst-3.ind	man	house	
	A man bought a house.			

Thus, either an object or a subject may follow the affixal predicate. Despite this flexibility in word order of subject and object, the affixation mechanism is invariant: an affixal predicate may only attach to mahta- "house" and not to capx- "man".

(64)	a.		maḥta?amit?iš maḥta- <u>'aap</u> -mit-?iiš house- <u>buy</u> -PST-3.IND A man bought a house.	čakup čakup man
	b.	*	čapxîaamitîiš čapx- <u><b>`aap</b>-mit-îiiš</u> man- <u>buy</u> -PST-3.IND A man bought a house.	maḥtii maḥtii house

This finding is at odds with the predictions of the RIGHT-directionality hypothesis. According to this hypothesis, if an element can occur right-adjacent to the affix, it should be eligible as a host. Thus, a subject which precedes an object, such as *čakup* "man" in (62b), is incorrectly predicted to act as a host.

Unlike with the RIGHT-directionality hypothesis, for the local spell-out hypothesis, the host for an affixal predicate is determined by derivational sisterhood. Local spell-out matches an affixal predicate and a host at an initial stage of the derivation: the point at which an affixal predicate is merged into the derivation. According to this analysis, the surface position of the arguments of the affixal predicate is irrelevant: affixation is determined before the relative ordering of the arguments is manipulated. Thus, the local spell-out hypothesis elegantly captures the generalization that the affixation pattern is invariant, despite variability in postpredicative word order.

A further argument against the RIGHT-directionality hypothesis comes from the make-up of the inventory of affixal predicates. Evidence will be shown in Chapter 4 that affixal predicates with the semantics of unergatives are systematically absent in Nuu-chah-nulth (see §4.2.3). Intransitive verbs with meanings such as *cry*, *run*, or *dance* occur only as independent predicates in the language, as in the example below with the non-affixal verb *huut*--"dance".

(65)	huułhuułamitk	?atḥiimit?i
	huu1-a[+R]-mit-k	?atḥii-mit-?ii
	dance-ITER-PST-2SG.Q	night-PST-DET
	Did you dance last night?	

In Chapter 4, I relate the absence of unergative affixal predicates to the requirement that a viable affix in Nuu-chah-nulth must find its host from within its derivational

sister. Unergatives, which inherently lack a suitable pairing of host and affix at the local spell-out of the predicate, fail as bound morphemes under the local spell-out analysis. By contrast, the RIGHT-directionality hypothesis predicts no similar restriction: choice of host is not sensitive to derivational sisterhood, and is therefore predicted to be insensitive to argument structure. Since unergatives may be followed by words (such as the temporal modifier in (65)), there is no explanation under the RIGHT-directionality hypothesis why an affixal predicate could not select a non-sister, right-adjacent element as a host.

In the next section, I demonstrate how the local spell-out hypothesis can be extended from affixal predicates to other affixes found in Nuu-chah-nulth. By the analysis, both clause-level and nominal-level inflectional clitics are linearized by local spell-out.

## 3.4 Cliticization Domains

There are two distinct cliticization domains in Nuu-chah-nulth. Clitic strings may be built up within a DP, or at a clausal level which excludes the DP(s). The bracketing in (66) illustrates these two zones of cliticization.

(66)	hiixtaqči[ <i>mitsiš</i> ] <sub>CP domain</sub>	huupuk <sup>w</sup> as[ <b>uk?itk</b> ] <sub>DP domain</sub>
	hiixtaq-čip- <b>mit-siiš</b>	huupuk <sup>w</sup> as- <b>uk-?iitk</b>
	have.accident-BEN-PST-1SG.IND	car-POSS-2SG.PS
	I had an accident with your car.	

In §3.4.1 and §3.4.2, I consider each of these cliticization domains in turn.

## 3.4.1 DP Domain

Within a Nuu-chah-nulth DP, functional morphemes appear in a strictly ordered clitic string suffixed to a root (Davidson 2002; Werle 2002; Ravinski 2005).

(67) Organization of the DP clitic sequence in Nuu-chah-nulth

= POSS=TENSE=AGR/DET

This clitic string includes the following enclitics: possessive markers, tense, possessive agreement, and the determiner -2ii.<sup>21</sup> These morphemes appear in the following examples, suffixed to a nominal (in brackets).

(68)	a.	'naatsiiči⊁itsiš	[maḥt'iimit?i]
		naatsii-ši⊁-mit-siiš	[maḥt'ii-mit-ʔii]
		see-PERF-PST-1SG.IND	house-PST-DET
		I saw the former house.	(Ravinski 2005: 16, ex.29)

<sup>&</sup>lt;sup>21</sup> Nuu-chah-nulth permits tense markings in the nominal domain, as well as in the clausal domain. This phenomenon is not uncommon in the Pacific Northwest *Sprachbund* (see Burton 1996 for discussion of the Salish language Halkomelem).

b.	≁iiḥum†it?iš	[huupuuk <sup>w</sup> asuk <sup>w</sup> itqs]
	λiiḥ-um1-mit-?iiš	[huupuukwas-uk-mit-qs]
	red-RD-PST-3.IND	car-POSS-PST-1SG.PS
	My former car was red.	

These inflectional morphemes encliticize to the leftmost root with a DP, illustrated below with the "second position" placement of the enclitic determiner -2ii. In (69a), the determiner suffixes to  $haak^waa\lambda$  "girl", while in (69b) it appears instead on the modifier  $q^wacat(aq)$  "(very) beautiful". In (70a), the determiner suffixes to pišmis "problem(s)", and in (70b) it attaches to 2aya "many".

(69)	a.	?u?uk <sup>w</sup> inkitsiš ?u-k <sup>w</sup> ink[+R]-mit-s ∅-talk.with-PST-1 I talked with the g	iiš SG.IND irl.	[ḥaakʷaaź [ḥaakʷaaź girl-DET	&?i] ₹-?ii]		
	b.	?u?uk <sup>w</sup> inkitsiš ?u-k <sup>w</sup> ink[+R]-mit-s ∅-talk.with-PST-1 I talked with the b	iiiš SG.IND eautiful	[q <sup>w</sup> acatao [q <sup>w</sup> acat-a beautiful- girl.	q?i lq[+S]-?i -AUG-DE	i T	ḥaakʷaa⊁] ḥaakʷaa⊁] girl
(70)	a.	?uucwa?iš ?u-ic-wa?iš Ø-own-3.QUOT Kay's the instigato	Kay Kay Kay or of the	q <sup>w</sup> ačii <del>1</del> q <sup>w</sup> a-čii <del>1</del> like-mako problems	2	[pišmis? [piš-mis bad-NOM	i] -?ii] 4-det
	b.	?uucwa?iš [ ?u-ic-wa?iš ] Ø-own-3.QUOT ] Kay's the instigato	Kay Kay Kay or of the	q <sup>w</sup> ačii <del>1</del> q <sup>w</sup> a-čii <del>1</del> like-make many pro	[?aya?i [?aya-? e many- blems.	ii DET	pišmis] piš-mis] bad-NOM

As can be seen in these examples, when the DP contains only a nominal, the determiner encliticizes to this word; however, when a modifier or quantifier takes on leftmost position in the DP, the placement consistently shifts to this leftmost element (Davidson 2002).

This "affix hopping" behaviour is ably handled by the local spell-out analysis. Let us take the positioning of the determiner in the examples in (69) as illustration of the spell-out properties of DP-level clitics. When the determiner is syntactically merged with a noun, such as  $haak^waa\lambda$  "girl" in (69a), local spell-out determines that the determiner and the noun must be linearized with respect to each other at spell-out to PF.



Because the determiner is a suffix, a linearization of  $< haak^{waa} - i > haak^{waa}$  is induced.

The placement of the determiner suffixed to the modifier in (69b) proceeds much the same way, although there is additional round of spell-out when the noun is first merged with the modifier  $q^wacat(aq)$  "(very) beautiful". Before the determiner may be merged with the noun phrase, the following syntactic operation takes place: Merge ( $q^wacataq$ ,  $haak^waax$ ). This builds a modified noun. (The category label of the noun is projected, as argued in Chapter 2.)



As derivational sisters, the adjective and noun are linearized with respect to each other at spell-out. A directionality convention establishes the modifier-initial pattern of Nuu-chah-nulth, setting the stage for a spell-out ordering of  $\langle q^w a \dot{c} a t(aq), haak^w aa \dot{x} \rangle$ .

In the next step of the syntactic derivation for (69b), the determiner is introduced by Merge. The determiner thus takes the NP as its derivational sister.



At spell-out, the determiner will need to be linearized with respect to its derivational sister. In particular, because the determiner is an affix, it must find a host from within its derivational sister. The earlier round of local spell-out established an ordering of  $\langle q^{w}a\dot{c}a^{+}(aq), haak^{w}aa\lambda \rangle$ . By the string adjacency effect, the host for the determiner  $-\partial ii$  is selected as the leftmost element of the NP:  $q^{w}a\dot{c}a^{+}(aq)$  "(very) beautiful". Thus, an ordering of  $\langle q^{w}a\dot{c}a^{+}(aq)-\partial i, haak^{w}aa\lambda \rangle$  results at spell-out.

The "affix-hopping" behaviour of the enclitic determiner can therefore be seen to be an interaction between the string adjacency effect and the syntactic composition of the derivational sister of the determiner. If the derivational sister is simplex, as in (69a), then the locality restriction on affixation is trivial: the determiner must be spelled-out with the single element in its derivational sister. If the derivational sister of the determiner is complex, as in (69b), then the locality restriction determines that the single leftmost element in the derivational sister is selected as a host. Note that the determiner has not actually "hopped": in both types of cases, the determiner consistently selects as a host the leftmost element in the string which is linearly adjacent to it at spell-out.

# 3.4.2 CP Domain

Functional morphemes outside of the DP are also subject to encliticization (see Klokeid 1978 for discussion of the Southern Wakashan language Ditidaht). Davidson (2002) identifies a range of inflectional morphemes which occur in a strictly ordered clitic sequence (simplified from Davidson 2002: 321).

#### (74) Organization of the CP clitic sequence in Nuu-chah-nulth

=TR=TEMP=PAS=POSS=TENSE=AGR/MOOD=PL=AGAIN=HAB

Parallel to cases of DP-level cliticization, the members of this sequence occur standardly in "second position" relative to a host morpheme at the left edge of the clause. This "second position" effect is exemplified in (75) with the positioning of the past tense morpheme -mit (PST) and the third person indicative subject agreement -2iis (3.IND). In (75a), these morphemes suffix to the verb *kamatq-uk* "run (DUR)". In (75b), however, their position "shifts" to the preverbal modifier  $\lambda a 2ix$  "fast".

(75)	a.	kamatquk <sup>w</sup> it?iš kamatq-uk-mit-?iiš run-DUR-PST-3.IND Florence was running.	Florence Florence Florence	
	b.	ửa?ixit?iš ửa?ix-mit-?iiš fast-PST-3.IND Florence was running fast	kamatquk kamatq-uk run-DUR	Florence Florence Florence

This suffixation pattern follows from the local spell-out hypothesis. We can consider the simpler case in (75a) first. Successive applications of Merge build the tree shown in (76), in which the tense and subject agreement morphemes occupy functional projections (Tense Phrase and Agreement Phrase) above the lexical projections of the verb *kamatq(uk)* "run". Note that the subject, *Florence*, is represented as a right-linearized specifier.



Each application of Merge is subject to local spell-out for the derivational sisters conjoined by Merge. Early rounds of local spell-out determine that the subject *Florence* is linearized to the right because of a right-branching specifier convention. When the past tense morpheme is added to the tree by Merge (T, vP), this has the result that *-mit* (PST) must be linearized with respect to its derivational sister *<kamatq(uk)*, *Florence>* at spell-out. Because *-mit* (PST) is an affix, it must find a host from within its derivational sister. The string adjacency effect determines that *kamatq(uk)*-*mit*, *Florence>*. The next morpheme to be linearized in accordance with the string adjacency effect is the subject agreement morpheme *-iiiš* (3.IND). As a suffix, it is tagged on at spell-out to the end of the leftmost element in its derivational sister. This yields the ordering *<kamatq(uk)mit-iiiš*, *Florence>*.

The tree in (77) is a representation of the derivation when the verbal predicate is modified by an adverbial, as in (75b). Here, the adverbial  $\frac{\partial^2 a}{\partial x}$  "fast" combines with *kamatq(uk)* "run" to form a complex verbal predicate.



Once again, successive applications of Merge determine that spell-out relationships are formed incrementally between derivational sisters. Starting with the lower portions of the tree, a directionality convention requires that the modifier  $\lambda a 2ix$  "fast" linearly precede kamatq(uk) "run" at spell-out. Similarly, the right-branching specifier convention entails that the subject *Florence* will be spelled-out the right of the rest of the vP. When it comes time for the past tense morpheme -mit (PST) to find a host at spell-out, the item selected as its host is the leftmost element in its derivational sister. Because the modifier linearly precedes the verb, it is the modifier which is determined to be the host for -mit (PST). This yields a linearization of  $<\lambda a 2ix-mit$ , kamatq(uk) Florence>. A final act of linearization suffixes the subject agreement morpheme -2iis (3.IND) to the tail end of the initial complex, resulting in an ordering of  $<\lambda a 2ixmit-2iis kamatq(uk)$  Florence>.

#### 3.5 Conclusion

This chapter has touched upon areas of Nuu-chah-nulth grammar which bear on the present analysis of affixal predicates. I have presented evidence for the configurationality of Nuu-chah-nulth syntax, and have represented this clausal structure within a right-linearized specifier system. By this analysis, Nuu-chah-nulth predicateinitial word order originates with a "basic" VOS system. I have identified two domains of cliticization in the language, linked to DP-level and clause-level inflection. Inflectional clitics in the language find their positions via the same spell-out principles responsible for the linearization of affixal predicates.

The empirical coverage for the remainder of this book corresponds to the "polysynthetic" realm canonically situated at the left edge of a Nuu-chah-nulth clause. It is in this morphologically complex sequence that affixal predicates may commonly be found united with their hosts.

(78)	a.	Sičpa+?in+?anitniš	kwaqmis	Mary
		Sič- <b>ṕa1</b> - <b>'in1</b> -'at-mit-niiš	k <sup>w</sup> aq-mis	Mary
		rotten-taste-serve-PAS-PST-11	PL.IND s.h.eggs-NOM	Mary
		We were served rotten-tasting	g spawned herring egg	s by Mary.
	h	huuhtakšiihmahsa?iš	Lucy annuar	u?aca

b. huuhtaksiihmahsahs Lucy quuquuhaca huhtak-šin-<u>iih[+L]-mahsa-</u>?iiš Lucy quuhac-[+R]-(y)a know-PERF-<u>try.to-want.to</u>-3.IND Lucy person-speak-CONT Lucy wants to learn how to speak Nuu-chah-nulth.

In the following chapters, I will present additional evidence for the syntactic structures underlying these morphologically complex sequences. In Chapter 4, the argument structure of affixal predicates which take nominal complements will be discussed. In Chapter 5, the argument structure of affixal predicates which take verbal complements will be discussed. Residual to my analysis are the factor(s) governing the word order variations found outside of the clause-initial polysynthetic complex.

# 4. Nominal Complements of Affixal Predicates

...where every word is at home, taking its place to support the others... ~T.S. Eliot Four Quartets no. 4

## 4.0 Introduction

The predicates in Nuu-chah-nulth which permit incorporation are a lexically specified set of affixal predicates. As discussed in earlier chapters, "incorporating" predicates in Nuu-chah-nulth are invariably bound. In this, Nuu-chah-nulth differs from incorporation languages such as Mohawk which do not have a lexically defined subclass of incorporating predicates. In Mohawk, a single predicate can show an alternation between an incorporating and a non-incorporating option (Baker 1988). In (1a), the inflected predicate *ye-nuhwe'-s* "like" incorporates its object *–nuhs* "house", while in (1b) it does not.

(1) *Mohawk* (examples from Postal 1962, as cited in Baker 1988: 81–82, ex. 14a, b)

ne baby house-l	ikes.		
ao-wir-a'a RE-baby-SUF	ye-nuhwe'-s 3FS/3N-like-ASP	ne DET	ka-nuhs-a' PRE-house-SUF
2	E-baby-SUF	E-baby-SUF 3FS/3N-like-ASP	E-baby-SUF 3FS/3N-like-ASP DET

For predicates in Nuu-chah-nulth, however, such an alternation is banned outright. Independent (non-affixal) predicates such as *maakuk* "buy" never permit incorporation. The example in (2a) illustrates the impossibility of incorporating the nominal *mahia*- "house" into the independent predicate *maakuk* "buy". The nominal must always occur separately from the independent predicate, as in (2b).

(2)	a. *	* maḥtamaakukwit?iš maḥta-maakuk-mit-?iiš house-buy-PST-3.IND A man bought a house.	čakup čakup man	
	b.	maakuk <sup>w</sup> it?iš maakuk-mit-?iiš buy-PST-3.IND A man bought a house.	čakup čakup man	maḥťii maḥťii house

Incorporation is an option exclusively reserved for affixal predicates in Nuu-chahnulth, such as  $\lambda u$ - $\lambda a p$  "buy". In (3a),  $\lambda u$ - $\lambda a p$  "buy" incorporates a nominal host, mahía- "house". As shown in (3b), it is impossible for an affixal predicate such as  $\lambda u$ - $\lambda a p$  "buy" to occur without a host.

(3)	a.		maḥťa?amit?iš maḥťa- <u>'aap</u> -mit- house- <u>buy</u> -PST-1 A man bought a	?iiš 3.IND house.	čakup čakup man
	b.	*	?aamit?iš <u>'aap-</u> mit-?iiš <u>buy</u> -PST-3.IND A man bought a	čakup čakup man house.	maḥťii maḥťii house

I argued in Chapter 1 that affixal and independent predicates in Nuu-chah-nulth are distinguished via specification of an [affix] requirement in the lexical entry of an affixal predicate. That is, affixal predicates constitute a lexically designated subclass of predicates.

We turn now to the *syntactic* characteristics of affixal predicates. What is the syntactic make-up of this lexically defined subclass? In this chapter, the syntactic structure of affixal predicates which take nominal complements is investigated. (Affixal predicates which take verbal complements, such as -qaath "claim", are discussed in Chapter 5.)

## (4) Classes of predicates in Nuu-chah-nulth

- A. Affixal predicates
  - (i) Nominal complements
  - (ii) Verbal complements
- B. Independent predicates
  - (i) Nominal complements
  - (ii) Verbal complements

The aim of this chapter is to develop an inventory of the argument structures which are available to affixal predicates which take nominal complements. Following Hale and Keyser (1993, 2002), I pursue a syntactic approach to argument structure in which positions for arguments are projected syntactically in accordance with the lexical properties of the head. Argument structure is what limits the number of arguments which exist for a given predicate. For example, a ditransitive verb such as  $\lambda u$ -yii "give" has a lexically licensed position for a benefactive argument, although a transitive predicate such as  $\lambda u$ -Zaap "buy" does not.

(5)	taanaqayimit?iš	čakup	?um?iiqsak
	taanaq- <b>ayii</b> -mit-?iiš	čakup	?um?iiqsu-?ak
	money-give-PST-3.IND	man	mother-POSS
	A man gave money to hi	s mother.	

(6)	a.	*	maḥťa?amit?iš	čakup	?um?iiqsak
			maḥťa- <u><b>'aap</b></u> -mit-?iiš	čakup	?um?iiqsu-?ak
			house- <u>buy</u> -PST-3.IND	man	mother-POSS
			A man bought a house for	or his motl	ner.

b.	maḥťa?apčip?iš	čakup	?um?iiqsak
	maḥťa- <b>'aap</b> -čip-?iiš	čakup	?um?iiqsu-?ak
	house-buy-BEN-3.IND	man	mother-POSS
	A man bought a house f	or his mot	her.

In (5), 2un2iiqsak "his mother" acts as the recipient of the predicate 2u-yii "give". The example in (6a) shows that such an argument is not directly licensed by the predicate 2u-2aap "buy". Instead, in order for a benefactive argument to appear with 2u-2aap, the predicate must be supplemented by the addition of the benefactive suffix -*čip* (BEN), as in (6b).

This chapter will demonstrate that the syntactic configuration of arguments of an affixal predicate plays a deterministic role in the pattern of incorporation in Nuu-chah-nulth. Specifically, the syntax conditions local spell-out operations by determining which elements will act as the derivational sister to the affixal predicate. As first discussed in Chapter 3, a host for an affixal predicate is chosen from the complement of an affixal predicate. In contrast, an element from the affixal predicate's specifier will be ineligible as a host because it is not a derivational sister of the affixal predicate. The syntactic limitations which are imposed on Nuu-chahnulth incorporation follow directly from the PF incorporation hypothesis in which incorporation occurs post-syntactically. According to the hypothesis, the linearization operation responsible for resolving the affixation requirement of an affixal predicate applies to the outputs of syntactic structure-building, at the point of spellout to PF. Only an element which is generated in the syntax in a position where it may be spelled-out with the affixal predicate is compatible with incorporation. Complements occupy a privileged position in that their syntactic sisterhood to the affixal predicate guarantees that the affixal predicate will be oriented with respect to this phrase by local spell-out.

## 4.1 Predication Configurations

Before turning to my diagnostics for Nuu-chah-nulth argument structure, I present an overview of the predication configurations which will be discussed over the course of this chapter.

## 4.1.1 Basic Structures

Argument structures of predicates are built by syntactic concatenation. The simplest possible argument structure for a predicate results from a single application of Merge. This one-place relation maps an argument to the complement position of the predicate. This is an *unaccusative* relation.

(7) Unaccusative syntax



A second application of Merge introduces another DP. This second argument is realized as a specifier of the predicate. This two-place relation may be referred to as an *extended unaccusative*.

#### (8) *Extended unaccusative syntax*



A central proposal of this chapter is that the argument structures available to affixal predicates reduce to these two basic configurations of arguments. Both basic types map an internal argument to complement position, although the types differ lexically in the possibility of projecting a specifier (Hale and Keyser 2002). An unaccusative predicate is a monadic relation which maps its single internal argument to its complement position. An extended unaccusative, in contrast, is an inherently dyadic relation which allows two internal arguments to occupy complement and specifier positions, respectively.

The affixal predicate *?u-nii* "arrive" is an example of a Nuu-chah-nulth unaccusative predicate. In the following example, this monadic predicate takes a single argument, *paastin?ath* "American(s)".

(9) ?uniii?iš ?u-<u>nii</u>-?i Ø-arriy

Puinns Pu-<u>nii</u>-?iiš Ø-<u>arrive</u>-3.IND Americans came. paastin?atḥ paastin?atḥ American unaccusative

In contrast, a locative predicate such as  $\lambda - k w_i^{*}$  in" is inherently "birelational" in the sense of Hale and Keyser (2002). The function mapped by the locative predicate is necessarily saturated by two arguments, a *locatum* (the element which is located) and a *location*. In the example below,  $\dot{c}a^2ak$  "water" corresponds to the locatum of  $\lambda - k w_i^{*}$  in", while  $\dot{c}ax^{*}ac$  "bucket" specifies the location of the water.

(10)  $2uk^{wi}iis$  ča?ak čax<sup>w</sup>ac  $2u-\underline{ci}-iis$  ča?ak čax<sup>w</sup>ac  $\emptyset-\underline{in}-3.IND$  water bucket The water is in the bucket. extended unaccusative

In the following sections, I discuss two factors which allow a six-way classification of predicates to be derived from these two basic unaccusative and extended unaccusative configurations. In §4.1.2, the topic of transitivization is introduced. This is followed in §4.1.3 by the proposal that inherently birelational predicates show flexibility in their theta-role mapping.

## 4.1.2 Transitivization

Transitive and ditransitive syntax is available to Nuu-chah-nulth affixal predicates; following Hale and Keyser (2002), I assume that these are derived structures. Embedded in the transitive and ditransitive configurations are the underlying syntax of unaccusatives and extended unaccusatives, respectively. A transitive predicate is created when a "light" verbal head, denoted as v, is merged with the basic unaccusative relation (Hale and Keyser 1993, 2002; Chomsky 1995). The verbal head v introduces an additional ("external") argument, yielding a derived dyadic verb.

## (11) Transitive syntax



Transitivization is associated with an agentive or causative interpretation which is not present with the basic predication configuration of an unaccusative (Hale and Keyser 1993, 2002; Kratzer 1994).

By this analysis, transitive predicates such as  $\lambda u - \dot{n}aah$  "look for" achieve their dyadicity in a manner distinct from locatives such as  $\lambda u - k^{w_i}$  "in". In the example in (12), the transitive affixal predicate  $\lambda u - \dot{n}aah$  "look for" takes a first person argument as the subject, and a second argument,  $\dot{c'upc'upsumtukqs}$  "my sweater" as an object.

(12)	?unaaḥsiš ?u- <b>naaḥ</b> -siiš	č'upč'upšum†ukqs č'upč'upšum†-uk-qs	transitive		
	Ø-look.for-1sg.ind	sweater-POSS-1SG.PS			
	I'm looking for my sweater.				

Unlike locative predicates, which are inherent two-place relations, transitives are composite predicates which are formed by implanting an unaccusative predicate configuration within the additional syntactic structure projected by the "light" verbal head v. By this view, only *čupčupšumłukqs* "my sweater" in (12) is an internal

argument. The first person subject is an external argument, made available by transitivization.

A parallel transitivization process is hypothesized to occur with ditransitive predicates. While transitives build on basic unaccusative syntax, ditransitives are formed from a basic extended unaccusative relation. As indicated in the diagram in (13), the ditransitive is formed when the verbal head v is merged with an extended unaccusative. This, in turn, is merged with a DP projected as the specifier of the v projection.

(13) *Ditransitive syntax* 



By this view, a ditransitive is conceptualized as a locative predicate which has an external argument added to the basic extended unaccusative relation (Freeze 1992; Hale and Keyser 2002). An example of a Nuu-chah-nulth ditransitive affixal predicate is  $\lambda u$ -yii "give". In (14),  $\lambda u$ -yii "give" takes three arguments: the second person subject, the recipient  $\lambda um\lambda$  "mother", and the theme *taana* "money".

(14)	?uyii?i	?um?i	taana
	?u- <u>a<b>yii</b></u> -'ii	?um?i	taana
	Ø- <u>give</u> -2sg.imp>30bj	mother	money
	Give mom money!		

## 4.1.3 Flexibility in Theta-role Mapping

Transitivization represents one means by which the basic predication configurations are used to build a wider array of argument structures. Another source of diversity is flexibility in the mapping between theta-roles and underlying structure. In particular, I follow Hale and Keyser (2002) in hypothesizing that when a locative lexical item projects two internal arguments, the arguments of the extended unaccusative may be realized in one of two possible orientations. Following Hale and Keyser (2002), these two types may be termed "locatum" and "location" predicates. (The topic of locative predicates is discussed in detail in §4.4.) For the class of locatum predicates, a locatum argument (the located element) is merged with the predicate as the basic step of the derivation; this is followed by another application of Merge in which the location argument is added to the structure. For location predicates, the basic step is uniting the location argument with the predicate by Merge; this precedes a secondary step in which the locatum is introduced.

(15) *Flexibility of theta-role mapping* 



The mechanisms of syntactic concatenation thus allow inherently dyadic predicates to show variability in their theta-role assignment to complement and specifier positions. By this means, extended unaccusative predicates come in two distinct "flavours", with inverse configurations of the internal arguments. In one, the locatum acts as the complement, and the location acts as the specifier; in the other, the order of Merge operations are reversed, and the relative positions are consequently the opposite.

The availability of inverse argument structures necessitates abandonment of a strict one-to-one mapping between syntactic structure and theta-role assignment (Baker 1988; Hale and Keyser 1993). According to Baker's Uniformity of theta assignment hypothesis, theta-roles have a unique structural realization.

## (16) Uniformity of theta assignment hypothesis (Baker 1988: 46)

Identical thematic relationships between items are represented by identical structural relationships between those items at the level of D-structure.

A variety of evidence in Nuu-chah-nulth points to the need for a more flexible mapping mechanism (see §4.4, §4.5). For example, subject agreement in locative predicates shows two distinct patterns: for one class of predicates (locatum predicates), subject agreement is linked to the location argument, as in (17a); for the other class (location predicates), subject agreement is determined by the locatum, as in (17b):

(17)	a.	?ukuxssiš ?u- <b>uxs</b> -sijš		℀i?ičum≁ ℀i?ičum≁	
		Ø- <u>on.head</u> -1sg	.IND	straw.hat	
		I'm wearing a st	traw hat.		
		(lit: "I'm headin	ig a straw	hat")	(locatum predicate)
	b.	?uk <sup>w</sup> isiš	ćucsac		
		?u- <b>či</b> -siiš	ćucsac		
		Ø- <b>in</b> -1sg.ind	tub		
		I'm in the tub.			(location predicate)

This pattern of subject agreement corresponds to a split between those locatives (locatum predicates) which suffix to a locatum argument, and those locatives (location predicates) which suffix to the location argument.

(18)	a.	ha?umcu?iš	qa?uuc?i	
		ha?um- <u><b>čuu</b></u> -?iiš	qa?uuc-?ii	
		food- <u>contain</u> -3.IND	burden.basket-I	DET
		There's food in the burde	en basket.	
		(lit: "The burden basket of	contains food")	(locatum predicate)
	b.	qa?uucči?iš	ýaṁa	
		qa?uuc- <u>či</u> -?iiš	yama	
		burden.basket- <u>in</u> -3.IND	salal.berries	
		The salal berries are in a	burden basket.	(location predicate)

I will argue in §4.4 that this contrast in incorporation behaviour follows naturally if locatum predicates such as 2u-cuu "contain" map a locatum argument to complement position, while location predicates such as 2u- $k^{w_i}$  "in" map a location to their complement. Such an analysis is incompatible with the rigid mapping mechanism of Baker's Uniformity of theta assignment hypothesis.

# 4.1.4 Predicate Inventory

In sum, there exists a six-way classification of affixal predicates which take nominal complements, which are built from the two basic predication configurations. Unaccusative and transitive predicate classes are composed from the basic unaccusative relation, while locatum, location, locatum-type ditransitive, and location-type ditransitive arise from the basic extended accusative relation.

Basic configuration	Transitivization	Orientation of arguments	Predicate type
Unaccusative	Underived	n/a	Unaccusative e.g., <i>?u-ńii</i> "arrive"
	Transitivized		Transitive e.g., <i>?u-ňaaḥ</i> "look for"
Extended unaccusative	Underived	Locatum-type	Locatum predicate e.g., <i>?u-ćuu</i> "contain"
		Location-type	Location predicate e.g., <i>?u-k<sup>w</sup>i</i> "in"
	Transitivized	Locatum-type	Locatum-type ditransitive e.g., <i>?u-yii</i> "give"
		Location-type	Location-type ditransitive e.g., <i>?u-?iip</i> "give to"

(19)	Classification	of affixal	predicates	which take	nominal	complements
()	erassigreenren	0, 0,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	preanes	minere reme		comprentients

#### 4.1.5 Unergatives are Necessarily Non-affixal

Absent from this inventory of affixal predicates are unergative predicates (Perlmutter 1978; Burzio 1981). That is, the Nuu-chah-nulth equivalents of intransitive verbs such as *cry*, *run*, or *dance* occur only as independent predicates in the language. As may be observed in the examples below, *Siih-šik* "cry-PERF", *kamatq-šik* "run-PERF", and *huut*-"dance" are not suffixed to any element.

#### (20) Unergative predicates: exclusively non-affixal

a.	Siiḥši?aq⊁k Siiḥ-ši⊁-?aq⊁-k cry-PERF-FUT-2S0 Are you going to	G.Q cry?	
b.	saya?ii?iš saya-?ii-?iiš much-go-3.IND S/he ran far.	kamatqa kamatq run-PER	šiλ šiλ F
с.	huu+huu+amitk huu+-a[+R]-mit-k dance-ITER-PST-2 Did you dance la	c 2SG.Q st night?	?atḥiimit?i ?atḥii-mit-?ii night-PST-DET

I relate this systematic absence of affixal unergative predicates to the requirement that affixal predicates must have an independent internal argument. Unergatives lack an independent internal argument. Following Hale and Keyser (1993, 2002), I adopt a *concealed transitive* analysis of unergatives.

(21) Unergative syntax



According to this *concealed transitive* analysis, unergatives are composite predicates which contain an inner VP layer embedded within a vP, as in (23). Unlike in transitives, however, the "phonological signature" (or "p-signature") of V is defective in unergatives. This defectiveness results in the internal argument undergoing *conflation* with the V. Hale and Keyser (2002: 63) describe this process in the following way: (22) Conflation consists in the process of copying the p-signature of the complement into the p-signature of the head, where the latter is "defective".

Thus, the internal argument of an unergative is not independently realized under this analysis. This lack of independence presents a problem for affixation.

By hypothesis, an affixal predicate requires a host chosen from its derivational sister. Two elements are required for this arrangement: an affix,  $-\alpha$ , and a host,  $\beta(-)$ , resulting in  $\beta-\alpha$ . The problem with an unergative affix is that the local spell-out domain of the predicate would contain only one phonologically contentful element, that of the NP complement. In (23), for example, V lacks a phonological form independent of the N(P). Thus, no host-affix dependency may be established between the V and N(P) at local spell-out.



Since affixation is inherently a binary relationship, affixal predicates are not predicted to be possible as the V of an unergative.<sup>1</sup> Affixal predicates require an independent internal argument so that they may find a host within the local spell-out domain.

The remainder of this chapter is organized as follows. The first two sections deal with predicates formed from the basic unaccusative configuration: unaccusatives and transitives. I begin in §4.2 with a discussion of intransitive predicates in Nuu-chah-nulth, and argue for the existence of a class of unaccusative affixal predicates. In §4.3, I discuss transitive affixal predicates, and illustrate syntactic diagnostics for their structure, including subject agreement and possessor raising. Next, I turn to the structure of predicates formed from the extended unaccusative configuration: locatives and ditransitives. In §4.4, it is shown how syntactic diagnostics motivate a distinction between two classes of locative predicates with inverse argument structures, which I label location and locatum predicates. The behaviour of ditransitives is discussed in §4.5, where I analyse these predicates as (di)transitivized locative verbs, which, like locative predicates, may be divided into two classes based on their asymmetrical argument structures. In §4.6, outstanding issues are discussed. Finally, §4.7 gives a summary of the findings.

#### 4.2 Unaccusatives

This section presents evidence for the existence of an unaccusative-unergative distinction in Nuu-chah-nulth, and proposes that there are no unergative affixal

<sup>&</sup>lt;sup>1</sup> I take this to be a diachronically relevant fact: I assume that the restriction was at play when the class of affixal predicates developed. The origins of a distinct class of affixal predicates may possibly be traced as far back as Proto-Wakashan. Perhaps the expletive host  $\partial \mu$ - was not available as a host at the time that the class of affixal predicates developed.

predicates in Nuu-chah-nulth. By my analysis, intransitive affixal predicates in Nuu-chah-nulth are exclusively unaccusative.

## 4.2.1 Incorporation

-

Predicates in the unaccusative class include 2u-nii "arrive", 2u-pa-t "be present", 2u-Saa?atu "move down". As indicated in the following (a) examples, these unaccusative predicates allow suffixation to their argument. The examples in (b) show affixation to the expletive host 2u-.

(24)	a.	paastinratinnris paastinrath- <u><b>áii</b>-</u> riiš American- <u>arrive</u> -3.IND Americans came.
	b.	?uniii?iš paastin?atḥ?i ?u- <u>nii</u> -?iiš paastin?atḥ-?ii Ø- <u>arrive</u> -3.IND American-DET The Americans came.
(25)	a.	tapałwa?iš pišaqaqwa?iš ta- <b>pał</b> -wa?iš piš-aq[+S]-aq[+S]-wa?iš sick- <b>present</b> -3.QUOT bad-AUG-AUG-3.QUOT There's sickness around, (and) it is really bad.
	b.	?upa†wa?iš tamis pišaqaqwa?iš ?u- <u>pa†</u> -wa?iš ta-mis piš-aq[+S]-wa?iš Ø- <u>present</u> -3.QUOT sick-NOM bad-AUG-AUG-3.QUOT There's sickness around, (and) it is really bad.
(26)	a.	ťaťuusSa?atumit?iš ťaťuus- <u>Saa?atu</u> -mit-?iiš star- <u>move.down</u> -PST-3.IND A star fell.
	b.	?uSaa?atumit?iš ťaťuus ?u- <u>Saa?atu</u> -mit-?iiš ťaťuus Ø- <u>move.down</u> -PST-3.IND star A star fell.

Incorporation of the argument of an unaccusative is predicted by the PF incorporation analysis. The internal argument is a derivational sister to the affixal predicate, and thus forms a local spell-out domain with the affixal predicate. When the affixal predicate reaches spell-out, it looks to this derivational sister for its host. For example, if the affixal predicate *?u-nii* "arrive" takes *paastin?ath* "American(s)" as its argument, then the reflex of spell-out will be a linearization of *paastin?ath-nii*, as in *paastin?athni?is* "Americans came" in (24a). The structural relationship underlying PF incorporation is given in (27).



The affixal predicate suffixes to its derivational sister by local spell-out.

The alternative to suffixation to a derivational sister is suffixation to a host inserted at spell-out, the expletive morpheme  $\lambda -$ . In (24b), the presence of the determiner  $-\lambda i$  signals saturation of the phrase, with the DP imposing a "border" between the affixal predicate and the members of its complement.

(28) -*i*ii DP arrive -*i*ii paastin?atḥ Americar

As noted in Chapters 2 and 3, DPs in Nuu-chah-nulth act as independent spell-out domains in that they form "islands" for affixation. Inflectional clitics, for example, are built up within a DP and do not cross it. In a context such as (28), the expletive  $\lambda u$ - "rescues" the stranded affixal predicate by acting as its host. PF incorporation is not possible in this context.<sup>2</sup>

In the next section, I discuss evidence for an unergative-unaccusative distinction in Nuu-chah-nulth.

#### 4.2.2 Intransitivity in Nuu-chah-nulth

According to the Unaccusativity Hypothesis (Perlmutter 1978; Burzio 1981), there are two subclasses of monadic predicates – unaccusatives and unergatives – which are associated with different underlying syntactic configurations. While the argument of an unaccusative verb such as "arrive" or "die" is an internal argument, the single (overt) argument of an unergative verb such as "cry" or "dance" is an external argument. In the framework which I have adopted, the difference between these two intransitive types may be represented according to an asymmetry as to which syntactic head introduces the argument. The argument of an unaccusative is generated as the complement of V, in the same position as the object of a transitive predicate. For unergatives, however, the argument is introduced by a v head, in the same position as the subject of a transitive predicate (Hale and Keyser 1994; Kratzer 1994; Chomsky 1995).

<sup>&</sup>lt;sup>2</sup> A topic for further research is why the expletive  $\lambda u$ - is employed in cases such as (26b), which lack a determiner. Incorporation is apparently optional in such cases. Textual analysis may shed light on whether use of  $\lambda u$ - correlates with specific stylistic effects.



Recall that I adopt the analysis that unergatives are "concealed" transitives which have a conflated internal argument. Thus, only the top argument (circled) of the unergative is overtly realized as an independent argument.

In Nuu-chah-nulth, unaccusatives and (non-affixal) unergatives receive the same type of subject agreement. A predicate such *siih* "cry" takes the same subject agreement as does a predicate such as *hinin* "arrive". In (31), both are inflected for the third person indicative subject agreement -2iis (3.IND).

(31)	a.	SiḥSiiḥamit?iš Siiḥ-a[+R]-mit-?iiš cry-IT-PST-3.IND My auntie was crying.	na?iiqsakqs na?iiqsu-?ak-qs aunt/uncle-POSS-1SG.PS
	b.	hinin?a⊁?iš hinin-?a⊁-?iiš arrive-TEMP-3.IND My auntie has arrived nov	na?iiqsakqs na?iiqsu-?ak-qs aunt/uncle-POSS-1SG.PS v.

Despite this superficial similarity of unergatives and unaccusatives in Nuu-chahnulth, I will advance two separate pieces of evidence for a distinction between unaccusative and unergative intransitives in the language. First, I propose that there is distributional evidence for unaccusativity, based on the class membership of affixal predicates (§4.2.3). Second, I propose that "long" possessor raising constitutes a reliable syntactic diagnostic for unaccusativity in Nuu-chah-nulth (§4.2.4).

# 4.2.3 Absence of Unergative Affixal Predicates

A systematic gap in the composition of the Nuu-chah-nulth lexicon supports a contrast in this language between the two monadic classes of unergatives and unaccusatives. While one-place predicates with the semantics of typical unaccusative verbs (e.g., *die*, *arrive*) are found amongst both the affixal and non-affixal classes in Nuu-chah-nulth, to the best of my knowledge, one-place predicates with the semantics of typical unergative predicates (e.g., *work*, *cry*, *dance*) exist only as non-affixal predicates.

The generalization that unergative predicates are absent from the affixal predicate inventory is supported by the suffix lists in Sapir and Swadesh (1939), and by the grammars of Rose (1981) and Davidson (2002). For example, consider Rose's (1981) description of the two classes of "verbal affixes" in Kyuquot, a northern dialect of Nuu-chah-nulth. Rose labels the two verbal classes "governing" or "restrictive". Amongst the "governing" category of verbal affixes that take an "NP object base" (rather than a sentential one), we find predicates with transitive and ditransitive syntax, according to the present classification. Unaccusatives are found amongst Rose's class of "restrictive verbal affixes". Importantly, there is no other category of "verbal affix" in Rose's list which plausibly resembles unergatives. The subclasses of "verbal affixes" discussed in Rose (1981) are illustrated in the following table with their correspondences to the present classification system.

Classification	Ahousaht examples	Kyuquot cognates	Label in Rose	
Unaccusative	<i>Pu-hii</i> "arrive" <i>Puu-Patu</i> "sink, go down" <i>Pu-suu</i> * "die, get destroyed"	- <i>ii</i> "come home, arrive" - <i>?ata</i> "sink, go down" - <i>suwi(</i> $\lambda$ ) (~ <i>suu</i> ( $\lambda$ )) "die, get destroyed"	Restrictive verbal affix	
	<i>?u-yii?i?</i> : "come into house"	- <i>ii?i(x̂)</i> "come into house"		
Transitive	<i>?uu-taq</i> "fix, work on" <i>?u-k<sup>w</sup>ii?</i> "make" <i>?u-?iic</i> "consume" <i>?u-ĉii?t</i> h "use as fuel"	<i>-taq</i> [+L] "work on" -(č) <i>ii†</i> "make" - <i>'iic</i> "eat" - <i>cith</i> "use as fuel"	Governing verbal affix -NP object base	
Ditransitive	<i>?u-?iip</i> "give" <i>?u-yii</i> "give" <i>?uu-kš</i> "ask for"	- <i>?iip</i> "give" - <i>aayi</i> "give" -(k)š [+L] "ask for"	Governing verbal affix -NP object base	
Auxiliary	- <i>sinḥi</i> [+L] "try to continue" - <i>maʕiiqૠ</i> "need to" (bodily functions)	<i>-sinh</i> [+L] "try to (be)" <i>-maʕaqૠ</i> "want to (be)"	Governing verbal affix -sentential base	

(32) Types of "Verbal affixes"

Note that a "governing verbal affix" such as *Auu-taq* "work on" is strictly transitive, with a meaning similar to "fix", rather than with a usage parallel to English intransitive *work*.

(33)	a.	?uutaqitsiš	muunaa
		?uu- <u>taq</u> -mit-siiš	muunaa
		Ø- <u>work.on</u> -PST-1SG.IND	engine
		I was working on an engin	ne.

b. \* ?uutaqitsiš ?uu-<u>taq</u>-mit-siiš
Ø-<u>work.on</u>-PST-1SG.IND I was working.
(*consultant's comment*: "you have to tell what you were fixing or working on")

In a similar vein, the affixal predicate *2u-2iic* "consume" has only a transitive usage. In the example below, the object *sapnii* "bread" is mandatorily expressed.

(34)	a.	?u?iiċamitsiš ?u- <u><b>'iic</b></u> -'ap-mit-siiš Ø- <u>consume</u> -TR-PST-1SG.IND I made Ken eat bread.	Ken Ken Ken	sapnii sapnii bread
	b. *	?u?iićamitsiš ?u- <u>'iic</u> -'ap-mit-siiš Ø- <u>consume</u> -TR-PST-1SG.IND I made Ken eat.	Ken Ken Ken	

This contrasts with the behaviour of the independent predicate *ha?uk* "eat", which, like English *eat*, allows for both an intransitive and transitive usage. In (35a), the object *sapnii* "bread" is expressed; in (35b), it is not.

(35)	a.	ha?uk̇̀*amitsiš	Ken	sapnii
		ha?uk-'ap-mit-siiš	Ken	sapnii
		eat-TR-PST-1SG.IND	Ken	bread
		I made Ken eat bread.		
	b.	ha?ukwamitsiš	Ken	
		ha?uk-'ap-mit-siiš	Ken	
		eat-TR-PST-1SG.IND	Ken	
		I made Ken eat.		

In the next section, we see further evidence that intransitive affixal predicates in Nuu-chah-nulth pattern as unaccusatives and not as unergatives.

#### 4.2.4 "Long" Possessor Raising as a Diagnostic for Unaccusativity

As first described in Chapter 2, Nuu-chah-nulth has a process of possessor raising in which the possessive marker -uk/-(2)ak (POSS) suffixes to a main predicate, instead of (or in addition to) suffixing to the possessum (Davidson 2002; Ravinski 2005). In (36a), the possessum  $k^waaluuc$  "grandchild" is suffixed by -uk (POSS). In the possessor raised (36b), -uk (POSS) suffixes to the predicate *talif* "sick".

(36) a.		ta?i <sup>+</sup> ?iš k <sup>w</sup> aa?uucukqs ta?i <sup>+</sup> -?iiš k <sup>w</sup> aa?uuc-uk-qs sick-3.IND grandchild-POSS-1SG.PS My grandchild is sick.		(unraised)
	b.	ta?i†uksiš ta?i†-uk-siiš sick-POSS-1SG.IND	k <sup>w</sup> aa?uuc k <sup>w</sup> aa?uuc grandchild	

In possessor raising, the possessor ends up determining subject agreement for the clause: in (36b), the subject agreement is -siis (1SG.IND) because it matches the first person possessor of  $k^{waa}auc$  "grandchild". Standardly, possessor raising targets the surface subject of the main predicate in Nuu-chah-nulth, including the derived subjects of passives and unaccusatives (Ravinski 2005). For main predicates, possessor raising is insensitive to the difference between the arguments of unergatives and unaccusatives. As shown in (37), possessor raising is permitted with the argument of an "unergative" intransitive such as *fiih* "cry", or the argument of an "unaccusative such as *hinin* "arrive".

(possessor raising)

(37)	a.	SiḥSiiḥakitsiš Siiḥ-a[+R]-?ak-mit-siiš cry-ITER-POSS-PST-1SG.IN My auntie was crying.	na?iiqsu na?iiqsu D aunt/uncle	
	b.	hinin?akitsiš hinin-?ak-mit-siiš arrive-POSS-PST-1SG.IND My auntie came	na?iiq na?iiq aunt/uncle	

My grandchild is sick.

However, when possessor raising applies in contexts of affixal auxiliaries such as -qaath "claim" or -mahsa "want to", a distinction emerges between unaccusatives and other types of predicates. When an unaccusative predicate combines with an affixal auxiliary, two possible interpretations are available in contexts of possessor raising. The example in (38) shows these two interpretations for the predicate taht "sick", which is suffixed by the affixal auxiliary -qaath "claim".<sup>3</sup> (In Nuu-chah-nulth, taht "sick" patterns with the class of unaccusative verbs.) In the first interpretation in (38), the one who doing the claiming is the same one who is sick ("my grandchild"). A second interpretation is also available in which the claimer and the one who is sick are disjoint: the "claimer" is a first person argument, and the one who is sick is "my grandchild". (For the second type of interpretation, the person

<sup>&</sup>lt;sup>3</sup> The interpretations are disambiguated by context. For example, in a sentence such as *ta?i+qathuk<sup>w</sup>itsiš k<sup>w</sup>aa?uuc čińuq*?  $\lambda$ *iis* $\lambda$ 

and number features of the "claimer" necessarily matches the person and number of the possessor of the one who is sick.)

(38)	ta?i1qatḥuk <sup>w</sup> itsiš	k <sup>w</sup> aa?u	uc
	ta?i1- <b>qaath</b> -uk-mit-siiš	k <sup>w</sup> aa?u	uc
	sick- <u>claim</u> -POSS-PST-1SG.IND	grandc	hild
	= (i) My grandchild claimed to b	be sick.	("short" possessor raising)
	= (ii) I claimed my grandchild w	as sick.	("long" possessor raising)

I refer to the first interpretation as a case of "short" possessor raising, and the second as an instance of "long" possessor raising, for reasons which will soon become apparent.

For unergative predicates, only a "short" possessor raising interpretation is available; "long" possessor raising is impossible. This is indicated in (40) with the unergative predicate *Saaq-ši* $\lambda$  "shout (PERF)". The "claimer" and the "shouter" must be the same person in this possessor raising context.

(39)	Saaqši <del>≿</del> qatḥuk <sup>w</sup> itsiš	naniiqsu
	Saaq-šiλ- <b>qaath</b> -uk-mit-siiš	naniiqsu
	shout-PERF-claim-POSS-PST-1SG.IND	grandparent
	= (i) My grandparent claimed to shout.	("short" possessor raising)
	$\neq$ (ii) I claimed my grandparent shouted.	("long" possessor raising)

Thus, the availability of a "long" possessor raising interpretation distinguishes between the class of unaccusatives and the class of unergatives: unaccusatives such as ta2it "sick" are compatible with "long" possessor raising, while unergatives such as faaq-sit "shout (PERF)" are not.

A sketch of my analysis of the two types of possessor raising follows. As I will argue in Chapter 5, affixal auxiliaries such as -qaath "claim" in (40) are raising verbs which do not project a subject of their own, but which license raising of an embedded argument.<sup>4</sup>

(40) nunuukqath?iš Florence nunuuk-**qaath**-?iišFlorence sing-**claim**-3.IND Florence Florence is pretending to sing.

The diagram in (41) represents how *Florence* takes on the role of "shared" subject of the auxiliary -qaath "claim" and the main predicate *nunuuk* "sing". As shown, *Florence* originates as the subject of the main predicate, and raises to specifier position of the auxiliary.

<sup>&</sup>lt;sup>4</sup> In Chapter 5, this raising analysis is motivated by the fact that affixal auxiliaries show a "same subject" effect, in which they require transitivization whenever a different subject is used with the main verb. The English gloss for the affixal auxiliary -qaath "claim" comes from its typical translation in these transitive contexts.

(41) Affixal auxiliaries as raising verbs



Following Ravinski (2005), I assume that the possessive morpheme -uk (POSS) licenses a position for raised possessors. This analysis is illustrated in (42) for the non-auxiliary case of possessor raising *ta?i+uksiš k<sup>w</sup>aa?uuc* "My grandchild is sick", from (36b). Here, the first person possessor of *k<sup>w</sup>aa?uuc* "grandchild" raises to specifier position of PossP.

(42) "Simple" possessor raising



In "short" and "long" possessor raising with auxiliaries, the syntax of the auxiliary interacts with the syntax of possessor raising.

(43)	ta?i <del>1</del> qatḥuk <sup>w</sup> itsiš	k <sup>w</sup> aa?uu	c
	ta?i1- <b>qaath</b> -uk-mit-siiš	k <sup>w</sup> aa?uu	c
	sick- <u>claim</u> -POSS-PST-1SG.IND	grandch	ild
	= (i) My grandchild claimed to b	e sick.	("short" possessor raising)
	= (ii) I claimed my grandchild w	as sick.	("long" possessor raising)

In the case of "short" possessor raising with auxiliaries, I propose that the entire possessed nominal ("my grandchild") raises to a "subject" position of the auxiliary, in specifier position of FP. This is followed by possessor-extraction, which raises the possessor to specifier of PossP. In "long" possessor raising, in contrast, the possessor is raised twice, on its own. A first move raises the possessor to "subject" position of the auxiliary, while the second move takes the possessor to specifier of PossP. The difference between the "short" and "long" possessor raising is illustrated below.

# (44) Possessor raising with unaccusative main predicate



b. "long" possessor raising



Thus, in the "short" case of (44a), "my grandchild" acts as the (derived) subject of -qaath "claim", while in the "long" case of (44b), -qaath "claim" has a first person subject (equivalent to the first person possessor). In both cases, the first person possessor occupies specifier position of PossP, and ultimately ends up determining the first person subject agreement of the clause, -siis (1SG.IND).

As noted in the previous discussion, only unaccusative predicates such as ta?if "sick" permit "long" possessor raising. As illustrated in (45), the argument of an unergative such as *Gaaq-šik* "shout (PERF)" is not compatible with "long" possessor raising.

#### (45) Illicit "long" possessor raising with unergative main predicate



The cause for this restriction is in need of further investigation. Whatever the grammatical motivations for this contrast, it serves as a reliable diagnostic for unaccusatives. The unaccusativity restriction is an empirically robust distinction, and holds for Nuu-chah-nulth speakers across a range of predicates.

Intransitive predicates such as *mamuuk* "work", *hita?ap* "win", and *yaac* "walk" all disallow a "long" possessor raising interpretation. The possible interpretations are shown below with the affixal auxiliary *–maḥsa* "want to".

(46)	mamuukmaḥsaksiš naʔiiq mamuuk- <u>maḥsa</u> -ʔak-siiš naʔiiq	
	work- <u>want</u> -POSS-1SG.IND aunt/uncle	
	= (i) My aunt/uncle wants to work.	("short" possessor raising)
	$\neq$ (ii) I want my aunt/uncle to work.	("long" possessor raising)
(47)	hita?apmaḥsaksiš naniiq	
. ,	hita?ap- <b>mahsa</b> -?ak-siiš naniiq	
	win-want-POSS-1SG.IND grandparent	
	= (i) My grandparent wants to win.	("short" possessor raising)
	$\neq$ (ii) I want my grandparent to win.	("long" possessor raising)
(48)	yaacši <del>x</del> maḥsaksiš na?iiq	Isu
	yaac-ši⊁- <u>maḥsa</u> -?ak-siiš na?iiq	lsu
	walk-PERF- <u>want</u> -POSS-1SG.IND aunt/	uncle
	= (i) My aunt/uncle wants to go for a wal	lk. ("short" possessor raising)
	$\neq$ (ii) I want my aunt/uncle to go for a wa	alk. ("long" possessor raising)

These unergative intransitives pattern together with transitive predicates such as  $\partial u$ -k mit (make" *his-sik* "hit (PERF)", and  $\partial u$ - $\partial aap$  "buy", which also disallow "long" possessor raising. In (49), only a "short" possessor raising interpretation is permitted with the transitive predicate  $\partial u$ -k mit "make". Likewise, in (50), a "long" possessor raising interpretation is shown to be impossible for the transitive predicate *his-sik* "hit (PERF)". Finally, the example in (51) shows this restriction against "long" possessor raising for the transitive predicate  $\partial u$ - $\partial aap$  "buy".

(49)	saapniqi†maḥsaksiš	naniiqsu	
	saapniq- <u>čii†</u> - <u>maḥsa</u> -?ak-siiš	naniiqsu	
	bread- <u>make</u> - <u>want</u> -POSS-1SG.IND	grandparent	
	= (i) My grandparent wants to make	ke bread. ("shor	rt" possessor raising)
	$\neq$ (ii) I want my grandparent to ma	ike bread. ("long	" possessor raising)
(50)	hisši <del>λ</del> qatḥuk <sup>w</sup> itsiš	ỷuk <sup>w</sup> iiqsu	Ray
	his-ši≁- <b>qaath</b> -uk-mit-siš	yuk <sup>w</sup> iiqsu	Ray
	hit-PERF- <u>claim</u> -POSS-PST-1SG.IND	y.sibling	Ray

= (i) My younger sibling claimed s/he hit Ray. ("short" possessor raising)

 $\neq$  (ii) I claimed my younger sibling hit Ray ("long" possessor raising)

(51) huupuk<sup>w</sup>as?apqatḥuk<sup>w</sup>itsiš yuk<sup>w</sup>iiqsu huupuk<sup>w</sup>as-<u>?aap-qaat</u>h-uk-mit-siiš yuk<sup>w</sup>iiqsu car-<u>buy-claim</u>-POSS-PST-1SG.IND y.sibling
= (i) My younger sibling claimed s/he bought a car. ("short" poss. raising) ≠ (ii) I claimed my younger sibling bought a car. ("long" possessor raising)

Conversely, unaccusative predicates consistently allow the "long" possessor raising interpretation. In the appropriate context, either a "short" possessor raising or a "long" possessor raising interpretation is allowed for the sentences below with the unaccusative predicate  $\underline{sahyut}$  "healthy". The sentence in (52a) shows both readings in the context of the auxiliary -mahsa "want", while (52b) illustrates parallel readings with the auxiliary -qaath "claim".

(52)	a.	šaḥyutmaḥsaksiš	naniiqsu	L
		šaḥyut- <u>maḥsa</u> -?ak-siiš	naniiqsu	L
		healthy-want-POSS-1SG.IND	grandpa	rent
		= (i) My grandparent wants to be	well.	("short" possessor raising)
		= (ii) I want my grandparent to be	well.	("long" possessor raising)
	b.	šahýutqathuk <sup>w</sup> itsiš	naniiqsu	
		šahýut- <b>qaath</b> -uk-mit-siiš	naniiqsu	
		healthy- <u>claim</u> -POSS-PST-1SG.IND	grandpa	rent
		= (i) My grandparent claimed to b	e well.	("short" possessor raising)
		(::) <b>T</b> -1-:	o o vrvo 11	("lawa" nananan unining)

For some, but not all, of consultants, the "long" possessor raising interpretation is highly salient for unaccusative examples such as (52). However, for all consultants there are some contexts in which only a "long" possessor raising interpretation arises. When an unaccusative main predicate takes an inanimate argument, a "short" possessor raising interpretation is ruled out – perhaps due to the pragmatic restriction that the inanimate cannot control a desiderative auxiliary such as –*maḥsa* "want". This is illustrated with the unaccusative main predicate *tuq-šiૠ* "melt (PERF)", which in the following sentence takes the inanimate argument *pata* "butter".

(53)	tuqši⊁maḥsaksiš	pata	
	tuq-ši⊁- <u><b>maḥsa</b></u> -?ak-siiš	pata	
	melt-PERF- <u>want</u> -POSS-1SG.IND	butter	
	$\neq$ (i) ! My butter wants to melt.		("short" possessor raising)
	= (ii) I want my butter to melt.		("long" possessor raising)

In this example, the only pragmatically available reading is a "long" possessor raising interpretation in which the controller of the auxiliary predicate *-mahsa* "want" is the same as the first person possessor of the main predicate's argument, *pata* "butter". A "short" possessor raising interpretation is impossible, since this entails an absurd reading in which the controller of the main and auxiliary predicates is the possessive nominal itself ("my butter"). Additional examples of "long" possessor raising with unaccusatives are shown in (54). Predicates which are compatible with "long" possessor raising include *puux-šik* "rise (PERF)", *nii?atu* "sink", *path-aa* "shine (CONT)", and *cah-aa* "leak (CONT)".

(54)	a.	puuxši⊁maḥsaksiš puux-ši⊁- <b>maḥsa</b> -ʔak-siiš rise-PERF- <u>want</u> -POSS-1SG.IND I want my bread to rise.	sapni sapni bread	i	
	b.	nii?atuqatḥuk <sup>w</sup> it?ick nii?atu- <b>qaatḥ</b> -uk-mit-?iick sink- <u>claim</u> -POSS-PST-2SG.IND You claimed your boat sank.	muut muut boat		
	с.	patḥaamaḥsaksiš patḥ-aa- <u>maḥsa</u> -ak-siiš shine-CONT- <u>want</u> -POSS-1SG.IND I want my fishing spoon to shine.	tii <del>1</del> a tii <del>1</del> a fishin	g.spoon	
	d.	caḥaaqatḥukʷitʔick caḥ-aa- <b>qaatḥ</b> -uk-mit-ʔiick leak-CONT- <u>claim</u> -POSS-PST-2SG.INI You claimed your boat is leaking,	muut muut Dboat and yet	?ata?ick ?ata-?iick but-2SG.IND you're trave	?uyiiq ?u-yiiq ∅-travel.in lling in it.

In sum, "long" possessor raising can be used as a diagnostic to separate unaccusative predicates from transitive and unergative classes, since only unaccusative main predicates ever allow "long" possessor raising. The results of this diagnostic are summarized in (55).

Main predicate	"long" possessor raising
A. Transitive	
<i>?u-k<sup>w</sup>ii1</i> "make"	×
hisšin "hit"	×
<i>?u-?aap</i> "buy"	×
B. Unergative	
mamuuk "work"	×
hita?ap "win"	×
<i>Saaqšii</i> t "shout"	×
<i>yaacšii</i> ? "walk"	×
C. Unaccusative	
<i>šahýut</i> "healthy"	$\checkmark$
<i>tuqšiit</i> "melt"	$\checkmark$
puuxšiit "rise"	✓
<i>nii?atu</i> "sink"	✓
pathaa "shining"	✓
cahaa "leaking"	✓ ✓
<i>?u-suu</i> <sup>*</sup> "die"	<b>√</b>
<i>?u-nii</i> "arrive"	~

(55)	"Long"	' possessor	raising	as a diagn	ostic for	unaccusativity
------	--------	-------------	---------	------------	-----------	----------------

If we apply the "long" possessor raising diagnostic to intransitive affixal predicates, we see that this test confirms that these affixal predicates behave similarly to non-affixal predicates such as  $\dot{s}ah\dot{y}ut$  "healthy" and  $tuq-\dot{s}i\lambda$  "melt (PERF)" in that they allow a "long" possessor raising interpretation. In other words, these intransitive affixal predicates pattern as unaccusatives. In the sentences below, the intransitive affixal predicate  $\partial u-suu\lambda$  "die" appears in complex predicates with the affixal auxiliary -qaath "claim". Both "short" and "long" possessor raising is permitted, dependent on the discourse context.

#### (56) a. "short" possessor raising

## b. "long" possessor raising

?usuu <del>λ</del> qatḥuk?ick	nani	?ata?iš	tiič
?u- <u>suu</u> 2-qaath-uk-?iick	nani	?ata-?iiš	tiič
Ø- <u>die</u> - <u>claim</u> -poss-2sg.ind	grandparent	but-3.IND	alive
You claim your grandparent	t died, but she is	alive.	

"Long" possessor raising is possible with other intransitive affixal predicates, such as  $\partial u$ -n'i" arrive".

#### (57) *"long" possessor raising*

Puniiqathuk?icknaniiqsuPatquuwikiitPu-<u>nii-qaath-uk-?iick</u>naniiqsuPat-quuwikiitØ-arrive-claimPOSS-2.INDgrandparentbut-3.CONDNEG-presentYou pretend your grandparent came, although she isn't here.NEG-presentNEG-present

In conclusion, the "long" possessor raising diagnostic supports a classification in which affixal predicates like 2u-suu $\hat{x}$  "die" and 2u- $\hat{n}ii$  "arrive" are unaccusative.

## 4.3 Transitives

This section considers the behaviour of transitive affixal predicates. According to the proposal in §4.1.2, these predicates project up to vP, via abstract transitivization of a basic unaccusative configuration. Representative examples of transitive affixal predicates are shown below, which illustrate suffixation of the predicate to a nominal host. The predicates 2u-2int "serve", 2u2u-2iih "gather", and  $2u-k^wiit$  "make" are each proposed to be transitive.

- (58) a. čamayin†itsiš čamas-<u>'in†</u>-mit-siiš sweets-<u>serve</u>-PST-1SG.IND I served sweets.
  - b. ťuťučiiḥ?iš nani ťučup-<u>'iiḥ[</u>+R]-?iiš nani sea.urchin-<u>gather</u>-3.IND grandparent Grandparent is gathering sea urchin.
  - c. saapniqii†?iš ?um?i saapniq-<u>čii†</u>-?iiš ?um?i bread-<u>make</u>-3.IND mom Mom is making bread.

In the absence of incorporation, these predicates suffix to the the expletive pronoun 2u-.

(59)	a.	?u?in∔itsiš ?u- <u><b>'in†</b>-</u> mit-siiš Ø- <u>serve</u> -PST-1S0 I served sweets.	G.IND	čamas čamas sweets	
	b.	?u?u?iiḥ?iš ?u- <u><b>`iiḥ</b>[+R]</u> -?iiš ∅- <u>gather</u> -3.IND Your grandparer	nt is gathe	nani nani grandparent ering sea urchin.	ťučup ťučup sea.urchin
	c.	?uk <sup>w</sup> ii†?iš ?u- <u>čii†</u> -?iiš ∅- <u>make</u> -3.ıND Mom is making	?um?i ?um?i mom bread.	sapnii sapnii bread	

I present evidence in §4.3.1 that only complements may act as the source of incorporation for transitive affixal predicates. Two tests are used to independently affirm the existence of a distinction between syntactic positions of the arguments of a transitive predicate: subject agreement (§4.3.2) registers an argument in subject position; possessor raising (§4.3.3) is possible only out of a subject, and not out of an object. Thus, in active contexts, incorporation is exclusively reserved for an argument which is incompatible with subject agreement or possessor raising.

## 4.3.1 Incorporation

There is an incorporation asymmetry in Nuu-chah-nulth between the two arguments of a transitive affixal predicate. Recent work has shown that Nuu-chah-nulth affixal predicates incorporate their object, and not their subject (Woo 2000; Davis and Sawai 2001; Stonham 2004). In Chapter 2, I termed this restriction on affixation the "complement" effect. The asymmetry is illustrated in the examples below, which show that the two arguments of a transitive predicate do not have equal ability to serve as the host for the affixal predicate. While  $\lambda u$ - $\lambda aap$  "buy" can suffix to the nominal mahta-"house", it cannot suffix to the nominal  $\lambda apx$ -"man".

130

(60)	a.		maḥťa?amit?iš	čakup
			maḥťa- <u><b>'aap</b></u> -mit-?iiš	čakup
			house- <u>buy</u> -PST-3.IND	man
			A man bought a house.	
	b.	*	čapx?aamit?iš	maḥťii
			čapx- <u><b>'aap</b>-mit-?iiš</u>	maḥťii
			man- <b>buy</b> -PST-3.IND	house
			A man bought a house.	

This incorporation asymmetry is robust. For example, in *wh*-questions (Davis and Sawai 2001) and relative clauses formed with an affixal predicate, incorporation of the *wh*- or relative-pronoun into the affixal predicate is obligatory for object-questions and object-relativizations, but is impermissible for subject-oriented ones. As Davis and Sawai (2001) describe, incorporation is mandatory for *wh*-pronouns that occur as the object of an affixal predicate. It is shown in (61) that the *wh*-object *?aqi*-"what" must incorporate into the affixal predicate *?u-?iic* "consume". In (61a), *?aqi*- "what" grammatically hosts the affixal predicate, while in the ungrammatical (61b) incorporation fails to occur as *?u*-support is employed instead.

(61)	a.		?aqiicith		John	
			?aqi- <b>`iic</b> -mit-ḥ		John	
			what- <u>consume</u> -	pst-3.q	John	
			What did John e	at?	(Davi	s and Sawai 2001:127; ex.11)
	b.	*	?aaqiči†h	?u?iic		John
			?aqi-či≁[+L]-ḥ	?u- <b>'iic</b>		John
			what-AUX-3.Q	Ø- <u>cons</u>	ume	John
			What did John e	at?	(Davi	is and Sawai 2001:127; ex.16

In contrast, incorporation into the affixal prediate is ruled out for *wh*-subjects. The example in (62a) shows that it is ungrammatical for the *wh*-subject *?ačaq-* "who" to incorporate into the affixal predicate *?u-?iic* "consume". As shown in (62b), a *wh*-subject must occur independently of the affixal predicate.

(62)	a.	*	?ačaŝiicitḥ ?ačaq- <u><b>`iic</b>-mit-ḥ</u> who- <u>consume</u> - Who ate the sal	PST-3.Q mon?	suuḥaa suuḥaa salmon (Davi	s and Sav	wai 2001: 129; ex. 19)	
	b.		?ačaqith ?ačaq-mit-h who-PST-3.Q Who ate this sa	?u?iic ?u- <mark>'iic</mark> ∅- <u>cons</u> lmon?	<u>sume</u> (cf. Davi	hiitas hiitas DEIC is and Sa	suuḥaaʔi suuḥaa-ʔii salmon-DET wai 2001: 130, ex. 22a	)

The same subject/object asymmetry is found with relative clauses, as illustrated below with the affixal predicate *2uu-2inhi* "wait for". The relative pronoun *yaq* 

"who" incorporates in an object relativization (63a), but not in a subject relativization (63b).

- (63) a. hačumsiqsaksiš haa čakup?i yaa\inhi?itq Mary hačumsiqsu-?ak-siiš haa čakup-?ii yaq-<u>?inhi[</u>+L]-?iitq Mary brother-POSS-1SG.IND DEIC man-DET REL-<u>wait.for</u>-3.RL Mary That man who Mary is waiting for is my brother.
  - b. hačumsiqsaksiš haa čakup?i yaq?itq ?uu?inhi Mary hačumsiqsu-?ak-siiš haa čakup-?ii yaq-?iitq ?u-<u>?inhi[+L]</u> Mary brother-POSS-1SG.IND DEIC man-DET REL-3.RL Ø-<u>wait.for</u> Mary That man who is waiting for Mary is my brother.

Under a PF incorporation analysis, the distinct patterns of subjects and objects are anticipated. An object of a transitive predicate, but not its subject, is generated in complement position. As the derivational sister of an affixal predicate, an object nominal undergoes local spell-out with the affixal predicate. This derives the "complement" effect.



At spell-out, a linearization of mahta-?aap "house-buy" is induced.

In the next section, we turn to our first of two syntactic diagnostics which corroborate the analysis that it is the syntactic complement of a transitive affixal predicate which serves as the source of incorporation.

## 4.3.2 Subject Agreement

Agreement in Nuu-chah-nulth corresponds to the subject of a transitive predicate, not to an object. The table below lists the Ahousaht agreement paradigm for the indicative mood, one of several mood inflections in the language (see Chapter 3; Nakayama 1997, 2001).

## (65) Indicative subject agreement

	Singular	Plural
1	-siiš	-niiš
2	-?iick	-?iicuuš
3	-1	liiš

Subject agreement registers the subjects of affixal and non-affixal predicates alike. For example, in (66a), the first person singular marker -siis references the (*pro*) subject of the affixal predicate 2uu-hcii "cook"; in (66b) the same marker applies to the non-affixal predicate *kith-sik* "phone (PERF)".

?uuhćiisiš čisqmis (66)a. ?u-hćii[+L]-siiš ćisamis Ø-cook-1SG.IND meat I am cooking meat. b. kithši?aq⊁siš suwa ?athii wikquus haana?a§as kith-ši<sup>2</sup>-?aq<sup>2</sup>-sii<sup>3</sup> suwa ?athii wik-quus haana?aq-'as ring-PERF-FUT-1SG.IND you tonight NEG-1SG.CON lahal-ASP I'll phone you tonight if I don't go to the lahal game.

I assume that subject agreement in Nuu-chah-nulth is licensed by the highest argument of a predicate (Ravinski 2005). Higher arguments are chosen over lower arguments due to the Minimal Link Condition (Chomsky 1995), in which short-distance relationships are preferred over long-distance ones. This is represented in (67), in which the external argument of a transitive is the closer to the agreement projection than the internal argument of the transitive. Subject agreement can be assumed to be licensed through an AGREE relation between a head (e.g., Agr) and the most local DP (Chomsky 1995).



Given the analysis that complements of an affixal predicate act as "incorporated" hosts, and given that complements of transitives are not the closest nominal to Agr, we expect that an incorporated nominal should not determine the subject agreement of a transitive predicate. This indeed holds. In (68) below, the nominal *cisquis* "meat" incorporates into *Auu-hcii* "cook", while it is the first person argument which determines subject agreement.

(68) čiisqmishćisiš čisqmis-<u>hčii[+L]-siiš</u> meat-<u>cook</u>-1SG.IND I am cooking meat.

In the next section, we see further evidence that complements of the transitive affixal predicate are the source of incorporation.

## 4.3.3 Possessor Raising

An additional diagnostic for the syntactic structure of transitive affixal predicates is supplied by the possessor raising construction. Recall from Chapter 3 that possessor raising only ever targets subjects in Nuu-chah-nulth (Davidson 2002; Ravinski 2005). This is illustrated in the sentence below with the predicate *hin-k<sup>wa</sup>?iih* "chase".

(69) hink<sup>w</sup>a?iiḥuksiš piišpiš maamaati hin-k<sup>w</sup>a?iiḥ-uk-siiš piišpiš maamaati LOC-after-POSS-3.IND cat bird = (i) My cat was after a bird.  $\neq$  (ii) A cat was after my bird.

In this example of possessor raising, the possessive morpheme -uk suffixes to the predicate, rather than to the possessum, and the possessor argument controls subject agreement (Davidson 2002; Ravinski 2005). What is noteworthy for our purposes is that the subject *piišpiš* "cat" is eligible to receive an interpretation as the possessum, but the object *maamaati* "bird" is not. Thus, we infer that only the subject is able to act as the source of possessor raising. The subject restriction on possessor raising is illustrated in (70).

(70) Subject restriction on possessor raising



Ravinski (2005) attributes this subject restriction on possessor raising to the Minimal Link Condition (Chomsky 1995). According to this condition, shorter moves are preferred over longer ones. As the highest of the two arguments, the subject has the shortest move to Spec, DP. Thus, possessor raising from objects is predicted to be ruled out.

Focusing now on affixal predicates, we find that the subject restriction on possessor raising distinguishes between the two arguments of these transitives. The

sentence below is an instance of possessor raising with the transitive affixal predicate 2u-yu2aat "find". As in the previous example, -uk (POSS) is suffixed to the predicate and it is the first person possessor which determines subject agreement.

(71)	?uyu?aa†uksiš	hupkumł	Sinii <del>λ</del>			
	?u- <u>u?aa1</u> -uk-siiš	hupk-um1	Sinii⊁			
	Ø- <u>find</u> -poss-1sg.ind	ball-RD	dog			
	= (i) My dog found the ball.					
	$\neq$ (ii) The dog found my ball.					

In this sentence, the only argument which may be interpreted as the possessum is  $Sinii\lambda$  "dog"; hupkumt "ball" is not interpretable as the possessum. This diagnoses  $Sinii\lambda$  "dog" as the subject of the predicate 2u-yu2aat, since only subjects in Nuuchah-nulth may act as the source of possessor raising.

Note that this restriction on interpretation holds regardless of the relative word orders of subject and object. (Recall from Chapter 3 that surface word order of arguments in Nuu-chah-nulth is often flexible.) In (72), the arguments are in reverse word order relative to (71).

(72)	?uyu?aa†uksiš	Sinii≁	hupkumł			
	?u- <b>u?aa1</b> -uk-siiš	Sinii⊁	hupk-um1			
	Ø- <u>find</u> -poss-1sg.ind	dog	ball-RD			
	= (i) My dog found the ball.					
	$\neq$ (ii) The dog found my ball.					

Here again, the only argument which is interpreted as the possessum is Sinii "dog".

Also note that the restriction on interpretation in cases of possessor raising holds in spite of the fact that in cases with no possessor raising, either the subject or the object of the affixal predicate is eligible as a possessum. The examples below are instances of unraised possessives with the same predicate,  $\lambda u$ -yulaat "find".

(73)	a.	?uyu?aa†?iš ?u- <u>u?aa†</u> -?iiš ∅- <u>find</u> -3.IND My dog found th	š ʕiniiλukqs iiš ʕiniiλ-uk-qs ND dog-POSS-1SG.PS und the ball.		hupkumł hupk-umł ball-RD	(unraised)
	b.	?uyu?aa+?iš ?u- <b>u?aa+</b> -?iiš Ø- <u>find</u> -3.IND The dog found m	Sinii⊁ Sinii⊁ dog ny ball.	hupkum hupk-un ball-RD-	tukqs nt-uk-qs POSS-1SG.PS	(unraised)

In (73a), the subject *Sinii* $\mathcal{X}$  is a possessum, and in (73b) it is the object *hupkum* $\mathcal{A}$  which is a possessum. The possessive marker -uk is suffixed to the possessum in each case.

The results of this possessor raising diagnostic can be used as support of the analysis that complements of a transitive affixal predicate are the source of
incorporation. What we anticipate for a transitive affixal predicate is that the argument which tests as a non-subject by the possessor raising diagnostic should be the same argument which permits incorporation. This predicted behaviour is shown to occur in (74).

(74)	a.	?uyu?aa†uksiš ?u- <u>u<b>?aa†</b></u> -uk-siiš Ø- <u>find</u> -pOSS-1SG.IND My dog found a bone.	Sinii⊁ Sinii⊁ dog	hamuut hamuut bone
	b.	hamuutu?aałuksiš hamuut- <b>u?aał</b> -uk-siiš bone- <u>find</u> -POSS-1SG.IND My dog found a bone.	Sinii⊁ Sinii⊁ dog	

Both examples illustrate possessor raising. In (74a), -u2aat "find" suffixes to the expletive  $\lambda t$ -, while in (74b), incorporation of the nominal *hamuut* "bone" occurs. In both cases, the argument *hamuut* "bone" tests as a non-subject since it fails to receive a possessum interpretation under possessor raising. Since complements of a transitive predicate are not subjects, this behaviour is predicted. Thus, the results of the possessor raising diagnostic coincide with the proposed complement effect on incorporation.

## 4.4 Extended Unaccusatives

This section examines the properties of locative affixal predicates in Nuu-chahnulth, which I analyse as having the syntax of extended unaccusatives. As first discussed in §4.1.3, these predicates project two internal arguments, realized in complement and specifier position, respectively.

Locative suffixes are abundant in the Wakashan languages (Sapir and Swadesh 1939; Boas 1947; Anderson 1985). In Nuu-chah-nulth, locative affixal predicates account for a substantial percentage of the affixal predicate inventory. For the Kyuquot dialect of Nuu-chah-nulth, Rose (1981: 293) estimates that there are 127 locative suffixes out of a set of 406 non-inflectional affixes in the language, amounting to approximately 31% of the total set. Locative affixal predicates denote a range of spatial relationships in Nuu-chah-nulth, including reference to physical locations (e.g.,  $\lambda u$ - $\lambda is$  "on the beach",  $\lambda uu$ -tsit "on the surface of a liquid") and body parts (e.g.,  $\lambda u$ -wik "on the head",  $\lambda u\lambda$ -qhta "on the foot/feet"), as well as more abstract relationships (e.g.,  $\lambda u$ -k wi "in",  $\lambda u$ -k čaas "beside") (Davidson 2002). Representative examples are given below.

(75)	a.	?u?is?iš	?aya	muks?i	ḥaa≁	ča?ak?i	
		?u- <u><b>'is</b></u> -?iiš	?aya	muks?i	ḥaa†	ča?ak-?ii	
		Ø- <u>on.beach</u> -3.ind	many	rocks	DEIC	island-DET	
		There's lots of rocks on the beach of that island.					
		(lit: "That island beaches/contains a lot of rocks")					

b.	?uutsit?iš	ćišxmis	•	niisyak?i
	?uu- <u>tsit</u> -?iiš	ćišx-mi	S	niisýak-?ii
	Ø- <u>in.water</u> -3.IN	D dirt-NO	М	pot-DET
	There's dirt (in t	he water) in the po	ot.	
	(lit: "The pot wa	ters/contains dirt"	')	
c.	?uuwik?iš	ċiisiicum?i		
	?u- <b>wik</b> [+L]-?iiš	čiisiicum-?ii		
	Ø-on.head-3.IN	D headband-DET		
	S/he's wearing a	headband.		
	( <i>lit</i> : "She's head	ing a headband")		
	(			
d.	?u?uqḥtinu?i		šuwisuk	x?itk
	?u-qhta[+R]-inu?	≂'ii	šuwis-u	k-?iitk
	Ø-on.feet-PERF-	2sg.imp>3.obj	shoes-P	OSS-2SG.PS
	Put your shoes of	on!		
	(lit: "Feet your s	hoes!")		
	<b>℃-1-w</b> : <b>℃</b> :≚			
e.	ruk"1ris	qaruuc	yama	
	ru- <u>ci</u> -riis	qaruuc	yama	
	Ø- <u>in</u> -3.IND	burden.basket	salal.be	rries
	The salal berries	are in the basket.		
f.	?ukčaasuksiš		maḥťii	saantiquwas
	?u- <u>čaas</u> -uk-siiš		maḥťii	saanti-quwas
	Ø- <u>beside</u> -poss-	1sg.ind	house	Sunday-building
	My house is bes	ide a church.		

Since the first study of Southern Wakashan languages in the early 20th century, researchers have observed that locative suffixes show contrasts in the type of relationship that holds between the suffix and its host (Sapir and Swadesh 1939; Swadesh 1939). Davidson (2002: 180–181) aptly notes that the locative suffixes  $-\dot{c}uu$  "contain" and  $-\dot{c}i$  "in" show opposite patterns with respect to the nominal they suffix to. In the examples below, the locative suffix 2u- $\dot{c}uu$  "contain" cannot suffix to the nominal *qa*2uuc "burden basket" (76b), while the 2u-k<sup>w</sup>i</sup> "in" does have the ability to do so (77b).

(76)	a.		ha?umću?iš	qa?uuc?i
			ha?um- <u><b>čuu</b></u> -?iiš	qa?uuc-?ii
			food- <u>contain</u> -3.IND	burden.basket-DET
			There's food in the burd	len basket.
			(lit: "The burden basket	contains food")
			(cf. Davidson 2002: 181	, ex. 277)
	b.	*	qa?uuccu?iš	ha?um
			qa?uuc- <u>čuu</u> -?iiš	ha?um
			burden.basket-contain-	3.IND food
			There's food in a burder	n basket.

(77)	a.	*	yamači?iš	qa?uuc
			ýama- <u>či</u> -?iiš	qa?uuc
			salal.berries- <u>in</u> -3.IND	burden.basket
			There's salal berries in a	burden basket.
	b.		qa?uucči?iš	ýama?i
			qa?uuc- <u>či</u> -?iiš	yama-?ii
			burden.basket- <u>in</u> -3.IND	salal.berries-DET
			The salal berries are in a	burden basket.
			(cf. Davidson 2002: 181,	ex. 276)

The claim that I develop in this section is that the suffixation patterns of locative suffixes in Nuu-chah-nulth fall out from conditioning effects of their argument structure. Under my analysis, the predicate  $\lambda u$ - $\dot{c}uu$  "contain" is classified as a *locatum predicate*, while  $\lambda u$ - $k^w i$  "in" is classified as *location predicate*. This terminology is borrowed from treatments of English denominal "location" and "locatum" verbs (Clark and Clark 1979; Hale and Keyser 2002). As Clark and Clark (1979) describe, English locatum verbs (such as *clothe* or *saddle*) are derivationally related to nominals which specify an object which is located (the "locatum"). In (78a), the locatum verb *saddle* references the object (the saddle) which is placed onto the horse. On the other hand, location verbs in English (such as *shelve* or *bottle*) are derived from nouns that specify the location of the object, and not the located object itself. In (78b), the location verb *shelve* references the position (the shelf) where the books end up.

(78) a. She <u>saddled</u> the horse. (locatum verb; Hale and Keyser 2002: 19, ex. 35b)
b. I <u>shelved</u> the books. (location verb; Hale and Keyser 2002: 19, ex. 35a)

Hale and Keyser (2002) present an analysis of English locatum and location verbs in which the two locative predicate types show inverse argument structures. Locatum predicates have a locatum argument in complement position of a covert preposition, and a location argument in specifier position of the preposition. Location predicates, in contrast, have a location argument in their complement, and locatum in their specifier.



b. Location predicate



(79)

In Hale and Keyser's (2002) analysis, the predicative function of these denominals is represented by a synthetic structure composed of stacked prepositional and verbal predicates headed by covert elements. The nominal head in each configuration ("saddle", "shelf") comes to be realized as a verb via a process of conflation which ties together the nominal with the verbal head.

Following Hale and Keyser's (2002) analysis of English locative denominals, I propose that the two locative classes in Nuu-chah-nulth differ in that locatum predicates such as  $\lambda u$ - $\dot{c}uu$  "contain" take a locatum argument as their complement, while location predicates such as  $\lambda u$ - $k^{wj}$  "in" take a location argument as their complement. Assuming that the second argument of these dyadic verbs is introduced in a right-branching specifier position in Nuu-chah-nulth, this yields the following syntactic representation:



This analysis of Nuu-chah-nulth locatives differs from Hale and Keyser's (2002) treatment of English denominal locatives with respect to the process of conflation. As noted earlier in the chapter, conflation copies the "p-signature" of a complement onto a defective head. Conflation accounts for the "denominal" characteristics of the English locatives. In Hale and Keyser's analysis of English locatives, the lexical head is introduced as a nominal complement of a phonologically "defective" (i.e., covert) preposition. This prepositional phrase is embedded as the complement of a covert verb. By conflation (represented by an arrow), the phonological "signature" of the nominal comes to be associated with a phonologically defective ( $\emptyset$ ) prepositional head, and in turn, a defective verbal head ( $\emptyset$ ). The "V" thus takes on the phonological characteristics of the N, *saddle*.

#### (81) Denominal-forming conflation



In this way, the process of conflation derives the verbal behaviour of the nominal head.

In Nuu-chah-nulth, a different set of empirical facts holds. Unlike the English denominal predicates ("shelve", "clothe") which show an overt correspondence to nominals ("shelf", "cloth(es)"), there is no evidence for a nominal alternation with Nuu-chah-nulth locatives. In other words, Nuu-chah-nulth locative predicates are not related to any free-standing nominals. Accordingly, I represent the lexical material of the Nuu-chah-nulth locative predicates as the predicate heads themselves, rather than as nominal heads embedded in additional structure. That is, I propose that no abstract conflation occurs in Nuu-chah-nulth to derive the predicates  $\lambda u$ -cuu "contain" and  $\lambda u$ -k<sup>w</sup>i "in" directly occupy a verbal head position, V. There is no mediating prepositional component to the predication. Thus, I adopt a direct predication analysis.



A further difference between Nuu-chah-nulth locatives and English locative denominals is that Nuu-chah-nulth locatives lack agentive force. In the Hale and Keyser analysis, the agentivity of English denominals is represented in the synthetic predicate structure by a tacit verbal element equivalent to "put". No such covert verbal head is warranted in the Nuu-chah-nulth cases, which lack agentivity. In the vP shell framework which I have adopted, the absence of agentive force of Nuu-chah-nulth locatives may be denoted by the absence of a vP projection. This allows for a contrast between stative locatives in Nuu-chah-nulth, and agentive locatives formed by overt causativization. The following example expresses a complex predicate equivalent to English "put", derived by causativization of the location predicate  $\lambda u - k^w i$  "in".

(83)	?uk <sup>w</sup> inup?aq⊁siš	καhiqs?i
	?u- <u>či</u> -nup-?aq <del>λ</del> -siiš	⊀ahiqs-?ii
	Ø- <u>in</u> -caus-fut-1sg.ind	box-DET
	I'll put it in the box.	

I assume that agentive locatives such as that in (83) have a *v*P projection, but stative locatives do not. This is illustrated in the following diagrams.



We now to turn to evidence which supports the syntactic structures of locatives proposed in (80). The diagnostics which I first introduced for transitives in §4.3 (subject agreement and possessor raising) will be shown to support an analysis in which locative predicates in Nuu-chah-nulth belong to two distinct classes which have inverse argument structures. In §4.4.1, I consider locatum predicates, and in §4.4.2, I turn to location predicates.

## 4.4.1 Locatum Predicates

This section presents diagnotics for the syntactic structure of locatum predicates such as 2u-cuu "contain" or 2u-hahut-" "on front".

(85)	a.	?ucuu?iš	č'a?ak						
		?u- <b>ćuu</b> -?iiš	č'a?ak						
		Ø- <u>contain</u> -3.ind	water						
		There's water in the	his.						
		(lit: "This contain	s water")	)					
	b.	č'a?akċu?iš							
		č'a?ak- <u>čuu</u> -?iiš							
		water-contain-3.II	water-contain-3.IND						
		There's water in the	his.						
		(lit: "This contain	s water")	)					
(86)	a.	?uḥaḥu <del>1</del> ?iš	Simtii?a	k?i		ḥaa	łuucma?i		
		?u- <b>ḥaḥu႑</b> -?iiš	Simtii-?	ak-?i		ḥaa	⁺uucma-?ii		
		Ø- <u>on.front</u> -3.ind	name-P	OSS-3.PS	5	DEIC	woman-DET		
		That woman's got her name written on her front.							
		( <i>lit</i> : "That woman is fronting her name")							
	b.	Simtiiḥaḥu <del>1</del> ?iš		ḥaa	łuucma	?i			
		Simtii- <b>ḥaḥu北</b> -?iiš		ḥaa	łuucma	-?ii			
		name- <u>on.front</u> -3.1	IND	DEIC	woman-	DET			
		That woman's got	a name	written o	on her fro	nt.			
					••				

(*lit*: "That woman is fronting her name")

I analyse these predicates as taking a locatum argument as a complement, and a location as a specifier.

### 4.4.1.1 Incorporation

According to the PF incorporation hypothesis, affixal predicates in Nuu-chah-nulth incorporate a host chosen from their derivational sister. Assuming the structure in (87), we predict that a locatum predicate such as  $\lambda u$ - $\dot{c}uu$  "contain" should be oriented at spell-out with respect to its locatum argument, and not to its location argument. It is the locatum argument of a locatum predicate which is a derivational sister of the affixal predicate.

(87)



This prediction holds, as indicated in the sentences in (88), repeated from (76). In (88a), the affixal predicate 2u-cuu "contain" suffixes to the locatum argument, *ha?um* "food". As shown in (88b), it is ungrammatical for this predicate to suffix to the location argument, *qa?uuc* "burden basket".

(88)	a.	ha?umċu?iš	qa?uuc?i
		ha?um- <u>čuu</u> -?iiš	qa?uuc-?ii
		food- <u>contain</u> -3.IND	burden.basket-DET
		There's food in the burde	en basket.
		(lit: "The burden basket	contains food")
	b. *	qa?uuccu?iš	ha?um
		qa?uuc- <u>čuu</u> -?iiš	ha?um
		burden.basket-contain-3	.IND food

There's food in a burden basket.

(89)	a.		∻i?ičum†uxs?iš	łuucma
			ℋiʔičum⁺- <u>uxs</u> -ʔiiš	łuucma
			straw.hat- <u>on.head</u> -3.IND	woman
			A woman is wearing a str	aw hat.
			(lit: "A woman is heading	a straw hat")
	b.	*	łučuxs?iš	<i>i</i> ?ičum†
			tuč- <u>uxs</u> -?iiš	λi?ičum↓
			woman- <u>on.head</u> -3.IND	straw.hat
			A woman is wearing a str	aw hat.

- (90) a. čiišxsit?iš ?uh niisýak?i huqsaapči čišx-sit[+L]-?iiš ?uh niisýak-?ii huq-saap-čii dirt-on.surface.of.liquid-3.IND DEIC pot-DET spill-CAUS-2SG.GO There's dirt in the pot. Go dump it out! (*lit*: "The pot surfaces/contains dirt. Go dump it out!")
  - b. \* niisýaksit?iš ćišxmis niisýak-<u>sit</u>-?iiš ćišxmis pot-<u>in.water</u>-3.IND dirt There's dirt in a pot.

The incorporation asymmetry derives from the properties of pairwise spell-out to PF. An affixal predicate finds a host from its derivational sister. As the derivational sister of a locatum predicate, the locatum argument acts as a host for the affix. Location arguments are ineligible as a host because they are not derivational sisters to the affixal predicate.

The incorporation asymmetry between the arguments of a locatum predicate extends predictably to the formation of *wh*-questions and relative clauses. For locatum verbs such as  $\lambda u - uq\lambda$  "inside" or  $\lambda u$ -*cuu* "in a container", a *wh*-word which corresponds to the locatum argument incorporates into the affixal predicate. In (91a), the locatum  $2aq\dot{r}$  "what" incorporates into the affixal predicate 2u- $uq\lambda$  "inside". In (91b), the same locatum incorporates into the affixal predicate  $\lambda u$ -*cuu* "in a container".

(91)	a.	?aqiq⊁ḥ ?aqi- <b>aqૠ</b> -ḥ what- <u>inside</u> -3.Q What's in the oven?	č'aṁaq <del>λ</del> ýak?i č'aṁaq <del>λ</del> ýak-?ii oven-DET
	b.	?aqicuḥ ?aqi- <b>cuu</b> -ḥ	?аḥkuu ?аḥkuu
		what- <u>contain</u> -3.Q What's in this?	DEIC

(*lit*: "What does this contain?")

Incorporation of a *wh*-location is disallowed by a locatum verb. As shown in (92a), it is impossible for the location *waayaq* "which" to incoporate into the affixal predicate *?u-cuu* "in a container". Instead, the location must remain unsuffixed to the locatum predicate, as in (92b).

(92)	a.	*	waayaqcuh	ha?um?akqs
			waayaq- <u><b>čuu</b></u> -ḥ	ha?um-?ak-qs
			which-in.container-3.Q	food-POSS-1SG.PS
			Which one is my food in?	

 b. waayaqh ?ucuu ha?um?akqs waayaq-h ?u-<u>cuu</u> ha?um?ak-qs which-3.Q Ø-<u>in.container</u> food-POSS-1SG.PS Which one is my food in? (*lit*: "Which contains my food?")

A parallel pattern is found with relative clauses formed with locatum predicates. A relative pronoun (yaq) which corresponds to the locatum argument incorporates into the affixal predicate 2u-kuxs "on the head" in (93a). As shown in (93b), a relative pronoun corresponding to the location argument does not incorporate into the locatum predicate.

(93)	a.	λułčuu?ak?iš λuł-čuu-?ak-?iiš	<sup>2</sup> <i>k</i> i?ičum <sup>1</sup> <sup>2</sup> <i>k</i> i?ičum <sup>1</sup>	yaquxs?itq yaq- <u>uxs</u> -?iitq	ḥaa ḥaa	łuucma?i łuucma-?ii
		The staw hat that lady	is wearing	is very nice.	locatun	<i>n-relative</i> )
	b.	ýuk <sup>w</sup> iiqsaksiš ýuk <sup>w</sup> iiqsu-?ak-siiš y.sibling-POSS-1SG.INI The woman who is we ( <i>location-relative</i> )	tuucmai tuucma- woman- aring a neo	li yaq?iitq -?ii yaq-?iitq DET REL-3.RL cklace is my your	nuutin <del>1</del> nuut- <u>wi</u> necklac nger sibl	i <b>nł</b> ce- <u>on.neck</u> ling.

## 4.4.1.2 Subject Agreement

The subject agreement for a locatum predicate corroborates an analysis in which the location argument occurs higher than the locatum. As the highest argument, the location is predicted to trigger the subject agreement in AgrP.

(94)



This prediction holds: for locatum predicates, the person agreement corresponds to the location argument, not the locatum argument. A first person location argument is registered by the first person indicative inflection -siis, as illustrated below.

(95)	a.	?ucuťumsiš	sačkahs
		?u– <u><b>cuťum</b></u> –siiš	sačkaņs
		Ø- <u>on.side.of.head</u> -1SG.IND	comb
		I've got a comb on the side of n	ny head.

- b. ?uuwintsiš nuutinum ?u?iihas puumat?iha ?uu-**wint**-siiš nuutinum ?u?iiha-s puumat?iha  $\varnothing$ -**on.neck**-1SG.IND necklace  $\varnothing$ -due.to-1SG.ABS itchy-feeling Because I'm wearing a necklace I'm itchy.
- c. Xiîičumłuxssiš
   Xiîičumł-<u>uxs</u>-siiš
   straw.hat-<u>on.head</u>-1SG.IND
   I'm wearing a straw hat.

Here, the locatum argument is not registered by subject agreement.

## 4.4.1.3 Possessor Raising

In contexts with no possessor raising, either argument of a locatum predicate may be marked with possessive morphology or receive an interpretation in which the possessive-marked nominal is a possessum. In (96a), the location argument  $\dot{cupcupsumf}$  "sweater" is possessive-marked as a possessum, while in (96b) it is the locatum *?imtii* "name" which acts as a possessum. (There is no restriction here as to whether the possessor *Lucy* must precede or follow the possessum: following Ravinski (2005), I assume this is a case of "scrambling".)

(96)	a.	?u?a†?iš	Simtii	Lucy	čuj	pč'upšum†uk?i
		?u- <u><b>?a1</b></u> -?iiš	Simtii	Lucy	č'u	pč'upšum <sup>1</sup> -uk-?i
		Ø- <u>on.flat.surface</u> -3.ind	name	Lucy	SW	eater-POSS-3.PS
		There is a name is on Lu	icy's sweat	ter. (poss	essu	m = location)
	b.	?u?a+?iš	Simtii?ak?i	i Lu	ıcy	č'upč'upšum†
		?u- <u><b>?a1</b></u> -?iiš	Simtii-?ak-	-?i Lu	icy	čupčupšum1
		Ø- <u>on.flat.surface</u> -3.iND	name-POS	s-3.ps Lu	ıcy	sweater
		Lucy's name is on a swe	eater.	(possess	sum	= locatum)

In possessor raising contexts, however, there is an asymmetry between the two arguments of the locatum predicate. Only the location argument of a locatum predicate may act as the source of possessor raising. This is reflected in the possessor raising sentences below, in which a possessive marker suffixes to the locatum predicate  $\lambda - 2a t$  "on a flat surface". Here, the location  $\dot{c} u \rho \dot{c} u \rho \dot{s} u m t$  "sweater" is obligatorily interpreted as the possessum. The sentence in (97a) shows the pattern with the locatum predicate suffixed to  $\lambda - \lambda d c$ , and (97b) shows the same restriction on interpretation when the affixal predicate has suffixed to the locatum argument.

(97)	a.	?u?a†uk?iš	Simtii	Lucy	č'upč'upšum†
		?u- <u><b>?a1</b></u> -uk-?iiš	Simtii	Lucy	čupčupšum1
		Ø- <u>on.flat.surface</u> -POSS-3.IND	name	Lucy	sweater
		= (i) There is a name on Lucy's s	weater.	(possess	um = location)
		$\neq$ (ii) Lucy's name is on a sweate	r.	(possess	$um \neq locatum)$

b.	Simtiqa†uk?iš	Lucy	čupčupšum†
	Simtii- <b>?<u>a1</u>-</b> uk-?iiš	Lucy	čupčupšum†
	name- <u>on.flat.surface</u> -POSS-3.IND	Lucy	sweater
	= (i) There is a name on Lucy's sweater.	(possess	sum = location)
	$\neq$ (ii) Lucy's name is on a sweater.	(possess	$um \neq locatum)$

In (97), an interpretation of "*Lucy's name is on a sweater*" is unavailable. Thus, the locatum (*Simtii* "name") proves to be ineligible as the source of possessor raising.

It is predicted by the analysis that only a location argument of a locatum predicate should act as a source of possessor raising. Possessor raising in Nuu-chahnulth is possible only out of subjects, and never objects. As the higher argument of a locatum predicate, the location takes on the role of a subject, and with this, the ability to serve as the source of possessor raising.

In the next section, we turn to the discussion of location predicates, which I argue have an orientation of arguments which is the inverse of that of locatum predicates.

### 4.4.2 Location Predicates

This section presents diagnotics for the syntactic structure of location predicates such as  $\lambda - k^{w_i}$  "in" or  $\lambda - k \dot{c} aas$  "beside". Examples of each of these affixal predicates are given below.

(98)	a.	?uk <sup>w</sup> i?iš ?u- <u>či</u> -?iiš Ø- <u>in</u> -3.IND The water is in tl	ča?ak ča?ak water he bucket	čax <sup>w</sup> ac čax <sup>w</sup> ac bucket	
	b.	č'ax <sup>w</sup> acči?iš č'ax <sup>w</sup> ac- <u>či</u> -?iiš bucket- <u>in</u> -3.IND The water is in tl	ča?ak ča?ak water he bucket		
(99)	a.	?ukčaas?iš ?u- <u>čaas</u> -?iiš Ø- <u>beside</u> -3.ıND Kay's sitting bes	ide a man	Kay Kay Kay L.	čakup čakup man
	b.	čapxčaas?iš čapx- <u>čaas</u> -?iiš man- <u>beside</u> -3.IN Kay's sitting bes:	D ide a man	Kay Kay Kay	

I analyse these predicates as taking a location argument as a complement, and a locatum as a specifier.

### 4.4.2.1 Incorporation

Only the location argument of a location predicate may incorporate; locata may not. This is illustrated in (100) for the predicate  $2u-k^{w_i}$  "in". In (100a), the predicate suffixes to the location  $\dot{cax}^{wac}$  "bucket". It is not possible for the locatum,  $\dot{ca2ak}$  "water", to serve as a host for the affixal location predicate, as indicated in (100b).

(100)	a.		č'ax <sup>w</sup> acči?iš	č'a?ak	
			č'ax <sup>w</sup> ac- <b>či</b> -?iiš	č'a?ak	
			bucket- <u>in</u> -3.IND	water	
			The water is in the	ne bucket	•
	b.	*	ča?akči?iš		č'ax <sup>w</sup> ac
			č'a?ak- <u>či</u> -?iiš		čax <sup>w</sup> ac
			water- <u>in</u> -3.IND		bucket
			The water is in the	ne bucket	

This incorporation asymmetry is readily observed in *wh*-questions, in which an incorporated *wh*-word references only a location, and not the locatum. In the following example, 2aqi "what" incorporates in the location predicate  $2u-k^{w}i$  "in". The *wh*-word corresponds only to the location argument, while the argument *qa2uuc* "basket" refers to the locatum.

(101)	?aqiči <u>ḥ</u>	qa?uuc	
	?aqi– <b>či</b> –ḥ	qa?uuc	
	what- <u>in</u> -3.Q	burden.basket	
	= (i) What's th	e burden basket in?	(wh = location)
	$\neq$ (ii) What's in	n the burden basket?	$(wh \neq locatum)$

When the locatum argument of a location predicate is *wh*-questioned, the locatum is not incorporated. Compare the locatum-question in (102a) to the location-question in (102b): only in the latter case is the *wh*-word *waayaq* "which" incorporated into the location predicate  $\lambda u$ -k<sup>w</sup>i "in".

(102)	a.	waayaqh	qa?uucči	
		waayaq-ḥ	qa?uuc- <u>či</u>	
		which-3.Q	burden.basket-in	
		Which one is in	the burden basket?	(wh = locatum)
	b.	waayaqčiḥ	ha?um?akqs	
		waayaq- <u>či</u> -ḥ	ha?um-?ak-qs	
		which-in-3.Q	food-POSS-1SG.PS	
		Which one has	my food in it?	(wh = location)

Given the proposed analysis of the syntactic orientation of arguments of a location predicate, this incorporation asymmetry follows. A location argument is predicted to act as the host for an affixal location predicate because this argument is the derivational sister of the predicate. This is illustrated in the diagram in (103), which represents the argument structure of the sentence *qa?uucči?iš yama?i* "The salal berries are in a burden basket" (from 77b). The location argument *qa?uuc* "burden basket" acts as host for the affixal predicate at local spell-out.



Spell-out induces a linearization of qa?uuc-či "in a burden basket".

## 4.4.2.2 Subject Agreement

For location predicates, subject agreement matches the locatum argument. This is illustrated in the question–answer pair in (104), in which the respondent specifies his/her location with the location predicate  $\partial u - k^{w_i}$  "in". The predicate is inflected with the first person subject agreement -siiš (1SG.IND), corresponding to the locatum argument of the predicate.

(104)	a.	waasik waasi-k where-2SG.Q Where are you?	
	b.	?uk <sup>w</sup> isiš ?u- <u>či</u> -siiš ∅- <u>in</u> -1sg.ınd	cucsac cucsac tub

I'm in the tub.

This pattern of subject agreement follows if the locatum is the highest argument of the location predicate. In the diagram below, the locatum argument establishes a "minimal link" to Agr.

(105) AgrP Agr VP locatum

### 4.4.2.3 Possessor Raising

For location predicates, locata act as the source of possessor raising. In the examples below, the possessum corresponds to the locatum, and not to the location. In (106), the locatum *halum* "food" is treated as the possessum; in (107), the locatum *mahtii* "house" is the possessum.

(106)	waayaqčakḥs	ha?um		
	waayaq- <b>či</b> -?ak-ḥs	ha?um		
	which- <u>in</u> -POSS-1SG.Q	food		
	= (i) Which one has my f	(possessum = locatum)		
	$\neq$ (ii) Which of mine has	food in it?	?	$(possessum \neq location)$
(107)	saantiquwasčasuksiš		maḥťii	
	saantiquwas- <u>čas</u> -uk-siiš		maḥťii	
	church-beside-POSS-1SG.	IND	house	
	= (i) My house is beside a church.			(possessum = locatum)
	$\neq$ (ii) The house is beside	(possessum $\neq$ location)		

This restriction on possessor raising holds in spite of the fact that either a locatum or a location may act as a possessum in contexts with no possessor raising. The example in (108a) is a case of possessor raising in which only the locatum yana "salal berries" may be interpreted as the possessum. The example in (108b) is an unraised example showing yana "salal berries" as a possessive-marked possessum. The example in (108c) shows that there is no restriction on the location qaluuc "burden basket" acting as possessive morpheme -uk (POSS), and no possessive marking appears on the predicate.

(108)	a.	qa?uucčaksiš		ýaṁa	
		qa?uuc- <u>či</u> -?ak	r−siiš	yama	
		basket- <u>in</u> -POS	ss-1sg.ind	salal.berries	
		= (i) My sala	l berries are in a b	urden basket.	(possessum = locatum)
		$\neq$ (ii) The sal	al berries are in m	y burden bask	et. (possessum $\neq$ location)
	b.	?uk <sup>w</sup> i?iš	qa?uuc	vama?akqs	
		?u- <b>či</b> -?iiš	qa?uuc	yama-?ak-qs	3
		Ø- <b>in</b> -3.ind	burden.basket	salal-POSS-1	SG.PS
		My salal ber	ries are in a burder	n basket.	(possessum = locatum)
	c.	?uk <sup>w</sup> i?iš	qa?uucukqs		vama
		?u- <b>či</b> -?iiš	qa?uuc-uk-qs		yama
		Ø- <b>in</b> -3.ind	burden.basket-P	OSS-1SG.PS	salal
		The salal ber	ries are in my bur	den basket.	(possessum = location)

Thus, it is only in possessor raising contexts that the location argument of a location predicate is barred from acting as a possessum. This pattern supports an analysis in

which the locatum argument is the subject of a location predicate, and not the object. Recall that possessor-extraction is possible only out of subjects in Nuu-chah-nulth.

In the next section, we turn to ditransitive affixal predicates, which I analyse as transitivized extended unaccusatives.

## 4.5 Ditransitives

There is evidence for two distinct classes of ditransitive affixal predicates in Nuuchah-nulth. The first class, exemplified by 2u-yii "give", suffixes to a theme argument. In (109a), the affixal predicate 2u-yii "give" incorporates the theme *taanaq*- "money". The second type, exemplified by 2u-2iip "give to", suffixes to a goal argument. This pattern is illustrated in (109b), in which the affixal predicate 2u-2iip "give to" incorporates sut- (2SG).

(109)	a.	taanaqayi?i taanaq- <b>yii</b> -'ii money- <b>give</b> -2SG.IMP>3.OBJ Give mom money!	?um?i ?um?i mother
	b.	sut?iimitsiš sut- <b>?iip</b> -mit-siiš 2SG-g <b>ive.to</b> -PST-1SG.IND I gave you money.	taana taana money

According to the analysis I proposed in §4.1.2, ditransitives are treated as extended unaccusatives which are abstractly transitivized when they are embedded within a vP "layer". I further propose that the two classes of ditransitives in Nuu-chah-nulth are distinguished according to the orientation of the arguments which belong to the extended unaccusative layer. Predicates such as 2u-yii "give" belong to a class I label "locatum-type" ditransitives, while predicates such as 2u-2iip "give to" belong to a "location-type" class. The incorporation patterns of each of these types of ditransitives are illustrated in the following sections.

## 4.5.1 Locatum-type Ditransitives

This section considers the characteristics of locatum-type ditransitives, which I analyse as having a configuration of arguments as in (110).



According to the analysis, these predicates take a theme in their complement position, while a goal argument occupies specifier position of the embedded extended unaccusative relation. Predicates belonging to this class are *?u-yii* "give", *?uu-kš* "ask for" and *?uu-pwin* "to owe".

(111)	a.	?uyii?i ?u- <b>ayii</b> -'ii Ø- <u>give</u> -2SG.IMP>3.OBJ Give mom money!		?um?i ?um?i mother	taana taana money
	b.	taanaqayi?i taanaq- <b>ayii</b> -'ii money- <u>g<b>ive</b></u> -2sg.IMP>3.01 Give mom money!	3J	?um?i ?um?i mother	
(112)	a.	?uukščii ?u- <u>kš[</u> +L]-čii Ø- <u>ask.for</u> -2SG.GO Go ask mother for water!	?um?i ?um?i mother	č'a?ak č'a?ak water	
	b.	č'aa?akščii č'a?ak- <u>kš[</u> +L]-čii water- <u>ask.for</u> -2SG.GO Go ask mother for water!	?um?i ?um?i mother		
(113)	a.	?uupŵin?ick ?uu- <b>pŵin</b> [+L]-?iick Ø- <u>owe</u> -2sG.IND You owe me money.	siỷa siỷa 1SG.PRO	1	taana taana money
	b.	taanaqapŵin?ick taanaq- <b>pŵin</b> [+L]-?iick money- <u>owe</u> -2SG.IND You owe me money.	siỷa siỷa 1sg.pro	1	

For locatum-type ditransitive, incorporation of a goal is not possible. Only a theme argument may act as a host for the affixal predicate. This is illustrated by the sentences in (114). In (114a), the theme hacaafyak "book" is the host for the affixal predicate hacaafyak "b

(114)	a.	načaa†yakayimit?iš	Robin	čakup?i
		načaa1 yak- <b>ayii</b> -mit-?iiš	Robin	čakup-?ii
		book-give-PST-3.IND	Robin	man-DET
		Robin gave a book to the man.		

b.	*	čapxayimit?iš	Robin	'načaa†ýak
		čapx- <b>ayii</b> -mit-?iiš	Robin	načaa 1 vak
		man-give-PST-3.IND	Robin	book
		Robin gave a man a book.		

This incorporation restriction is reflected in the formation of relative clauses. A relative pronoun corresponding to a goal argument may not incorporate into a locatum-type ditransitive. As shown in (115a), for goal relativizations, the relative pronoun *yaq* "who, which" occurs independently of the affixal predicate  $\lambda u$ -*yii* "give". The ungrammaticality of (115b) demonstrates that the relative pronoun cannot serve as the host for the affixal predicate  $\lambda u$ -*yii* "give" in a goal relativization.

- (115) a. ýuk<sup>w</sup>iiqsaksiš haa yaq?iitq ?uyii taana Robin ýuk<sup>w</sup>iiqsu-?ak-siiš haa yaq?iitq ?u-yii taana Robin younger.sibling-POSS-1SG.IND DEIC REL-3.RL Ø-give money Robin That is my younger sibling, who Robin gave money to.
  - b. \* ýuk<sup>w</sup>iiqsaksiš haa yaqayii?itq Robin taana ýuk<sup>w</sup>iiqsu-?ak-siiš haa yaq-**yii**-?iitq Robin taana younger.sibling-POSS-1SG.IND DEIC REL-<u>give</u>-3.RL Robin money That is my younger sibling, who Robin gave money to.

The incorporation asymmetry for locatum-type ditransitives follows from the complement effect in PF incorporation. If a theme occupies complement position of a locatum-type ditransitive, then it is the derivational sister of the predicate, and is directed to act as the host for an affixal predicate at spell-out. Goal arguments, in contrast, do not have the privilege to form a local spell-out domain directly with the affixal predicate. This is illustrated for *čaa?akščii ?um?i* "Go ask mother for water!", from (112b):



The reflex of this local spell-out is a linearization of *ča?ak-kš*.

## 4.5.2 Location-type Ditransitive

The second type of ditransitives has a configuration of internal arguments which is the inverse of that of locatum-type ditransitives.



By this analysis, the goal argument acts as the complement of the ditransitive, while the theme is in specifier position. The predicate *2u-2iip* "give to" is a location-type ditransitive.

(118)	a.	?u?iimitsiš ?u- <u>?iip</u> -mit-siiš Ø- <u>give.to</u> -PST-1SG.IND I gave you money.		suẁa suẁa you	taana taana money
	b.	sut?iimitsiš sut- <u>?iip</u> -mit-siiš you- <u>give.to</u> -PST-1SG.IND I gave you money.	taana taana money		

Location-type ditransitives incorporate a goal argument. For example, the goals *si*-"me" and *sut*- "you" incorporate into *?u-?iip* "give to" in the following examples.

(119)	a.	si?iip̀is siv́a- <b>?iip</b> -'iis	?ayapẁin?a⊀ats ?aya-pẁin-'a⊁-'at-s		taana taana
		me- <u>give.to</u> -2SG.IMP>1SG Give it to me! He owes n	many-o ne lots of	we-TEMP-PAS-1SG.ABS money.	money
	b.	sut?iimit?iš	Robin	taana	

).	summens	KOUIII	taana
	sut- <u><b>?iip</b>-mit-?iiš</u>	Robin	taana
	you-give-PST-3.IND	Robin	money
	Robin gave you money.		-

In relative clauses formed with 2u-2ip "give to", a relative pronoun (yaq) corresponding to the goal incorporates into the affixal predicate. This is illustrated below.

(120)	ỷuk <sup>∞</sup> iiqsaksiš	ḥaa	yaSiip?itq	Robin	taana
	yuk™iiqsu−?ak-siiš	ḥaa	yaq- <u><b>?iip</b></u> -?iitq	Robin	taana
	y.sibling-POSS-1SG.IND	DEIC	REL- <u>give</u> -3.RL	Robin	money
	That is my younger sibling	ng, who l	Robin gave money	to.	

For this location-type ditransitive, incorporation of a theme is not possible. As shown in (121), it is ungrammatical for the theme *taanaq*- "money" to act as the host for the predicate *Au-Aip* "give to".

(121) \* taana\imit?i\si Robin su\u00fca taanaq-?iip-mit-?ii\si Robin su\u00fca money-give.to-PST-3.IND Robin you Robin gave you money.

The restriction that themes may not serve as the host for a location-type ditransitive follows from the analysis I have presented. PF incorporation is sensitive to the argument structure of the affixal predicate. I have analysed location-type ditransitives as having a goal as a complement. The theme, in contrast, appears as an internal argument in specifier position. At local spell-out, the affixal predicate is linearized with respect to the goal argument, its derivational sister. This spell-out domain excludes the theme, and thus the possibility of the affixal predicate taking the theme as its host. This is illustrated in (122) for the sentence *sut?iimitsiš taana* "I gave you money". Here, the goal *sut*- "you" is the derivational sister of the affixal predicate taking the fully of the affixal predicate taking predicate taking the theme as its host. This is illustrated in (122) for the sentence *sut?iimitsiš taana* "I gave you money". Here, the goal *sut*- "you" is the derivational sister of the affixal predicate taking predicate taking predicate taking the theme and the sentence the affixal predicate taking predicate taking predicate taking the theme as its host. This is illustrated in (122) for the sentence sut?iimitsiš taana "I gave you money". Here, the goal *sut*- "you" is the derivational sister of the affixal predicate taking predicate taking predicate taking the theme and the sentence sut?iinitsis taana "I gave you money".



The reflex of spell-out for (130) is sut-2iip.

## 4.6 Serial Verb Affixation

This section discusses a usage of the affixal predicates discussed in this chapter which is in need of further research. A subset of affixal predicates show the ability to participate in a complex predicate strategy which I will refer to as "serial verb affixation". This predication construction is illustrated in (123), in which the unaccusative affixal predicate 2u-wahsut "go out (of opening)" combines with the verbal host sa "crawl", forming the complex predicate sa-wahsut "crawl out of an opening".

(123)	sawaḥsu†?iš	histaqši⊁	ſaʕi⊁?i		
	sa- <b>waḥsu1</b> -?iiš	histaq-ši⊁	ſaʕi <del>λ</del> -?ii		
	crawl-go.out-3.IND	from-PERF	cave-DET		
	S/he crawled out from the cave.				

These complex predicates disallow "decomposition": the affixal predicate may not be separated from the verbal host. The example in (124a) shows serial verb affixation involving the affixal predicate  $\lambda u$ -*Saa?atu* "move down". In (124b), it is shown to be ungrammatical for  $\lambda u$ - to be used as the host for -*Saa?atu* "move down" when it is combined with the predicate *mat*- "fly".

(124)	a.		matSaa?atumit?iš mat- <u>Saa?atu</u> -mit-?iiš fly- <u>move.down</u> -PST-3.IND	maamaati maamaati bird	
			A bird flew down.	(cf. Davidson 2)	002: 198, ex. 292c)
	b.	*	?uSaa?atumit?iš ?u- <u>Saa?atu</u> -mit-?iiš Ø- <u>move.down</u> -PST-3.IND A bird flew down.	mataa mat-aa fly-CONT	maamaati maamaati bird

This lack of decompositionality stands in contrast to cases in which the affixal predicate takes a nominal complement. As shown in the "noun incorporation" examples in (125), the predicate *- Saa?atu* "move down" can be separated from the nominal host *tatuus* "star" if the expletive host *?u*-appears.

- (125) a. ťaťuusſa?atumit?iš ťaťuus-<u>Saa?atu</u>-mit-?iiš star-<u>move.down</u>-PST-3.IND A star fell.
  - b.  $2usaa^{2}atumit^{2}is$  tatuus  $2u-saa^{2}atu$ -mit-?iiš tatuus  $\emptyset$ -move.down-PST-3.IND star A star fell.

Unaccusative verbs of motion frequently occur in this serial verb construction, as do locatum predicates. The examples in (126) show affixation of  $-iiii \mathcal{X}$  "go inside house" and *-saaatu* "move down". The examples in (127) illustrate affixation of *-citum* "on side of the head" and *-wint* "on the neck".

- (126) Serial verb affixation with unaccusatives
  - a. kamatqii?i⊁?iš Tom kamatq-<u>ii?i⊁</u>-?iiš Tom run-<u>go.inside.house</u>-3.IND Tom Tom ran inside.
  - b. tisaa?atumit?iš matuk ti-<u>saa?atu</u>-mit-?iiš matuk fall-<u>move.down</u>-PST-3.IND plane A plane crashed.

#### (127) Serial verb affixation with locatum predicates

a.	κikciťum⊁siš	Ken
	≁ik− <u>ciťum</u> -u≁-siiš	Ken
	punch- <u>side.of.head</u> -PERF-1SG.IND	Ken
	I punched Ken on the side of the he	ad.
b.	taaq <sup>w</sup> in†?anitsiš	Ken
	taaq- <b>win1</b> -?at-mit-siiš	Ken
	squeeze- <u>on.neck</u> -PAS-PST-1SG.IND	Ken

The morpheme -yaq- often intervenes between an initial predicate and the locative affixes  $-'i \neq "$ in the house", -'as "on ground", -'is "on beach". In such contexts, the initial predicate may be marked for aspect. This pattern is illustrated in (128) with the locative affix  $-'i \neq "$ in the house". The example in (128a) shows serial verb affixation without the use of the morpheme -yaq-. In (128b), the morpheme -yaq-appears and the initial predicate is marked with continuative aspect.

(128)	a.	hapti <del>1</del> ?iš Ken	
		hapt- <u>ʻit</u> -?iiš Ken	
		hide- <u>in.house</u> -3.IND Ken	
		Ken is hiding in the house.	
	b.	haptaayaqi <del>1</del> 7iš	Ken
		hapt-aa-yaq- <b>'it</b> -?iiš	Ken
		hide-CONT-?-in.house-3.IND	Ken
		Ken's been hiding in the house.	

The syntactic constraints on this process of serial verb affixation require additional research. The question should be tackled from two angles, to determine any restrictions on which predicate may act as the initial element in a serial verb, and which affixal predicates may act as the second element. Preliminary research into the latter suggests that the ability of affixal predicates to act as the second element may be tied to the absence of an external argument. Indeed, the distinctive ability of unaccusative and locative affixal predicates to appear as second elements in these serial verbs sets these classes apart under traditional descriptions as "restrictive" suffixes (Sapir and Swadesh 1939; Rose 1981; Davidson 2002), a topic discussed in Chapter 6. Further investigation is needed into the syntactic roles which arguments in these constructions play.

Anderson (1985) discusses similar complex predicates in the Northern Wakashan language Kwakw'ala, which are formed using cognates of Nuu-chahnulth locatives. Anderson argues that in such complex predicates, the Kwakw'ala locative suffixes describe the position of subjects for an initial intransitive predicate, and the position of objects for an initial transitive predicate. For the Kwakw'ala complex predicate  $k^{w}\partial' l \cdot i t$  "lie down in the house", for example, Anderson (1985: 31) argues that the locative suffix -it "(on the floor) in the house" describes the position of the subject of the intransitive predicate  $k^{\mu} i' l'$  "lie down". The restrictions on examples with the cognate affixal predicate -ii + "(on the floor) in the house" in Nuu-chah-nulth are unclear at present, however. In the examples in (129), -ii + "(onthe floor) in the house" references the position of the subject of the initial predicate.<sup>5</sup> In (129a), it is the subject of *SaaqSaaqa* "shout (ITER)" which is described by -ii +"(on the floor) in the house". In (129b), -ii + "(on the floor) in the house" references the position of the subject of *huirqaa* "stare (CONT)".

(129)	a.	SaaqSaaqayaqi <del>1</del> ?iš	ťaňaak
		Saaq−a[+R]−yaq− <u><b>ʻi1</b></u> −?iiš	ťaňa-?ak
		shout-ITER-?- <u>in.house</u> -3.IND	child-POSS
		She is shouting at her child in the l	nouse.
		= (i) mother is inside, shouting at h open window.	ner child outside through an
		$\neq$ (ii) mother is outside, shouting a	t her child inside.
	b.	hu⊁qaayaqi†?iš hu⊁q-aa-yaq- <b>'i†</b> -?iiš	nuŵiiqsak nuŵiiqsu-?ak
		stare-CONT-?- <b>in.house</b> -3.IND	father-POSS
		She's staring at her father inside.	

= (i) she is inside and her dad is outside.

 $\neq$  (ii) she is outside and her dad is inside.

In superficially similar examples, however, a different effect is found. In the examples in (130), -ii "(on the floor) in the house" does not reference the position of the subject of the initial predicate. In (130a), the position of the child who is spitting is irrelevant, so long as the spit ends up on the floor. A similar effect is shown in (130b), in which -ii "(on the floor) in the house" references the aim of the bullets, and not the position of the shooter.

(130)	a.	taaxtaaxayaqi <del>1</del> uk?ick	ťana?is
		taax-a-yaq-ʻ <b>it</b> -uk-?iick	ťana-?is
		spit-ITER-?- <u>in.house</u> -POSS-2SG.IND	child-DIM
		Your child keeps spitting on the floor.	
		= (i) child is on the floor inside	
		= (ii) child is elsewhere	

b. kiikumyiłitwa?iš Ken hiqaačip ?um?iiqsak maḥťii kiikum-'<u>it</u>-mit-wa?iš Ken hiqaa-čip ?um?iiqsu-?ak maḥťii shoot-<u>in.house</u>-PST-3.QUOT Ken wreck-BEN mother-POSS house Ken was shooting onto the floor. He wrecked his mother's house.
= (i) Ken is inside.
= (ii) Ken is elsewhere.

<sup>&</sup>lt;sup>5</sup> The morpheme -yaq- is obligatory in these examples. This matter requires future research.

The restrictions on interpretation may ultimately be clarified by further research on the argument structure of the initial independent predicates involved (e.g., *SaaqSaaqa* "shout (ITER)" and *taaxtaaxa* "spit (ITER)").

## 4.7 Conclusion

In this chapter, the argument structure of affixal predicates was shown to be linked to a restricted set of attested incorporation patterns. I attributed this restricted pattern to a conditioning effect of the syntax at local spell-out, whereby only an argument introduced as a derivational sister to the affixal predicate is able to be spelled-out as the host for the affix. This induces the "complement" effect of PF incorporation in Nuu-chah-nulth. This effect was demonstrated to hold across a variety of affixal predicates, which have a range of argument structures – from unaccusative to transitive, extended unaccusative to ditransitive. For all these predicates, only derivational sisters to the affix may act as hosts.

The local spell-out hypothesis predicts the absence of unergative affixal predicates in the language. According to the analysis, affixal morphemes require linearization with respect to a host at the point of spell-out. Unergative predicates, which lack a pairing of a phonologically contentful verb and a complement, fail to meet the binary requirement for affixation. For affixation, two distinct elements are required at spell-out: a host, and an affix.

# 5. Verbal Complements of Affixal Predicates

Through the unknown, unremembered gate... Is that which was the beginning... ~T.S. Eliot Four Quartets no. 4

## 5.0 Introduction

In Chapter 4, an analysis was presented of the argument structure of affixal predicates which take nominal complements. The suffixation pattern of these affixal predicates often results in "noun incorporation" – although the process is not limited to targeting a single syntactic category. Rather, the choice of host is sensitive to string adjacency to the affixal predicate within its local spell-out domain.

We now turn to the suffixation pattern of affixal predicates which take verbal complements, with particular consideration given to how the syntactic structure of these predicates interacts with restrictions on PF incorporation. As we will see in this chapter, when an affixal predicate which takes a verbal complement suffixes to a host, the outcome may resemble "verb incorporation". In (1), for example, the affixal predicate -*atah* "try to" incorporates the verb *suk*- "reach" as its host.

 susuk<sup>w</sup>ataḥ?iš kiλuuk suk-<u>ataḥ[</u>+R]-?iiš kiλuuk reach-<u>try</u>-3.IND glass S/he's trying to grab a glass.

As in cases with "noun incorporating" affixal predicates, however, the process of "verb incorporation" does not strictly select for syntactic category: non-verbal elements may also serve as the host for these affixal predicate, as first discussed in Chapter 2. By the string adjacency effect, an affixal predicate which takes a verbal complement consistently incorporates whichever element is leftmost in the string adjacent to it at spell-out. When the complement of -mahsa "want to" contains just the verb wat-si $\lambda$  "want to (PERF)", it is this verb which serves as the host for the affix, as in (2a). If a pre-verbal modifier, such as *wityax* "slowly" is present in the complement, however, this modifier takes on the role of host, as in (2b). In contexts with pre-verbal modifiers, incorporation of the verb is ruled out, as shown in (2c). Here, the affixal predicate consistently suffixes to the leftmost element of the complement, regardless of its syntactic structure.

(2) a. wałśiλmaḥsasiś
 wał-šiλ-maḥsa-siiš
 go.home-PERF-want.to-1SG.IND
 I want to go home.

b.		wityaxmaḥsasiš	waa†ši⊁	
		wityax- <u>mahsa</u> -siiš	wa≁-[+L]-ši⊁	
		slow- <u>want.to</u> -1SG.IND	go.home-CONT	-PERF
		I want to go home slowly	·.	
c.	*	waa†ši⊁maḥsasiš		wityax
		wa₁-ši⊁[+L]- <u>maḥsa</u> -siiš		wityax
		go.home-CONT-PERF-wai	nt.to-1sg.ind	slow

I want to go home slowly.

The aim of this chapter is to investigate the syntactic structures which underlie the suffixation pattern of affixal predicates which take verbal complements. By "verbal" complement, I refer to the syntactic frames projected by verbal heads, including any modifiers and functional projections associated with these verbs.

The remainder of this chapter is organized in the following way. First, I argue in §5.1 that the class of "verb incorporating" affixal predicates must be subdivided into categories of main predicates and auxiliary predicates, which show contrastive thematic properties. In §5.2, I present an analysis in which PF incorporation is consistently barred across full clausal (CP) complements of affixal predicates. I relate this prohibition to an opacity effect induced by a saturated domain. In §5.3, I discuss a variety of morphological evidence that the complement in incorporation configurations fails to project up to a CP, and is smaller than a Tense Phrase (TP). Syntactic evidence for a lack of clause-boundedness effects in these constructions is given in §5.4. The chapter concludes with §5.5.

# 5.1 Two Classes of "Verb Incorporating" Affixal Predicates

This section presents evidence that affixal predicates which allow "verb incorporation" are not a homogenous group, and must be divided into categories of main and auxiliary predicates. These classes show distinct behaviours with respect to complementation alternations and a same-subject requirement.

# 5.1.1 Affixal Main Predicates

"Verb-incorporating" affixal predicates vary in their ability to select a fully inflected complement. The class of verbs which I will term *affixal main predicates* alternate between an incorporation strategy, and a strategy in which the affixal predicate takes a full complement which is inflected for subject/mood agreement. In the latter case, the affixal predicate suffixes to the expletive morpheme 2u-, rather than to the embedded verb. The (a) examples below illustrate the incorporation strategy, while the (b) examples show 2u-support in the case of an inflected complement.

(3)	a.	tuuxtuux <sup>w</sup> aSi⊁itsiš	ťaatňa?is
		tux <sup>w</sup> -a[+R]- <u><b>Siλ</b></u> -mit-siiš	ťaatňa-?is
		jump-ITER- <u><b>find</b></u> -PST-1SG.IND	children-DIM
		I came upon the children jumping.	(incorporation)

	b.	?uʕiλitsiš ?u- <u>ʕiħ</u> -mit-siiš Ø- <u>find</u> -PST-1SG.II I came upon the c	?in ?in ND COMP hildren ju	tuuxtu tux <sup>w</sup> -a jump-1 mping.	iux <sup>w</sup> aḥuk I[+R]-ḥuuk ITER-3.DEP		ťaatňa?is ťaatňa-?is children-DIM (?u-support)
(4)	a.	qaqaḥatuɬitsiš qaḥ- <b>atuɬ</b> [+R]-mit- die- <b>dream.of</b> -PST I dreamt that gran	siiš -1SG.IND dparent pa	assed av	naniiq naniiq grandparen way.	t (ir	ncorporation)
	b.	?u?uutu∱itsiš ?u- <u>atu∱[</u> +R]-mit-s: Ø- <u>dream.of</u> -PST- I dreamt that gran	iiš 1sg.ind dparent pa	?in ?in COMP assed av	qaḥšiૠḥuk qaḥ-šiૠ-ḥuu die-PERF-3.1 way.	ık DEP	naniiq naniiq grandparent ( <i>?u-support)</i>
(5)	a.	cuk <sup>w</sup> i⊁cuk?iš cu-k <sup>w</sup> i≁- <u>cuk</u> -?iiš wash-PERF- <u>need</u> -? The potatoes need	3.IND I washing.		qaawic?i qaawic-?ii potato-DET	(in	corporation)
	b.	?ucuk?iš ?u- <b>cuk</b> -?iiš $\emptyset$ - <b>need</b> -3.IND It is best to wash t ( <i>lit</i> : "It is necessar	cuk <sup>w</sup> i?atq cu-k <sup>w</sup> iλ-? wash-PER he potato ry that the	uu at-quu F-PAS-3 es. potatoo	3.COND	qaawio qaawio potato l'')	z-?i >-?ii -DET ( <i>?u-support)</i>

The inflected complement of matrix affixal predicates such as  $\lambda u$ -*Sik* "find" in (3) and  $\lambda u \lambda u$ -*utu* "dream of" in (4) is headed by the complementizer  $\lambda in$  (COMP) and is marked with dependent mood morphology. For the the affixal predicate  $\lambda u$ -*cuk* "need" in (5), there is a conditional complement which is inflected with conditional morphology –*quu* (3.COND).

# 5.1.2 Affixal Auxiliary Predicates

The class of verbs which I label *affixal auxiliary predicates* are incompatible with an inflected complement, and rigidly select only the incorporation strategy. The examples in (a) below illustrate the incorporation strategy, while the (b–c) examples show the impermissibility of the auxiliary taking a fully inflected complement and being "rescued" by  $\lambda t$ -support. Auxiliary predicates may never select an inflected complement, whether it is a dependent clause (6b–8b) or a conditional clause (6c–8c).

(6)	a.		wałaakmaḥsasiš wałaak- <u>maḥsa</u> -siiš go.to- <u>want.to</u> -1SG.IND I want to go to Victoria.	mituuni mituuni Victoria		(incorporation)
	b.	*	?umaḥsasiš ?u- <u>maḥsa</u> -siiš Ø- <u>want.to</u> -1sg.ıND I want to go to Victoria.	?in ?in COMP	wa†aaksa wa†aak-sa go.to-1SG.DEP	mituuni mituuni Victoria ( <i>?u-support)</i>

### THE LINEARIZATION OF AFFIXES

	c.	*	?umaḥsasiš	wataak	quus	mituuni
			ru- <u>maņsa</u> -siiš	wataak-	-quus	mituuni
			Ø- <u>want.to</u> -ISG.IND	go.to-15	G.COND	Victoria
			I want to go to victoria.			(ru-support)
(7)	a.		waa?ičsinhi?iš	John		
			wa?ič- <u>sinhi[</u> +L]-?iiš	John		
			sleep-try.to.stay-3.IND	John		
			John is trying to stay sle	eping.		(incorporation)
	b.	*	?usinḥi?iš	John	?in	wa?ičḥuk
			?u- <u>sinḥi</u> [+L]-?iiš	John	?in	wa?ič-ḥuuk
			Ø- <u>try.to.stay</u> -3.ind	John	COMP	sleep-3.DEP
			John is trying to stay sle	eping.		(Pu-support)
	c.	*	?usinḥi?iš	John	wa?ičqu	u
			?u- <u>sinḥi</u> [+L]-?iiš	John	wa?ič-qu	u
			Ø- <u>try.to.stay</u> -3.ind	John	sleep-3.	COND
			John is trying to stay sle	eping.		(Pu-support)
(8)	a.		?ucači⊁wiťasitsiš		mituuni	
			?u-ca-či⊁- <b>wiťas</b> -mit-siiš		mituuni	
			Ø-go.to-PERF-gonna-PST	Γ-1SG.IND	Victoria	
			I was gonna go to Victor	ria.		(incorporation)
	b.	*	?uwitasmitsiš	?in ?uca	ıči⊁sa	mituuni
			?u-witas-mit-siiš	?in ?u-c	a-či⊁-sa	mituuni
			Ø- <u>gonna</u> -PST-1SG.IND	сомр Ø-д	o.to-PER	F-1SG.DEP Victoria
			I was gonna go to Victor	ria.		(Pu-support)
	c.	*	?uwitasmitsiš ?	ucači⊁quu	5	mituuni
			?u-witas-mit-siiš ?	u-ca-či <del>λ</del> -q	uus	mituuni
			$\emptyset$ -gonna-PST-1SG.IND (	Ø-go.to-₽E	RF-1SG.C	OND Victoria
			I was gonna go to Victor	ria.		(Pu-support)

Unlike main predicates, these auxiliary verbs have a same-subject restriction which requires that the notional subjects of the two predicates in the construction match. This is illustrated in the following examples with the affixal auxiliary predicate -qaath "claim", which incorporates the predicate 2u-uc "own".<sup>1</sup>

# (9) Same-subject requirement

a.	?uucqatḥ?iš	Florence šuwis
	?u-ic- <b>qaath</b> -?iiš	Florence šuwis
	Ø-own- <u>claim</u> -3.IND	Florence shoes
	Florence is pretending s	she owns the shoes.

 $<sup>^1</sup>$  The  $\mathcal{U}\!\!-\!$  in these examples occurs as a host to the affixal predicate  $\mathcal{U}\!\!-\!\!uc$  "own".

b.	*	?uucqathitsiš	Florence šuwis
		?u-ic- <b>qaath</b> -mit-siiš	Florence šuwis
		Ø-own- <u>claim</u> -PST-1SG.IND	Florence shoes
		I pretended Florence owns the	shoes.

Here, the same-subject requirement determines that the "claimer" and the "owner" must corefer. In (9a) a "match" occurs, while the ungrammatical (9b) illustrates a "mismatch".

## 5.1.3 Affixal Auxiliaries are Non-thematic

I propose that the difference between main and auxiliary predicates in Nuu-chahnulth is linked to their thematic properties. Specifically, their behaviours fall out from an analysis in which auxiliaries are functional elements, while main verbs are lexical (Cinque 2001; Wurmbrand 2004). Under this view, auxiliaries such as -qaath"claim" are non-thematic raising verbs which do not project a subject of their own.<sup>2</sup>

(10)	nunuukqatḥʔiš	Florence
	nunuuk- <b>qaath</b> -?iiš	Florence
	sing- <u>claim</u> -3.IND	Florence
	Florence is pretending	to sing.

The "same subject" effect of these auxiliaries arises when the subject of the main verb raises to specifier position of the auxiliary. In effect, the subject is "shared" between the auxiliary and the main verb. The diagram in (11) shows that the subject of the main predicate, *Florence*, raises to the specifier of the auxiliary.

(11) Auxiliaries as raising verbs



<sup>&</sup>lt;sup>2</sup> A direct English translation of Nuu-chah-nulth -qaath "claim" is problematic. The morpheme is irrealis, and is often translated as "seem" or "pretend"; Sapir and Swadesh (1939: 329) gloss it as "pretendedly". For consistency, I have elected to translate it as "claim" because that is a typical translation when the predicate is transitivized. Affixal auxiliaries require transitivization for different subjects (see §5.1.3.2).

In what follows, I will introduce evidence in favour of this analysis of auxiliaries as non-thematic verbs. In §5.1.3.1, I show that auxiliaries have rigid complementation. This is followed in §5.1.3.2 with an illustration of how transitivization allows an auxiliary to license an "external" argument. In §5.1.3.2, I discuss the lack of citation forms for these functional affixes.

### 5.1.3.1 Rigid vs. Alternating Complementation

As functional morphemes, auxiliaries are predicted to show rigid complementation, in which they select only an infinitival complement (Cinque 2001). In (12), the infinitival complement of the auxiliary is represented as a *v*P. This *v*P lacks higher functional projections for tense (TP), agreement (AgrP), and complementizers (CP).

### (12) Auxiliary predicate: strict selection of infinitival complement



Main predicates, in contrast, are lexical verbs which permit lexically specified alternations in complementation. For Nuu-chah-nulth, I propose that these main predicates may take either an infinitival or clausal complement. In (13a), the main verb selects an infinitival ( $\nu$ P) complement, while in (13b), it selects a "full" CP complement.

### (13) Main predicate: alternations in complement selection



### 5.1.3.2 Ability to License a "Mismatched" Subject

A distinction between the lexical and functional characteristics of affixal predicates has implications for these predicates' argument-taking abilities. As functional elements, auxiliary predicates are anticipated to be non-thematic (Cinque 2001; Wurmbrand 2004). This meshes with the observation that the valency-increasing morpheme – ap (TR) is employed in Nuu-chah-nulth in contexts in which the same-subject requirement of auxiliaries is overridden (see also Rose 1981).<sup>3</sup> In the

<sup>&</sup>lt;sup>3</sup> Rose (1981: 306) notes that -ap is used with Kyuquot desideratives "to indicate that the subject (the desirer) is not coreferential to the complement subject". Evidence from Ahousaht

following (a) examples, -ap(TR) is used to permit a "different subject" reading. The examples in (b) show the illicit outcome without use of -ap(TR).

(14)	a.		?uucqath?apsiš		Florence	šuwis
			?u-ic- <u>qaath</u> -'ap-siiš		Florence	šuwis
			Ø-own- <u>claim</u> -tr-3.ind		Florence	shoes
			I'm pretending Florence o	wns the s	shoes.	
	b.	*	?uucqatḥsiš	Florence	e šuwis	
			?u-ic- <b>qaath</b> -siiš	Florence	e šuwis	
			Ø-own- <u>claim</u> -1sg.ind	Florence	e shoes	
			I'm pretending Florence o	wns the s	shoes.	
(15)	a.		m҆iλšiλmaḥsapsiš mํiλ-šiλ- <u>maḥsa</u> -'ap-siiš			
			rain-PERF- <u>want.to</u> -TR-1SC	J.IND		
			I want it to rain.			
	b.	*	mii⊁ši⊁maḥsasiš			
			mi⊁-ši≁- <u>maḥsa</u> -siiš			
			rain-PERF- <u>want.to</u> -1SG.IN	D		
			I want it to rain.			
(16)	a.		λuu†sinḥap?iš	John	šuwisuk?i	
			λut- <u>sinhi</u> -'ap-?iiš	John	šuwis-uk-?i	
			good-try.to.stay-TR-3.IND	) John	shoes-POSS-3.PS	5
			John is trying to keep his	shoes nic	e.	
	b.	*	⊁uu†sinḥi?iš	John	šuwisuk?i	
			≁u1- <u>sinhi</u> -?iiš	John	šuwis-uk-?i	
			good-try.to.stay-3.IND	John	shoes-POSS-3.PS	5
			John is trying to keep his	shoes nic	e.	
			, , , , , , , , , , , , , , , , , , ,			

Note that this behaviour contrasts with that of main predicates, which do not employ -ap (TR) in cases of subject mismatches. In (17a), the "dreamer" and the "one who dies" are disjoint, without use of -ap (TR). The example in (17b) shows that -ap (TR) is impossible in this environment.

(17)	a.	qaqaḥatuᆉitsiš	suwa
		qaḥ- <b>atu¹</b> [+R]-mit-siiš	suwa
		die-dream.of-PST-1SG.IND	2sg
		I dreamt that you died!	

indicates that – *ap* has a broad usage in which is it employed with the full range of auxiliary predicates, not simply desideratives.

b.	*	qaqahatu 1 ?amitsiš	suwa
		qaḥ- <b>atuł</b> [+R]-'ap-mit-siiš	suwa
		die-dream.of-TR-PST-1SG.IND	2sg
		I dreamt that you died!	

Although -ap is standardly labelled a "causative" (Sapir and Swadesh 1939), I contend that it has a more general transitivizing function than this term suggests (see also Rose 1981). With certain psych predicates (such as *yaa?ak* "feeling", or *?iiḥmis* "be treasured"), usage of this morpheme converts an intransitive to a transitive. The examples in (18) show an intransitive usage of *yaa?ak* "feeling" and *?iiḥmis* "be treasured", while the examples in (19) illustrate a transitive usage employing – *ap*. In (18a), the predicate is translated as "sore" and it takes a single argument (*ħisħin?atqs* "my foot/feet"); in the transitivized (19a), it is translated as "love" and takes two arguments. In (18b), the predicate is translated as "be treasured" and it takes a single propositional argument; in the transitivized case of (19b), the predicate is translated roughly as "won't part with" and it takes two arguments (*ʔux "aapi* "paddle" and *naniiq* "grandparent").

- (18) Intransitive
  - a. yaa?ak?iš  $\lambda$ iš $\lambda$ in?atqs yaa?ak-?iiš  $\lambda$ iš $\lambda$ in-?at-qs feeling-3.IND lower.leg-IPOSS-1SG.PS My feet are sore.
  - b. ?iiḥmis?iš kʷakuucnakat qʷaaqḥ +ačyuu ?uḥ Kay ?iiḥmis-?iiš kʷakuuc-naak-'at qʷaa-qḥ-Ø +ačyuu ?uḥ Kay treasured-3.IND grandchildren-have-PAS how-AUX-3.ABS let.go DEIC Kay Having grandchildren is something to be treasured, and yet Kay lets them go.

## (19) Transitivized

a.	yaa?akapsiš	suwa
	yaa?ak-'ap-siiš	suwa
	feeling-TR-1SG.IND	2sg
	I love you; I care for you.	

b.	?iiḥmis?ap?iš	?ux <sup>w</sup> aapi	naniiq		
	?iiḥmis-'ap-?iiš	?ux <sup>w</sup> aapi	naniiq		
	treasured-TR-3.IND	paddle	grandparent		
	Grandparent won't part with a paddle (i.e., he's possessive of it).				

The transitivizing function of -ap (TR) is represented in the diagrams below. In (20a), the predicate *yaa?ak* "feel" is shown as an unaccusative, projecting a single, internal argument. In (20b), -ap (TR) introduces an additional argument.



For affixal auxiliary predicates, I propose that the morpheme -ap (TR) serves a similar transitivizing function when it licenses a subject "mismatch". Recall that affixal auxiliary predicates show a same-subject restriction (as in 21a), and disallow mismatched subjects (as in 21b). The morpheme -ap must be used in cases in which the subjects do not match (as in 21c).

a.		wataakmaḥsaʔiš	Kay	mituuni	
		wataak- <u>mahsa</u> -?iiš	Kay	mituuni	
		go.to- <u>want.to</u> -3.IND	Kay	Victoria	
		Kay wants to go to Victori	a.		(same subject)
b.	*	wataakmaḥsasiš	Kay	mituuni	
		wataak- <u>mahsa</u> -siiš	Kay	mituuni	
		go.to- <u>want.to</u> -1SG.IND	Kay	Victoria	
		I want Kay to go to Victor	ia.		(different subjects)
c.		wataakmaḥsapsiš	Kay	mituuni	
		wataak- <b>maḥsa</b> -'ap-siiš	Kay	mituuni	
		go.to-want.to-TR-1SG.IND	Kay	Victoria	
		I want Kay to go to Victor	ia.		(transitivization)
	а. b. c.	a. b. * c.	<ul> <li>a. wałaakmaḥsaʔiš wałaak-maḥsa-ʔiiš go.to-want.to-3.IND Kay wants to go to Victori</li> <li>b. * wałaakmaḥsasiš wałaak-maḥsa-siiš go.to-want.to-1SG.IND I want Kay to go to Victor</li> <li>c. wałaakmaḥsapsiš wałaak-maḥsa-'ap-siiš go.to-want.to-TR-1SG.IND I want Kay to go to Victor</li> </ul>	<ul> <li>a. wałaakmaḥsaʔiš Kay wałaak-maḥsa-ʔiiš Kay go.to-want.to-3.IND Kay Kay wants to go to Victoria.</li> <li>b. * wałaakmaḥsasiš Kay wałaak-maḥsa-siiš Kay go.to-want.to-1SG.IND Kay I want Kay to go to Victoria.</li> <li>c. wałaakmaḥsapsiš Kay wałaak-maḥsa-ʾap-siiš Kay go.to-want.to-TR-1SG.IND Kay I want Kay to go to Victoria.</li> </ul>	<ul> <li>a. wałaakmaḥsa?iš Kay mituuni wałaak-maḥsa-?iiš Kay mituuni go.to-want.to-3.IND Kay Victoria Kay wants to go to Victoria.</li> <li>b. * wałaakmaḥsasiš Kay mituuni go.to-want.to-1SG.IND Kay Victoria I want Kay to go to Victoria.</li> <li>c. wałaakmaḥsapsiš Kay mituuni wałaak-maḥsa-'ap-siiš Kay mituuni go.to-want.to-TR-1SG.IND Kay Victoria I want Kay to go to Victoria.</li> </ul>

As noted in the earlier discussion, I propose that affixal auxiliaries are raising verbs which do not introduce an external argument (see Cinque 2001). The auxiliary "shares" the thematic subject projected by the embedded verb. With the introduction of the transitivizer – *ap*, however, the auxiliary predicate inherits transitive syntax, and thus the same-subject restriction is overcome. In the same-subject case in (22a), the argument *Kay* is shared by the auxiliary –*mahsa* "want to" and the main predicate *wataak* "go to". In (22b), in contrast, the affixal predicate –*mahsa* "want to" inherits a first person singular argument by virtue of the transitivizer. Therefore, (22b) allows a different subject than the one (*Kay*) used by the main predicate *wataak* "go to".



No derived transitivization occurs for affixal main predicates, because the status of these predicates as (thematic) lexical verbs allows them to introduce an external argument directly. Recall that there is no same-subject requirement for a main predicate such as  $\partial u$ -Si $\lambda$  "come upon".

(23)	wa?ičSi⊁itsiš	Ken
	wa?ič- <u><b>Siૠ</b></u> -mit-siiš	Ken
	sleep- <u>come.upon</u> -PST-1SG.IND	Ken
	I found Ken sleeping.	

In the following diagram, the affixal main predicate  $\partial u$ -Si $\lambda$  "come upon" is represented as a thematic verb with inherent transitivity. The transitivizer – ap (TR) does not appear in this construction.



The predicate 2u-Sik "find" takes a first person singular argument which is distinct from the one contained in its complement (*Ken*).

### 5.1.3.3 Citation Forms

Additional support for a split between functional and lexical characteristics of affixal predicates may be adduced from native speakers' judgements about the citation forms of these verbs. Affixal auxiliary predicates, which I analyse as functional elements, are unrecognizable to Nuu-chah-nulth speakers without an incorporated verb, and therefore lack an independent citation form. Affixal main predicates, on the other hand, are pronounced in isolation as forms affixed to the expletive morpheme  $\lambda u$ -. Thus, while speakers of Nuu-chah-nulth reject "words" such as \* $\lambda u$ -witas "gonna" or \* $\lambda u$ -qaath "claim", they freely accept  $\lambda u \lambda u$ -utut "dream of" or  $\lambda u$ -Si $\lambda$ " "come upon".<sup>4</sup> I suggest that this difference in isolatability is linked to the lexical status of these affixal predicates.

### 5.1.4 Summary

To recap, evidence has been presented for two distinct varieties of verbincorporating affixal predicates. I have proposed that the behaviours of affixal main and auxiliary predicates are tied to the thematic properties which derive from their functional or lexical status. In the next section, I turn to an analysis of the suffixation patterns of main and affixal auxiliary predicates, which states that PF incorporation is only permitted in infinitival environments.

### 5.2 PF Incorporation Occurs Only in Infinitival Environments

For affixal predicates that take verbal complements, incorporation arises only in monoclausal configurations in which the complement of an affixal predicate is an infinitival smaller than a TP. PF incorporation is ruled out across biclausal structures in which the embedded clause is demarcated by CP (cf. Li 1990). As first discussed in Chapter 2, this may be attributed to an opacity effect, induced by the status of CPs as saturated domains which constitute self-contained units of the derivation.

### (25) a. Incorporation configuration b. Full CP complementation



<sup>&</sup>lt;sup>4</sup> A superficial exception is the affixal predicate -mahsa "want to", which I analyse as a auxiliary. Speakers accept  $\lambda u$ -mahsa as a "word". However, in this case it appears that the auxiliary -mahsa is homophonous with the non-auxiliary form  $\lambda u$ -mahsa "want", which takes a nominal complement, as in  $\lambda u$ -mahsais has  $\lambda a$  capac $\lambda a$ " "I want that canoe". The form  $\lambda u$ -mahsa is not accepted in contexts of verbal complementation (ex. 5).

In contexts in which an affixal predicate selects a full complement, a stranded-affix "repair" strategy of suffixation to the expletive morpheme  $\lambda \mu$  is implemented, rather than suffixation via incorporation. As was discussed in Chapter 2, a host may not be chosen from within the CP, because the fully interpreted CP is closed off from the active workspace of the derivation.

Let us sketch the analysis of the linearization strategy which occurs when the complement of an affixal main or auxiliary predicate is infinitival. When a main predicate such as  $\partial u$ -*sik* "come upon" incorporates a host, the choice of host is determined by string adjacency to the affixal predicate at spell-out. In (26a), the affixal predicate suffixes to the verb *wa?ič* "sleep", while in (26b), it suffixes to the modifier *hacuk* "(sleep) deeply", stranding the verb *wa?ič* "sleep".

(26)	a.	wa?ičSiૠitsiš wa?ič- <u>Siૠ</u> -mit-siiš	Ken Ken	
		sleep- <u>come.upon</u> -PST-1SG.IND	Ken	
		I found Ken sleeping.		
	b.	ḥacukSi⊁itsiš	wa?ič	Ken
		ḥacuk- <b>Siૠ</b> -mit-siiš	wa?ič	Ken
		deeply- <u>come.upon</u> -PST-1SG.IND	sleep	Ken
		I came upon Ken in a deep sleep.		

The syntactic positions of the verb and the pre-verbal modifier are illustrated in (27). In (27a),  $-\Im \lambda$  "come upon" takes a vP complement with an unmodified verb, wa?ič "sleep", as the embedded predicate. In (27b), this embedded predicate is modified by *hacuk* "(sleep) deeply".



In each case, the embedded subject, *Ken*, appears as a right-linearized specifier of the embedded *v*P. In (27a), the embedded *v*P is linearized as *<wa?ič*, *Ken>*, while in (27b), the embedded *v*P is linearized as *<hacuk wa?ič*, *Ken>*. According to the local

spell-out proposal, the linearization of the affixal predicate for (26a-27a) and (26b-27b) follows from the orderings of these embedded phrases. For (27a), it is the verb *wa?ič* "sleep" which is string adjacent to the affixal predicate  $-Si\lambda$  "come upon" when it attains spell-out. For (27b), in contrast, it is the modifier *hacuk* "(sleep) deeply" which has this privileged status of string adjacency to the affix. Thus, verb incorporation is induced for (27a), while adverbial incorporation is induced for (27b).

A parallel process of linearization is proposed to occur for affixal auxiliary predicates, such as  $-\dot{witas}$  "gonna". Just like affixal main predicates, these auxiliaries allow suffixation to either a verbal or adverbial host. The example in (28a) shows  $-\dot{witas}$  "gonna" incorporating the verb *wa?ič* "sleep", while (28b) shows incorporation of the temporal adverbial *čaani* "first".<sup>5</sup>

- (28) a. wa?ičŵiťassiš wa?ič-<u>ŵiťas</u>-siiš sleep-<u>gonna</u>-1SG.IND I'm gonna sleep.
  - b. čaaniwitassiš wa?ič
     čaani-<u>witas</u>-siiš wa?ič
     first-gonna-1SG.IND sleep
     I'm gonna sleep first (i.e., before doing something else)

The syntactic representations I assume for these examples are given in (29). The vP complements of the auxiliary  $-\dot{witas}$  "gonna" contain a null first person singular pronominal as subject. In (29a), the verb *wa?ič* "sleep" is unmodified, while in (29b) the modifier *čaani* "first" occupies a pre-verbal adjunct position.<sup>6</sup>



<sup>&</sup>lt;sup>5</sup> As noted in Chapter 2, the temporal adverbal *čaani* "first" is "flexibly positioned": outside of incorporation contexts, it allows either a pre-verbal or post-verbal positioning. Accordingly, verb incorporation is an alternative to the adverbial incorporation pattern of (28b).

<sup>&</sup>lt;sup>6</sup> I adopt an analysis in which the temporal adverbial *čaani* "first" occupies the same preverbal adjunct position to the verb as a manner adverbial. This is not a necessary assumption, so long as each occur pre-verbally (see Chapter 2 for discussion). The syntactic position of adverbials in Nuu-chah-nulth is an understudied area in need of future research.
The pre-verbal position of the adverbial *čaani* has consequences for the linearization of the affixal auxiliary  $-\dot{witas}$  "gonna". In (29a), the derivational sister of the auxiliary is equivalent to  $\langle wa2ic\rangle$ , while for (29b), the derivational sister is the linearized object  $\langle caani, wa2ic\rangle$ . At local spell-out of  $-\dot{witas}$  "gonna", wa2ic "sleep" is selected as the host for (29a), while *čaani* "first" takes on the role of host in (29b) due to its string adjacency to the affix.

The proposed analysis states that while main and auxiliary predicates differ in their lexical properties, they share the ability to take an infinitival  $(\nu P)$ complement. It is in these infinitival environments that incorporation occurs. As will be shown in the following sections, a range of evidence exists for the analysis that the syntactic prerequisite for PF incorporation is an infinitival complement smaller than a TP. In the following sections, I consider three sets of arguments in favour of this analysis: the absence of embedded clausal morphology; "restructuring" effects which indicate a lack of clause-boundedness; and finally, evidence for affixal main and auxiliary predicates being raising verbs.

#### 5.3 Absence of Clausal Morphology

This section examines morphological evidence for the analysis that PF incorporation occurs in infinitival contexts. If the "clauselet" in incorporation contexts is smaller than a TP, then this complement is predicted to lack specifications for tense, person/mood agreement and complementizers. If this reduced clause is equivalent to vP, then only aspectual morphology (occupying v) is predicted to be present. This section demonstrates that this prediction regarding the absence of higher-level clausal morphology holds. In incorporation contexts, the verbal host of an affixal auxiliary may only be inflected for aspect: independent specifications for tense, subject/mood agreement and complementation are systematically absent. For example, the verbal host wat-sik "go home (PERF)" in (30) contains the perfective marking -sik (PERF), but no tense marker -2aqk (FUT), subject/mood agreement -sa (1SG.DEP) or complementizer 2in (COMP).

(30)	watšiλ(*?aqλ)(*sa)maḥsasiš	(*?in)
	wat-šik-(*?aqk)-(*sa)-mahsa-siiš	(*?in)
	go.home-PERF-(FUT)-(1SG.DEP)- <u>want.to</u> -1SG.IND	(COMP)
	I want to go home.	

This follows from an analysis in which the complement of -mahsa "want to" is a reduced clause equivalent to vP, with perfective aspect shown to occupy v.<sup>7</sup>

<sup>&</sup>lt;sup>7</sup> It is not a crucial assumption for aspect to occupy v. I propose that the reduced clause is smaller than a TP, but it is possible that projections intermediate to TP and vP exist which could house aspectual morphology. The syntactic representation of aspectual morphology in Nuu-chah-nulth is in need of further research, as this will shed light onto how much smaller than TP the "clauselet" is. See §6.2.1.3 for related discussion of the status of the vP domain.



In the following subsections, I will present in detail the evidence for a lack of tense (§5.3.1), person/mood agreement (§5.3.2) and complementizers (§5.3.3) in the reduced clause.

## 5.3.1 No Tense

An incorporated verb is systematically free of tense morphology. Neither the past tense marker -mit (PST) nor the future tense marker  $-2aq\lambda$  (FUT) can occur with an incorporated verb. This is true for verbs hosting either an auxiliary or main predicate. In (32), with the affixal auxiliary predicate -qaath "claim", it is shown to be ungrammatical for the verbal host *nunuuk* "sing" to be inflected for the future tense  $-2aq\lambda$  (FUT). In (33), with the main verb 2uu-nakuuh "observe", it is illicit for the host *tuuxtuux<sup>w</sup>a* "jump (ITER)" to be marked for past tense -mit (PST).

(32) Incorporation with auxiliaries: absence of embedded tense

a.		nunuuk	qatḥ?iš	]	Florence	e	
		nunuuk	- <b>qaatḥ</b> -ʔiiš	J	Florence	e	
		sing- <u>cla</u>	<u>aim</u> -3.ind	J	Florence	e	
		Florenc	e is pretendi	ng to si	ng.		
b.	*	nunuuk	?aq <del>x</del> qath?iš			Florence	e
		nunuuk	-?aqx-qaath-	?iiš		Florence	e
		sing-FU	T-claim-3.IN	D		Florence	e
		Florenc	e is pretendi	ng she's	s going	to sing.	
		cf.	nunuukqath	h?aq <del>x</del> ?iš	ŝ		Florence
			nunuuk-gaa	<b>ath</b> -?aq?	₹-?iiš		Florence
			sing-claim	-FUT-3.	IND		Florence
			Florence w	ill prete	end to si	ng.	
-					c		

(33) Incorporation with main verbs: absence of embedded tense

a.	tuuxtuux <sup>w</sup> anakuuhitsiš	suwa
	tuux <sup>w</sup> -a[+R]- <u><b>'nak̈uuḥ[</b>+L]-mit-siiš</u>	suwa
	jump-ITER- <u>observe</u> -PST-1SG.IND	you
	I observed you jumping.	

#### THE LINEARIZATION OF AFFIXES

b.	*	tuuxtuux <sup>w</sup> amitnakuuhitsiš	suwa
		tuux <sup>w</sup> -a[+R]-mit- <u><b>ṅakuuh</b>[</u> +L]-mit-siiš	suwa
		jump-ITER-PST- <u>observe</u> -PST-1SG.IND	you
		I observed you jumping.	

This ban on embedded tense morphology stands in contrast to the availability of tense specification in full complements. In (34), the full complement *in tuuxtuux<sup>w</sup>amitsuk* "that you were jumping" contains the past tense marker *-mit* (PST).

(34)	?uunakuuḥitsiš	?in	tuuxtuux <sup>w</sup> amitsuk
	?u- <u><b>ňakuuḥ[</b>+L]-mit-siiš</u>	?in	tuux <sup>w</sup> -a[+R]-mit-suuk
	∅– <u>observe</u> -pst-1sg.ind	COMP	jump-ITER-PST-2SG.DEP
	I observed you jumping. (lit: "	I observed th	at you were jumping")

The lack of independent embedded tense morphology in incorporation contexts is predicted by an analysis in which the embedded clause does not project a Tense Phrase.

## 5.3.2 No Person/Mood Agreement

The complex predicate formed by incorporation is marked with only a single set of portmanteau person/mood inflection: the incorporated verb is not independently inflected. In (35), it is shown for the auxiliary predicate -mahsa "want to" that the host is not inflected for the person/mood agreement -suuk (2SG.DEP). In (36), this same restriction is demonstrated for the main verb  $2u-2ii-či\lambda$  "hear, find out (PERF)".

(35) Incorporation with auxiliaries: absence of embedded person/mood inflection

a.	?acši⊁maḥsak
	?ac-ši <del>λ</del> - <u>maḥsa</u> -k
	go.fishing-PERF- <u>want.to</u> -2SG.Q
	Do you want to go fishing?

 b. \* ?acši⊁suukmaḥsak ?ac-ši⊁-suuk-<u>maḥsa</u>-k go.fishing-PERF-2SG.DEP-<u>want.to</u>-2SG.Q Do you want to go fishing?

(36) Incorporation with main verbs: absence of embedded person/mood inflection

a.	ta?ił?iiči <del>λ</del> itsiš	suwa
	ta?i₁- <u><b>?ii</b></u> -či⊁-mit-siiš	suwa
	sick- <u>hear</u> -PERF-PST-1SG.IND	you
	I heard you were sick.	

b. *	ta?i∱suuk?iiči⊁itsiš	(suwa)	
		ta?i₁-suuk- <u><b>?ii</b></u> -či⊁-mit-siiš	(suwa)
		sick-2sg.dep- <u>hear</u> -perf-pst-1sg.ind	(you)
		I heard you were sick.	

An embedded verb is inflected within a full complement, however. In (37), the main verb  $\partial u$ - $\partial i$ - $\partial i$ - $\partial i$ , "hear, find out (PERF)" takes a full complement,  $\partial i$  ta $\partial i$ +suuk "that you were sick", which contains the dependent person/mood agreement –suuk (2SG.DEP).

(37)	?u?iiči⊁itsiš	?in	ta?i <del>1</del> suuk
	?u- <u><b>?ii</b></u> -či <del>λ</del> -mit-siiš	?in	ta?i1-suuk
	Ø- <u>hear</u> -perf-pst-1sg.ini	D COMP	sick-2SG.DEP
	I heard that you were sick	ζ.	

The lack of inflectional morphology on a verbal host falls out directly from an analysis in which the "clauselet" containing the embedded verb does not project up to an Agr Phrase.

# 5.3.3 No Complementizer

(39)

There is no complementizer 2in in incorporation contexts. In (38), it is shown that the complementizer 2in (COMP) is obligatorily absent in contexts with the auxiliary -mahsa "want to". In (39), the complementizer is shown to be illicit when the affixal main predicate 2uu-nakuuh "observe" suffixes to the verbal host muu2akwa-cix "burn (PERF)".

(38)	Incorporation	with	auxiliaries:	absence	of com	plementizer
(30)	meorporation	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	construction rest.	ubbenee	oj com	rementizer

a.		saapniqiiłmaḥsaḥ saapniq-čiił- <u>maḥsa</u> -ḥ bread-make- <u>want.to</u> -3.Q Does Kay want to make b	Kay Kay Kay read?	
b.	*	saapniqiiłmaḥsaḥ saapniq-čiił- <b>maḥsa</b> -ḥ bread-make- <b>want.to</b> -3.Q	?in ?in COMP	Kay Kay Kay
		Does Kay want to make b	read?	5
Inco	orpo	ration with main verbs: ab	sence of	complementizer
a.		mu?ak <sup>w</sup> ači≁nakuuhitsiš		łayiipt

murakwačinnakuuhitsis	Tayupt
mu?ak <sup>w</sup> a-či <del>λ</del> - <u>nໍakuuḥ[</u> +L]-mit-siiš	łayiipt
burn-PERF- <u>observe</u> -PST-1SG.IND	leaves
I was watching leaves burn.	

b. *	*	mu?ak <sup>w</sup> ači <del>λ</del> nakuuḥitsiš	?in	łayiipt
		mu?ak <sup>w</sup> a-či <b>~-<u>nໍakuuḥ[</u>+L]-mit-sii</b> š	?in	łayiipt
		burn-PERF- <u>observe</u> -PST-1SG.IND	COMP	leaves
		I was watching leaves burn.		

In contrast, within a full complement, complementizers are present. In (40), the affixal main predicate  $\partial uu$ -nakuuh "observe" is hosted by the expletive  $\partial u$ . Here, the complementizer appears with the full complement.

(40)	?uunakuuhitsiš	?in	mu?ak <sup>w</sup> ači <del>λ</del> ḥuk	łayiipt
	?u- <u>ňakuuh</u> [+L]-mit-siiš	?in	mu?ak <sup>w</sup> a-či≁-ḥuuk	łayiipt
	Ø- <u>observe</u> -PST-1SG.IND	COMP	burn-perf-3.dep	leaves
	I was watching leaves but	rn.		

The absence of a complementizer in incorporation contexts follows if there is no Complement Phrase demarcating the boundary between the affixal predicate and its complement. According to the saturated domain hypothesis introduced in Chapter 2, CPs are independent domains for affixation. Thus, incorporation of a host is only possible when a CP "border" does not intervene between the affixal predicate and its potential host.

# 5.4 Lack of Clause-boundedness Effects

A diagnostic property of infinitival constructions are transparency effects in which matrix and embedded constituents form a unitary domain for otherwise clausebound processes (Cinque 2001). These "restructuring" phenomena (also known as "clause union" or "reanalysis") have been widely documented in languages, including Romance (e.g., Aissen and Perlmutter 1983; Rizzi 1982; Roberts 1997; Rooryck 2000; Cinque 2002) and Germanic (e.g., Evers 1975; Wurmbrand 2001). To illustrate an example, consider the optional process of clitic-climbing in Romance, which occurs in environments in which the matrix predicate belongs to a restricted class of "restructuring" auxiliaries. In (41), the "restructuring" verb is *voudrais* "would like".

(41) *Clitic-climbing* (Cinque 2002: 620, ex. 4b, 5b)

a.	Je voudrais y I would-like there I would like to go ther	aller. e go e.	
b.	J'y voudrais I- <i>there</i> would-like I would like to go ther	aller. go e.	(French)

With clitic-climbing, the pronominal clitic associated thematically with an infinitival verb is promoted to a position within the matrix clause. In (41b), the pronominal y "there" abandons its position within the embedded clause (*y aller* "go there") and surfaces instead as part of the pronominal proclitic string attached to *voudrais* "would like". In this case of "restructuring", the infinitival does not constitute an independent cliticization domain distinct from the matrix verb.

In this section, I provide evidence that cases of incorporation in Nuu-chahnulth display a similar lack of clause-boundedness. Thus, affixal main and auxiliary predicates are "restructuring" verbs. The two cases I consider here are "long" *wh*questions and "long" possessor raising constructions which are formed with affixal predicates. Following Cinque (2001) and Wurmbrand (2001), I argue that these types of "restructuring" effects in Nuu-chah-nulth are indicative of the reduced clausal status of the embedded complement: no clause-boundedness effects are found with incorporation because no clause boundary (i.e., CP) is projected between the matrix predicate and its infinitival complement.

### 5.4.1 "Long" wh-movement

As first noted by Davis and Sawai (2001), *wh*-movement in Nuu-chah-nulth is strictly clause-bound. This is shown in the examples below, in which *wh*-extraction out of the complement of the non-affixal predicate *taaquk* "believe" is disallowed. The example in (42a) shows a grammatical case with no *wh*-extraction. In (42b, c), it is shown to be ungrammatical to question the subject of the embedded verb phrase *kuuŵit čapac* "steal the canoe".

- (42) a. ťaaquk<sup>w</sup>i?aλ?iš John ?in kuuwiłłithuk Mary čapac ťaaquk-čiλ-'aλ-?iiš John ?in kuuwiłł-mit-huuk Mary čapac believe-PERF-TEMP-3.IND John COMP steal-PST-3.DEP Mary canoe John believes that Mary stole the canoe. (cf. Davis and Sawai 2001:133)
  - b. \* ?ačaqḥ ťaaquk<sup>w</sup>i?aλ John ?in kuuŵiłitḥuk čapac ?ačaq-ḥ ťaaquk-čiλ-'aλ John ?in kuuŵił-mit-ḥuuk čapac who-3.Q believe-PERF-TEMP John COMP steal- PST-3.DEP canoe Who does John believe stole the canoe? (cf. Davis and Sawai 2001:133)
  - c. \* ?ačaqh ťaaquk<sup>w</sup>i?a $\lambda$  John (?in) kuuwiłith čapac ?ačaq-h ťaaquk-či $\lambda$ -'a $\lambda$  John (?in) kuuwiłi-mit-h čapac who-3.Q believe-PERF-TEMP John (COMP) steal-PST-3.Q canoe Who does John believe stole the canoe?

A different pattern is displayed by affixal predicates, however. "Long" *wh*-movement is permitted out of the complement of an affixal auxiliary. In (43a), it is shown to be grammatical to question the subject of the verb phrase *kuuŵit čapac* "steal the canoe", when the auxiliary –*qaath* "claim" is used. In (43b), a similarly grammatical example is shown with the auxiliary –*maḥsa* "want to": here, the object of the predicate *maakuk* "buy" is questioned.

(43)	a.	?ačaqqatḥ?apḥ	John	kuuwi1	čapac
		?ačaq- <b>q<u>aath</u>-'</b> ap-ḥ	John	kuuwi1	čapac
		who- <u>claim</u> -TR-3.Q	John	steal	canoe
		Who does John claim st	ole the car	noe?	

b. ?aaqičiłmaḥsak maakuk ?aqi-čił[+L]-**maḥsa**-k maakuk what-AUX-<u>want.to</u>-2SG.Q buy What do you want to buy?

The contrast between ungrammatical interclausal *wh*-movement and grammatical "long" *wh*-movement across an infinitival can also be observed with affixal main predicates. *Wh*-movement is barred across a full clausal complement of an affixal main predicate, but "long" *wh*-movement is allowed when the complement is infinitival.

### (44) a. Full complement: no long-range wh-movement

\* ?aačinakuuhith Florence ?in tuuxtuuxwamithuk ?ačaq-či1{[+L]-nakuuh[+L]-mit-h Florence ?in tuuxw-a[+R]-mit-huuk who-AUX-observe-PST-3.Q Florence COMP jump-ITER-PST-3.DEP Who was Florence watching jumping?

#### b. Infinitival complement: "long" wh-movement

?aačinakuuhith	Florence	tuuxtuux <sup>w</sup> a
?ačaq-či†[+L]- <u>ňakuuh</u> [+L]-mit-ḥ	Florence	tuux <sup>w</sup> -a[+R]
who-AUX-observe-PST-3.Q	Florence	jump-ITER
Who was Florence watching jumping?		

This type of "long" *wh*-movement with affixal predicates avoids true long-range movement (crossing a CP), as indicated in the diagrams below.

(45) a. Ungrammatrical long-range wh-extraction



b. grammatical "long" wh-extraction across an infinitival complement

-qaatḥ	[ <sub>vP</sub>	?ačaq	kuuwi1	č'apac]
claim		who	steal	canoe
•		]		

*Wh*-movement in Nuu-chah-nulth does not cross a CP (Davis and Sawai 2001). However, in contexts in which an affixal predicate takes what I analyse to be an infinitival complement, movement out of the complement is unrestricted. This transparency effect is predicted if the infinitival complement of an affixal predicate lacks higher clausal projections such as CP.

#### 5.4.2 "Long" Possessor Raising

This section argues that "long" possessor raising in Nuu-chah-nulth is indicative of the reduced clausal status of complements in incorporation contexts. As first described in §2.2.2.1, Nuu-chah-nulth exhibits a process of possessor raising in which the possessive morpheme  $-uk/-\partial ak$  (POSS) appears on the predicate rather than the possessum (Davidson 2001; Ravinski 2005). In a possessor raising construction, the subject agreement of the clause matches the possessor. The example in (46a) shows no possessor raising: the possessive morpheme -uk (POSS) suffixes to the possessum  $k^waa\partial uuc$  "grandchild". In the possessor raising example (46b), the predicate  $ta\partial t$  "sick" is suffixed by -uk (POSS), and the subject agreement is first person singular -siis(1SG.IND), to match the features of the possessor.

(46)	a.	ta?ił?iš ta?ił-?iiš sick-3.IND	kʷaaʔuucukqs kʷaaʔuuc-uk-qs grandchild-POSS-1SG.PS		
		My grandchild is sick.		(unraised)	
	b.	ta?iłuksiš ta?ił-uk-siiš sick-POSS-1SG.IND My grandchild is s	k <sup>w</sup> aa?uuc k <sup>w</sup> aa?uuc grandchild sick.	(possessor raising)	

As discussed in Chapter 4, this pattern can be accounted for under an analysis in which the possessive morpheme -uk (POSS) licenses a position for a raised possessor (Ravinski 2005). In the unraised example (46a–47a), the possessor remains within the possessive NP. In the raised example (46b–47b), the possessor raises to Spec, PossP where, as the highest DP, it takes on the role of subject, and determines subject agreement.



There is evidence that possessor raising is only permitted within a clause. Possessor raising cannot cross a clause boundary marked by the complementizer 2in (COMP). The example in (48a) shows an unraised example for a sentence in which the matrix

predicate taqaak "believe" takes a complement containing the embedded nominal k "aa?uucukqs" "my grandchild". In (48b), intraclausal possessor raising occurs, in which the possessive morpheme -uk (POSS) is suffixed to the embedded verb ta?i? "sick". This case can be contrasted with the ungrammatical example in (48c), in which possessor raising crosses the CP boundary of the embedded clause, to suffix the possessive morpheme -uk (POSS) to the matrix predicate taqaak "believe".

(48)	a.	ťaqaaksiš ťaqaak-siiš believe-1SG.IND I believe my gran	?in ?in COMP ndchild i	ta?i+h ta?i+- sick-3 is sick.	uk ņuuk .DEP	k <sup>w</sup> aa?uuc k <sup>w</sup> aa?uuc grandchil	ukqs -uk-qs  d-POSS-]	l SG.PS (unraised)
	b.	ťaqaaksiš ťaqaak-siiš believe-1sG.IND I believe my grai	?in ?in COMP ndchild i	taʔiɬu taʔiɬ- sick-P is sick.	ksa uk-sa OSS-1SG. (ii	DEP ntraclausa	k <sup>w</sup> aa?uuo k <sup>w</sup> aa?uuo grandchi <i>il posses</i> .	c c ld sor raising)
	c. *	* ťaqaakuksiš ťaqaak-uk-siiš believe-POSS-1S I believe my gra	?i ?i G.IND C ndchild	n n OMP is sick	ta?i1(uk ta?i1-(ul sick-(PO (in	)sa x)-sa ss)-1sG.D sterclausal	EP ! possess	k <sup>w</sup> aa?uuc k <sup>w</sup> aa?uuc grandchild or raising)

From the ungrammaticality of (48c), it is evident that possessor raising must be contained within a CP.

# 5.4.2.1 "Long" Possessor Raising with Affixal Auxiliaries

Despite this restriction on interclausal possessor raising, "long" possessor raising can be observed in environments in which an affixal predicate incorporates a host. As was initially described in Chapter 4, for affixal auxiliaries suffixed by -uk (POSS), either a "short" possessor raising or "long" possessor raising interpretation is generally possible with unaccusative hosts. In (49a), the affixal auxiliary -qaath "claim" is hosted by the unaccusative predicate sahyut "healthy". The possessor raising interpretation, "my grandparent" is construed as the notional subject of both -qaath "claim" and sahyut "healthy." In the "long" possessor raising interpretation, "my grandparent" is construed as the notional subject of both -qaath "claim" and sahyut "healthy." In the "long" possessor raising interpretation, "my grandparent" is construed as the subject of sahyut "healthy", while the first person possessor of "grandparent" corresponds to the first person "subject" of -qaath "claim". The example in (49b) indicates that when the argument of the unaccusative is inanimate (sapnii "bread"), only a "long" possessor raising interpretation is felicitious.

(49)	a.	šaḥyutqatḥukʷitsiš	naniiqsu
		šaḥỷut- <b>qaatḥ</b> -uk-mit-siiš	naniiqsu
		healthy- <u>claim</u> -POSS-PST-1SG.IND	grandparent
		= (i) My grandparent claimed to be well.	('short' possessor raising)
		= (ii) I claimed my grandparent was well	l. ('long' possessor raising)

b. pux<sup>w</sup>aaqathuk<sup>w</sup>it?ick sapnii ?ata?iš wikyuu pux<sup>w</sup>-aa-**qaath**-uk-mit-?iick sapnii ?ata-?iiš wik-yuu rise-CONT-<u>claim</u>-POSS-PST-2SG.IND bread still-3.IND NEG-RES ≠ (i) ! Your bread claimed it was rising, but it still hasn't. (*'short' possessor raising*)
= (ii) You claimed your bread was rising, but it still hasn't.

('long' possessor raising)

The syntactic prerequisite for "long" possessor raising is feature matching between the notional subject of the auxiliary and of the possessor. For example, the sentence in (50) is ungrammatical if the "claimer" and the possessor show a feature mismatch between a first person singular "claimer" and a second person singular "possessor".

 (50) \* šahýutqathuk<sup>w</sup>itsiš naniiqs(ak?itk) šahýut-qaath-uk-mit-siiš naniiqsu(-?ak-?iitk) healthy-claim-POSS-PST-1SG.IND grandparent(-POSS-2SG.PS) I claimed your grandparent was well.

In contexts of feature mismatches, possessor raising is impossible: instead, -'ap (TR) is employed to license the "mismatch".<sup>8</sup>

(51)	šaḥyutqatḥ?amitsiš	naniiqsak?itk
	šaḥýut- <b>qaatḥ</b> -'ap-mit-siiš	naniiqsu-?ak-?iitk
	healthy- <u>claim</u> -TR-PST-1SG.IND	grandparent-POSS-2SG.PS
	I claimed your grandparent was v	vell.

<sup>&</sup>lt;sup>8</sup> This transitivization strategy is consistently available as an alternative to possessor raising with auxiliaries, whether or not possessor raising would be illicit in that context. In (i), transitivization expresses a meaning parallel to the "long" possessor raising in (ii).

 <sup>(</sup>i) ?uniiqath?ap?ick naniiqsak?itk ?atquu wikiit
 ?u-<u>nii-qaath</u>-'ap-?iick naniiqsu-?ak-?iitk ?at-quu wik-'iit
 Ø-<u>arrive-claim</u>-TR-2sG.IND grandparent-POSS-2SG.PS but-3.COND NEG-EXIS
 You claimed your grandparent came, although she isn't here.

(ii)	?uniiqathuk?ick	naniiqsu	?atquu	wikiit		
	?u- <u><b>ňii</b></u> - <u>qaath</u> -uk-?iick	naniiqsu	?at-quu	wik-'iit		
	Ø- <u>arrive</u> - <u>claim</u> -poss-2sg.ind	grandparent	but-3.COND	NEG- EXIS		
	= (i) <i>"short" possessor raising</i> Your grandparent claimed she came, although she isn't here.					
	= (ii) "long" possessor raising	g				

You claimed your grandparent came, although she isn't here.

"Long" possessor raising is only possible when an auxiliary takes an unaccusative verb as its complement. When the embedded verb is transitive or unergative, use of -ap (TR) is mandatory for disjoint readings of the subject of the auxiliary and of the embedded verb.

There is a contrast in Nuu-chah-nulth between illicit interclausal possessor raising and "long" possessor raising in contexts with affixal auxiliaries. This contrast is represented in (52). Raising of a DP possessor across a CP boundary is ruled out in Nuu-chah-nulth, as indicated in (52a). According to the analysis that the "clauselets" of affixal auxiliaries lack CPs, movement to a "raised" position is predicted to be possible. In (52b), "long" possessor raising does not cross a CP, because no CP is present.

#### (52) Interclausal vs. "long" possessor raising

a. Ungrammatical interclausal movement

b. Grammatical "long" movement



Thus, the possibility of "long" possessor raising in auxiliary environments can be viewed as evidence for the lack of clause-boundedness between the possessive argument of the embedded verb and the auxiliary.

### 5.4.2.2 No "Long" Possessor Raising with Affixal Main Predicates

Consider now the case of "long" possessor raising with affixal main predicates. All else being equal, we would anticipate that "long" possessor raising should be possible out of an infinitival complement of an affixal main predicate, as it is out of the complement of an auxiliary affixal predicate. However, what is instead found is that possessor raising can cross neither the full nor infinitival complement of an affixal main predicate. The examples below show this restriction for the main predicate  $\partial u$ - $\partial u$ - $\partial u$ - $\partial u$  (PERF)". The examples in (53) show that interclausal possessor raising (crossing a full complement clause) is impossible, as it is in all contexts in the language. In (54), "long" possessor raising is shown to also be illicit. The example in (54a) shows a grammatical unraised example, while (54b) shows the ungrammatical outcome when the -uk (POSS) suffixes to the predicate complex.

(53)	a.	?u?iiči⊁itwa?iš	Lucy ?in	ta?ił k	waa?uucu	k?i
		?u- <u><b>?ii</b></u> -či⊁-mit-wa?iš	Lucy ?in	ta?ił k	waa?uuc-u	ık-?i
		Ø- <u>hear</u> -PERF-PST-3.QUO	T Lucy COMP	sick gi	randchild-	-poss-3.ps
		Lucy heard that her gran	dchild is sick.	-		(unraised)
	b. *	?u?iiči∻itukwa?iš	Lucy	?in	ta?i <del>1</del>	k <sup>w</sup> aa?uuc
		?u- <u><b>?ii</b></u> -či <del>λ</del> -mit-uk-wa?iš	Lucy	?in	ta?i <del>1</del>	k <sup>w</sup> aa?uuc
		Ø- <u>hear</u> -PERF-PST-POSS-	3.QUOT Lucy	COMP	sick	grandchild
		Lucy heard that her grar	dchild is sick.		(int	erclausal PR)
(54)	a.	ta?i1?iiči⁢?iš		Lucy	k <sup>w</sup> aa?uu	ucuk?i
. ,		ta?i <b>+-?ii</b> -či <del>⊁</del> -mit-?iiš		Lucy	k <sup>w</sup> aa?uu	uc-uk-?i
		sick- <u>hear</u> -PERF-PST-	3.IND	Lucy	grandch	nild-POSS-3.PS
		Lucy heard that her	grandchild is s	sick.	C	(unraised)
	b.	* ta?i1?iiči⊁ituk?iš		Lucy	k <sup>w</sup> aa?uu	1
		ta?i₁- <b>?ii</b> -či≁-mit-uk-	?iiš	Lucy	k <sup>w</sup> aa?uu	ıc
		sick- <u>hear</u> - PERF-PST	-poss-3.ind	Lucy	grandcl	nild
		Lucy heard that her	grandchild is s	sick. ('l	long' poss	sessor raising)

A priori, the impermissibility of "long" possessor raising out of an infinitival complement of an affixal main predicate is surprising given the transparency effects found with affixal auxiliary predicates. However, I propose that this behaviour is a direct consequence of the absence of a TP node in the complement of an affixal main predicate.

Based on the proposal that nominative case is assigned by T (Chomsky 1995), the absence of TP predicts that an embedded subject is not case-licensed within an infinitival clause (Wurmbrand 2001). According to this proposal, main verbs such as  $\lambda u$ - $\lambda u$ - $\lambda i$ / $\lambda c$  (PERF) "hear, find out",  $\lambda u$ - $\lambda i \lambda$  "come upon" and  $\lambda u u$ - $\hbar a k u u h$  "observe" are analysable as verbs which govern promotion of an infinitival subject to a matrix object position. Under this analysis, an argument generated within the infinitival is unlicensed for nominative case, and so must raise to the matrix clause to receive abstract accusative case via the matrix  $\nu$  (Chomsky 1995). For a sentence such as in (55), this entails that  $k^waa \lambda u u c k q s$  "my grandchild" is realized as an object of the matrix verb, although its genesis is in the subject position of the embedded "clauselet".

(55)	tuuxtuux <sup>w</sup> anakuuhitsiš	k <sup>w</sup> aa?uucukqs
	tuux <sup>w</sup> -a[+R]- <u><b>'nakuuḥ</b></u> -mit-siiš	kwaa?uuc-uk-qs
	jump-ITER- <u>observe</u> - PST-1SG.IND	grandchild-POSS-1SG.PS
	I observed my grandchild jumping.	

This analysis is represented in (56). Here, the nominal phrase  $k^waa \partial uucukqs$  "my grandchild" moves from its base position as an argument of the embedded verb *tuuxtuux*<sup>w</sup>a "jump (ITER)" to the matrix *v*P projection of the affixal main predicate  $\partial uu-\dot{n}a\dot{k}uu\dot{h}$  "observe".

(56) Affixal main verb governing raising-to-object



The analysis treats these Nuu-chah-nulth main verbs as parallel to "Exceptional Case Marking" (ECM) verbs such as English "believe", which license accusative case assignment for the argument of an embedded infinitival.

(57)	a.	I believe <i>him</i> to be sleeping.	(ECM)
	b.	I believe that <i>he</i> is sleeping.	

Once we assume the raising-to-object analysis for this main verb, the "long" possessor raising facts directly fall out. Recall that "long" possessor raising is not possible with affixal main predicates, as indicated by (58b).<sup>9</sup>

(58)	a.		tuuxtuux <sup>w</sup> anakuuhitsiš	k <sup>w</sup> aa?uucukqs
			tuux <sup>w</sup> -a[+R]- <u><b>'nakuuḥ</b></u> -mit-siiš	kwaa?uuc-uk-qs
			jump-ITER- <u>observe</u> -PST-1SG.IND	grandchild-POSS-1SG.PS
			I observed my grandchild jumping.	(unraised)
	b.	*	tuux <sup>w</sup> tuux <sup>w</sup> anakuuhuk <sup>w</sup> itsiš	k <sup>w</sup> aa?uuc
			tuux <sup>w</sup> -a[+R]- <u><b>'nakuuh</b></u> -uk-mit-siiš	k <sup>w</sup> aa?uuc
			jump-ITER- <u>observe</u> - POSS-PST-1SG.	IND grandchild
			I observed my grandchild jumping.	('long' possessor raising)

Also recall from Chapters 3 and 4 that possessor raising is possible only out of subjects in Nuu-chah-nulth, not out of objects (Ravinski 2005). This is reflected in the interpretation of the possessor raised example of (59). Note that only the subject *piišpiš* "cat" is amenable to an interpretation as the possessum.

<sup>&</sup>lt;sup>9</sup> This restriction holds whether the embedded predicate is unaccusative (as with *ta?i*+ "sick" in 54b), or unergative (as with *tuux*<sup>w</sup>-"jump" in 58b).

(59) hink<sup>w</sup>a?iiḥuksiš piišpiš maamaati hin-k<sup>w</sup>a?iiḥ-uk-siiš piišpiš maamaati LOC-after-POSS-3.IND cat bird = (i) My cat was after a bird.  $\neq$  (ii) A cat was after my bird.

The unavailability of "long" possessor raising in the main verb environment is directly predicted by an analysis which states that the argument generated within the infinitival is behaving as an object, rather than a subject. Under a raising-to-object analysis, the promoted argument is predicted to be incompatible with possessor raising, since it fails to occupy a subject position. Possessor raising is inapplicable for objects in Nuu-chah-nulth. This analysis is represented in the diagram below. Here, only the subject of the matrix predicate is shown to allow possessor raising.

(60) Raising-to-object verbs are incompatible with "long" possessor raising



### 5.5 Conclusion

In summary, this chapter has argued that affixal predicates which take verbal complements fall into two general classes: affixal main predicates, and affixal auxiliary predicates. The two classes differ empirically in a variety of ways, including their ability to select a CP complement and to license a subject "mismatch". Moreover, only auxiliary affixal predicates permit "long" possessor raising. These behaviours were accounted for under an analysis in which main predicates are lexical verbs which govern raising-to-object, while auxiliary predicates are non-thematic functional verbs which govern raising-to-subject. I presented evidence for the infinitival nature of the complement of an affixal predicate in incorporation environments. The "clauselet" behaves as a monoclausal configuration which lacks clausal demarcation between the affixal predicate and its complement. According to the analysis, the complement of an affixal predicate which governs incorporation is a phrase smaller than a TP. Thus, in these reduced clause contexts, no CP "border" is imposed between an affixal predicate and its host. Opacity effects fail to apply to these reduced clauses because they are unsaturated domains.

This study expands the cross-linguistic typology of "restructuring" effects. The manifestation of these effects differs considerably across languages, although they share the common property of a lack of clausal demarcation between matrix and embedded constituents (Wurmbrand 2001; Cinque 2001). For Romance languages, primary examples of "restructuring" effects are clitic-climbing, long NP movement, and auxiliary selection (see Rizzi 1982), while Germanic "restructuring" behaviours include long object movement, long-distance scrambling, and verb raising (see Wurmbrand 2001). This study demonstrates that in Nuu-chah-nulth, "restructuring" effects include incorporation, "long" *wh*-movement and "long" possessor raising. These "restructuring" effects were previously unrecognized in the Wakashan literature.

# 6. Implications

We shall not cease from exploration and the end of all our exploring will be to arrive where we started and know the place for the first time. ~T.S. Eliot Four Quartets no. 4

### 6.0 Introduction

This concluding chapter first summarizes the effects of local spell-out in Nuu-chahnulth, and then moves on to discussion of key theoretical and typological implications of the analysis. The chapter closes with concluding remarks.

### 6.1 Summary

This book has proposed that the positioning of affixal predicates in Nuu-chah-nulth is a reflex of the need to linearize these bound morphemes. I have advanced an argument for the way in which syntactic objects are mapped to phonological linearization. According to the proposal, the linearization of syntactic constructs is established incrementally over the course of the derivation, in minimal domains equivalent to a single cycle of Merge. The local spell-out hypothesis determines that an affixal predicate finds a host from within its derivational sister. This derivational sister is a linearized object at spell-out, leading to the restriction that suffixation operates on the basis of string adjacency.

The consequence of local spell-out of affixal predicates in Nuu-chah-nulth is a phenomenon termed PF incorporation. Through PF incorporation, an affixal predicate suffixes to a host, in a process unselective for syntactic category, and insensitive to the syntactic structure of the derivational sister. The syntax and the linearization mechanism are not fully independent, however; syntax plays a conditioning role in the linearization process, through the composition of local spell-out domains. For affixal predicates which take nominal complements, I presented evidence that the argument structure of the affixal predicate constrains the choice of host for the affix. Whether an affixal predicate is unaccusative, extended unaccusative, transitive, or ditransitive, the position in which arguments are introduced has an influence on the pattern of incorporation. In each case, it is the argument which is introduced as the derivational sister to the affixal predicate which may serve as host. Higher arguments are ineligible as hosts.

#### (1) Nominal complements of affixal predicates



Moreover, the syntax was shown to condition spell-out of affixal predicates through the creation of "self-contained" derivational units. According to the saturated domain hypothesis, hosts for an affixal predicate are never selected from across a DP, because a DP is an independent phonological unit. Instead, in these contexts, the expletive  $\lambda u$ -is inserted to host the affix.

For affixal predicates which take verbal complements, a similar analysis for the conditioning effect of the syntax was given. I argued that the inventory of affixal predicates in Nuu-chah-nulth includes both main and auxiliary verbs, which permit incorporation of a host chosen from its complement. The incorporation pattern occurs with infinitival (*v*P) complements, and not with CP complements. By the saturated domain hypothesis, affixation is barred from crossing a CP.

### (2) Verbal complements of affixal predicates

a. Auxiliary predicate

```
b. Main predicate
```



Under the analysis I have proposed, spell-out to PF effects the ordering of syntactic elements when linearization is induced. This model is incompatible with a characterization of the derivational capacity of PF which states that PF amounts to a "second syntax" or to a "hybrid" branch sharing both syntactic and phonological sensitivities (cf. Embick and Noyer 2001; Rivero 1999). The properties I have

188

#### IMPLICATIONS

attributed to syntactic and PF operations are distinct. The syntax, under my analysis, plays a structure-building role in which linearization is irrelevant: syntactic constructs are inherently unordered. In contrast, at spell-out to PF, the structures of the syntax are inconsequential, while relationships based on string adjacency take on import. Under the present proposal, seemingly "syntactic" restrictions at PF are by-products of the role which syntax plays in conditioning the input to PF. The Nuu-chah-nulth facts indicate two ways in which syntax has a conditioning effect: the structure-building component is responsible for the creation of local spell-out domains (derivational sisters), as well as the creation of computationally independent opaque constituents.

(3) Conditioning effects of the syntax
 (i) Determining derivational sisterhood
 (ii) Creation of saturated domains

The first effect is responsible for the "complement" restriction on Nuu-chah-nulth incorporation. The second effect determines that incorporation is impossible in Nuu-chah-nulth across a DP or CP. In Chapter 2, I proposed that the opacity effects associated with DP and CP in Nuu-chah-nulth arise due to the derivational inertness which is induced upon syntactic saturation. In the discussion to follow, the nature of these opacity effects are further investigated, as the local spell-out model is contrasted with the phasal spell-out hypothesis of Chomsky (2001, 2005).

## 6.2 Theoretical Implications

This section presents three theoretical implications of the local spell-out hypothesis. In §6.2.1, I distinguish the local spell-out model from "phasal" spell-out (Chomsky 2001, 2005). Next, §6.2.2 discusses how PF incorporation in Nuu-chah-nulth fits in to the broader question of the locus of head movement operations in the grammar. Finally, §6.2.3 illustrates how the linearization effected by local spell-out necessarily induces an "outside-in" dependency.

# 6.2.1 Spell-out is Strictly Interpretive

According to the local spell-out hypothesis, the spell-out mechanism applies iteratively over the derivation, directing the outputs of syntactic concatenation to interpretation at the LF and PF interfaces. A characteristic of this model is that the mapping mechanism of spell-out does not remove the interpreted elements from the derivation. As first discussed in Chapter 1, the Order Preservation hypothesis (Fox and Pesetsky 2005) states that information established by spell-out is not deleted over the course of the derivation. As was shown in Chapter 2, the proposed iterative spell-out model allows a build-up of affixes, which occurs when one affix finds a host at spell-out, and this affix-host complex in turn serves as a host for another affixal at a subsequent application of spell-out. In (4), if  $-\alpha$  and  $-\theta$  are each suffixes, then a linearization of  $\phi$ - $\theta$  is induced in a first round of local spell-out. Next, when  $-\alpha$  is spelled-out, its host is this complex, yielding a linearization of  $\phi$ - $\theta$ - $\alpha$ .



If spell-out removed an affix and its host from the derivation, then such strings of affixes would be predicted to be impossible with iterative spell-out. That is, if spell-out occurred at  $\beta$ , and if this spell-out removed the contents of  $\beta$  from the derivation, then  $-\alpha$  and  $-\theta$  would be predicted to belong to distinct phonological domains. Instead, I hypothesize that spell-out enriches the derivation by establishing cumulative linearization (see also Fox and Pesetsky 2005). By this view, interpretation at the interfaces and removal from the derivation are necessarily distinct: spell-out is strictly interpretive.

This division of labour contrasts with the conflation of the interpretive and removal functions of spell-out in Chomsky (2001, 2005). According to Chomsky's dual-function model of spell-out, portions of the derivation are interpreted as distinct "phases", concomitant with their removal from the active derivation. In what follows, the local spell-out model is compared to phasal spell-out, noting conceptual and empirical shortcomings of the phasal model.

#### 6.2.1.1 Comparison to "Phasal" Spell-out

Chomsky (2001, 2005) proposes a model of multiple spell-out in which chunks of syntactic derivations are mapped cyclically to the interfaces as "phases". Spell-out occurs at designated points of the derivation, triggered by the presence of a "phasal head". Once a phasal head is reached in the derivation, the sister to this head is spelled-out. In the illustration in (5), C and D are represented as phasal heads.<sup>1</sup>



This model of cyclic spell-out partitions the derivation into discrete units. Within each phase, Chomsky (2005: 9) hypothesizes that all syntactic movement is driven

190

<sup>&</sup>lt;sup>1</sup> We set aside here the issue of v, which is also taken to be a phasal head (Chomsky 2001, 2005). See §6.2.1.3 for discussion of the vP domain.

by the phase head. Memory is also phase-bound, since information that is spelledout is inaccessible to later phases. This opacity effect is imposed by *Phase Impenetrability* (Chomsky 2001), which may be defined as below.

#### (6) *Phase impenetrability*

If XP is spelled-out, the sister to X may not be accessed later in the derivation.

The diagram in (7) illustrates how C(P) imposes a boundary between two phases of the derivation.



In (7), the phasal head C remains active in the derivation, while the sister to C - TP – is made inaccessible to operations after spell-out. According to this analysis, the spell-out node of C(P) is an "escape hatch" for movement in that it forms a potential link for movement across the phasal boundary: elements within the CP phase may move to C(P) to carry on in the derivation. Thus, long-distance movement, as in (8), proceeds in cycles by accessing the "escape hatch" at CP.

### (8) Who<sub>i</sub> do you think $[_{CP} t_i [we can trust t_i]?$

The *wh*-word, which originates in the embedded clause as object of the verb, makes an intermediate step to the "escape hatch" in the embedded CP before remerging in the matrix CP.

Before turning to the empirical adequacy of the Phase Impenetrability hypothesis, we first ask the following question: is this stipulation necessary, or does opacity follow from a primitive of the computation? According to the Phase Impenetrability hypothesis, opacity is inherently tied to spell-out itself: once a portion of the derivation is spelled-out, it is no longer retrievable. Under the phasal spell-out model, the source of opacity effects is therefore linked to the designation of spell-out nodes themselves. By this model, only spell-out nodes trigger opacity. However, it remains to be shown under this model what forces the special status of phasal heads as spell-out triggers. The existence of a limited set of spell-out triggers is left as a stipulation. Motivation for the existence of opacity effects is therefore a conceptual shortcoming of this model, to the extent that the choice of spell-out triggers is unexplained.

In contrast, according to the local spell-out model, phrases such as DP and CP are not special spell-out units: every instance of syntactic Merge induces spellout. By this assumption, the source of opacity effects therefore cannot be in spell-out itself. Under the local spell-out model, spell-out plays a purely interpretive role, as it fails to induce removal from the derivation. In Chapter 2, the saturated domain hypothesis was advanced regarding the source of opacity effects.

## (9) Saturated domain hypothesis: the contents of DP and CP are fully interpreted

Under the proposal, the inertness of the contents of DP and CP follows from their saturation. This inertness derives from economy in that only unsaturated portions of the derivation command attention; saturated portions of the derivation are neglected. What sets D and C apart as "borders" of the derivation may ultimately reduce to the role which these heads play in driving movement: Chomsky (2005: 9) hypothesizes that all syntactic movement is forced by these heads ("phasal heads") which mark the edge of an opaque domain. Thus, introduction of a node such as C allows for syntactic saturation of the sister to C. Under the saturated domain hypothesis, opacity effects thus come naturally from the same derivational primitives which drive interpretation and movement.

It is also worth noting an empirical flaw of the Phase Impenetrability hypothesis as applied to Nuu-chah-nulth. Assuming that the complementizer 2in (COMP) in Nuu-chah-nulth occupies C, the Phase Impenetrability hypothesis is empirically inadequate in capturing the spell-out properties of affixal predicates. According to this hypothesis, the head of a spell-out domain should be available to the higher phase. For example, in (7), C will attain spell-out along with V when the next phase is reached. This predicate behaviour does not occur in Nuu-chah-nulth. When an affixal predicate takes a CP complement, the complementizer is not available as a host to the affixal predicate. In (10a), it is shown to be ungrammatical for 2in (COMP) to host the affixal predicate -nakuuh "observe". Instead, when the affixal predicate takes a CP complement, the expletive 2u- mandatorily hosts the affix, as in (10b).

(10)	a.	*	?iinnakuuhitsiš ?in- <u>nakuuh</u> [+L]-mit-siiš COMP- <u>observe</u> -PST-1SG.IND I observed that you were jumping.	tuuxtuu tuux <sup>w</sup> -a jump-IT	x <sup>w</sup> amitsuk [+R]-mit-suuk `-PST-2SG.DEP
	b.		?uunakuuhitsiš ?u- <u>nakuuh[</u> +L]-mit-siiš Ø- <u>observe</u> -PST-1SG.IND I observed that you were jumping.	?in ?in COMP	tuuxtuux <sup>w</sup> amitsuk tuux <sup>w</sup> -a[+R]-mit-suuk jump-IT-PST-2SG.DEP

This behaviour is unexpected if the affixal predicate and the complementizer were spelled-out together. Thus, the technical implementation of the Phase Impenetrability hypothesis for CPs is problematic for Nuu-chah-nulth. This problem does not arise under the saturated domain hypothesis, since by this analysis, CP is set apart as a distinct domain of the derivation. (11) a. Saturated domain hypothesis b. Phase Impenetrability hypothesis



Let us now turn to discussion of cross-linguistic variation in "escape hatches" at the borders of opaque domains.

#### 6.2.1.2 Variation in Permeability of CP Domains

The previous chapters illustrated a range of cases in Nuu-chah-nulth in which CPs and DPs form impermeable domains. In the context of affixation, when an affixal predicate takes a CP or DP as its sister, a host for the affixal predicate may not come from these saturated domains. As first noted in Chapter 2, a similar constraint exists on syntactic movement in the language. *Wh*-movement in Nuu-chah-nulth is barred from crossing a CP (Davis and Sawai 2001), as indicated in (12). A *wh*-question formed via extraction from an embedded clause, as in (12b, c), is ungrammatical.

y čapac
y čapac
y canoe
2001:133)
2

- čapac b. \* ?ačaqh ťaaguk<sup>w</sup>i?a<del>λ</del> John ?in kuuwi1-ithuk ?ačaq-h ťaaquk-či<del>x</del>-'a<del>x</del> John kuuwił-mit-huuk čapac ?in who-3.0 believe-PERF-TEMP COMP steal-PST-3.DEP John canoe Who does John believe stole the canoe? (cf. Davis and Sawai 2001: 133)
- c. \* ?ačaqh ťaaquk<sup>w</sup>i?a<sup>\*</sup> John (?in) kuuŵiłith čapac ?ačaq-h ťaaquk-či<sup>\*</sup>/a<sup>\*</sup> John (?in) kuuŵił-mit-h čapac who-3.Q believe-PERF-TEMP John (COMP) steal-PST-3.Q canoe Who does John believe stole the canoe?

This restriction on cross-domain movement in Nuu-chah-nulth stands in contrast to those languages which are known to allow long-range movement. In English, for example, an "escape hatch" at CP permits movement from one clause to another.

### (13) Who<sub>i</sub> do you think $[_{CP} t_i [we can trust t_i]?$

What ensures that a CP border is tightly closed in Nuu-chah-nulth – barring longrange movement – while a more permeable edge is allowed in those languages which permit cross-clausal movement?

I suggest that the possibility of cross-clausal extraction reduces to incomplete interpretation of CPs – a lack of saturation. In the case of long-range movement, an

"escape hatch" is created if and only if the CP carries a [wh] feature which remains unsaturated at this border. This permeable border is represented in (14).

#### (14) Unsaturated [wh] feature at a CP border



By hypothesis, the lack of saturation which is inherent to (14) leaves the edge of the CP domain "active" in the derivation. In contrast, the edge is rendered "inert" if the [wh] feature is saturated at the border – barring any cross-clausal extraction from this domain. Two types of fully interpreted CPs are illustrated in (15).

#### (15) Saturated [wh] features at a CP border



One means by which an inert edge may arise is in contexts of "selectional saturation", in which a verb, such as English *wonder*, is lexically specified for an interrogative complement. In (15a), the lexically specified [wh] feature of the verb matches the [wh] feature of the interrogative complement, allowing for saturation of the embedded [wh]. The complements of these inherently interrogative verbs are known to be islands for movement, as indicated by the ungrammaticality of (16b). By hypothesis, the opacity of these lexically selected interrogative CPs follows from their complete interpretation – as fully saturated domains, their edges are inert. The ban on cross-clausal movement from the interrogative complement of a lexically marked [wh] verb stands in contrast to the cross-clausal movement permitted when the verb is not inherently interrogative, such as English "think" in (17). Questions such as (17b), which show no lexical selection for a [wh] complement, match the configuration in (14). Cross-clausal movement is permitted via the "escape hatch" which arises due to the incomplete interpretation of the CP.

(16)	a.	I wonder whether John saw Mary.
	b.	* Who do you wonder whether John saw?
(17)	a.	I think John saw Mary.
	b.	Who do you think John saw?

#### IMPLICATIONS

I hypothesize that the featural composition of the CP may also be saturated intraclausally (15b), rather than via lexical saturation (15a). The inert status of CPs in Nuu-chah-nulth is proposed to arise from the intraclausal saturation illustrated in (15b). According to the proposal, the [*wh*] feature of an interrogative clause in Nuu-chah-nulth need not look outside its domain for saturation – a possibility which I suggest may be tied to the obligatory interrogative "mood" marking within the clause. As first described in Chapter 3, clause types in Nuu-chah-nulth are marked in a clause via fusional "mood" enclitics which simultaneously encode subject agreement. Interrogative mood is one paradigm of inflectional endings, as are moods such as indicative, relative, or conditional. A partial mood paradigm is shown in (18); see §3.2.1 for a complete set of mood distinctions.

Mood/Person	lsg	2sg	3	1pl	2pl
INDICATIVE	-siiš	-?iick	-?iiš	-niiš	-?iicuuš
INTERROGATIVE	-ḥs	-k	-ḥ	-ḥin	-ḥsuu
SIMPLE RELATIVE	-qs	-?iitk	-?iitq	-qin	-?iitqsuu
CONDITIONAL	-quus	-quuk	-quu	-q <sup>w</sup> in	-quusuu

## (18) *Mood/person distinctions (partial set)*

A clause containing a *wh*-word in Nuu-chah-nulth is obligatorily marked for interrogative "mood". In (19a), the portmanteau "mood" marker -h simultaneously marks interrogative clause type and third person. It is ungrammatical for alternative "moods" to occur in a sentence with a *wh*-word, as is indicated by the ungrammatical presence of the third person indicative mood ending  $-\lambda i i i$  in (19b).

(19)	a.		?aqi?amitḥ	John
			?aqi- <u>?aap</u> -mit-ḥ	John
			what- <b>buy</b> -PST-3.Q	John
			What did John buy?	(Davis and Sawai 2001: 123, ex. 1)
	b.	*	?aqi?amit?iš	John
			?aqi- <b>?aap</b> -mit-?iiš	John
			what- <b>buy</b> -PST-3.IND	John
			What did John buy?	

For an interrogative clause in Nuu-chah-nulth, I hypothesize that the interrogative agreement saturates the [wh] feature of a *wh*-word. This is illustrated in (20), where the third person interrogative morpheme is shown housed in C<sup>2</sup>:

 $<sup>^2</sup>$  Other analyses of the location of this fusional interrogative marker within CP (such as Agr) are compatible with the intraclausal saturation analysis. However, interrogative "mood" endings do not co-occur with the complementizer 2in- this complementary distribution would follow if they occupied the same C slot.



By this mechanism, the features of the CP domain are fully saturated. The complete interpretation of the [wh] features within CP in Nuu-chah-nulth renders the domain ineligible for cross-clausal extraction. If an interrogative clause is necessarily marked with interrogative mood in Nuu-chah-nulth, then saturation of the [wh] feature in a *wh*-word is induced within the CP – preventing the creation of an "escape hatch" in cross-clausal contexts.

## 6.2.1.3 The Status of the vP Domain

Although CP and DP pattern as opaque domains in Nuu-chah-nulth, a distinct set of properties holds for the vP domain. In Nuu-chah-nulth, vP fails to exhibit opacity effects similar to that of CP and DP, as might be anticipated under analyses such as Chomsky (2001, 2005) which posit v as a privileged head on par with C or D. The vP domain is transparent for affixation in Nuu-chah-nulth: there is a cumulative build-up of affixes and hosts in Nuu-chah-nulth which is not halted at the vP "border". As illustrated in Chapter 3, an affixal predicate and its host forms a single word together with a range of affixal morphemes whose syntactic origin is above the vP, including tense and agreement clitics. In (21), for example, the affixal predicate -int"serve" and its host *čamas* "sweets" forms a word with the tense morpheme -mit (PST) and the agreement morpheme -siiš (1SG.IND).

(21) čamayintitsiš čamas-<u>'int</u>-mit-siiš sweets-<u>serve</u>-PST-1SG.IND I served sweets.

#### IMPLICATIONS

Such examples are distinct from opacity effects involving CP or DP, as it has been shown that a host for an affixal predicate cannot cross either of these saturated domains. Likewise, a related problem for the notion that vP constitutes an opaque domain in Nuu-chah-nulth is the possibility of affixation across a vP phrase. When an affixal predicate takes a verbal complement, a host for the affixal predicate is chosen from the complement, as discussed in Chapter 5. In (22), the affixal predicate -*Sir* "come upon" suffixes to its host *hacuk* "deeply".

(22)	ḥacukʕi⊁itsiš	wa?ič	Ken
	ḥacuk- <u><b>Siૠ</b></u> -mit-siiš	wa?ič	Ken
	deeply- <u>come.upon</u> -PST-1SG.IND	sleep	Ken
	I came upon Ken in a deep sleep.	_	

Syntactically, this complex may be represented as in (23), in which the predicate  $\lambda u$ -*li* $\lambda$  "come upon" belongs to a different verbal projection than *hacuk* "deeply".



If vP imposed an impermeable border between the matrix affixal predicate and its complement, affixation would not be predicted between  $-Si\lambda$  "come upon" and its attested host, *hacuk* "deeply".

Despite this evidence that vP constitutes a transparent domain for affixation, there is nonetheless suggestive evidence that the vP may contain a unique phonological domain in Nuu-chah-nulth. The inner layer of suffixes in a Nuu-chah-nulth word – namely, affixal predicates and aspectual suffixes – display phonological effects which can be contrasted to that of the outer suffixes.

(24) "Inner" and "outer" layers of suffixes in Nuu-chah-nulth

[=PREDICATE=ASP] =TR=TEMP=PAS=POSS=TENSE=AGR/MOOD=PL=AGAIN=HAB

"core" suffixes

(23)

"peripheral" suffixes

Phonological processes in Nuu-chah-nulth are traditionally classified into two classes or strata: those which apply to "core" or "stem"-level suffixes, and those phenomena which occur peripherally at "word" level (see Sapir and Swadesh 1939; Nakayama 2001; Stonham 2007, among others).<sup>3</sup> In Chapter 2, it was noted that affixal predicates and aspectual suffixes trigger non-local effects on their host morpheme, conditioning vowel length and reduplication on the initial syllable of a host (see §2.1). In contrast, the "peripheral" inflectional suffixes, such as tense markers or agreement morphemes, never induce reduplication or condition vowel length of a host. As Nakayama (2001: 17) notes, any phonological change caused by "peripheral" suffixes is limited to the host segment directly adjacent to the suffix. One such local change is a consonant mutation or "hardening" effect which induces glottalization of a neighbouring consonant. For example, in (25), the "glottalizing" temporal suffix -  $a \star$  (TEMP) causes the final stop in *Sičaak* "rotten" to surface as a glottalized [k'].

(25) Sičaakaλ?iš
 Sičaak-'aλ-?iš
 rotten-TEMP-3.IND
 They're rotten.

(See Sapir and Swadesh 1939; Nakayama 2001; Kim 2003, among others, for detailed discussion of glottalization and other local effects). Although both "core" and "peripheral" suffixes display these local glottalizing effects, the two classes of suffixes differ in their range of targets – while "core" suffixes glottalize a neighbouring stop, affricate, or fricative, the "peripheral" suffixes affect only the first two classes, failing to mutate an adjacent fricative. The differing treatment of the lateral fricative [ $\frac{1}{2}$ ] is shown in (26). In (26a), the fricative is adjacent to the glottalizing affixal predicate –  $\frac{a}{hs}$  "in a vessel", and is realized as the glottalized glide [ $\frac{v}{2}$ ]. In (26b), the lateral fricative fails to undergo mutation when it occurs with the glottalizing temporal suffix –  $\frac{a}{k}$  (TEMP), a "peripheral" suffix.

(26)	a.	hiỷaḥs hiᆉ-'aḥs-∅ LOC-in.vessel 'inside a canoe'	(Nakayama 2001: 16, ex. 35)
	b.	hił?aૠ hił-'aૠ–Ø LOC-TEMP-3.ABS 'He was there.'	(Nakayama 2001: 16, ex. 36)

Nakayama (2001: 16) summarizes this phonological difference between "core" and "peripheral" suffixes by noting that "the effect on the stem triggered by the core suffixes is stronger than that triggered by peripheral suffixes."

<sup>&</sup>lt;sup>3</sup> Sapir and Swadesh (1939) refer to "core" and "peripheral" suffixes as "formative" and "incremental" suffixes, respectively.

#### IMPLICATIONS

Additional research is required to determine whether this traditional distinction between "core" and "peripheral" suffixes may be recast as corresponding to the boundary between intra-vP and extra-vP projections. Under the present analysis, (non-auxiliary) affixal predicates and aspectual markers are contained within the vP, while the "outer" suffixes are assumed to be housed in higher projections.<sup>4</sup> This clausal demarcation came into play in Chapter 5, in which it was shown that only the inner layer of suffixes are included together with a verb when the verb is "incorporated" into an affixal predicate. The example in (27) illustrates "verb incorporation" of the verbal complex wat-sit" "go home (PERF) into the affixal predicate –mahsa "want to". While wat- "go home" may be suffixed by the aspectual suffix -sit (PERF), it may not be suffixed by peripheral suffixes, including the tense morpheme - $iaq\dot{x}$ (FUT).

(27) wałśiλ(\*?aqλ)maḥsasiš
 wał-šiλ-(\*?aqλ)-<u>maḥsa</u>-siiš
 go.home-PERF-(FUT)-<u>want.to</u>-1SG.IND
 I want to go home.

Thus, there is suggestive evidence from both affixation and other phonological processes that the intra-vP suffixes in Nuu-chah-nulth form a more tightly integrated domain with their host than the suffixes which originate outside of vP. It remains to be shown how vP compares to the saturated domains of CP and DP, and whether the possibility of cumulative affixation may be attributed to the presence of an "escape hatch" at the border of a vP.

In sum, the Nuu-chah-nulth data constitute novel support for the derivational independence of CP and DP. Further investigation is necessary to determine the status of the vP domain. In the next section, an additional theoretical implication of the study is assessed.

#### 6.2.2 The Head Movement Question

In the wake of Chomsky's (1995, 2001) suggestion that head movement may be best analysed as a PF phenomenon, much attention has been paid to theoretical motivations for eliminating this operation from the narrow syntax. A well-noted objection to a standard syntactic treatment of head movement (e.g., Baker 1988; Chomsky 1993) is its countercyclic violation of the Extension Condition, in which head movement builds the tree at a non-root node (Chomsky 2001, among others). According to a traditional syntactic treatment of head movement, a head X is moved to adjoin to an immediately c-commanding node Y, as in (28a). In (28a), the Y node at which structure-building is effected is not the root node. In cyclic movement, structurebuilding occurs at the root, as in (28b).

<sup>&</sup>lt;sup>4</sup> More research is needed on the status of auxiliary affixal predicates. In Chapter 5, it was proposed that these predicates were housed in a functional projection located directly above vP.



Countercyclicity entails that the moved element does not c-command its "trace", under standard definitions of c-command. An additional problematic aspect of a traditional analysis of syntactic head movement is the stipulative nature of the Uniformity Condition on the phrasal status of chains (Surányi 2003; Matushansky 2006).

# (29) Uniformity condition (Chomsky 1995: 253) A chain is uniform with regard to phrase structure status

However, a variety of analyses have indicated that these problems posed by head movement within a Minimalist syntax may be reconciled through an adjustment in the syntactic machinery of head movement (Donati 2003; Surányi 2003; Matushansky 2006). For example, Matushansky (2006) argues that phrasal movement and head movement have identical landing sites, in which both extend the projection at the root. By this analysis, the issue of apparent countercyclicity evaporates.

The residue of this theoretical debate on the admissibility of head movement as a syntactic operation is the question of empirical support for the notion that head movement occurs in PF (Boeckx and Stjepanovic 2001). In this light, the empirical basis for diagnosing the symptoms of a PF operation takes on considerable theoretical import (see also Rivero 1999; Embick and Noyer 2001). Specifically, if head movement is a PF operation, then what features should it be expected to have? This examination of incorporation data from Nuu-chah-nulth expands the empirical coverage of the "head movement question". As I have shown, incorporation in Nuuchah-nulth has the hallmark properties of a post-syntactic, PF process. The phenomenon of PF incorporation in Nuu-chah-nulth demonstrates adherence to string adjacency, and shows an insensitivity to syntactic category and constituency, and as well as an absence of LF effects. Moreover, the opacity effects associated with DP and CP constituents in Nuu-chah-nulth provide novel support for the computational independence of these phrases at a phonological level (cf. Chomsky 2001; Matushansky 2006).

In what follows, I make explicit the claim that while the "head movement" operation of affixal predicates in Nuu-chah-nulth is strictly phonological, other types of head movement show contrasting properties. In §6.2.2.1, I summarize the argument that the linearization mechanism for Nuu-chah-nulth affixes is achieved at spell-out to PF. In §6.2.2.2, this linearization mechanism is contrasted with syntactic head movement.

#### 6.2.2.1 Linearization is Purely Phonological

In a strict sense, the placement of affixal predicates in Nuu-chah-nulth is purely phonological. Although syntactic composition conditions the input to the linearization mechanism at local spell-out, the syntax plays no role in the means by which an affix is attached with its host. As I have argued, an affixal predicate suffixes to its host as a reflex of the spell-out mechanism. For a syntactic construct such as (30), the suffix  $-\alpha$  is the linearized object  $\beta-\alpha$ .



A characteristic of the linearization of Nuu-chah-nulth affixal predicates is that it is conditioned by the "base" position of affixes and their arguments, via aggressive interpretation achieved through local spell-out. That is, as soon as an affixal predicate is merged into the derivation, its linearization is determined: the affixal predicate is hosted by a string adjacent element from its derivational sister. Under the simplest assumptions, it follows from this analysis that subsequent syntactic movement in the derivation, of either the affix or its derivational sister, should not affect this relationship between the affix and its host. This prediction is upheld in Nuu-chah-nulth. For example, when the nominal object of an affixal predicate is passivized, the "complement" effect in affixation remains. Whether an affixal predicate is active or passive, the affix suffixes to a host from its derivational sister, its complement.<sup>5</sup> The examples in (31) show this pattern for the transitive affixal predicate *Pu-Piic* "consume". The active sentence in (31a) shows the affixal predicate suffixing to the nominal  $k^{waq}$  "spawned herring eggs", which the predicate takes as its syntactic object (as argued in Chapter 4). In (31b), the predicate has been passivized: this is indicated morphologically by the addition of the suffix -at (PAS). What is relevant is that passivization has no consequence for the choice of host for the affixal predicate. In (31b), the host for the affixal predicate is  $k^{w}aq$  "spawned herring eggs", just as it is in the active case of (31a).

(31)	a.	k <sup>w</sup> aʕiicʔiš k <sup>w</sup> aq- <u>'<b>iic</b></u> -ʔiiš s.h.eɛgs <b>-consume</b> -3.IND	k <sup>w</sup> aa?uuc k <sup>w</sup> aa?uuc grandchild	
		Grandchild is eating spawned herri	ng eggs.	(active)
	b.	k <sup>w</sup> aʕiicck <sup>w</sup> atʔiš k <sup>w</sup> aq- <u>'iic</u> -ck <sup>w</sup> i-'at-ʔiiš s.h.eggs- <u>consume</u> -EVID-PAS-3.IND Spawned herring eggs must have b	?uušḥ?at ?uuš-qḥ-'at some-do.by-PAS een eaten by some	one. (passive)

<sup>&</sup>lt;sup>5</sup> This claim directly contradicts Waldie's (2004) assertion that affixal predicates with nominal hosts cannot be passivized in the Ahousaht dialect of Nuu-chah-nulth. The Ahousaht speakers who I have worked with consistently allow passivization in appropriate discourse contexts, with semantically appropriate predicates.

According to a syntactic treatment of Nuu-chah-nulth passivization (Kim 2001), the object of the passivized predicate is promoted to a surface subject position. In (32), the passivized object k''aq "spawned herring eggs" moves to specifier position of the projection headed by – *at* (PAS).



Crucially, the syntactic movement of  $k^{waq}$  does not affect its spell-out position hosting the affixal predicate.

Furthermore, the relationship between an affixal predicate and its host is not affected by other types of word order manipulations which disrupt the surface adjacency between the predicate and its syntactic complement. This effect may be observed in contexts in which the complement of the affixal predicate contains multiple words, such as when the object contains a modifier. In the following examples, the modified nominal  $\dot{cus}(uk)$   $\dot{suwis}$  "new shoes" is the complement of the affixal predicate 2u2u- $\dot{ciga}$  "lose". As shown in the examples in (33), in which the affixal predicate is hosted by the expletive 2u-, the subject *Louis* and the object  $\dot{cusuk}$   $\dot{suwis}$  "new shoes" show a variable word order. (As described in Chapter 3, subjects and objects often show variable word order when the object is inanimate and indefinite.) The crucial examples are given in (34). In (34), the affixal predicate 2u2u- $\dot{ciga}$  "lose" is hosted by  $\dot{cus}$  "new", which originates syntactically in object position as the modifier of the nominal  $\dot{suwis}$  "shoes". In (34a), the subject *Louis* 

(33)	a.	?u?učiqa?iš	Louis	ćušuk		šuwis
		?u- <b>ċiqa</b> [+R]-?iiš	Louis	ċuš-uk		šuwis
		Ø- <u>lose</u> -3.ind	Louis	new-DUI	R	shoes
		Louis lost a brand new pai	r of shoe	s.		
	b.	?u?uciqa?iš	ċušuk		šuwis	Louis
		?u- <u><b>čiqa</b>[</u> +R]-?iiš	ċuš-uk		šuwis	Louis
		Ø- <u>lose</u> -3.ind	new-DU	R	shoes	Louis
		Louis lost a brand new pai	r of shoe	s.		
(34)	a.	ćućušćiga?iš	Louis	šuwis		
<b>`</b>		ćuš- <b>ćiqa</b> [+R]-?iiš	Louis	šuwis		
		new-lose-3.IND	Louis	shoes		
		Louis lost a brand new pai	r of shoe	s.		

(32)

b.	ċuċušċiqa?iš	šuwis	Louis
	ċuš- <u>ċiqa</u> [+R]-?iiš	šuwis	Louis
	new- <u>lose</u> -3.IND	shoes	Louis
	Louis lost a brand new pair of shoes.		

What the example in (34a) suggests is that surface position of the predicate and its object is irrelevant for the affixation mechanism. Instead, affixation is determined at the earliest possible stage of the derivation, with the object adjacent to the predicate in complement position. For the syntactic structure in (35), local spell-out determines that the affixal predicate  $-\dot{ciqa}$  "lose" selects  $\dot{cus}$  "new" as its host because of the string adjacency the two share.



When movement derives an object-final order, this affixation relationship is unaffected. In the diagram in (35), the object-final word order of (34a) is shown to be derived by movement of the object to a right-linearized specifier position. Under this analysis, I assume that it is the bottom "copy" of the moved element  $\dot{c}u\ddot{s}$  "new" which is pronounced in Nuu-chah-nulth. Note that it is otherwise impossible to separate the modifier  $\dot{c}u\ddot{s}(uk)$  "new (DUR)" from the nominal  $\dot{s}uwis$  "shoes", as shown in (36).

(36) *	?u?ućiqa?iš	ćušuk	Louis	šuwis
	?u- <b>ćiqa</b> [+R]-?iiš	ċuš-uk	Louis	šuwis
	Ø- <u>lose</u> -3.ind	new-DUR	Louis	shoes
	Louis lost a brand new pair of shoes.			

The ungrammaticality of such an example suggests that *cus(uk) suwis* "new shoes" moves only as a unitary constituent in Nuu-chah-nulth, lending support to the analysis in (35) that it is the NP which moves rightward, and not simply the nominal head *suwis* "shoes".

In summary, syntactic movement appears to have no consequence for the linearization mechanism of affixal predicates in Nuu-chah-nulth. The host for an affixal predicate is determined by local spell-out of the affix at the stage in the derivation in which it is first introduced. Linearization of the affix and its host is not disturbed by subsequent syntactic movement of the syntactic category containing the host. Thus, syntactic manipulations play no role in the linearization of Nuu-chahnulth affixal predicates, beyond the irreducible conditioning effect of the syntax in forming the local spell-out domain of the affixal predicate.

In the next section, I argue that the linearization mechanism of Nuu-chahnulth affixal predicates may be contrasted with syntactic head movement, which is achieved by an interplay between syntactic movement and spell-out.

#### 6.2.2.2 Syntactic Head Movement is More Complex

As described in Chapter 2, PF incorporation in Nuu-chah-nulth is sensitive to linear adjacency: an affixal predicate suffixes to a string adjacent element from its derivational sister. In this, Nuu-chah-nulth differs from those languages in which incorporation operates on the basis of hierarchical adjacency. As noted in Chapter 2, the contrast between sensitivity to linear adjacency and hierarchical adjacency is readily observed in environments in which the complement of the "incorporating" verb is modified. With PF incorporation, it is a linearly adjacent modifier which is targeted for suffixation by an affixal predicate. In (37), the affixal predicate *Au-Dic* "consume" suffixes to the modifier *haAum* "tasty", stranding the nominal head *Taapinis* "apple".

(37)	ha?um?ic?iš?a+	?aapinis
	ha?um- <u><b>'iic</b></u> -?iiš-?a†	?aapinis
	tasty- <u>consume</u> -3.IND-PL	apples
	They are eating delicious apples.	

In contrast, in syntactic incorporation, a modifier is necessarily inaccessible for incorporation (Baker 2003). As argued in Chapter 2, the prohibition on syntactic adjective incorporation is a consequence of the Head Movement Constraint, which determines that movement targets the head of a complement.

With respect to adjective incorporation, Nuu-chah-nulth may be contrasted with Mohawk (Iroquoian). As Mithun (1984: 870) describes, when a verb in Mohawk suffixes to a noun, it may strand a modifier.<sup>6</sup> In (28), the verb *ú:ni* "make" suffixes to *-akya'tawi'tsher-* "dress", stranding the modifier *kanekwarúnyu* "polka-dotted".

(38) *Mohawk* (Mithun 1984: 870, ex. 106)

kanekwarúnyu wa'-k-akya'tawi'tsher-ú:ni it.dotted.DIST PST-I-dress-make I made a polka-dotted dress.

In what follows, I suggest that the difference between Nuu-chah-nulth and Mohawk amounts to whether affixation is achieved through local spell-out, or via a more complex derivation.

<sup>&</sup>lt;sup>6</sup> Mithun (1984) explicitly argues against a syntactic treatment of noun incorporation. However, I follow Baker (1988, 2003) in assuming that Mohawk incorporation is syntactic.

#### IMPLICATIONS

This section hypothesizes that syntactic head movement arises through an interaction between syntax and local spell-out. According to this notion, syntactic head movement reflects the spell-out of a head not in its base position, but in the position it has attained via a local form of syntactic movement. This falls in line with other work which proposes that syntactic head movement is a multi-step process with individual components of syntactic movement and affixation (Donati 2003; Matushansky 2006).

I adopt Matushansky's (2006) proposal that head movement is a cyclic operation which remerges a head at the root node. For Matushansky (2006), the prerequisite for this local movement is an AGREE relation equivalent to c-selection. In (39a), the head X selects the category Y as its complement. In (39b), Y is (re)Merged at the root node X(P).



As Matushansky argues, in syntactic head movement, the movement in (39b) is followed up by an affixation process which attaches the moved head Y to the head X.

A syntactic analysis of Mohawk incorporation (Baker 1988) may thus be implemented within the Minimalist framework in which remerge is effected at the root node. This analysis translates into the following movement operation for noun incorporation which strands a modifier.

#### (40) Syntactic incorporation in Mohawk



Here, the noun *akya'tawi'tsher*- "dress" is (re)Merged at the root node, as a projection of V(P).

How does the movement operation in (40) result in suffixation of the verb to the noun? I suggest that the affixation reflex for this head movement operation relates to the bound status of Iroquoian nominals. With few exceptions, nominals in Iroquoian languages are obligatorily morphologically bound, and cannot occur independently (Bryan Gick, Marianne Mithun, p.c.).<sup>7</sup> Let us therefore assume that *akya'tawi'tsher-* "dress" is subject to an affixation requirement. On the assumption that a host for this affixal noun must be a linearly adjacent element from its derivational sister, then the verb  $\hat{u}:ni$  "make" will be selected as host due to its position as leftmost element of the derivational sister. The choice of  $\hat{u}:ni$  "make" as host of *akya'tawi'tsher-* "dress" would therefore immediately follow. Thus, the linearization of *akya'tawi'tsher-\hat{u}:ni* "dress-make" results.

In the next section, we turn to an inherent restriction on the affixation relations created by local spell-out.

### 6.2.3 Local Spell-out Creates "Outside-in" Dependencies

According to the local spell-out hypothesis, the relationship between a Nuu-chahnulth affixal predicate and its host is necessarily "outside-in".<sup>8</sup> A suffix ( $-\alpha$ ) is always "outside", or higher than, the syntactic terminals contained within its derivational sister ( $\beta$ ).

(41)



In Nuu-chah-nulth, an affixal predicate  $(-\alpha)$  must be linearized with respect to a host from its derivational sister  $\beta$ . If its derivational sister is linearized as  $\langle \delta, \pi \rangle$ , then the host for an affixal predicate is determined to be  $\delta$ , due to its string adjacency to this morpheme ( $\langle (\delta - \alpha), \pi \rangle$ ).

What about "inside-out" dependencies? In this configuration, an affix "climbs up" to find a host higher in the tree. In the illustration below, let us take  $-\alpha$  to be a suffix, and  $\delta$  to be its host. In an "inside-out" dependency, an affix  $-\alpha$  attaches to the host  $\delta$ , even though  $\delta$  is not contained within its derivational sister. In the tree below,  $\pi$  is the derivational sister of  $\alpha$ , not  $\delta$ .

<sup>&</sup>lt;sup>7</sup> I return to this topic in §6.3.1. Outside of incorporation contexts, noun stems in Mohawk are most commonly suffixed by -a', a "noun suffix" which contributes no discernible meaning (Marianne Mithun, p.c.).

<sup>&</sup>lt;sup>8</sup> Thanks to Gunnar Hansson for suggesting this term to me.



### 6.2.3.1 "Inside-out" Dependencies are not Possible via Local Spell-out

I hypothesize that the effect of an "inside-out" dependency will never arise from local spell-out. That is, "inside-out" affixation will never satisfy a lexical requirement of an affix. Instead, I suggest that this arrangement is necessarily achieved later in the derivation than what is possible by the "aggressive" linearization algorithm of local spell-out. "Inside-out" arrangements arise via prosody. This type of affixation is not local in the same sense as local spell-out, since it is done on full assemblies of derivations, rather than in the incremental steps taken by local spell-out. In the following section, I present the notion that "inside-out" linearizations apply in later stages of the derivation than local spell-out, due to prosodic requirements.

### 6.2.3.2 Evidence from Kwakw'ala (Northern Wakashan)

Evidence for the existence of "inside-out" dependencies comes from the Northern Wakashan language Kwakw'ala (Anderson 1984, 2005; Klavans 1985). As Anderson (1984) notes, in Kwakw'ala, determiners are enclitics, and they attach to the right edge of a *preceding* word. The following example is slightly modified from Anderson (1984: 21, ex. 1) to match the morpheme gloss conventions of this book.

#### (43) Kwakw'ala "inside-out" enclitic determiner

k<sup>w</sup>ix?id-ida bəg<sup>w</sup>anəma-x-a qasa-s-is təlwag<sup>w</sup>ayu clubbed-DET man-OBJ-DET otter-INST-POSS club The man clubbed the sea-otter with his club.

In this example, the determiner -ida (DET) is semantically construed as the marker for the nominal b g wan a ma "man". Yet instead of attaching to this nominal, it attaches leftward to the preceding word, the verb k wix 2id "clubbed". The same leftward pattern is also exhibited by the other determiner in the sentence, -a (DET). Although this determiner should be semantically construed with the object nominal  $\dot{q}asa$ "otter", it nonetheless attaches to the right edge of a different nominal, the subject nominal b g wan a ma "man", which happens to precede the object.

The Kwakw'ala example can be contrasted with the following example from Nuu-chah-nulth, which shows an "outside-in" dependency. In Nuu-chah-nulth, the determiner -2ii reliably suffixes to the right edge of the first word in its complement. (This distribution is identical to that of a Nuu-chah-nulth affixal predicate.) Here, -2ii suffixes to the nominal *huupuuk*<sup>w</sup>as "car".
(44) Nuu-chah-nulth "outside-in" enclitic determiner

kuuwiłit?iš	čakup	huupuuk <sup>w</sup> as?i
kuuwi1-mit-?iiš	čakup	huupuuk <sup>w</sup> as-?ii
steal-PST-3.IND	man	car-DET
A man stole the	car.	

The cases in (43) and (44) constitute a minimal pair for the "inside-out"/"outside-in" distinction. The difference between these two types of dependencies is represented syntactically below.

(45) a. *Kwakw'ala "inside-out" affixation* 

b. Nuu-chah-nulth "outside-in" affixation



In (45a), the Kwakw'ala determiner orients itself leftward to attach to the verb. In (45b), the Nuu-chah-nulth determiner positions itself rightward to attach to the nominal under its semantic scope.

In each case, the determiner takes a single step to attach to a neighbouring word. Why, then, would only (45b) qualify as a "local" linearization? Recall that local spell-out applies only to derivational sisters. Only in (45b) does the affix attach to a host within its derivational sister: in (45a), the derivational sister of the enclitic determiner D is N, not the V which it takes as a host. Therefore, (45a) does not satisfy this strict definition of derivational locality.

In the following section, I propose that linearization induced by local spellout applies only in cases in which the affix is lexically specified for an affixation requirement.

### 6.2.3.3 Affixation as Primitive or Derived

What may condition the choice between linearization which is effected by local spell-out and linearization which occurs later in the derivation? I suggest that it may relate to whether affixal status is *primitive* or *derived* (cf. Anderson's (1992) "special" or "simple" clitics). Recall that in Nuu-chah-nulth affixal predicates are lexically specified as affixes. This lexical distinction serves to differentiate them from independent predicates in the language, which never occur as suffixes.

208

### IMPLICATIONS



The difference between affixal and independent predicates is not independently reducible to a factor distinct from affixhood, such as prosodic weight. With respect to the prosodic heaviness, affixal predicates come in a range of types – from the non-syllabic to the polysyllabic – and, as such, overlap with the syllabic and polysyllabic forms of independent (non-affixal) predicates. Examples of the different weights of affixal and independent roots are given below.

(47)	Polysyllabic affixal predicates	(48) Polysyllabic independent predicates			
	a. <i>-'nakuuḥ</i> [+L] "observe" b. <i>-ḥaḥu⁺</i> [+L] "on front"	a. <i>kuuŵi</i> ł "steal" b. <i>paŵa</i> ł "to lose something"			
(49)	Monosyllabic affixal predicates	(50) Monosyllabic independent predicates			
	a. <i>-naah</i> "trying to locate" b. <i>-pa +</i> "be present"	a. <i>maa</i> "to bite" b. <i>k<sup>*w</sup>ix</i> "to kiss"			

(51) Non-syllabic affixal predicates

a. -q[+R] "travelling with in vessel"
b. -š[+L] "asking for"

Affix, in Nuu-chah-nulth, must therefore be a lexically specified *primitive*. It is not the case that affixation takes place because of a deficiency in prosodic weight: affixal predicates may be weightly or weightless.<sup>9</sup>

If affixal status is a lexical property of an affix in Nuu-chah-nulth, rather than a prosodically conditioned characteristic, then it constitutes a bare output requirement *for that morpheme*. It is a tenet of the Minimalist grammar that the lexically specified properties of an element must be satisfied by the point of spellout, so that the features of the lexical item may receive an appropriate interface interpretation. Under this view, it is lexical specification that forces affixation in Nuu-chah-nulth to emerge via local spell-out.

According to this line of reasoning, the lexical entry for the Nuu-chah-nulth determiner comes equipped with an affix specification, in the same way that affixal

<sup>&</sup>lt;sup>9</sup> In Anderson's (1992) terminology, Nuu-chah-nulth affixal predicates thus cannot be "simple" clitics.

predicates' do. In other words, affixal status in Nuu-chah-nulth is primitive. For Kwakw'ala, conversely, the implication of this analysis is that the affixal status of determiners in Kwakw'ala must be derived, not inherent. Only for derived affixes will "inside-out" dependencies be possible. Only for derived affixes will affixation *not* be a requirement at the stage of local spell-out. Instead, if affixal status is prosodically determined, and established at a derivationally later stage, then prosodic manipulations will apply to find a host for the derived affix.

There is independent evidence that affixation of the Kwakw'ala determiner applies later in the derivation than affixation in Nuu-chah-nulth. The key to this idea is the observation that a Kwakw'ala determiner construed with the subject attaches leftward *to the verb*. That is, in the VSO word order of Kwakw'ala, the determiner of the S attaches to the V, as we saw in example (43), repeated here below.

(52) k<sup>w</sup>ix?id-ida bəg<sup>w</sup>anəma-x-a qasa-s-is təlwag<sup>w</sup>ayu clubbed-DET man-OBJ-DET otter-INST-POSS club The man clubbed the sea-otter with his club.

Crucially, as I noted in Chapter 3, VSO is not possible as an underived word order for a language, since the verb and object necessarily form a constituent. Instead, VSO order must always be derived. The fact that affixation takes place in Kwakw'ala only after VSO order has been established entails that it must be occuring at a derivationally later stage than in Nuu-chah-nulth. In Nuu-chah-nulth, an affixal predicate can only ever find a host from the object, never the subject, even though either VOS or VSO surface permutations are permitted.

The analysis of the affixation patterns of Kwakw'ala rests on empirical confirmation of the prosodic characteristics of Kwakw'ala determiners. Evidence in this direction comes from Anderson (2005). According to Anderson (2005: 14–22), these elements are prosodically deficient. Anderson presents an analysis in which affixation of these elements to a host arises via prosodic constraints on "stray adjunction" of weak elements. Future research is required in ascertaining the defining prosodic characteristics of this class of clitics.

The contrast between the "inside-out" linearization of Kwakw'ala determiners and the "outside-in" pattern of Nuu-chah-nulth affixes implies a typological split between the Northern and Southern branches of the Wakashan family. This typological contrast may be modelled through distinct lexical specifications of the affixes. In the following section, I discuss additional typological implications for my proposal.

## 6.3 Typological Implications

This section presents three typological implications for my analysis of PF incorporation. I address implications for the typology of noun incorporation in §6.3.1, followed in §6.3.2 by discussion of the traditionally assumed "governing/restrictive" distinction in Wakashan. In §6.3.3, I argue that it is inaccurate to portray "lexical suffixation" as an uniform areal feature of the Pacific Northwest.

# 6.3.1 The Typology of Noun Incorporation

A claim of this book is that the affixation phenomenon descriptively labelled "noun incorporation" is not a uniform construction (cf. Mithun 1984, 1986; Sadock 1986; Baker 1988). According to my analysis, a noun may come to be suffixed by a verb via four distinct "routes" of bound/free dependencies. As first discussed in Chapter 2, the noun may be free or bound, and the verb may itself be free or bound. This four-way typology is given in (53).

# (53) Typology of noun–verb dependencies

	Bound noun	Free noun
Bound verb	Nuu-chah-nulth	Nuu-chah-nulth
Free verb	Mohawk	English

This section reviews the affixation patterns for each of these four underlying noun-verb dependencies.

Nuu-chah-nulth exhibits two of these four relationships. In Nuu-chah-nulth, bound predicates mandatorily require suffixation to a host, and never occur independently. The inability of an affixal predicate to separate from its host is illustrated in (54) for the predicate *?u-siik* "make". In (54a), the affixal predicate suffixes to *tuč?in* "dress". In (54b), suffixation fails to occur and the result is ungrammaticality.

(54)	a.	nupititsa nupit-mit-sa once-PST-1SG.ABS I made a dress once.	⁺uč?ins ⁺uč?in- dress- <u>n</u>	⁺uč?insiik ⁺uč?in- <u>siik</u> dress- <u>make</u>		
	b. *	<sup>:</sup> nupititsa nupit-mit-sa once-PST-1SG.ABS I made a dress once.	siik <u>siik</u> make	⁺uč?in ⁺uč?in dress		

This lack of isolatability of the predicate sets the suffixation strategy in Nuu-chahnulth apart from noun incorporation languages such as Mohawk. In Mohawk, a verb can be grammatically separated from an incorporated noun. In (55a), the inflected predicate *ye-nuhwe'-s* "like" incorporates the nominal *nuhs-* "house", while in (55b) it does not.

(55) *Mohawk* (from Postal 1962, as cited in Baker 1988: 81–82, ex. 14a, b)

a. Yao-wir-a'a ye-nuhs-nuhwe'-s PRE-baby-SUF 3FS/3N-house-like-ASP The baby house-likes. b. Yao-wir-a'a ye-nuhwe'-s ne ka-nuhs-a' PRE-baby-SUF 3FS/3N-like-ASP DET PRE-house-SUF The baby likes the house.

From this, I deduce that incorporation in Mohawk satisfies no boundedness requirement on the verb.

An opposite effect is found with Mohawk nominals, however. In Mohawk, a noun separated from an incorporating verb typically cannot stand on its own. In Mohawk, the majority of nominals are obligatorily bound (Marianne Mithun, p.c. 2005). Outside of incorporation contexts, the regular pattern is that a noun stem is suffixed by a neuter prefix and a "noun suffix".<sup>10</sup>

(56) NEUTER-noun.stem-NOUN.SUFFIX

An example of this pattern is given below for the Mohawk noun *ohnennà:ta*' "potato".

(57) ohnennà:ta' o-hnenna't-a' NEUT-potato-NOUN.SUFFIX 'potato'

(Marianne Mithun, p.c. 2005)

This example employs the most common noun suffix, -a', which contributes no discernible meaning (Marianne Mithun, p.c.). This noun suffix appears to bear a formal resemblance to the expletive morpheme  $\lambda - in$  Nuu-chah-nulth. In Nuu-chah-nulth,  $\lambda - acts$  as a "placeholder" for an affixal predicate requiring a host which it may suffix to. In Mohawk, a plausible hypothesis is that -a' acts as a "placeholder" for an affixal nominal requiring a host which it may prefix to. Note that this morpheme does not surface when the noun has undergone incorporation, as indicated in (55) for the nominal *nuhs-(a')* "house".

The final dependency is one in which a free noun suffixes to a free verb. This option corresponds to compounding in English. In (58), the noun *file* is free, as is the verb *share*.

- (58) a. They are sharing files.
  - b. They are file-sharing.

As each of the components of this compound are morphologically free, the absence of a morphological "placeholder" akin to Nuu-chah-nulth 2u- or Mohawk -a' is predicted.

<sup>&</sup>lt;sup>10</sup> There are some odd words used as lexical nouns which do not show this structure, but they are quite rare, and reportedly are never incorporated (Marianne Mithun, p.c. 2005).

### 6.3.2 The Governing/Restrictive Hypothesis in Wakashan

In proposing that the combinatory properties of Nuu-chah-nulth affixal predicates are conditioned by their argument structure, the present analysis differs markedly from traditional treatments of these suffixes (e.g., Sapir and Swadesh 1939; Swadesh 1939; Rose 1981; Nakayama 1997, 1998; Davidson 2002). In existing literature on the language, differences in these morphemes' suffixation patterns have been noted, but it has not previously been recognized that these "lexical suffixes" serve as predicates which show a range of distinctive argument structures. Under traditional classifications, affixal predicates are subject to a Wakashan-specific dichotomy between "governing" and "restrictive" suffixation. This distinction amounts to whether the suffix has a predicative or modificational role, respectively.

Suffixes which I have analysed as transitive and location predicates are treated as predicative "governing" lexical suffixes which take their morphological base as their object (Rose 1981). On the other hand, predicates which I classify as unaccusative and locatum predicates are grouped together with an assortment of other suffixes (such as plural markers) as modificational "restrictive" lexical suffixes. According to Rose (1981: 313), these "restrictive" suffixes "have an appositive relation to their base and do not influence the semantic class of the stem in which they occur." Syntactically, these suffixes are not considered to take complements. It is claimed that when a "restrictive" suffix attaches to a nominal, the nominal does not serve as the object of the suffix, but rather as the main predicate. For example, Rose (1981) analyses the locative construction in (59) as being syntactically headed by the nominal predicate *siičpax* "cougar". Crucially, by traditional analyses, the nominal *siičpax* "cougar" does not serve as the object of the locative suffix.

(59) siičpaxnaq
 siičpax-<u>naq</u>-Ø
 cougar-<u>on.top</u>-3.ABS
 There's a cougar on top.
 (Kyuquot dialect; adapted from Rose 1981: 314, ex. 447)

In contrast, the proposed analysis states that the locative  $-\dot{naq}$  "on top" is a locatum predicate which takes  $s\ddot{ncpax}$  "cougar" as its syntactic object. (The subject is a null third person pronominal, registered by absolutive agreement.) This analysis is represented below.

(60) VP -*naq siičpax* on.top cougar

Thus, by my analysis, it is not the case that the nominal *siičpax* "cougar" serves as the main predicate.

At the heart of the governing/restrictive hypothesis is the notion that "restrictive" suffixes, unlike "governing" suffixes, do not syntactically head a predicate phrase. The proposal presented in this book offers an alternative to the languagespecific governing/restrictive hypothesis, and derives the behaviours of affixal predicate subclasses through universal tenets of argument structure. According to this analysis, the syntax of affixal predicates is built from two basic underlying syntactic configurations of unaccusative or extended unaccusative structures. The host for affixal predicates is chosen from their complement.

## 6.3.3 Lexical Suffixation as an Areal Feature of the Pacific Northwest

There is a lengthy tradition of viewing the phenomenon of lexical suffixation as an areal feature of languages of the Pacific Northwest (see, e.g., Sapir 1911; Swadesh 1948; Gerdts and Hinkson 1996; Kinkade 1998; Mithun 1999). Kinkade (1963) coined the term "lexical suffix" due to "the semantic similarity between these suffixes and usual lexical items" (Kinkade 1998: 266). These bound morphemes bear a lexical load parallel to that of free morphemes. Under a strict definition, lexical suffixes are thus simply suffixal morphemes which convey lexical content (see Swadesh 1939). Within the Salishanist literature, however, the term "lexical suffix" has come to be synonymous with the noun-like bound elements found in these languages (see Saunders and Davis 1975; Gerdts and Hinkson 1996). For Salishan languages, Gerdts (1998; citing Carlson 1989) notes that there is support for the notion that "lexical suffixes can be regarded as incorporated nouns that have lost their status as free-standing nominals" (Gerdts 1998: 97). For Wakashan affixal predicates, however, the inverse is true: these "lexical suffixes" pattern productively as incorporating predicates in Nuu-chah-nulth. For this reason, the "lexical suffixes" of Wakashan may be viewed as only superficially similar to their counterparts in Salish (Wojdak 2003b).

Gerdts and Hinkson (1996) provide a survey of Salishan lexical suffixes and identify two basic categories of lexical suffixes in Salish. The authors argue that both types are derived historically from nouns, but have undergone differing degrees of grammaticalization. In the first case, exemplified by noun compounds and suffixation paralleling compounding incorporation, the lexical suffix retains its categorial status as a noun. The example in (61) shows an instance of lexical suffixation in Lillooet Salish (St'at'imcets), in which the lexical suffix corresponds to the theme of the root verb. (Data is from Gerdts and Hinkson 1996: 168 (11); originally van Eijk 1985).

(61) Lillooet Salish ník-<u>**1ča?**</u> cut-<u>**flesh**</u> "cut meat"

In the second type, represented by applicatives and suffixation resembling classificatory noun incorporation, the lexical suffixes are semantically bleached and behave acategorially. An example of this highly grammaticized form of lexical suffixation is shown below with data from Halkomelem Salish. (Data is from Gerdts and Hinkson 1996: 172 (20)).

(62) Halkomelem Salish

te?cs-<u>élə</u> k<sup>w</sup>θə nə mémənə eight-<u>people</u> DET 1POS children I have eight children.

This book presents evidence against a cross-linguistically uniform treatment of Pacific Northwest lexical suffixes. I have demonstrated that morphemes which have been analysed as lexical suffixes in Nuu-chah-nulth are fundamentally distinct from their noun-like counterparts in Salishan languages. In Nuu-chah-nulth, these bound morphemes are affixal predicates which productively incorporate a host selected from their complements (see also Woo 2000; Yiu 2001; Davis and Sawai 2001; Stonham 2004). They are not, as has been proposed for Salish, degenerate nominals which occur in compounds. This contrast entails a strong formal distinction between those morphemes in Salishan and Wakashan languages which have been labelled "lexical suffixes".

## 6.4 Conclusion

This book has introduced typologically rare affixation effects to the debate surrounding the division of labour between the modules of the grammar. Evidence from Nuu-chah-nulth suggests that syntactic composition indirectly conditions linearizations which are established by the phonological component. In Nuu-chah-nulth, the argument structure of affixal predicates serves to restrict the source of "host" for affixation: only complements of affixal predicates are eligible. Hand-in-hand with this sensitivity to argument structure, however, Nuu-chah-nulth affixation also shows an insensitivity to syntactic dominance relations in that hosts are selected on the basis of string adjacency to the predicate. The local spell-out model successfully captures these superficially paradoxical effects. This Minimalist approach to the linearization of affixal predicates presents Nuu-chah-nulth affixation as proof-ofconcept for the hypothesis that syntax is "spelled out" to phonology in strictly minimal syntactic domains.

In the introductory chapter to this book, affixation in Nuu-chah-nulth was discussed as a "displacement" effect. Affixation in Nuu-chah-nulth allows a host from the complement of an affixal predicate to be removed from the post-predicative position it would otherwise occupy. It may be more accurate, however, to describe this phenomenon more literally as a "placement" effect. That is, affixal status serves to induce a linearization on unordered elements. According to the local spell-out hypothesis, linearization is induced each time the syntactic tree is expanded, when  $\alpha$  and  $\beta$  are spelled-out to PF.

(63) 
$$\alpha \beta$$
 local spell-out

By this linearization algorithm, an affix ( $\alpha$ ) must look to its derivational sister ( $\beta$ ) in order for it to be linearized. This results in a "complement" effect in affixation.

The string adjacency effect in Nuu-chah-nulth affixation arises when the derivational sister to an affixal predicate is internally complex. In such contexts in Nuu-chah-nulth, an affixal predicate  $(-\alpha)$  attaches to the element at the left periphery of the linearized derivational sister.

(64)



By attending to the periphery of this derivational sister, the linearization mechanism avoids processing the internal components of the complex phrase. Hence, this linearization mechanism is sensitive to linear positioning rather than internal syntax composition.

Under this model of spell-out, the syntax is "phonologized" over the course of the derivation, in minimal stages induced by application of Merge (Epstein 1999). Consequently, interface requirements are assessed aggressively throughout the derivation. The cumulative build-up of affixes in Nuu-chah-nulth was advanced as indication of the strictly interpretive nature of the spell-out mechanism. By this analysis, the inertness of interpreted constituents arises not via spell-out inself, but instead through saturation (cf. Chomsky 2001, 2005). The constituents of CP and DP were shown to bear the characteristics of saturated domains in Nuu-chah-nulth, as evidenced by opacity effects for affixation and syntactic movement.

A central ramification of the local spell-out proposal is that spell-out to PF manipulates the order of syntactic elements by imposing linearizations (see also Ndayiragije 2000). This analysis entails a dichotomy between PF linearizations and head movement. In affixation which occurs strictly as a linearization reflex of spell-out, internal dominance relations are ignored in place of restrictions on string adjacency. In head movement, in contrast, dominance relations are respected. Thus, head movement cannot be strictly phonological (contra Chomsky 1995, 2001).

# References

- Abney, Steven. 1987. The English noun phrase in its sentential aspect. Doctoral dissertation, MIT.
- Anderson, Stephen R. 1984. Kwakwala syntax and the Government-Binding theory. In Eung-Do Cook and Donna B. Gerdts (eds.), The syntax of Native American Languages, *Syntax and Semantics* 16: 21–75. Orlando: Academic Press.
- Anderson, Stephen. R. 1985. Typological distinctions in word formation. In Timothy Shopen (ed.), Language Typology and syntactic description, *Grammatical Categories and the Lexicon* 3: 3–56. Cambridge: Cambridge University Press.
- Aissen, Judith and David Perlmutter. 1983. Clause reduction in Spanish. In David Perlmutter (ed.), *Studies in Relational Grammar* 1: 360–403. Chicago: University of Chicago Press.
- Arima, Eugene Y. 1983. The West Coast People: The Nootka of Vancouver Island and Cape Flattery. British Columbia Provincial Museum Special Publication No. 6. Victoria.
- Arima, Eugene Y., Denis St. Claire, Louis Clamhouse, Joshua Edgar, Charles Jones and John Thomas. 1991. Between Ports Alberni and Renfrew: Notes on West Coast Peoples. Canadian Museum of Civilization, Canadian Ethnology Service Mercury Series Paper 121. Ottawa.
- Bach, Emmon. 1968. Nouns and noun phrases. In Emmon Bach and Robert T. Harms (eds.), *Universals in Linguistic Theory*: 90–122. New York: Holt, Rinehart and Winston.
- Bach, Emmon. 2004. Northern Wakashan Languages: Some Comparisons. Paper presented at the Wakashan Linguistics Conference (Vancouver, BC).
- Baker, Mark C. 1988. *Incorporation: A Theory of Grammatical Function Changing*. Chicago: University of Chicago Press.
- Baker, Mark C. 1996. *The Polysynthesis Parameter*. Oxford Studies in Comparative Syntax. New York: Oxford University Press.
- Baker, Mark C. 2003. *Lexical Categories: Verbs, Nouns and Adjectives*. Cambridge: Cambridge University Press.
- Behrend, Olaf and Henry Kammler. 2003. Fieldwork in a contemporary first nations community. *European Review of Native American Studies* 17: 19–26.
- Boas, Franz. 1891. The Nootka. In Second generation report of the Indians of British Columbia, *British Association for the Advancement of Science Report* 60 (1890): 582–604, 668–679. London.
- Boas, Franz. 1947. Kwakiutl Grammar, with a glossary of the suffixes. *Transactions* of the American Philosophical Society 37(3): 201–377.
- Bobaljik, Jonathan David. 1994. What does adjacency do? In *MIT Working Papers in Linguistics* 21: *The morphology-syntax connection*: 1–32. Cambridge, MA: MITWPL.
- Boeckx, Cedric and Sandra Stjepanović. 2001. Head-ing toward PF. *Linguistic Inquiry* 32: 345–355.

- Boyd, Robert. 1999. The Coming of the Spirit of Pestilence: Introduced Infectious Diseases and Population Decline among Northwest Coast Indians: 1774– 1874. Vancouver, BC/Seattle, WA: University of British Columbia Press – University of Washington Press.
- Braithwaite, Ben. 2003. Syntactic approaches to possessive constructions in Nuuchahnulth. In J.C. Brown and Michele Kalmar (eds.), *Papers for the 38th International Conference on Salish and Neighbouring Languages*. UBCWPL 11: 7–22.
- Burton, Strang. 1996. Past tense on nouns as death, destruction and loss. In Kiyomi Kusumoto (ed.), *Proceedings of NELS* 27: 65–77.
- Burzio, Luigi. 1981. Intransitive verbs and Italian auxiliaries. Doctoral dissertation, MIT.
- Carlson, Barry F. 1989. Compounding and lexical affixation in Spokane. *Anthropological Linguistics* 31: 69–82.
- Chomsky, Noam. 1977. On wh-movement. In Peter Culicover, Tom Wasow and Adrian Akmajian (eds.), *Formal Syntax*: 71–132. New York: Academic Press.
- Chomsky, Noam. 1993. A minimalist program for linguistic theory. In Kenneth Hale and Samuel J. Keyser (eds.), *The View from Building* 20: 1–49. Cambridge, MA: MIT Press.
- Chomsky, Noam. 1995. The Minimalist Program. Cambridge, MA: MIT Press.
- Chomsky, Noam. 2001. Derivation by phase. In Michael Kenstowicz (ed.), Ken Hale: A life in Language: 1–52. Cambridge, MA: MIT Press.
- Chomsky, Noam. 2005. On phases. Ms., MIT.
- Chung, Sandra. 1991. Functional heads and proper government in Chamorro. *Lingua* 85: 85–134.
- Chung, Sandra. 1998. *The Design of Agreement: Evidence from Chamorro*. Chicago: University of Chicago Press.
- Chung, Sandra. 2004. Restructuring and verb-initial order in Chamorro. *Syntax* 7: 199–233.
- Cinque, Guglielmo. 1999. Adverbs and Functional Heads: A Cross-linguistic Perspective. Oxford Studies in Comparative Syntax. Oxford: Oxford University Press.
- Cinque, Guglielmo. 2001. Restructuring and functional structure. In Laura Bruge (ed.), *University of Venice Working Papers in Linguistics* 11: 45–127. Venice: University of Venice, Department of Linguistics.
- Cinque, Guglielmo. 2002. A note on restructuring and quantifier climbing in French. *Linguistic Inquiry* 33: 617–636.
- Cook, Eung-Do and Darin Howe. 2004. Aboriginal Languages of Canada. In William O'Grady and John Archibald (eds.), *Contemporary Linguistic Analysis* (5th edition): 294–309. Toronto: Addison Wesley Longman.
- Davidson, Matthew. 2002. Studies in Southern Wakashan (Nootkan) Grammar. Doctoral dissertation, SUNY at Buffalo.
- Davis, Henry. 2005. Constituency and Coordination in St'át'imcets (Lillooet Salish). In Andrew Carnie, Sheila Anne Dooley and Heidi Harley (eds.), Verb First: on the Syntax of Verb Initial Languages: 31–64. Amsterdam: John Benjamins.

- Davis, Henry and Naomi Sawai. 2001. Wh-movement as noun incorporation in Nuu-chah-nulth. In Karine Megerdoomian and Leora Anne Bar-El (eds.), *WCCFL 20 Proceedings*: 123–136. Somerville, MA: Cascadilla Press.
- Davis, Henry, Ryan Waldie and Rachel Wojdak. 2007. Condition C effects in Nuuchah-nulth. In *Canadian Journal of Linguistics: Special Edition on Southern Wakashan* 52.
- Davis, Henry and Rachel Wojdak. 2007a. Canadian Journal of Linguistics: Special Edition on Southern Wakashan 52.
- Davis, Henry and Rachel Wojdak. 2007b. Southern Wakashan: descriptive and theoretical perspectives. In *Canadian Journal of Linguistics: Special Edition on Southern Wakashan* 52.
- Déchaine, Rose-Marie and Martina Wiltschko. 2002. Decomposing pronouns. *Linguistic Inquiry* 33: 409–442.
- Demirdache, Hamida and Lisa Matthewson. 1995. On the Universality of Syntactic Categories. *Proceedings of NELS* 25: 79–93.
- Dewhirst, John. 1978. *The Yuquot Project*. Ottawa: National Historic Parks and Sites Branch, Parks Canada, Environment Canada.
- Dobashi, Yoshihito. 2003. Phonological phrasing and syntactic derivation. Doctoral dissertation, Cornell University.
- Donati, Caterina. 2003. An Extensive View on Head Movement. Paper presented at the 26th GLOW Colloquium (Lund, Sweden).
- Drucker, Philip. 1951. *The Northern and Central Nootkan Tribes*. Washington, DC: Bureau of American Ethnology, Bulletin 144.
- van Eijk, Jan P. 1985. The Lillooet language. Doctoral dissertation, Universiteit van Amsterdam.
- Embick, David. 1997. Voice and the interfaces of syntax. Doctoral dissertation, University of Pennsylvania.
- Embick, David and Rolf Noyer. 2001. Movement operations after syntax. *Linguistic Inquiry* 32: 555–595.
- England, Nora. 1991. Changes in basic word order in Mayan languages. International Journal of American Linguistics 57: 446–486.
- Epstein, Samuel David, Erich Groat, Ruriko Kawashima and Hisatsugu Kitahara. 1998. *A Derivational Approach to Syntactic Relations*. Oxford: Oxford University Press.
- Epstein, Samuel David. 1999. Un-principled syntax: The derivation of syntactic relations. In Samuel David Epstein and Norbert Hornstein (eds.), *Working Minimalism*: 317–345. Cambridge, MA: MIT Press.
- Evers, Arnold. 1975. The transformational cycle of Dutch and German. Doctoral dissertation, University of Utrecht.
- Fox, Danny and David Pesetsky. 2005. cyclic linearization of syntactic structure. In Katalin É. Kiss (ed.), *Theoretical Linguistics*: 1–46. Special issue on Object Shift in Scandinavian.
- Freeze, Ray. 1992. Existentials and other locatives. Language 68: 553-595.
- Gerdts, Donna B. 1998. Incorporation. In Andrew Spencer and Arnold M. Zwicky (eds.), *The Handbook of Morphology*: 84–100. Oxford: Blackwell.

- Gerdts, Donna B. and Mercedes Q. Hinkson. 1996. Salish lexical suffixes: a case of decategorialization. In Adele Goldberg (ed.), *Proceedings of the Conference on Conceptual Structure, Discourse, and Language*: 163–176. Stanford, CA: CSLI.
- Golla, Susan M. 1987. He has a name: history and social structure among the Indians of western Vancouver Island. Doctoral dissertation, Columbia University.
- Greenberg, Joseph H. 1966. Some universals of grammar with particular reference to the order of meaningful elements. In Joseph H. Greenberg (ed.), *Universals* of grammar: 73–113 (2nd edition). Cambridge, MA: MIT Press.
- Haas, Mary R. 1969. Internal reconstruction of the Nootka-Nitinat pronominal suffixes. *International Journal of American Linguistics* 35: 108–124.
- Haas, Mary R. 1972. The structure of stems and roots in Nootka-Nitinat. *International Journal of American Linguistics* 38: 83–92.
- Hale, Ken. 1983. Warlpiri and the grammar of non-configurational languages. *Natural Language and Linguistic Theory* 1: 5–47.
- Hale, Ken and Samuel Jay Keyser. 2002. Prolegomenon to a Theory of Argument Structure. Cambridge, MA: MIT Press.
- Halle, Morris and Alec Marantz. 1993. Distributed morphology and the pieces of inflection. In Kenneth Hale and Samuel Jay Keyser (eds.), *The View from Building* 20. Cambridge, MA: MIT Press.
- Halpern, Aaron. 1992. On the Placement and Morphology of Clitics. Stanford, CA: CSLI.
- Hoover, Alan L. (ed.) 2000. *Nuu-chah-nulth Voices, Histories, Objects and Journeys*. Victoria: Royal British Columbia Museum.
- Howe, Darin. 2000. Oowekyala segmental phonology. Doctoral dissertation, University of British Columbia.
- Jacobsen, William H., Jr. 1969. Origin of the Nootka pharyngeals. *International Journal of American Linguistics* 35: 125–153.
- Jacobsen, William H., Jr. 1979. Noun and verb in Nootkan. In Barbara S. Efrat (ed.), *The Victoria Conference on Northwestern languages* (British Columbia Provincial Heritage Record, No. 4): 83–155. Victoria: British Columbia Provincial Museum.
- Jacobsen, William H., Jr. 1993. Subordination and cosubordination in Nootka: clause combining in a polysynthetic verb-initial language. In Robert D. Van Valin, Jr. (ed.), *Advances in Role and Reference Grammar*: 235–274. Amsterdam: John Benjamins.
- Jacobsen, William H., Jr. 2007. The subclassification of Southern Wakashan. In *Canadian Journal of Linguistics: Special edition on Southern Wakashan* 52.
- Jelinek, Eloise and Richard Demers. 1994. Predicates and pronominal arguments in Straits Salish. *Language* 70: 697–736.
- Kayne, Richard S. 1994. The Antisymmetry of Syntax. Cambridge, MA: MIT Press.
- Kess, Joseph Francis and Anita Copeland Kess. 1986. On Nootka baby talk. International Journal of American Linguistics 52:201–211.
- Kim, Eun-Sook. 2001. The morphology and syntax of -*?at* in Nuu-chah-nulth. Syntax generals paper, University of British Columbia.

- Kim, Eun-Sook. 2003. Theoretical issues in Nuu-chah-nulth phonology and morphology. Doctoral dissertation, University of British Columbia.
- Kim, Eun-Sook and Rachel Wojdak. 2002. A survey of Nuu-chah-nulth reduplication. In Carrie Gillon, Naomi Sawai and Rachel Wojdak (eds.), Papers for the 37th International Conference on Salish and Neighbouring Languages. UBCWPL 9: 189–202.
- Kinkade, M. Dale. 1963. Phonology and morphology of Upper Chehalis: II. International Journal of American Linguistics 29: 345–356.
- Kinkade, M. Dale. 1998. Origins of Salish lexical suffixes. In Papers for the 33rd International Conference on Salish and Neighbouring Languages: 266–295.
- Kitahara, Hisatsugu. 1994. Target α: a unified theory of movement and structurebuilding. Doctoral dissertation, Harvard University, Cambridge, MA.
- Kitahara, Hisatsugu. 1995. Target α: deducing strict cyclicity from derivational economy. *Linguistic Inquiry* 26: 47–77.
- Klavans, Judith L. 1985. The independence of syntax and phonology in cliticization. *Language* 61: 95–120.
- Klokeid, Terry. 1976. Encliticization in Nitinaht. Paper presented at the 11th International Conference on Salishan Languages.
- Klokeid, Terry. 1978a. Surface structure constraints and Nitinaht enclitics. In Eung-Do Cook and Jonathan Kaye (eds.), *Linguistic Studies of Native Canada*: 157–176. Vancouver, BC: University of British Columbia Press.
- Klokeid, Terry. 1978b. Syntactic and conceptual relations in Nitinaht. In *Papers for the 12th International Conference on Salishan Languages*: 1–68. Colville, WA: Colville Federated Tribes.
- Kratzer, Angelika. 1994. Severing the external argument from its verb. In Johan Rooryck and Laurie Zaring (eds.) *Phrase structure and the lexicon*. Dordrecht: Kluwer.
- Kratzer, Angelika. 1995. Stage-level and individual-level predicates. In Gregory N. Carlson and Francis Jeffry Pelletier (eds.), *The Generic Book*. Chicago: University of Chicago Press.
- Kroeber, A. L. 1909. Noun incorporation in American languages. XVI Internationaler Amerikanisten-Kongress: 569–76.
- Kroeber, A. L. 1911. Incorporation as a linguistic process. *American Anthropologist* 13: 577–584.
- Lakoff, George. 1970. Irregularities in Syntax. New York: Holt, Rinehart and Winston.
- Lasnik, Howard. 1981. Restricting the theory of transformations. In Norbert Hornstein and David Lightfoot (eds.), *Explanation in Linguistics*. London: Longmans.
- Lasnik, Howard. 2000. Syntactic Structures Revisited: Contemporary Lectures on Classic Transformational Theory. Cambridge, MA: MIT Press.
- Lee, Felicia. 2000. VP Remnant movement and VSO in Quiaviní Zapotec. In Andrew Carnie and Eithne Guilfoyle (eds.), *The Syntax of Verb Initial Languages*: 143–162. Oxford: Oxford University Press.
- Li, Yafei. 1990. X<sup>0</sup>-binding and verb incorporation. *Linguistic Inquiry* 21: 399–426.
- Lieber, Rochelle. 1992. Deconstructing Morphology: Word Formation in Syntactic Theory. Chicago: University of Chicago Press.

- Marantz, Alec. 1988. Clitics, morphological merger, and the mapping to phonological structure. In Michael Hammond and Michael Noonan (eds.), *Theoretical Morphology: Approaches in Modern Linguistics*: 253–270. San Diego, CA: Academic Press.
- Marantz, Alec. 1989. Clitics and phrase structure. In Mark Baltin and Anthony Kroch (eds.), *Alternative Conceptions of Phrase Structure*. Chicago: University of Chicago Press.
- Marshall, Yvonne. 1993. A political history of the Nuu-chah-nulth people. Doctoral dissertation, Simon Fraser University.
- Massam, Diane. 2000. VSO and VOS: aspects of Niuean word order. In Andrew Carnie and Eithne Guilfoyle (eds.), *The Syntax of Verb Initial Languages*: 97–116. Oxford: Oxford University Press.
- Matushansky, Ora. 2005. Going through a phase. In Martha McGinnis and Norvin Richards (eds.), *Perspectives on Phases. MIT Working Papers in Linguistics* 49. MITWPL.
- Matushansky, Ora. 2006. Head-movement in linguistic theory. In *Linguistic Inquiry* 37.
- McCloskey, James. 1991. Clause structure, elipsis, and proper government in Irish. In *The Syntax of Verb Initial Languages: Lingua Special Edition* 85: 259–302.
- Mithun, Marianne. 1984. The evolution of noun incorporation. Language 60: 847-894.
- Mithun, Marianne. 1986. On the nature of noun incorporation. Language 60: 32-37.
- Mithun, Marianne. 1999. *The Languages of Native North America*. Cambridge: Cambridge University Press.
- Nakayama, Toshihide. 1997. Discourse-pragmatic dynamism in Nuu-chah-nulth (Nootka) Morphosyntax. Doctoral dissertation, University of California.
- Nakayama, Toshihide. 1998. Lexical suffixes in Nuuchahnulth: an overview. In Osahito Miyaoka (ed.), *Languages of the North Pacific Rim* 5: 19–41. Suita, Japan: Osaka Gakuin University.
- Nakayama, Toshihide. 2001. Nuuchahnulth (Nootka) Morphosyntax. University of California Publications in Linguistics 134. Berkeley: University of California Press.
- Nakayama, Toshihide (ed.). 2003. Caroline Little's Nuu-chah-nulth (Ahousaht) Texts with Grammatical Analysis. Endangered Langauges of the Pacific Rim Publications Series A2-027. Suita, Japan: Osaka Gakuin University.
- Ndayiragije, Juvenal. 2000. Strengthening PF. Linguistic Inquiry 31: 485–512.
- Newmeyer, Frederick J. 2005. Against a parameter-setting approach to language variation. In Pierre Pica (ed.), *Language Variation Yearbook 2004*: 181–234. Amsterdam: John Benjamins.
- Paik, Tai Shung. 1968. Clayoquot phonology. MA thesis, University of Washington.
- Pearson, Matthew. 1998. Two types of VO languages. Ms., University of California, Los Angeles.
- Perlmutter, David. 1978. Impersonal passives and the unaccusative hypothesis. Papers from the Annual Meeting of the Berkeley Linguistics Society 4: 157–189.
- Postal, Paul. 1962. Some syntactic rules of Mohawk. Doctoral dissertation, Yale University.

- Powell, Jay (ed.). 1991. *Our World Our Ways: T'aat'aaqsapa Cultural dictionary*. Port Alberni, BC: Nuu-chah-nulth Tribal Council.
- Rath, John C. 1974. On the phonological description of the Heiltsuk language. In *Dutch Contributions to the 9th International Conference on Salish languages*: 39–75.
- Ravinski, Christine. 2005. Grammatical possession in Nuu-chah-nulth. MA thesis, University of British Columbia.
- Reinhart, Tanya. 1979. Syntactic domains for semantic rules. In Franz Guenther and S.J. Schmidt (eds.), *Formal Semantics and Pragmatics for Natural Language*: 107–130. Dordrecht: Reidel.
- Renker, Ann M. 1987. Rethinking noun and verb: an investigation of AUX in a Southern Wakashan language. Doctoral dissertation, The American University.
- Rivero, Maria Luisa. 1999. Stylistic verb movement in yes-no questions in Bulgarian and Breton. In István Kenesei (ed.), *Crossing boundaries: Advances in the theory of Central and Eastern European languages*: 67–90. Amsterdam: John Benjamins.
- Rizzi, Luigi. 1982. Issues in Italian Syntax. Dordrecht: Foris.
- Rizzi, Luigi. 1986. Null objects in Italian and the theory of pro. *Linguistic Inquiry* 17: 501–557.
- Roberts, Ian. 1997. Restructuring, head movement and locality. *Linguistic Inquiry* 28: 423–460.
- Rooryck, Johan. 2000. Configurations of Sentential Complementation: Perspectives from Romance languages. New York: Routledge.
- Rose, Suzanne M. 1981. Kyuquot grammar. Doctoral dissertation, University of Victoria.
- Rose, Suzanne M. and Barry F. Carlson. 1984. The Nootka-Nitinaht passive. *Anthropological Linguistics* 26: 1–12.
- Ross, John R. 1967. Constraints on variables in syntax. Doctoral dissertation, MIT.
- Sadock, Jerrold M. 1980. Noun incorporation in Greenlandic: a case of syntactic word formation. *Language* 56: 300–319.
- Sadock, Jerrold M. 1986. Some notes on noun incorporation. Language 62: 19-31.
- Sapir, Edward. 1911a. The problem of noun incorporation in American languages. *American Anthropologist* 13: 250–282
- Sapir, Edward. 1911b. Some aspects of Nootka language and culture. American Anthropologist 13: 15–28.
- Sapir, Edward. 1921. *Language: An Introduction to the Study of Speech*. New York: Harcourt, Brace and World.
- Sapir, Edward. 1924. The rival whalers a nitinat story. *International Journal of American Linguistics* 3: 76–102.
- Sapir, Edward. 1929. Nootka baby words. International Journal of American Linguistics 5: 118–119.
- Sapir, Edward and Morris Swadesh. 1939. Nootka Texts, Tales and Ethnological Narratives. Philadelphia/Baltimore, MD: Linguistic Society of America.
- Sapir, Edward and Morris Swadesh. 1955. Native accounts of Nootka ethnology. International Journal of American Linguistics 21, part 2.

- Sapir, Edward, Morris Swadesh, Alexander Thomas, John Thomas and Frank Williams. 2000. *The Whaling Indians: Tales of Extraordinary Experience*, Part 10 of the Sapir-Thomas Nootka Texts, Eugene Arima, In Terry Klokeid and Katherine Robinson (eds.), Canadian Museum of Civilization Mercury Series CANES 134, Ottawa.
- Sapir, Edward, Morris Swadesh, Alexander Thomas, John Thomas and Frank Williams. 2004. *The Whaling Indians: Legendary Hunters*, Part 9 of the Sapir-Thomas Nootka Texts, In Eugene Arima, Terry Klokeid and Katherine Robinson (eds.), Canadian Museum of Civilization Mercury Series Ethnology Paper 139, Ottawa.
- Saunders, Ross and Philip W. Davis. 1975. Bella Coola referential suffixes. International Journal of American Linguistics 41: 355–368.
- Sproat, Gilbert. 1868. Scenes and Studies of Savage Life. London: Smith, Elder. (Reprinted in 1987 as The Nootka: Scenes and Studies of Savage Life, Charles Lilliard (ed.), Victoria: Sono Nis).
- Stonham, John. 1998. Numerals and incorporation in Nootka. In Papers for the 33rd International Conference on Salish and Neighbouring Languages: 384–394. Washington, DC: University of Washington.
- Stonham, John. 1999. Aspects of Tsishaath Nootka Phonetics & Phonology. München: Lincom Europa.
- Stonham, John. 2004. *Linguistic Theory and Complex Words: Nuuchahnulth Word Formation.* New York: Palgrave MacMillan.
- Stonham, John. 2005. A Concise Dictionary of the Nuuchahnulth Language of Vancouver Island. New York: The Edwin Mellen.
- Stonham, John. 2007. Nuu-chah-nulth double reduplication and stratal OT. In *Canadian Journal of Linguistics: Special edition on Southern Wakashan* 52.
- Surányi, Balázs. 2003. Multiple operator movements in Hungarian. Doctoral dissertation, Utrecht University.
- Svenonius, Peter. 2001. Locality, phases, and the cycle. Ms., University of Tromsø.
- Svenonius, Peter. 2004. On the edge. In David Adger, Cécile de Cat and George Tsoulas (eds.), *Peripheries: Syntactic Edges and their Effects*.
- Swadesh, Morris. 1939. Nootka internal syntax. International Journal of American Linguistics 9: 77–102.
- Swadesh, Morris. 1948. A structural trend in Nootka. Word 4: 106–119.
- Swadesh, Morris. 1953. Mosan 1: a problem of remote common origin. *International Journal of American Linguistics* 19: 26–44.
- Szabolcsi, Anna. 1994. The noun phrase. In Ferenc Kiefer and Katalin É. Kiss (eds.), Syntax and Semantics 27: The Syntactic Structure of Hungarian: 179–274. San Diego, CA: Academic Press.
- Travis, Lisa D. 1984. Parameters and effects of word order variation. Doctoral dissertation, MIT.
- Uriagereka, Juan. 1999. Multiple spell-out. In Samuel David Epstein and Norbert Hornstein (eds.), *Working Minimalism*: 251–282. Cambridge, MA: MIT Press.

- Uriagereka, Juan. 2002. Derivations: Exploring the Dynamics of Syntax. London: Routledge.
- Waldie, Ryan. 2004. Nuu-chah-nulth denominal verbs. MA thesis, University of Victoria.
- Werle, Adam. 2002. The Southern Wakashan one-foot word. In Carrie Gillon, Naomi Sawai and Rachel Wojdak (eds.), Papers for the 37th International Conference on Salish and Neighbouring Languages. UBCWPL 9: 382–397.
- Whistler, Kenneth W. 1985. Focus, perspective, and inverse person marking in Nootkan. In Johanna Nichols (ed.), *Grammar Inside and Outside the Clause:* Some Approaches to Theory from the Field. Cambridge: Cambridge University Press.
- Wojdak, Rachel. 2000. Nuu-chah-nulth modification: syntactic evidence against category neutrality. In Suzanne Gessner and Sunyoung Oh (eds.), Papers for the 35th International Conference on Salish and Neighbouring Languages. UBCWPL 3: 269–281.
- Wojdak, Rachel. 2001. An argument for category neutrality? In Karine Megerdoomian and Leora A. Bar-El (eds.), WCCFL 20 Proceedings: 621–634. Somerville, MA: Cascadilla.
- Wojdak, Rachel. 2002. Dissimilation in Nuu-chah-nulth reduplicative fixed segmentism. Paper presented at the 38th Meeting of the Chicago Linguistic Society: Main Session.
- Wojdak, Rachel. 2003a. PF incorporation: evidence from Wakashan. Paper presented at the 26th GLOW Colloquium (Lund, Sweden).
- Wojdak, Rachel. 2003b. Predicative lexical suffixes in Nuu-chah-nulth. In J.C. Brown and Michele Kalmar (eds.), Papers for the 38th International Conference on Salish and Neighbouring Languages. UBCWPL 11: 275–289.
- Wojdak, Rachel. 2004. On the classification of Wakashan lexical suffixes. Paper presented at the 30th Meeting of the Berkeley Linguistics Society.
- Wojdak, Rachel. 2005. The linearization of affixes: evidence from Nuu-chah-nulth. Doctoral dissertation, University of British Columbia.
- Wojdak, Rachel and Florence Woo. 2004. Deriving the definiteness effects in Nuuchah-nulth locatives. In Yunhee Chung and Christine Ravinski (eds.), *Proceedings of WSCLA 9*. UBCWPL 14.
- Woo, Florence. 2000. Predicative governing suffixes and incorporation in Nuuchah-nulth. Ms., University of British Columbia.
- Woo, Florence. 2004. Word order. Ms., University of California at Santa Cruz.
- Wurmbrand, Susanne. 2001. *Infinitives: Restructuring and Clause Structure*. Berlin: Mouton de Gruyter.
- Wurmbrand, Susanne. 2004. Two types of restructuring lexical vs. functional. *Lingua* 114: 991–1014.
- Yiu Sze Man. 2001. Incorporation in Nuuchahnulth. MA thesis, University of Manchester.

# Appendix

## Orthography

Although Nuu-chah-nulth is traditionally an oral language, written forms of the language are currently in use by linguists, educators, and some other community members. I adopt a modified IPA orthography, following the practice of the Haa-Huu-Payak school in Port Alberni, BC. This orthography appears widely in materials developed by the Nuu-chah-nulth communities (e.g., *The Nuu-chah-nulth Alphabet Book*, created by the Nuu-chah-nulth-eets Tsitsiqi Language and Cultural Resource Project, and the *Ha-shilth-sa* community newspaper). The orthography is closely based on the system represented in standard sources such as Sapir and Swadesh (1939).

The orthography captures a three-way contrast in vowel quality, and a distinction between short and long vowels: (i, ii), (u, uu), (a, aa). A lexically determined contrast between long and "persistently" long vowels is standardly assumed for Southern Wakashan (Sapir and Swadesh 1939). "Persistently" long vowels escape a vowel-shortening rule which targets long vowels outside of the first foot of a word (Jacobsen 1979; Wilson 1986; Stonham 1990; Werle 2002). I represent both long and "persistently" long vowels as long vowels in the second line of my four-line glosses; however, only "persistently" long vowels surface as long outside of the first foot.

The consonant inventory is complex, and distinguishes several places of articulation. The orthographic representations for each phoneme are listed in the table below.

	Labial	Apical	Alveolar	Lateral	Palatal	Velar	Labio-velar	Uvular	Labio-uvular	Pharyngeal	Glottal
Stops	р	t	с	r	č	k	kw	q	$q^w$	ſ	?
Ejectives	ģ	ť	ċ	Ĥ	č	ķ	κ <sup>w</sup>		(ở <sup>w</sup> )		
Fricatives			S	ł	š	Х	Xw	( <u>x</u> )		h	h
Resonants	m	n			у		W				
Glottal	'n	'n			ý		ŵ				
resonants											

## (1) *Consonant inventory* (Nakayama 1997: 9)

# Abbreviations

ABS	absolutive mood	S	vowel shortening
AGR	agreement	SG	singular
ASP	aspect	SPOR	sporadic
AUG	augmentative	SUB	subordinate mood
AUX	auxiliary	SUF	suffix
BEN	benefactive	TR	transitivizer
CAUS	causative	TEMP	temporal
COMP	complementizer	TENSE	tense
COND	conditional mood		
CONF	confirmation mood		
CON1	conjunction		
CONT	continuous		
DEIC	deictic		
OEP	dependent mood		
DET	determiner		
DIM	diminutive		
DIR	directive		
DUR	durative		
FUT	future tense		
GO	"go" imperative		
HAB	habitual		
IMP	imperative		
IND	indicative mood		
INST	instrumental		
INV	invisible		
IPOSS	inalienable possessive		
IREL	"indefinite" relative mood		
ITER	iterative		
L	vowel lengthening		
LOC	locative		
NEG	negative		
NOM	nominaliser		
OBI	object		
PAS	passive		
PERF	perfective		
PL	plural		
POSS	possessive		
PRE	prefix		
PS	possessive mood		
PST	past tense		
0	interrogative mood		
QUOT	quotative mood		
R	reduplication		
RD	round		
REL	relative pronoun		
RES	resultative		
RL	(simple) relative mood		

# Index

adjective, 43-49, 94, 101-102, 202-203, 204-205 adverb, 49-53, 57, 78, 103-104, 159-160, 170-172 affixal predicates as obligatorily affixal, 2, 13-18, 29, 107-108 as suffixes. 78 classes of, 108, 114, 136 hosts for, 67-68 affixation as a lexical requirement, 17, 108, 208-210 as a prosodic deficiency, 25, 207-210 affix-hopping, 101-102 Ahousaht dialect (see also Nuu-chah-nulth), 20, 23, 55, 79, 85-86, 120, 132, 164-165n., 201n. Anderson, S., 156, 207-210 argument structure basic configurations, 109-111 conditioning affixation, 138, 142-143, 147-148, 154, 158, 187, 213-214 flexibility in mapping, 112-114 transitivization, 111-112 asymmetry, incorporation, 83, 97, 130-131, 143, 147, 152 auxiliary, 161 Baker, M., 28n., 29n., 31, 42, 46, 80, 107, 113, 199, 205 bare nominal, 63-64 bare output conditions, 4, 7, 9, 27, 209 benefactive, 108-109 biclausal, 169 binary concatenation, 5-6, 7, 27, 45, 67 Bobaljik, J., 9 bound noun, 28, 206, 211 bound verb, 28-30, 211 c-command, 71-73, 199-200 Chomsky, N., 3-5, 7, 11-12, 32, 61-62,

Chomsky, N., 3-5, 7, 11-12, 32, 61-62, 71, 111, 133, 183, 190-191, 199-200 Chung, S., 86-87 Cinque, G., 49, 163, 176 citation form, 30, 169 clause-boundedness, 176 clitic DP-level, 100-102, 118, 207-210 CP-level, 79, 100, 103-104, 195 second-position, 18, 50, 91-92 string, 22-23, 62 clitic climbing, 176, 186 complement effect, 83, 96-98, 130-132, 152, 201 complementizer, 64, 161, 172, 175, 179, 192. 195n. concealed transitive, 115, 119 conditioning, prosodic, 32-37, 198 conditioning, syntactic, 67, 138, 187-189 configurationality, 78-84 conflation, 115-116, 139-140 Coordinate Structure Constraint (CSC), 18, 40, 53-56, 67 countercyclicity, 42, 199-200 c-selection, 45, 205 Davidson, M. on clitics, 91, 100-101, 103 on possessor raising, 81-83, 134 on locatives, 136-138 on word order. 94 on word structure, 21-22 Davis H. and N. Sawai, 21, 44, 65, 77, 83, 92n., 130-131, 177, 193 decompositional, 154 derivational sisterhood, 5, 98-100 determiner and word order effects, 86, 89, 96n. as clitic, 100-102, 207-210 as inducing opacity effects, 31-32, 63-64, 66, 118 directionality conventions, 73-78 Distributed Morphology, 6, 29n. ditransitive, 108-109, 111-112, 150-154 Dobashi, 71 do-support, 31 dyadic, 110-113, 139

Embick, D. and R. Noyer, 6, 9, 18, 188 English locatives, 138-140 prepositions, 74-75 Epstein, S., 71n., 216 Epstein, S. et al., 2, 5-6, 9-10, 12, 18 escape hatch for movement, 25, 62, 67, 191, 193-196 expletive host, 24, 29-32, 33, 62, 169, 212 Fox, D., and D. Pesetsky, 10, 41, 189 Gerdts, D. and M. Hinkson, 214-215 "governing" affix, 120, 213-214 Greenlandic, suffixal verbs in, 30 Hale, K. and S. J. Keyser on argument structure, 100-112, 115, 118 on locatives, 112, 138-140 Halle, M. and A. Marantz, 6 Halpern, A., 67 head-initial. 75-77 head-marking, partial, 79 head movement, 45, 67, 199-200 hierarchical adjacency, 45-49 Hungarian, postpositions in, 74-75 incorporation (see also PF incorporation) of adjectives, 43-49, 56, 204-205 of adverbs, 49-53, 170-172 of negative particles, 57-58 of nouns, 14, 21, 27-29, 43-44, 56-57, 83, 155, 211-212 of quantifiers, 44, 56 of relative pronouns, 56-57, 144, 152.153 of verbs, 14, 16, 21, 22, 30, 40, 57, 64-65, 159-163, 170-171 of wh-pronouns, 56-57, 143-144, 147 independent predicates, 15-18 infinitival, 92, 164, 169 infix, 7-8 Iroquoian (see also Mohawk), 206 Jacobsen, W. H., 19, 84, 88 Kayne, R., 7, 71 Kim, E.-S., 21, 32, 198, 202 Kinkade, D., 214

Klavans, J., 39-40 Kyuquot dialect (see also Nuu-chah-nulth), 20, 21, 85, 120, 136, 164n., 213 Kwakw'ala clitics. 207-210 genetic affiliations, 19 locatives. 156 Lasnik, H., 3, 9, 31, 45 lexical suffixes, 21, 214-215 linear adjacency (see also string adjacency), 45-49 Linear Correspondence Axiom (LCA), 71-73 linearization as a phonological requirement, 7, 73.201-204 as cumulative, 10, 40-43 attending to periphery of string, 39-40 of bound morphemes, 7-9, 76 of syntactic terminals, 70-75 local spell-out (see also spell-out) iterative application, 10, 40-43 minimal syntactic units, 9, 36-37 "outside-in" dependencies, 206 phonological evidence, 32-37 location predicates (see also locatives), 146-150 locatives, 112-114, 136-141, 156-157 locatum predicates (see also locatives), 141-146, 155-156 long-distance movement, absence of, 65, 193-96 "long" possessor raising, 119, 121-129 main predicate, 160 Marantz, A., 9, 40 Massam, D., 87, 95 Matushansky, O., 45, 62, 200, 205 Merge, 5 Minimal link condition, 133, 134 Mirror Principle, 42 Mithun, M., 28, 59, 204, 211, 214 Mohawk, 30-31, 204-206 monadic predicate, 110, 118-119 mood inflection, 174, 195 Morphological merger, 9, 40 Morphological visibility condition, 81 Move, 5-6

Nakayama, T., 15n., 19, 21-22, 79, 132, 198, 213 negation, 57-58 Newmeyer, F., 76 nominal predicate, 93-94 Nuu-chah-nulth, genetic affiliations, 19-20 opacity effects (see also saturated domain) with CP, 61-64, 169 with DP, 31-32, 61-63, 64-65 with vP. 196-199 Order Preservation hypothesis, 10, 41, 189 passive, 22, 122, 201-202 PF Incorporation (see also incorporation), 18, 24, 27 phase, 12, 32, 61, 190-193 Phase impenetrability, 191 Phonetic Form (PF), 4 phonological effects (see also conditioning, prosodic), 197-198 placeholder (see also expletive host), 29-30, 212 polysynthesis, 21, 80, 105 possessor raising absence of determiner, 66 doubling of morphology, 82 feature matching, 181 subject restriction, 81-84, 134-136, 145-146, 149-150 "long" movement, 121-129, 177, 179-185 postposition, 74-75 predicate-initial (see also word order), 84-87 predicate inventory, 114, 136, 188 prefix, 7-8, 76 preposition, 74-75, 138-140 pro-drop, 79-81 prosodic weight, 209 p-signature, 115-116 psych predicate, 166 quantifier scope (see scopal effects)

raising-to-object, 183 raising-to-subject, 163 Ravinski, C., 66, 81-83, 90-91, 94, 100, 121-124, 133-134, 145, 179, 184

reduplication, 23, 33-35, 198 referentiality of incorporee, 58-60 Reinhart, T., 71 relative clause, 93, 131-132, 144, 152, 153 "restrictive" affix, 120, 156, 213-214 restructuring effects, 24, 64-66, 172, 176-186 rigid complementation, 164 Rose, S. on bound nominals, 14-15 on classes of suffixes, 120, 136. 213on "leftmost" hosts, 43, 45, 51, 58 on word order, 84-85, 89 Sadock, J., 28, 30, 211 same subject effect, 162-168 Sapir, E., 19, 20-21, 28, 80n., 88 saturated domain (see also opacity effects), 61-67, 190-196 scopal effects, absence of, 60-61 second position effect (see also cliticization), 50, 62, 91-92, 101, 103 serial verb, 154-158 specifier, branching scheme for, 88 spell-out (see also local spell-out) mapping requirements, 4, 7, 39 phases, 190-193 strictly interpretive, 189-190 timing, 10-13 Stonham, J., 7, 20-21, 47, 77, 130, 198 stranded affix filter, 9 string adjacency, 18, 37-56, 159, 170-172, 203 subject agreement clitics, 91-92, 103-104 in locatives, 113, 144-145, 148 in possessor raising, 82-83, 122, 125, 179 paradigms, 79, 132-133, 195 suffix bound morpheme, 7-8 directionality convention, 76-77 sequence, 21-23, 197 triggering phonological effects, 197-198 Surányi, B., 45n., 200 syntactic category, insensitivity to, 56-57

tense, absence of, 22, 65, 173-174, 199 theta-role, 112-114 transitive, 111-112, 129-136 transitivizer, 22, 164-168 typology of bound/free dependencies, 14 of noun incorporation, 211-212 tucking in, 42 unaccusative basic, 109-111, 116-118 diagnostic for, 121-129 extended, 110, 112-114, 136-141 serial verb, 154-156 transitivization of, 111-112, 150, 166-167 Unaccusativity hypothesis, 118 unergative absence of affixal unergatives, 24, 37, 98-100, 115-116, 119-121, 158 possessor raising, 121-129, 181n., 184n. subject agreement, 119 Uniformity condition, 200

Uniformity of theta assignment, 113 Uriagereka, J., 11, 73 verbal affix, 120 verbalizing suffix, 28 Wakashan, 19 Waldie, R., 21, 30, 201n. weak crossover, 24, 84 Werle, A., 19, 68, 100 wh-movement clause-bound, 65-66, 193-196 "long" movement, 66, 177-178 showing incorporation asymmetry, 131, 143-144, 147 showing weak crossover effects, 84 stranding restriction, 44-45 Woo, on word order, 85, 95 word order as derived via movement, 87, 210 as head-initial, 77 as variable, 82n., 84-86, 98-99, 135, 145, 202-203 as VOS, 24, 69, 87-89

# Managing Editors

Liliane Haegeman, University of Geneva Joan Maling, Brandeis University James McCloskey, University of California, Santa Cruz

# **Publications**

1.	L. Burzio: Italian Syntax. A Government-binding Approach. 1986
	ISBN Hb 90-277-2014-2; Pb 90-277-2015-0
2.	W. D. Davies: Choctaw Verb Agreement and Universal Grammar. 1986
	ISBN Hb 90-277-2065-7; Pb 90-277-2142-4
3.	K. É. Kiss: Configurationality in Hungarian. 1987
	ISBN Hb 90-277-1907-1; Pb 90-277-2456-3
4.	D. Pulleyblank: Tone in Lexical Phonology. 1986
	ISBN Hb 90-277-2123-8; Pb 90-277-2124-6
5.	L. Hellan and K. K. Christensen: Topics in Scandinavian Syntax. 1986
	ISBN Hb 90-277-2166-1; Pb 90-277-2167-X
6.	K. P. Mohanan: The Theory of Lexical Phonology. 1986
	ISBN Hb 90-277-2226-9; Pb 90-277-2227-7
7.	J. L. Aissen: Tzotzil Clause Structure. 1987
	ISBN Hb 90-277-2365-6; Pb 90-277-2441-5
8.	T. Gunji: Japanese Phrase Structure Grammar. A Unification-based Approach.
	1987 ISBN 1-55608-020-4
9.	W. U. Wurzel: Inflectional Morphology and Naturalness. 1989
	ISBN Hb 1-55608-025-5; Pb 1-55608-026-3
10.	C. Neidle: The Role of Case in Russian Syntax. 1988
	ISBN 1-55608-042-5
11.	C. Lefebvre and P. Muysken: <i>Mixed Categories</i> . Nominalizations in Quechua.
	1988 ISBN Hb 1-55608-050-6; Pb 1-55608-051-4
12.	K. Michelson: A Comparative Study of Lake-Iroquoian Accent. 1988
	ISBN 1-55608-054-9
13.	K. Zagona: Verb Phrase Syntax. A Parametric Study of English and Spanish.
	1988 ISBN Hb 1-55608-064-6; Pb 1-55608-065-4
14.	R. Hendrick: Anaphora in Celtic and Universal Grammar. 1988
	ISBN 1-55608-066-2
15.	O. Jaeggli and K. J. Safir (eds.): The Null Subject Parameter. 1989
	ISBN Hb 1-55608-086-7; Pb 1-55608-087-5
16.	H. Lasnik: Essays on Anaphora. 1989
	ISBN Hb 1-55608-090-5; Pb 1-55608-091-3
17.	S. Steele: Agreement and Anti-Agreement. A Syntax of Luiseño. 1990
	ISBN 0-7923-0260-5

18.	E. Pearce: <i>Parameters in Old French Syntax</i> . Infinitival Complements. 1990 ISBN Hb 0-7923-0432-2 <sup>•</sup> Pb 0-7923-0433-
19.	Y. A. Li: Order and Constituency in Mandarin Chinese. 1990
20.	ISBN 0-7923-0500- H. Lasnik: <i>Essays on Restrictiveness and Learnability</i> . 1990
21.	ISBN 0-7923-0628-7; Pb 0-7923-0629- M. J. Speas: <i>Phrase Structure in Natural Language</i> . 1990
22.	H. Haider and K. Netter (eds.): <i>Representation and Derivation in the Theor</i>
23.	J. Simpson: Warlpiri Morpho-Syntax. A Lexicalist Approach. 1991 ISBN 0-7923-1130- ISBN 0-7923-1292
24.	C. Georgopoulos: Syntactic Variables. Resumptive Pronouns and A' Binding in Palauan. 1991 ISBN 0-7923-1293-
25.	K. Leffel and D. Bouchard (eds.): Views on Phrase Structure. 1991 ISBN 0-7923-1295-
26.	C. Tellier: Licensing Theory and French Parasitic Gaps. 1991 ISBN 0-7923-1311-9; Pb 0-7923-1323-
27.	SY. Kuroda: Japanese Syntax and Semantics. Collected Papers. 1992 ISBN 0-7923-1390-9; Pb 0-7923-1391-
28.	I. Roberts: Verbs and Diachronic Syntax. A Comparative History of Englis and French. 1992 ISBN 0-7923-1705-2
29.	A. Fassi Fehri: Issues in the Structure of Arabic Clauses and Words. 1993 ISBN 0-7923-2082-
30. 31.	M. Bittner: <i>Case, Scope, and Binding.</i> 1994 ISBN 0-7923-2649- H. Haider, S. Olsen and S. Vikner (eds.): <i>Studies in Comparative German</i>
32.	Syntax. 1995ISBN 0-7923-3280-N. Duffield: Particles and Projections in Irish Syntax. 1995
33.	ISBN 0-7923-3550-3; Pb 0-7923-3674- J. Rooryck and L. Zaring (eds.): <i>Phrase Structure and the Lexicon</i> . 1996
34.	ISBN 0-7923-3745-2 J. Bayer: <i>Directionality and Logical Form</i> . On the Scope of Focusing Particle
35.	and Wh-in-situ. 1996 ISBN 0-7923-3752- R. Freidin (ed.): <i>Current Issues in Comparative Grammar</i> . 1996
36.	ISBN 0-7923-3778-6; Pb 0-7923-3779- CT. J. Huang and YH. A. Li (eds.): <i>New Horizons in Chinese Linguistic</i> .
37.	ISBN 0-7923-3867-7; Pb 0-7923-3868-           A. Watanabe: Case Absorption and WH-Agreement. 1996
38.	H. Thráinsson, S. D. Epstein and S. Peter (eds.): <i>Studies in Comparativ</i> <i>Germanic Syntax</i> . Volume II. 1996 ISBN 0-7923-4215-

39.	C. J. W. Zwart: <i>Morphosyntax of Verb Movement</i> . A Minimalist Approach to the Syntax of Dutch 1007 ISBN 0 7023 4263 1; Pb 0 7023 4264 X
40	T Silani, New Diverse and Newinglighting The Sector of DDs 1007
40.	1. Shoni: Noun Phrases and Nominalizations. The Syntax of DPS. 1997
	ISBN 0-7923-4608-4
41.	B. S. Vance: Syntactic Change in Medieval French. 1997
	ISBN 0-7923-4669-6
42.	G. Müller: Incomplete Category Fronting. A Derivational Approach to
	Remnant Movement in German. 1998 ISBN 0-7923-4837-0
43.	A. Alexiadou, G. Horrocks and M. Stavrou (eds.): Studies in Greek Syntax.
	1998 ISBN 0-7923-5290-4
44.	R. Sybesma: The Mandarin VP. 1999 ISBN 0-7923-5462-1
45.	K. Johnson and I. Roberts (eds.): Beyond Principles and Parameters. Essays
	in Memory of Osvaldo Jaeggli. 1999 ISBN 0-7923-5501-6
46.	R. M. Bhatt: Verb Movement and the Syntax of Kashmiri. 1999
	ISBN 0-7923-6033-8
47.	A. Neeleman and F. Weerman: Flexible Syntax. A Theory of Case and
	Arguments, 1999 ISBN 0-7923-6058-3
48	C. Gerfen: Phonology and Phonetics in Coatzospan Mixtee 1999
101	ISBN 0-7923-6034-6
49	I Paul V Phillins and I Travis (eds.): Formal Issues in Austronesian Linguistics
чу.	2000 ISBN 0-7923-6068-0
50	M Frascarelli: The Syntax-Phonology Interface in Focus and Tonic Constructions
50.	in Italian 2000 ISBN 0-7022-6240-2
51	In Italian, 2000 ISBN 0-7923-0240-5
51.	1. Landau. Elements of Control. Structure and Meaning in Infinitival
50	Constructions. 2000 ISBN 0-7923-0620-4
52.	W. D. Davies and S. Dubinsky (eds.): <i>Objects and other Subjects</i> . Grammatical
	Functions, Functional Categories and Configurationality. 2001
	ISBN 1-4020-0064-2; Pb 1-4020-0065-0
53.	J. Ouhalla and U. Shlonsky (eds.): <i>Themes in Arabic and Hebrew Syntax</i> . 2002
	ISBN 1-4020-0536-9; Pb 1-4020-0537-7
54.	E. Haeberli: Features, Categories and the Syntax of A-Positions. Cross-Linguistic
	Variation in the Germanic Languages. 2002
	ISBN 1-4020-0854-6; Pb 1-4020-0855-4
55.	J. McDonough: The Navajo Sound System. 2003
	ISBN 1-4020-1351-5; Pb 1-4020-1352-3
56.	D. E. Holt (ed.): Optimality Theory and Language Change. 2003
	ISBN 1-4020-1469-4; Pb 1-4020-1470-8
57.	J. Camacho: The Structure of Coordination. Conjunction and Agreement
	Phenomena in Spanish and Other Languages, 2003
	ISBN 1-4020-1510-0: Pb 1-4020-1511-9
58.	L Toivonen: Non-Projecting Words, A Case Study of Swedish Particles 2003
20.	ISBN 1-4020-1531-3. Ph 1-4020-1532-1
	1551(1 1626 1551 5, 10 1 4626 1552 1

59.	D. Adger, C. de Cat and G. Tsoulas (eds). <i>Peripheries. Syntactic Edges and their Effects.</i> 2004 ISBN 1-4020-1908-4; Pb 1-4020-1909-2
60.	C. Goria: Subject Clitics in the Northern Italian Dialects. A Comparative
	Study Based on the Minimalist Program and Optimality Theory. 2004
	ISBN 1-4020-2736-2; Pb 1-4020-2737-0
61.	V. Dayal and A. Mahajan (eds.): Clause Structure in South Asian Languages.
	2004 ISBN 1-4020-2717-6; Pb 1-4020-2718-4
62.	P. Kempchinsky and R. Slabakova (eds.): Aspectual Inquiries. 2005
	ISBN 1-4020-3032-0; Pb 1-4020-3035-5
63.	M. Arad: Roots and Patterns. Hebrew Morpho-syntax. 2005
	ISBN 1-4020-3243-9; Pb 1-4020-3245-5
64.	P. Štekauer and R. Lieber (eds.): Handbook of Word-Formation. 2005
	ISBN 1-4020-3595-0; Pb 1-4020-3597-7
65.	A. Johns, D. Massam and J. Ndayiragije (eds.): Ergativity. Emerging Issues.
	2006 ISBN Hb 1-4020-4186-1; Pb 1-4020-4187-X
66.	F. Lee: Remnant Raising and VSO Clausal Architecture. A Case Study from
	San Lucas Quiavini Zapotec. 2006 ISBN 1-4020-4300-7
67.	Olga M: Balkam Sprachbund Morpho-Syntactic Features. 2006
	ISBN 978-1-4020-4487-8
68.	Kiss, Katalin É. (eds.): Event Structure and the left Periphery. 2006
	ISBN 978-1-4020-4753-4
69.	Harbour, Daniel: Morphosemantic Number, from Kiowa Noun Classes to
	UG Number Features. 2007 ISBN 978-1-4020-5037-4
70.	Pereltsvaig, Asya: Copular Sentences in Russian A Theory of Intra-Clausal
	Relations. 2007 ISBN-978-1-4020-5792-2
71.	William D. Davies & Stanley Dubinsly (eds.): New Horizons in the Analy-
	sis of Control and Raising. 2007 ISBN-978-1-4020-6175-2
72.	Helen de Hoop and Peter de Swart (eds.): Differential Subject Marking,
	2008 ISBN-978-1-4020-6498-2
73.	Rachel Wojdak: The Linearization of Affixes: Evidence from Nuu-chah-
	nulth. 2008 ISBN 978-1-4020-6549-1